

Durham E-Theses

An Investigation into the Effects of Electronic Storybooks on Language and Literacy Outcomes

SAVVA, MARILENA

How to cite:

SAVVA, MARILENA (2024) An Investigation into the Effects of Electronic Storybooks on Language and Literacy Outcomes, Durham theses, Durham University. Available at Durham E-Theses Online: http://etheses.dur.ac.uk/15528/

Use policy



This work is licensed under a Creative Commons Attribution Non-commercial No Derivatives 2.0 UK: England & Wales (CC BY-NC-ND)

Academic Support Office, The Palatine Centre, Durham University, Stockton Road, Durham, DH1 3LE e-mail: e-theses.admin@durham.ac.uk Tel: +44 0191 334 6107 http://etheses.dur.ac.uk



Durham E-Thesis

An Investigation into the Effects of Electronic Storybooks on Language and Literacy Outcomes

MARILENA SAVVA

Use policy



This work is licensed under a Creative Commons Attribution Non-commercial 3.0 (CCBY-NC)

Academic Support Office, Durham University, University Office, Old Elvet, Durham DH1 3HP e-mail: e-theses.admin@dur.ac.uk Tel: +44 0191 334 6107 <u>http://etheses.dur.ac.uk</u>

DURHAM UNIVERSITY

AN INVESTIGATION INTO THE EFFECTS OF ELECTRONIC STORYBOOKS ON LANGUAGE AND LITERACY OUTCOMES

A Thesis presented for the degree of Doctor of Philosophy School of Education, Durham University

> By Marilena Savva 2023

Table of Contents

| Table of Contents | 4 |
|-------------------------------------------------------------------------------------------------------------|----|
| List of Tables | 6 |
| List of Figures | 9 |
| List of Appendices | 9 |
| Abstract | 10 |
| Statement of Declaration | 13 |
| Statement of Copyright | 13 |
| Acknowledgements | 14 |
| Introduction | 15 |
| Outline of the Thesis | 18 |
| Chapter 1: Traditional Print Books and Electronic Storybooks (E-books) | 20 |
| 1.1 Introduction | 20 |
| 1.2 Storybook Reading | 21 |
| 1.3 Emergent Literacy | 23 |
| 1.4 What is an E-book? | 25 |
| 1.5 Learning with E-books – the Role of Digital Features | 26 |
| 1.6 Recognising the Overarching Aims of the Thesis | 29 |
| 1.7 Summary | 31 |
| Chapter 2: Teacher Scaffolding During Storybook Reading | 32 |
| 2.1 Introduction | 32 |
| 2.2 Background | 32 |
| 2.3 Scaffolding During Storybook Reading | 33 |
| 2.4 Scaffolding and Literacy Development | |
| 2.5 Interactive and Performance Teaching Styles | 40 |
| 2.6 Scaffolding and E-books | 43 |
| 2.7 Summary | 46 |
| Chapter 3: Young Children Learning English as an Additional Language (EAL) and Storybooks (Print & E-books) | 48 |
| 3.1 Introduction | 48 |
| 3.2 Characteristics of EAL Learners | 49 |
| 3.3 The Case of Cyprus | 50 |
| 3.4 Storybook Reading and Scaffolding for Children Learning EAL | 52 |

| 3.5 E-books and Young Children Learning EAL | 55 |
|------------------------------------------------------------------------------------|---------------|
| 3.6 E-books and Adult Scaffolding for Young Children Learning EAL | 56 |
| 3.7 Summary | 58 |
| 3.8 An Overview of the Thesis's Research Ouestions | 58 |
| Chapter 4: Study 1 – Meta-Analysis | 61 |
| 4.1 Research Questions | 67 |
| 4.2 Provide Mate Analyses of the Effects of E books Compared to Print Books | 02 67 |
| 4.2 M-4h - J | |
| 4.5 Method 4.3.1 Inclusion and Exclusion Criteria | 04 |
| 4.3.2 Literature Search | |
| 4.3.3 Study Coding | 67 |
| 4.3.4 Methodological Rigour of Studies | |
| 4.3.5 Data Analysis | 84 |
| 4.4 Results | 85 |
| 4.4.1 Descriptive Statistics | 85 86 |
| 4.4.2 Overall Effects | |
| 4.4.4 Publication Bias | |
| 4.4.5 Effect Sizes for Each Outcome | 92 |
| 4.4.6 Multimedia versus Interactive E-books versus "Non-Embellished" E-books | |
| 4.4.7 Animation and Hotspots | |
| 4.4.8 Adult Scaffolding/Mediation Effects | |
| | |
| 4.5 Discussion | 106 |
| 4.5.1 Limitations | 109 110 |
| 4.5.3 Implications for Future Research | |
| 4.6 Conclusion | 112 |
| Chapter 5: Experimental Study | 114 |
| 5 1 The "Develocity" E heads | ד 11 <i>1</i> |
| 5.1 The "Perfect" E-Dook | 114 |
| 5.2 Research Questions | 122 |
| 5.3 Method | 123 |
| 5.3.1 Participants | |
| 5.3.2 Covid-19 Pandemic | |
| 5.3.3 Materials | 120 128 |
| 5.3.5 Design | |
| 5.3.6 Procedure | |
| 5.3.7 Statistical Analyses | |
| 5.4 Results | 136 |
| 5.4.1 Preliminary Analyses | |
| 5.4.2 Research Question 1 - Comparison of Book Mediums: E-book versus Print Format | |
| 5.4.5 Kesearch Question 2 - Interactive versus Performance Teaching Style | 144 150 |
| 5.4.5 Bonferroni Correction | |
| 5 5 Discussion | 150 |
| 5.5 1 Fhook versus Print Book | 154 153 |
| 5.5.2 Interactive Teaching Style versus Performance Teaching Style | |
| | |

| 5.5.3 Limitations of the Study | 157 |
|------------------------------------------------------------------------|------------|
| 5.5.4 Theoretical Implications | 157 |
| 5.5.5 Practical Implications | 158 |
| 5.5.6 Implications for Future Research | 160 |
| 5.5.7 Conclusion | 161 |
| Chapter 6: Discussion and Conclusion | |
| 6.1 Returning to the Purpose of the Thesis: Connecting the Two Studies | 162 |
| 6.2 The Outcomes of the Study for EAL Learners in Cyprus | 166 |
| 6.2 Lessons Learned from the Studies Reported in this Thesis | 168 170 |
| 6.3 Contributions to the research literature | |
| 6.4 Conclusion | 175 |
| References | 177 |
| Appendices | 216 |

List of Tables

| Table 4.1 Inclusion and Exclusion Criteria 66 |
|-----------------------------------------------------------------------------------------------------------------------------|
| Table 4.2 Study Coding |
| Table 4.3 Assessment of Methodological Quality Based on Selected Internal and ExternalValidity Criteria from Troia (1999)82 |
| Table 4.3 Assessment of Methodological Quality Based on Selected Internal and ExternalValidity Criteria from Troia (1999) |
| Table 4.4 Statistics for Each Study 87 |
| Table 4.4 Statistics for Each Study 88 |
| Table 4.5 Overall Results of the 29 Studies and Moderator Analyses#89 |
| Table 4.6 Overall Results for Outcome Measures 93 |
| Table 4.7a Statistics of Each Study for Comprehension and Outcome Measures Descriptions.94 |
| Table 4.7a Statistics of Each Study for Comprehension and Outcome Measures Descriptions.95 |
| Table 4.7b Statistics of Each Study for Vocabulary and Outcome Measures Descriptions 96 |
| Table 4.7b Statistics of Each Study for Vocabulary and Outcome Measures Descriptions |
| Table 4.7b Statistics of Each Study for Vocabulary and Outcome Measures Descriptions 98 |
| Table 4.7c Statistics of Each Study for Decoding and Outcome Measures Descriptions |
| Table 4.7c Statistics of Each Study for Decoding and Outcome Measures Descriptions 100 |
| Table 4.8 E-books With and Without Multimedia and Interactive Features on Language andLiteracy Outcomes103 |
| Table 4.9 Studies With and Without Adult Scaffolding / Mediation in Print and E-books105 |
| Table 5.1 Vocabulary Development and Story Comprehension - What Worked?116 |

| Table 5.2 Vocabulary Development and Story Comprehension - What Did Not Work?117 |
|-------------------------------------------------------------------------------------------------------------------------------------------|
| Table 5.2 Vocabulary Development and Story Comprehension - What Did Not Work? |
| Table 5.3 Description of Conditions 132 |
| Table 5.4 An Example of the Schedule of the Experiment 134 |
| Table 5.5 Means (and standard deviations) of Basic Characteristics per Condition137 |
| Table 5.6 The Number of Participants Who Responded to Each Measure for Each Story138 |
| Table 5.7 Repeated Measures ANOVA Results for Receptive Vocabulary Based on BookMedium (pre-test versus post-test 1)140 |
| Table 5.8 Repeated Measures ANOVA Results for Expressive Vocabulary Based on BookMedium (pre-test versus post-test 1) |
| Table 5.9 One-way ANOVA Results for Implicit and Explicit Story Comprehension (post-test 1) |
| Table 5.10 One-way ANOVA Results of PLS-5 Based on Book Medium (pre-test and post-test) |
| Table 5.11 Repeated Measures ANOVA Results for PLS-5 Based on Book Medium (pre-testversus post-test) |
| Table 5.12 Repeated Measures ANOVA Results for Receptive Vocabulary Based on TeachingStyle (pre-test versus post-test 1)145 |
| Table 5.13 Repeated Measures ANOVA Results for Expressive Vocabulary Based on TeachingStyle (pre-test versus post-test 1)147 |
| Table 5.14 One-way ANOVA Results of Implicit, Explicit and Overall Story ComprehensionResults Based on Teaching Style (post-test 1)148 |
| Table 5.15 One-way ANOVA Results of PLS-5 Based on Teaching Style (pre-test and post-test) |
| Table 5.16 Repeated Measures ANOVA Results for PLS-5 Based on Teaching Style (pre-test versus post-test) 149 |
| Table 5.17 Two-way ANOVA Results for Transfer Task of Book Medium by Teaching Style (post-test 1) 150 |
| Means (and standard deviations) of Basic Characteristics per Condition258 |
| Table M1 One-way ANOVA Results for Receptive and Expressive Vocabulary Based on BookMedium (pre-test, post-test 1, post-test 2) |
| Table M2 One-way ANOVA Results for Receptive and Expressive Vocabulary Based onTeaching Style (pre-test, post-test 1, post-test 2)273 |
| Table M3 Repeated Measures ANOVA Results for Receptive Vocabulary of Book Medium byTeaching Style (pre-test versus post-test 1)275 |
| Table M4 Repeated Measures ANOVA Results for Receptive Vocabulary of Book Medium byTeaching Style (pre-test, post-test 1 and post-test 2) |
| Table M5 Repeated Measures ANOVA Results for Expressive Vocabulary of Book Medium byTeaching Style (pre-test versus post-test 1)277 |
| Table M6 Repeated Measures ANOVA Results for Expressive Vocabulary of Book Medium by |

| Table M7 Two-way ANOVA Results for Implicit Story Comprehension of Book Medium byTeaching Style (post-test 1) | 0 |
|-----------------------------------------------------------------------------------------------------------------------------------------|---|
| Table M8 Two-way ANOVA Results for Explicit Story Comprehension of Book Medium byTeaching Style (post-test 1) | 1 |
| Table M9 Repeated Measures ANOVA Results for Standardised Test PLS-5 UK of BookMedium by Teaching Style (pre-test versus post-test 1)28 | 3 |
| Table L1 Research Tools used in studies related to e-books and young children | 7 |

List of Figures

| Figure 4.1 Funnel Plot of Standard Error by Hedge's g | 91 |
|------------------------------------------------------------------------------------------------------------------------------------------------------|-------------|
| Figure 4.2 Trim and Fill Funnel Plot Hedges's g and Standard Error | 91 |
| Figure 4.3 Effect Sizes and 95% CIs of Outcome Measures | 93 |
| Figure 5.1 Repeated Measures for Receptive Vocabulary Based on Book Medium | .140 |
| Figure 5.2 Repeated Measures for Expressive Vocabulary Based on Book Medium | .141 |
| Figure 5.3 Repeated Measures for Receptive Vocabulary Based on Teaching Style | .145 |
| Figure 5.4 Repeated Measures for Expressive Vocabulary Based on Teaching Style | .147 |
| Figure M1 Repeated Measures for Expressive Vocabulary of Book Medium by Teaching Sty for the Performance Teaching Style (pre-test to post-test 1) | vle .277 |
| Figure M2 Repeated Measures for Expressive Vocabulary of Book Medium by Teaching Sty for the Interactive Teaching Style (pre-test to post-test 1) | vle .278 |
| Figure M3 Implicit Story Comprehension of Book Medium by Teaching Style at Post-test 1 | .280 |
| Figure M4 Explicit Story Comprehension of Book Medium by Teaching Style at Post-test 1 | .282 |

List of Appendices

| Appendix A: PRISMA Diagram of the Literature Search | 216 |
|---------------------------------------------------------------------------|-----|
| Appendix B: Forest Plot – Hedges's g and 95% CI | 217 |
| Appendix C: Consent Form | 218 |
| Appendix D: Discussion Points and Questions Asked During the Intervention | 220 |
| Appendix E: English Language 5-Point Scale | 233 |
| Appendix F: Receptive Vocabulary Tests | 234 |
| Appendix G: Expressive Vocabulary Tests | 238 |
| Appendix H: Implicit Story Comprehension Tests | 244 |
| Appendix I: Explicit Story Comprehension Tests | 250 |
| Appendix J: Transfer Task Text Analysis | 254 |
| Appendix K: Intraclass Correlation Coefficient (ICC) | 258 |
| Appendix L: Analysis of Covariance (ANCOVA) Results | 261 |
| Appendix M: Exploratory Analysis | 270 |
| Appendix N: Research Tools Selection Process | 284 |

Abstract

The array of availability of diverse digital reading applications, the mixed results emerging from small-scale experimental studies, as well as the long-standing tradition and range of known positive developmental outcomes gained from adult-child storybook reading warrant an investigation into electronic storybooks (e-books). In doing so, the thesis looks to provide the reader with both a meta-analysis and an experimental study of e-book and print book reading for young children. To achieve this, the thesis comprises of two studies, which complement each other.

Study 1 comprises of a meta-analysis of examining the effects of e-books on language and literacy outcomes for young children. The overall purpose of the meta-analysis was to examine the impact of e-book reading on language and literacy development of young children when compared to traditional reading of print books with or without adult scaffolding in a structured and controlled environment. The meta-analysis included 29 experimental studies published between 2008 and 2021 with a target population of 3-8year-olds. Analyses indicated a small positive effect for e-books when compared to print books on language and literacy development (g = 0.25; 95% CI = [0.09, 0.42]). A moderate positive effect was found for vocabulary learning (g = 0.40; 95% CI = [0.10, 0.69]), especially in relation to expressive vocabulary (g = 0.54; 95% CI = [0.08, 1.00]). In addition, a significant positive correlation was found between multimedia e-books and the development of code-related skills (g = 0.63, 95% CI = [0.28, 0.99]). However, no significant differences were found between e-book and print book reading in relation to story comprehension. Overall, findings showed that digital features combined with adult scaffolding produced significant positive effects when compared to traditional print book reading with adult support. The findings have practical ramifications, since they can help researchers and educators identify which digital features have the greatest influence on improving children's language and literacy skills when engaging with e-books.

Study 2 collected data from sixty children between the ages of 3 and 7 years old over the school year 2020-2021. The children attended a private English school in Cyprus where English was taught as a main language, but it was considered an additional language for the participants. The purpose of the second study was to employ the findings from the first study and apply them to a particular demographic, namely children who learn English as an

additional language, in order to assess the impact of e-books in contrast to print books on the learning outcomes of this specific group of learners. The intervention took place at their school, where two print books and two e-books were compared using two different teaching styles – an interactive style and a performance style. The children listened to each story twice. The statistical analyses were conducted using the IBM SPSS Statistics 28 software; particularly, repeated measures ANOVA analyses were conducted. The results of the study indicated that children who are acquiring English as an additional language may enhance their vocabulary and comprehension skills by engaging in storybook reading with the support of interactive adult scaffolding. Specifically, the findings demonstrated a significant statistical difference in expressive vocabulary between the interactive teaching style and the performance teaching style, with the interactive teaching style yielding higher mean scores $(F(1, 58) = 5.014, p=0.02, \eta_p^2=0.08)$. Additionally, it was observed that the interactive teaching style yielded greater mean scores compared to the performance teaching style (F(1,56)=4.15, p=0.04, $\eta_p^2 = 0.06$) in story comprehension outcomes (although the results were deemed non-significant after the Bonferroni adjustment). In terms of e-books versus print books, there was no interaction effect for receptive vocabulary (F(1, 58) = 0.177, p=0.67, $\eta_p^2 = 0.00$) and expressive vocabulary (F(1, 58) = 2.490, p = 0.12, $\eta_p^2 = 0.04$), which shows that everyone improved from pre-test to post-test regardless of condition. While analysing the results retrieved from book medium analyses, post-scores revealed a statistically significant result for explicit story comprehension (p=0.04, $\eta_p^2=0.07$), favouring the print condition (before Bonferroni correction).

Together, the two studies indicate that e-books can support language and literacy outcomes. Multimedia features incorporated in e-books are found to benefit children more than interactive features, which seem to distract children from the storyline and as a result from their development. This thesis also highlights the importance of adult scaffolding during storytelling sessions, since it benefits children more than independently listening to an e-book. Adult support during e-book reading can support both typical developing learners, as well as children learning English as an additional language. Findings indicated that the utilisation of interactive adult scaffolding is beneficial for children who are learning English as an additional language. This approach involves engaging children in discussions about the story while they are reading, as opposed to the performance teaching style, which involves skilful storytelling with minimal interruptions in the text. Overall, multimedia e-books with adult support have the potential to support young children in both vocabulary

and story comprehension, regardless of their native language.

Statement of Declaration

I declare that this work is my own. Material in this thesis has been published.

Statement of Copyright

The copyright of this thesis rests with the author. No quotation from it should be published without the author's prior written consent and information derived from it should be acknowledged.

Acknowledgements

I wish to use this space to take the opportunity to say thank you for all the people who have helped me to complete my doctoral study.

First and foremost, I would like to express my gratitude to my supervisors, Professor Nadin Beckmann and Professor Steve Higgins. Their guidance has been invaluable in shaping this piece of work, and I will be forever thankful for the feedback they have provided.

I would, of course, also like to extend my gratitude to all the participants who partook in the study. Without them, this research project simply would not have been possible. Similarly, I would like to say thank you to the teachers and the school that supported me during the intervention process.

Lastly, my gratitude must also extend to my close friends, family and partner, whose support and patience during the doctoral study has been immeasurable.

Introduction

There is ample evidence that storybook reading is one of the most important sources of language and literacy development during nursery, preschool and primary school years (Bus et al., 1995; Hindman et al., 2014; Mol & Bus, 2011). However, in a world in which media dominates our daily lives, young children spend a lot of time with on-screen activities (e.g., playing online games, watching cartoons) at the expense of reading print books. "Modern" classrooms are elevating by following the social norm as schools have integrated the use of tablets into their curriculum by downloading various educational and non-educational applications as well as e-books (Apple, 2013). In addition, the unexpected arrival of the COVID-19 pandemic has necessitated significant changes in school institutions throughout the world. It has resulted in the closing of schools and the reform of teaching and learning practices, with technology being used more today than ever before. This development holds great promise and makes it imperative to determine how much the use of this software affects young children's language and emergent literacy levels. How well can an e-book develop children's language, story comprehension and other early literacy skills depends on the quality and how well e-books have been designed to specifically meet the children's developmental needs (Shamir & Korat, 2006).

Since 2010, research on e-books has expanded substantially. E-books are digital storybooks that are easily accessible via mobile applications and other electronic devices (e.g., Kindle, tablet) and include various digital components. Most of the e-books aimed at young children are equipped with interactive and multimedia features, e.g., oral reading, background music and sounds, hotspots activating animation, on-demand word definitions, and other functions that may support or hinder language and literacy development. Based on research over the last decade, well-designed e-books for young children's language (vocabulary learning and story comprehension) and literacy development (phonological awareness, print awareness, word reading and writing, alphabet knowledge). To date, there have been conflicting findings with some studies finding that e-books may hinder learning (e.g., Estefani & Queiroz, 2018; Gong & Levy, 2009) while others demonstrate that they may help (Bus et al., 2015; Shamir et al., 2012). In addition, the fast development of technology and software has reduced the generalisability of findings between generations. Thus, it is crucial to assess and evaluate this book medium that children use today and how well their included features

can increase children's language and literacy development.

Study 1 aims at analysing the efficacy of e-books in facilitating young children's language and literacy development when compared to more traditional presentations of stories, such as narrating a story aloud or reading a print storybook, by performing a meta-analysis. The overall purpose of the meta-analysis is to examine the impact of e-book reading on language and literacy development of young children when compared to traditional reading of print books with or without adult scaffolding in a structured and controlled environment. When comparing print books and e-books as a medium, multimedia, and interactive features embedded in e-books may tax working memory compared to paper books, or conversely, paper books may necessitate significant more adult input for new learning to be acquired by young children. Several experimental studies have investigated the factors that influence children's learning via e-books and story applications (e.g., Bus et al., 2020; Hassinger-Das et al., 2020; Lawrence & Choe, 2021; Neumann, 2020b). As the e-books in question integrate a variety of characteristics, the variety of research published over the previous decade yields mixed results. Consequently, it is essential to review and evaluate the e-books used by young children and the extent to which the digital elements contained in e-books promote children's language and literacy development. To that end, the meta-analysis (Study 1) seeks to investigate the embedded features within e-books, which may resemble adult scaffolding, and to what extent these digital features can promote learning when compared to traditional reading of print books.

Furthermore, the purpose of this thesis is to gain a better understanding of adult scaffolding via the usage of various teaching styles while reading a print book and an e-book. Although positive correlations have been found between variation in the quality of adult scaffolding during shared reading and children's language development (e.g., Hindman et al., 2008; Hindman et al., 2014), further research is needed to examine the effectiveness of these scaffolding techniques in relation to young children's language and story comprehension development. The two scaffolding techniques proposed to be evaluated by this thesis (in Study 2) are the interactive teaching style and the performance teaching style. During the interactive teaching style adults initiate a discussion and ask questions before, during and after the storytelling session. Within a performance teaching style adults and learners engage in a brief conversation before the reading begins. During the story listening session, children remain silent in order to concentrate on listening to the story. At the end of the story, the

discussion continues, and children may ask any questions or initiate conversation. These two scaffolding techniques, as well as book mediums (e-book versus print book) are assessed in Study 2.

Book medium (e-book versus print book) as well as teaching style (interactive versus performance) are evaluated for children learning English as an additional language. It is essential to evaluate the effectiveness of e-books against print books with the assistance of adults by comparing the two aforementioned teaching styles for children whose native language is not English. In the past decade, the majority of research on the effectiveness of e-books has concentrated on first-language learners, whereas relatively few studies have included individuals learning English as an additional language. Therefore, the second study assessed these variables at a school-based intervention employing an e-book and a print book for the development of vocabulary and story comprehension in children learning English as an additional language. The participants were children (3-7 years) residing in Cyprus and attending an English private school. English was either the second, third or fourth language of the participants. The choice of school and country for this study was based on the author's home country and their school of employment.

Cyprus, the third largest island in the Mediterranean, is located in the north-eastern part of the Mediterranean Sea and encompasses a total land area of 9,251 square kilometres. Cyprus is a relatively newly formed sovereign state, established in 1960, and a member of the European Union since 2004. Despite the fact that the island has historically had a multilingual and multicultural profile, cultural variety has risen significantly since the late 1990s as a result of economic growth. According to the preliminary census results of the Statistical Service of Cyprus (2021, October 1), the population in the Government controlled areas of Cyprus on the 1st of October 2021 was 923,272. The total number of foreign nationals amounts to 193,300 and corresponds to 21% of the total population. Article 20 of the Constitution of Cyprus (1960, April 6) protects the right of every individual to receive instruction or education, and the right of any person or institution to provide it. This right is safeguarded for any person, irrespective whether Cypriot or foreigner, to the extent that they reside in the Republic. To this end, migrant students can enrol in public as well as private schools, even if their parents do not have the status of legal migrants. The language taught at public schools is Greek and they follow the Greek national curriculum. Private schools can adjust their curriculum and timetable according to the

educational system of another country. Therefore, many foreigners choose for their children to attend English schools and be taught primarily in English rather than in Greek. The English private school where the author works accepted pupils from a variety of socioeconomic backgrounds, and English language proficiency is not a criterion.

In addition to the broader contextual framework outlined above, this research also included a personal dimension. As a preschool teacher for the past ten years, it has been possible to professionally engage and reflect upon issues of effective language and literacy teaching in preschools with the use of technology. Incorporating non-traditional teaching approaches into my lessons has always piqued my interest, particularly when technology is involved. Therefore, the purpose of the thesis was to provide answers to critical questions that I, as an educator, had encountered when attempting to include technology in my teaching practice. Overall, this thesis is divided into two studies:

- Study 1: Meta-analysis examining the effects of electronic storybooks on language and literacy outcomes for children aged 3 to 8 years old.
- Study 2: The "perfect" e-book versus print book: A comparison of different types of media and reading styles for vocabulary and story comprehension for children learning English as an additional language.

Outline of the Thesis

The thesis is structured in six chapters. The introductory section describes the context of the study, as well as its aims and significance.

Chapters 1, 2 and 3 provide the theoretical foundation. Chapter 1 presents the background and recent research on electronic storybooks and traditional reading of print books for language and literacy development of young children. Chapter 2 examines adult scaffolding during print book reading and e-book reading and presents the two teaching styles used in Study 2. In Chapter 3, research on young children learning English as an additional language and storybooks (print and e-books) are presented, as well as the characteristics of children learning English as an additional language.

Chapter 4 present the meta-analysis (Study 1), which aimed at examining the impact of ebook reading on language and literacy development of young children when compared to traditional reading of print books with or without adult scaffolding in a structured and controlled environment. The chapter presents the research questions, inclusion and exclusion criteria and results of the meta-analysis.

Chapter 5 describes the experimental study (Study 2), including its research design, the research procedures followed throughout the intervention, and the presentation of the results.

Finally, Chapter 6 presents the research findings and the relationship between the findings of the two studies, their embodiment in prior research literature, as well as the lessons learned from the two studies. This chapter offers an insight into the contribution this study makes to the field.

Chapter 1: Traditional Print Books and Electronic Storybooks (E-books)

This chapter examines the background and benefits of reading traditional print books. In addition, after assessing the advantages for language and literacy development, the design and features of e-books that may aid or hinder children's growth are explored.

1.1 Introduction

The sale of e-books is a growing trend and its market share is steadily increasing (Korat & Falk, 2019). E-books can be instantly purchased online and they are available via apps, tablet or e-reader devices, and current digital sales are increasing across the board. In the present study, e-books are defined as a digital form of a print book and a number of terms are being used to refer to e-books, such as electronic storybooks, digital books, storybook apps, digital storybooks, e-readers. According to a report by the Association of American Publishers (AAP, 2021), e-book sales increased by 15.2% in the first eleven months of 2020, generating \$1.0 billion United States dollars. Harper Collins has reported that in the first quarter of 2020, e-book sales were up 26% compared to the prior year (Kozlowski, 2020). The big genre sales within the digital division were children and general fiction e-books (Kozlowski, 2021). There are various formats available for the purchaser and consequently the ease of access has paved the way for the application of e-books within the early years' curriculum.

Since the advent of tablet computers and smartphone devices, young children tend to spend more and more of their time on on-screen activities (e.g., playing online games, watching online videos). During the COVID-19 pandemic, screen time has increased dramatically for everyone - including children, many of whom were receiving school instruction remotely (Hassinger-Das et al., 2020). Ofcom (2020) in the United Kingdom reported that 24% of 3–4-year-old children had their own tablet and 42% of 5-7 years old owned their own tablet or smartphone device. Ofcom's most recent investigation, conducted in 2023, reveals that 40% of children aged 3 to 4 and 61% of children aged 5 to 7 own a mobile device. According to the data, tablets and telephones are utilised by children aged 3-4 and 5-7 years for various online activities, such as watching videos or playing online games. Specifically, tablets emerged as the predominant device utilised by

children aged 3-4 years for accessing the internet (75%), engaging in gaming activities (50%), and viewing television content (excluding traditional television sets) (67%). Tablets were also the most popular devices among children aged 5-7 years, with 86% of this age group using them to access the internet, 67% to watch television programmes or films (excluding traditional television sets), and 67% to engage in gaming activities (Ofcom, 2023). These findings further amplify the concerns already expressed by several studies on the impact of digital media on children's learning and development. The driving force behind these concerns is that young children today have their own digital devices and are more likely to be using them on a daily basis, which may potentially result in the replacement of printed storybooks by digital versions. Thus, e-books are a ubiquitous part of our society and may have a direct and immediate effect on the rate of young children's language development and literacy acquisition.

1.2 Storybook Reading

Storybook reading is well-established in the literature, and it is regarded as a leading avenue for building early academic skills (Bus et al., 1995; Bus et al., 2000; Dickinson & Tabors, 2001; Gonzalez et al., 2014; Whitehurst & Lonigan, 2003). A large body of work supports the finding that storybook reading facilitates emergent literacy skills (Bus et al., 2000; Chomsky, 1979; Clark, 1978; Davidse et al., 2011; Durkin, 1966; Hindman et al., 2014; Hindman et al., 2012; Mol & Bus, 2011; National Research Council, 1998; Scarborough & Dobrich, 1994; Teale, 1987). The range of positive developmental outcomes gained from early exposure to books persist beyond early childhood and continue into adolescence and young adulthood (Mol & Bus, 2011; Rose et al., 2018). Early childhood education can provide children with opportunities to develop essential foundation skills in language and literacy that directly translate into later school success (Burchinal et al., 2002; McCardle et al., 2001). Storybooks are by far the most prevalent text genre that preschool teachers choose to read aloud in their classrooms (Gerde & Powell, 2009; Hindman et al., 2008; Stone & Twardosz, 2001). Preschool teachers consider reading aloud to be an important activity in their classrooms (Hindman & Wasik, 2008; Lonigan et al., 2000), and indeed reading aloud helps young children to develop the oral language and early literacy skills needed for later literacy success (e.g., Dickinson et al., 1992; Wasik et al., 2006). To achieve this, teachers employ extratextual discourse, which pertains to conversations that take place outside the realm of reading, as a method

to promote children's engagement, improve their language and literacy skills, and offer the necessary support for children to successfully tackle tasks that are slightly beyond their independent capabilities.

Storybook reading is a popular practice in preschool classrooms, and it can offer the reader/listener a number of educational benefits (Bus et al., 1995; Gonzalez et al., 2014). Early childhood education offers young children an opportunity to enhance their communication and language skills, including the development of grammar, phonology, vocabulary, and pragmatics. Storybook reading activates the development of emergent literacy skills (Bus & Van IJzendoorn, 1988; Inoue et al., 2018), such as concepts of print (e.g., Nevo & Vaknin-Nusbaum, 2018; Justice & Ezell, 2000, 2004), phonological awareness (Chow et al., 2008; Lefebvre et al., 2011), and alphabet knowledge (Aram, 2006). The enhancement of these concepts can be achieved by engaging in high-quality interactions between adults and children during storybook reading (Justice & Ezell, 2002). These interactions have been found to facilitate the development of various early language skills, including expressive and receptive vocabulary (Farrant & Zubrick, 2012; Flack et al., 2018; Lenhart et al., 2020), as well as decontextualised language (Beck & McKeown, 2001; Seven et al., 2020), which involves the use of unfamiliar words (Marulis & Neuman, 2010; Montag et al., 2015).

Research results on storybook reading with children from low socioeconomic status (SES) families were found to be equally beneficial as results found in medium and high SES families (Shahaeian et al., 2018; Wasik & Bond, 2001). Low SES children often enter the early years' classroom demonstrating a gap in their oral language production and weaker print and phonological awareness skills as compared to their middle-class peers (Dickinson & Snow, 1987; Justice & Ezell, 2002). For example, Lefebvre et al. (2011) studied low-income preschoolers' language and literacy development while undergoing a storybook-reading intervention programme. According to the findings, the experimental group outperformed the control group in phonological awareness. The results of the analysis were also compared with a group of higher-income preschoolers who did not receive the intervention. Children from low-income families in the experimental group outperformed their higher-income counterparts on phonological awareness, vocabulary, and print awareness scores. Neuman and Kaefer (2018) worked in a year-long storybook intervention with low-income children on their oral language vocabulary and content

knowledge. Results of their analysis showed that the young children learned significantly more words than the control group. These results indicate that storybook interventions have the potential to develop young children's language and literacy development regardless of their background.

1.3 Emergent Literacy

Literacy - the ability to read and write - is without a doubt the most important skill required for academic and personal success (Snow, 1983). Literacy skills emerge even before children begin formal school (Burgess et al., 2002). From birth till the beginning of formal education, crucial language and literacy development milestones are accomplished. In the first 5 years of life, children acquire the skills and knowledge about language that are developmental precursors to conventional forms of reading and writing (Whitehurst & Lonigan, 1998). This group of skills and knowledge, including language development, vocabulary, print awareness, a core of basic world knowledge, and motor skills necessary for writing, is referred to as emergent literacy.

Emergent literacy was introduced by Clay (1966) and then restated by Sulzby and Teale (1996):

"Emergent literacy is concerned with the earliest phases of literacy development, the period between birth and the time when children read and write conventionally. The term emergent literacy signals a belief that, in a literate society, young children – even 1- and 2-year-olds – are in the process of becoming literate" (p. 728).

This suggests that language and literacy development has its roots from the most basic biologically and socially based human activities (Dickinson & McCabe, 2001). Emergent literacy concerns all of the different ways that humans communicate – through reading, writing, speaking, and listening – in real life situations. When a child is engaged with a picture book or drawing on a tablet, they are becoming a reader and a writer, engaging in these activities at their own level of competence (Strouse & Ganea, 2017).

The Early Years Foundation Stage (Department for Education, 2021) (UK Curriculum) statutory framework specifies four areas of emergent literacy development children require before learning to read and include (a) communication and language, (b) phonological awareness, (c) print awareness and (d) alphabet knowledge. Communication and language

includes vocabulary, grammar, phonology, and comprehension. Phonological awareness "refers to the ability to generate and recognize rhyming words, to count syllables, to separate the beginning of a word from its ending, and to identify each of the phonemes in a word" (Yopp & Yopp, 2000, p.130). Poor phonological awareness skills are a strong predictor of future reading difficulties in pupils (Storch & Whitehurst, 2002) making explicit instruction of phonological skills essential (Ehri, 1989). Print awareness is knowledge of the form and function of print and the association between written and oral language units (Ehri & Sweet, 1991; Snow et al., 1998). The acquisition of print concepts throughout the preschool period enhances a child's capacity to connect written and oral language (Adams, 1990). Finally, alphabet knowledge refers to a child's ability to recognise letters as distinct units known as graphemes and to name those letters (Worden & Boettcher, 1990). The ability to identify letters is regarded as a reliable indicator of one's knowledge of letter sounds, a crucial skill in comprehending the relationship between graphemes and phonemes and subsequently decoding words (Ehri & Sweet, 1991). Thus, an early year's teacher has an important goal to fulfil by helping every child achieve its full potential and to promote literacy in these crucial years. A highly researched activity supporting literacy development is shared storybook reading due to its association with emergent literacy and language development (Bus et al., 1997).

Reading books aloud not only promotes good reading habits, but also listening and speaking abilities. Children who are frequently exposed to storybook reading are more likely to use complex sentences, have enhanced literal and comprehension skills, develop a deeper understanding of story concepts, increase letter and symbol recognition, and develop positive attitudes about reading (Silvern, 1985). Reading to young children can enhance their communication skills as they discuss the story and illustrations (Kaderavek & Justice, 2002). According to Huck et al. (1989), children who exhibit advanced linguistic skills are often those who have been exposed to a significant amount of reading. Language expert, Chomsky (1972), believed that the development of syntactic complexity and expansion of vocabulary in young children could be attributed to the adult-child reading experience. Thus, reading a print storybook with an adult is a vital practice for promoting language and literacy development. The following section contrasts studies on traditional print book reading versus electronic storybooks.

1.4 What is an E-book?

E-books appeared as early as 1967 when Andries van Dam, a technology professor at Brown University, developed hypertext on a computer (Anuradha & Usha, 2006). Although he conceptualised e-books so many years ago, the electronic storybook has made its appearance in the early year's classroom in the mid-1990s when educational researchers began exploring their value as computers became smaller and more affordable. Since then, the availability of e-books has grown rapidly, resulting in e-books that are available in many languages and contain a variety of multimedia and instructional features that can be tailored to individual needs. E-books are characterised as a form of electronic text that contains key features of traditional print books such as a main idea or theme and pages that turn, but e-books may also contain multimedia, interactive, animated features and follow up activities too (Zucker et al., 2009). This complicates things even more as today's children are being introduced to various forms of e-books more than ever before.

Storybooks delivered via digital devices have been referred to by a variety of terms (e.g., electronic storybooks, digital books, storybook apps, digital storybooks, e-readers) (Reich et al., 2016). In the present study, the term *e-book* will be used when referring to the abovementioned terms. E-books can be described as digital representations of books that possess similar characteristics to traditional printed books, such as "turning" pages, and digital features that can assist the reader, such as audio narration, word pronunciations, text highlighting, text-to speech options and gamification (Dore et al., 2018; Takacs et al., 2015; Yin & Hwang, 2018). E-books can incorporate a variety of digital features with little uniformity across currently available software (Korat & Falk, 2019; Papadakis et al., 2018). Innovative technological hypermedia, such as multimedia (e.g., animation, music and sound effects) and interactive features (e.g., games, hotspots, dictionary function) add complexity to the reading experience as e-books contain a variety of elements and functions that traditional reading on printed pages do not (Chen et al., 2022; Zhang et al., 2020). Thus, there is a need to research and evaluate the effectiveness of this educational medium, as it seems that the use of the e-book will continue to expand and make its way into the early year's classroom.

Young children today can listen to storybooks not only when an adult reads to them from a printed version, but they can also "read" by themselves using e-books appearing on a computer, a tablet, a smartphone, or some other electronic device (Korat & Falk, 2019). As a result of the growing awareness and concern of parents and teachers for the educational quality of e-books, researchers in several countries have studied this issue (e.g., in The Netherlands: De Jong & Bus, 2003; Smeets & Bus, 2015; in the US: Ingram, 2020; Roskos et al., 2009; Schugar et al., 2013; in the UK: Karemaker et al., 2010; Kucirkova & Flewitt, 2020; Underwood & Underwood, 1998; and in Israel: Segal-Drori & Shabat, 2021; Shamir & Korat, 2015). Furthermore, it is imperative for curriculum planners, e-book designers, and software developers to consider the quality of e-books (Korat & Falk, 2019; Van Daal et al., 2019). The possible discrepancy between the objectives of e-book developers and the cognitive development of children is a matter of great significance. An aim of this thesis is to analyse experimental studies involving interventions with e-books in an effort to evaluate the specific features embedded in e-books that may simulate extraneous support (e.g., adult scaffolding) provided to children while listening/reading a story for language and literacy development via performing a meta-analysis (Study 1).

1.5 Learning with E-books – the Role of Digital Features

Investigating the quality of app and e-book content is a continuing concern of researchers as they argue that digital and media developers are prioritising marketing and digital sales rather than pursuing educational objectives and values (Hiniker et al., 2019). Commercial e-books available in app stores might be highly entertaining, but they may fall outside the scope of what researchers consider educational (Hirsh-Pasek et al., 2015). In order for children to learn, comprehend and develop their literacy and language skills while engaging with an e-book they have to show understanding that goes beyond words, focus on the content of the story, listen clearly to the dialogues, add their own perceptions and knowledge to foresee what will happen next, and preserve information they have already read as the story progresses (Newmann, 2020). The arguments regarding the design properties of e-books should be focused on the principles of Mayer's multimedia learning (Mayer, 2005), Paivio's dual coding theory (Paivio, 1986), and Sweller's cognitive load theory (Sweller, 1988; Sweller et al., 2011) and how words (printed or audible or both) and pictures (static or dynamic) should be presented to provide meaningful and effective learning.

Paivio's dual coding theory explains that the human mind simultaneously processes two types of information: visual information-images (nonverbal) and verbal information-

logogens (Paivio, 1986; Sadoski & Paivio, 2013), both of which are represented within the content of e-books (Wong & Neuman, 2019). Mayer's (2005) cognitive theory of multimedia learning argues that the combination of multimedia narrations and visuals facilitates the creation of verbal and visual mental representations, which are then further developed in working memory and can be combined with prior information to form new knowledge. Dual coding theory as applied in Takacs and Bus's (2016) study states that when two sources of information (such as narration and animation) are compatible, they "can be processed simultaneously without causing cognitive overload" (Takacs & Bus, 2016, p.2). In line with Mayer's (2005) multimedia learning theory, it appears that the incorporation of narrative text accompanied by illustrations is beneficial for young children, just as images are beneficial when applied to text. The multimodal features of ebooks may facilitate simultaneous processing of verbal and nonverbal information, thereby enhancing story comprehension (Sadoski & Paivio, 2013). Reviews of apps for young children have shown that, in addition to static images, animated pictures, in particular, can be useful additions to stories (Van Daal & Sandvik, 2011; Zucker et al., 2009). Multimedia elements (e.g., extra images, animation) may be used to illustrate important events in a story or include details about the context of a word in e-books. For example, animation embedded in an e-book might better demonstrate the meaning of a verb than a static illustration.

The incorporation of engaging features (e.g., decorative illustrations) can improve student interest but may also adversely affect learning (Mayer, 2014). According to cognitive load theory (Sweller, 1988), the human cognitive system is limited and the amount of information that learners need to process affects learning new knowledge (Sweller et al., 2019). The auditory and visual channels have limited capacity in terms of working memory and too much information or unnecessary/distracting information might create a cognitive overload which interferes with learning (Sweller et al., 2011). Consequently, the limited capacity of young children's working memory and the various extraneous features embedded in e-books may lead to cognitive overload, and as a result this would interfere with children's understanding of the storyline and prevent them from absorbing important information related to the educational activity (Bus et al., 2015; Sweller, 2016). Thus, the design and the number of multimedia components in e-books are critical. In order to mitigate cognitive overload, the assessment of e-book features is crucial for determining their impact on language and literacy development. The presence of multimedia

components in e-books that are unrelated to the main educational objective can impede children's learning, as these components possess the capacity to divert attention or disrupt the learning process (Kelley & Kinney, 2017).

When comparing print books and e-books as a medium, multimedia and interactive features embedded in e-books might tax working memory compared to paper books or conversely paper books without any extra features that might require a lot more adult input in order for new learning to pass from working memory to long-term memory. Furthermore, both formats incorporate active learning, as e-books offer various interactive features, such as hotspots and dictionaries and paper books usually go hand-in-hand with adult scaffolding. Finally, the success of any implementation of these concepts would be determined by the cognitive load that the medium and story material place on young children with limited concentration span (Richter & Courage, 2017). Due to the apparent inconsistencies between these viewpoints, additional investigation is warranted to explore the discrepancies in the cognitive processing of e-books and traditional print books among young children.

E-books have a choice of facilities such as multimedia effects, hidden hotspots and built in dynamic visuals. Multimedia effects include starting and restarting the oral reading, printed text that brightens or changes colour or size while the text is narrated, music, sound effects and animation. Hotspots are devices embedded in various screen locations and are intended to provide supplementary information such as illustrations that activate animations, hotspots in text that activate the pronunciation of words or phrases, explanation of a word, an option that provides an overview of all screens, and games and songs that are more or less related to the story content (De Jong & Bus, 2003). Activating e-books that incorporate a built-in dictionary and highlighted text may provide young children with a valuable learning experience, including the acquisition of new words, as well as the development of word recognition and reading skills (Korat, 2010). Built-in dynamic visuals may also expand on the story's content beyond what appears in the original text and may aid in a better understanding of the narrative (Korat, 2010). Thus, the research suggests that quality, as it pertains to the combination of considerate and interactive multimedia features, is a factor that potentially influences literacy development. E-books are viable tools to support literacy development in early childhood, but also specific factors or conditions have the potential to impact outcomes positively or negatively. Thus, there is a need to examine and evaluate the e-books that teachers use in the class to support children's emergent literacy.

Furthermore, a significant number of studies focusing on e-books have been conducted on the basis of altered e-books (Korat et al., 2013; Smeets & Bus, 2012). These were commercial e-books altered in order to correspond to the study's needs or e-books created by the authors to carry out their research on language and literacy (e.g., Silverman, 2013; Segers & Verhoeven, 2002; Korat, 2010; Shamir, 2009; Korat & Shamir, 2012, 2008, 2007; Shamir et al., 2012). For instance, the study conducted by Korat et al. (2014a) utilised an e-book created by the researchers. The e-book was developed by scanning the pages of the print book that served as the comparative condition. To test vocabulary acquisition in the context of e-book reading, they embedded a dictionary with dynamic and static visuals into the e-book. Thus, more research needs to be done in order to assess and evaluate commercially available e-books used by children to promote language and literacy.

In conclusion, factors found to impact children's learning from e-books and story apps have been explored in several experimental studies (e.g., Bus et al., 2020; Hassinger-Das et al., 2020; Lawrence & Choe, 2021; Neumann, 2020b). The array of studies published in the past decade offer mixed results as the e-books in question incorporate various and diverse features. To that end, the present thesis, within Study 1, investigates the embedded features within e-books, which may resemble adult scaffolding, and to what extent these digital features can promote learning when compared to traditional reading of print books. After evaluating the digital skills offered in e-books, the author moves on to the second study of the thesis by selecting e-books with digital features that are considered supportive based on the findings of the meta-analysis (Study 1), in order to evaluate its effectiveness in regard to story comprehension and vocabulary learning specifically in children learning English as an additional language.

1.6 Recognising the Overarching Aims of the Thesis

The above literature outlined the different areas that e-books may have an impact on, specifically literacy development, vocabulary enhancement, increase of attention span, and child engagement. The findings are considered encouraging; nonetheless, additional investigation is warranted due to the rapid evolution of electronic stories in the market,

including new features, new platforms and eventually quite novel reading experiences. Thus, the subject of e-books as well as the advantages they provide in education is a fertile and rich ground for further research and analysis. The numerous viewpoints and attitudes that are presented in this thesis are just a starting point. As seen in the literature, researchers agree that the enjoyment of e-books by the children may just be a novelty factor, but there are also several documented cases where e-books support literacy as well as further development of language. The added functionality as well as interactivity that is available in e-books may greatly benefit children (Machado, 2015).

It is clear that growth of literacy skills is typically a vital part of the overall development of a child. Literacy skills are the foundation for developing independence, doing well at school and socialising with others. However, before the children learn how to write and read, it is imperative that the children develop the building blocks for literacy, which are the ability to listen, speak, understand, observe as well as draw (Cole, 2008). Thus, reading books to young children is one of the most important activities for developing language and early literacy skills.

All the studies included in this literature review are small-scale experimental studies. Research regarding the effects of e-books may not be numerous but existing studies cover a range of participant variables such as age, gender, socioeconomic status and bilingualism. An array of experimental conditions have been investigated such as participants working in pairs, adult reading of a print book, adult mediation of an e-book, comparison of different e-books, and the use of various e-book functions, such as hot spots and on-screen dictionaries.

Digital storybooks, including those created for modern devices such as smartphones and tablets, include a variety of enhancements that are changing children's early experiences with books (Lantz et al., 2020). The literature review presented here indicates that additional enhancements can improve the comprehension and retention of story details by children as young as 3 years old. Research findings indicate that the utilisation of animated visuals has been shown to decrease the cognitive effort needed to associate nonverbal cues with story language. This reduction in cognitive load subsequently enhances the processes of vocabulary acquisition and comprehension of stories (Schnotz & Rasch, 2005). There is evidence that animated images may be particularly beneficial for young children and children with language-delays (Bus et al., 2015). However, the inclusion of extraneous

visualisations and/or a variety of digital features might impede the learning process by depleting the mental capacities dedicated to information processing. In particular, certain digital components incorporated within storybook applications may lack a deliberate design and purpose aimed at directing attention and facilitating learning towards specific words and topics. These attractive and exciting digital features may attract children's attention to certain visuals that may be irrelevant to the story. Consequently, there exists the possibility that these e-books could hinder rather than improve story comprehension and vocabulary acquisition. Moreover, Bus et al. (2015) have provided empirical support for the notion that music and auditory stimuli have the potential to cause interference and decrease performance, particularly among children who have language deficits. The primary objective of Study 1 is to assess the effectiveness of e-books and their associated digital components in promoting or hindering the learning process.

In conclusion, well-designed e-storybooks provide scaffolding which supports young children who are developing literacy skills. Although many hopes have been raised with reference to the potential of e-book reading by young children, their quality as language and early literacy support is controversial. There is a concern that many e-books found on the commercial market are targeted more towards amusement than promoting young children's learning (Korat & Segal-Drori, 2016). Consequently, further research is needed to examine and evaluate commercially accessible e-books utilised by children to develop language and literacy.

1.7 Summary

This chapter examined the usage of e-books by young children in terms of their language and literacy development, as well as the digital aspects featured in e-books. The digital features are further examined with a meta-analysis performed in Study 1 (Chapter 4). Furthermore, when a child utilises an e-book, additional aspects of this activity, such as adult scaffolding, may be deemed essential. Chapter 2 explores teacher scaffolding during storybook reading, which is explored in both Study 1 and Study 2 of this thesis.

Chapter 2: Teacher Scaffolding During Storybook Reading

This chapter includes an overview of teacher scaffolding during traditional storybook reading as well as research assessing scaffolding while children are listening to electronic books. Two scaffolding strategies, performance and interactive, are introduced in Chapter 2 and evaluated in Study 2 (Chapter 5).

2.1 Introduction

Young children today are exposed to tablets, telephones, computers, and television from the very first months of their birth. Information is presented not only as words printed on paper but as digital images, sounds, animations, and texts. Materials such as e-books and applications are being used at schools to further teachers' goals in language and literacy development. Teachers at times may need to utilise different teaching resources to support and enhance reading. Thus, schools can provide children with innovative ways to use this technology. A number of studies have shown that digital literacies have the potential to provide young children with valuable learning experiences that enhance their learning and development (Beschorner & Hutchison, 2013; Clarke & Abbott, 2016). Moreover, technology has been found to have a positive impact on developing children's early literacy skills (Crescenzi et al., 2014; Neumann, 2014) as well as children's acquisition of a second language (Pellerin, 2013). However, the types of strategies educators can use to scaffold children's learning and understanding of the story content of e-books and print books requires further examination.

2.2 Background

According to Holdaway (1979) the "shared book experience" strategy is an interactive "reading experience" where the adult reads the story to the child and following the stories plot or memorable character, they both discuss the book's plot or topic (Holdaway, 1979; Reid, 2016). In this study, shared book reading is described as verbal alternations between the reader (i.e., adult, teacher) and the children in order to encourage the children to engage in lengthy discussions, ultimately fostering the children's literacy and language development (Dickinson & Morse, 2019). Storybook reading is a regular activity in the

preschool classroom and has many benefits for early learning (Bus et al., 1995; Dickinson & Tabors, 2001; Gonzalez et al., 2014). The adult may ask questions about the story or vice versa and, as a result, the adult is increasing the child's vocabulary, sentence structure, syntax, and print concepts (Reid, 2016). Adult-child shared storybook reading has been shown to promote the language and literacy of children considered at risk for learning difficulties (Justice et al., 2005). Zucker et al. (2013) examined the frequency (repetitions) of shared reading sessions and features (teachers' extra-textual discussions) of reading experiences within the preschool classroom with respect to children's language and literacy outcomes. Participants were 28 preschool teachers and 178 children (at risk). The researchers found that the frequency of shared classroom reading was positively and significantly related to children's receptive vocabulary development, as was the inclusion of extra-textual conversations around the text. Similarly, Aram and Biron (2004) compared the effectiveness of a joint storybook reading intervention for literacy development by using 11 children's books. The children involved were aged 3-5 years from low socioeconomic backgrounds in Israel (n=35, control=24). The children in the reading programme performed significantly better than a control group (no intervention) on measures of orthographic awareness (d=0.91), word writing (d=0.13), and listening comprehension (d=0.30).

Traditionally, shared book reading has functioned as a chance for social contact to assist children's emerging literacy skills (Dickinson & Smith, 1994), and the majority of this research has historically been conducted using traditional paperbound books. As the use of e-books increases, so does the amount of research examining book reading in an electronic format. Theories of collaborative book reading using an electronic medium are likewise grounded within sociocultural theory (Neumann & Neumann, 2014; Yelland & Masters, 2007). These theories propose that children may learn from e-books through scaffolding. The current thesis sought to better understand scaffolding with the use of different teaching styles while reading a print book and an e-book.

2.3 Scaffolding During Storybook Reading

There is an extensive body of research demonstrating the impact of storybook reading on language-related outcomes, including a number of meta-analyses that evaluated the effects of storybook reading with young children in early years classrooms (e.g., Blok, 1999; Mol et al., 2009; Noble et al., 2019). Blok (1999), for example, conducted a meta-analysis and

reviewed ten studies, published between 1968 and 1994, regarding the impact of reading to young children in schools. The dependent variables were divided into two categories: oral language and reading skills. The total effect size for the domains of oral language and reading was d = 0.63 and d = 0.41, respectively. Despite the fact that these impact sizes may appear optimistic, it is important to note that the number of included studies was relatively small, and the majority of the included studies were of poor quality, e.g., the studies did not utilise an experimental design and a common deficiency in the majority of the included studies was the lack of information on the reliability of the dependent variables. Mol et al. (2009) conducted a meta-analysis to determine the extent to which interactive storybook reading improves vocabulary and print knowledge. The authors examined 31 (quasi-experimental) studies (n=2,049 children) in which teachers were trained to encourage children's active participation before, during, and after shared book reading. The authors categorised their included studies into three groups (i.e., dialogic, interactive, and interactive with extra activities) but did not provide more information relating to the characteristics of these groups, although the dialogic and interactive teaching style have many similarities. The effect sizes for language-related outcomes ranged from small to moderate: d = 0.45 for receptive vocabulary, d = 0.62 for expressive vocabulary, and d = 0.39 for print knowledge. In addition to the meta-analyses reported above, Noble et al. (2019) examined studies conducted between 1980 and 2017 that focused on shared reading interventions for typically developing children aged 7 years or younger. While there is an impact of shared reading on language development, it is less than that found in prior meta-analyses (g = 0.19, p = 0.002) and is close to zero when control groups were active and not just following the daily school curriculum (g = 0.028, p = 0.703). These results indicate that the current evidence for the efficacy of shared reading treatments is substantially lower than previously recognised.

The existing body of literature related to storybook reading suggests that engaging in this activity is associated with a variety of positive developmental outcomes. Among the most important are advances in language (expressive, receptive) and in emergent literacy (e.g., print awareness, vocabulary acquisition, letter knowledge, phonological awareness) that are fundamental to learning to read, engagement in reading, and ultimately contributing to academic achievement in school (Bus et al., 1995; Fletcher & Reese, 2005; Mol & Bus, 2011; Mol et al., 2008; Whitehurst & Lonigan, 1998). The benefits derived from early exposure to books last beyond the early childhood and extend into the periods of

adolescence and young adulthood (Mol & Bus, 2011). Over the past several decades, numerous studies have shown preschoolers' exposure to shared reading is associated with their development of language and early literacy development (e.g., Bus et al., 1995; Hindman et al., 2012; Mol et al., 2008; Mol & Bus, 2011; Payne et al., 1994; Zucker et al., 2013). Many of these positive outcomes are mediated by the adult child interaction that occurs during reading (Bus et al., 1995; Fletcher & Reese, 2005). Adults' talk to children is more complex during storybook reading than it is in other contexts (Richter & Courage, 2017). The approach in which young children are read to correlates with the language skills they acquire from shared book reading (Zevenbergen & Whitehurst, 2003) compared to an adult reading a story to a child without actively engaging the child (Arnold & Whitehurst, 1994; Whitehurst et al., 1994). Although positive correlations have been found between variation in the quality of adult scaffolding during shared reading and children's language development (e.g., Hindman et al., 2008; Hindman et al., 2014), further research is needed to examine the effectiveness of these scaffolding techniques in relation to young children's language and story comprehension development. Hence, the second study presented in this thesis evaluated and compared two teaching styles of an adult reading a story to a group of young children.

Several factors have been reported to influence language and literacy development from storybooks, such as who reads the story (Hindman et al., 2008), the type of scaffolding style employed (Bus et al., 1995; Reese & Cox, 1999), the child's age (Hargrave & Sénéchal, 2000), and the number of times the story is read to the children (e.g., Sénéchal, 1997). During the storybook reading session, adult scaffolding may occur through the use of various approaches and techniques. Scaffolding can occur prior to reading a story, during the story, and after the story is finished. As a start, adults have the opportunity to begin a story in ways that promote and ensure children's engagement in the session prior to shared reading. A traditional approach is to begin the story by examining the front and back cover of the storybook and make predictions concerning the story plot (Hindman et al., 2012; van Kleeck, 2008; Zucker et al., 2013). The synchronisation of attention between adult and child towards an object (in this case the story) prior to reading a story is referred to as joint attention (Tomasello & Farrar, 1986). According to Hoel et al. (2020), in the context of shared reading, creating a frame of joint attention prior to the reading activity provides an opportunity to scaffold literacy acquisition and can play a crucial role in sustaining children's attention, which may aid in vocabulary acquisition and story comprehension. In
addition, joint attention can continue while reading the story with the use of "extratextual" talk during shared storybook reading (Hindman et al., 2014), as well as at the end of the story. During and after a storybook reading, teachers frequently engage in conversations concerning the story (Fisher et al., 2004; Hisrich & McCaffrey, 2021; McCaffrey & Hisrich, 2017).

An example of extratextual talk during a shared-reading intervention is the *dialogic reading* approach developed by Whitehurst et al. (1988) as a language intervention for young children. The adult/educator is trained to utilise particular strategies to engage children in story-related discussions that are intended to foster their language development (Zevenberg & Whitehurst, 2003). These include asking wh-questions (e.g., "what is that?") and openended inquiries (e.g., "what is going on on this page?") and proving feedback (e.g., by expanding on what the child has said and/or described). Interactive (or dialogic) reading, for example, entails using a series of strategies to facilitate an interaction or dialogue between an adult and a child about the story at hand (Noble et al., 2019; Whitehurst et al., 1988). This common activity promotes conversation (Hoffman & Paciga, 2014) and appears to be a particularly rich source of verbal interaction between the adult and the child (e.g., Gilkerson et al., 2017; Mol et al., 2008; Whitehurst et al., 1988).

There is a substantial body of correlational, experimental, and intervention research that provides evidence for the advantages of dialogic reading (e.g., Flack et al., 2018; Mol et al., 2008; Noble et al., 2019). The meta-analysis by Mol et al. (2008) was undertaken to investigate the additional benefits of an interactive shared book reading that prioritises active engagement by the child, as compared to non-interactive involvement. They included 16 studies that reported vocabulary as an outcome measure and whose (quasi-) experimental conditions included a dialogic reading intervention group and a reading-as-usual control group with parents and their children. For expressive vocabulary their results revealed a moderate effect size d=0.59, with a sample size of 322 participants, favouring the dialogic reading intervention group (95% CI = [0.44, 0.75]; SE = 0.08; p< .001). However, the researchers noted that the effect size significantly decreased in older children (aged 4 to 5 years; n=275; d= 0.14; SE = 0.07; 95% CI = [-0.10, 0.37]) or in those who were at risk for language and literacy difficulties (n=208; d=0.13; SE=0.11; 95% CI = [-0.08, 0.35]). In general, the studies included in this meta-analysis have demonstrated that engaging in dialogic reading has the potential to influence the home literacy practises of

families with children between the ages of 2 and 3 years. However, it does not appear to have the same impact on families with children who are at a higher risk of experiencing academic difficulties. In another meta-analysis conducted by Flack et al. (2018), a total of 38 studies including 2,455 children were examined to explore the potential moderating effects of reading styles, story repetitions, and associated variables on children's word comprehension. The findings of their study demonstrated that the implementation of dialogic reading, such as the utilisation of gestures, offering definitions, or engaging children in questioning during shared storybook reading, had a notable impact on the acquisition of new vocabulary by children. The use of dialogic reading techniques yielded a statistically significant increase of 1.22 words in children's vocabulary acquisition compared to non-dialogic reading approaches (p < .001). Therefore, in the context of acquiring vocabulary through storybooks, the manner in which stories are read holds greater significance than who reads them. The most recent meta-analysis conducted within the scope of dialogic reading was the study by Dowdall et al. (2020), aimed at evaluating the effects of dialogic book-sharing interventions on the language development of young children (ages 1-6 years). Their meta-analysis included 20 studies and their results revealed that shared reading interventions with parents produced a small positive effect size for expressive (n=1.664; d=0.41; 95% CI = [0.20, 0.61]) and receptive vocabulary (d=0.26; d=0.26; d=0.41; 95% CI = [0.20, 0.61])95% CI = [0.12, 0.40]) of children. This suggests that interactive book sharing may be effective in enhancing both receptive and expressive vocabulary skills; however, these results were compared to control conditions that did not involve book reading (i.e., no intervention). Similarly, the meta-analysis of Flack et al. (2018) did not include a control group. In general, the results from various meta-analyses yielded inconsistent findings. Mol et al. (2008) reported that the beneficial effects of storybook reading were less evident in studies involving older preschool children (4-5 years old). Conversely, Dowdall et al. (2020), Flack et al. (2018) and Noble et al. (2019) suggested that the effects of storybook reading do not appear to differ based on children's age. In general, the existing body of literature indicates that dialogic reading is a generally effective approach for fostering receptive and expressive vocabulary. Hence, it is essential to explore potential variances in outcomes when comparing different scaffolding techniques employed during shared storybook reading with young children. The purpose of Study 2 is to examine and contrast two different scaffolding strategies, namely interactive and performance teaching styles, with the aim of assessing their efficacy in facilitating language acquisition among young children.

Overall, research has shown that during adult-child storybook reading, teachers are more likely to engage children in discussions relating to the instruction of new vocabulary (Dickinson et al., 2014), and this is supported by sociocultural theory (Vygotsky, 1978). A recent study by Clemens and Kegel (2021) found that engaging in book reading serves as a highly effective activity of boosting language skills in toddlers as early as 9 months of age. Furthermore, the study revealed that this particular activity evokes a greater degree of language usage when compared to other common parent-infant activities with languagerelated goals. The content of the storybooks appears to offer a valuable resource for parents, educators, and young children, stimulating rich language from both adults and children (Sosa, 2016). Furthermore, storybook reading allows adults to personalise story material by linking the text to the children's prior experience, enabling them to elicit conversations revolved around their own personal experiences, which improves story comprehension (Hoffman & Paciga, 2014), language exposure and the development of narrative and conversation skills (e.g., Grolig et al., 2019; Morrow, 1988). Preschool teachers apply these interactive techniques while reading to young children, and although discussion and questioning during shared book reading have been proven to be beneficial to children's literacy and language development, the amount and type of discussion used by adults is found to vary widely. Despite the substantial research base on storybook reading, there are several remaining areas to be investigated. The amount and type of dialogue initiated by the teacher while reading, as well as the manner in which a teacher reads, are two crucial parts of shared book reading that have a significant effect on the outcomes for children (Teale, 2003).

2.4 Scaffolding and Literacy Development

Scaffolding involves education-oriented support, discussions, and interactions that occur between a teacher and a learner with the aim of facilitating learning. It is closely connected to Vygotsky's social constructivist view of learning and his concept of Zone of Proximal Development (ZPD, Vygotsky, 1978) as well as the constructivist learning theories of Bruner (1966), Dewey (1923), and Piaget (1973). Constructivism's central idea is that learning is constructed, and learners develop new knowledge by building on their prior knowledge and experiences. According to Vygotsky (1978), learning takes place within the ZPD, that is, an area in which cognitive development is still in progress. The ZPD delineates the gap between a student's independent capabilities and the level of support required to accomplish a learning activity successfully. Students experience success in the ZPD, which is facilitated by the provision of instructional scaffolding, which is widely recommended as a highly effective and varied constructivist teaching strategy (Clack & Graves, 2005). By employing Vygotsky's theoretical framework, educators have the ability to provide instructional scaffolding by adapting the amount of assistance provided throughout a teaching session to align with the student's present level of performance (Verenikina, 2008). A constructivist approach fosters a collaborative learning environment in which teachers and students co-operate and share their knowledge (Nicaise & Barnes, 1996). During shared book reading, which is a student-centred activity, both the child and adult are active participants in the construction of a dialogue surrounding the storybook (Whitehurst et al., 1988). Research greatly supports adult guidance while reading/listening to a storybook and the importance of adult scaffolding during storytelling to develop young children's literacy skills (Homer et al., 2014; Korat et al., 2011; Moody et al., 2010; Rvachew et al., 2017).

Studies on scaffolding during storybook reading fall into two broad categories: descriptive studies (i.e., not experimental) focusing on how adults read storybooks to children (e.g., Hindman et al., 2012; Zucker et al., 2021) and experimental studies analysing reading styles to determine how adults may read most effectively with young children (e.g., Wasik et al., 2006; Strasser et al., 2013). Walsh and Hodge (2018) conducted a review assessing the questioning techniques of 20 experimental studies during shared book reading with young children. The studies were examined in terms of their focus, quality, and questioning techniques. The reported studies appeared to focus on the effect of questioning techniques on language and literacy development, while the majority of studies used a low level of cognitive demand questioning to scaffold children's learning. Indeed, findings from other studies (e.g., Dickinson & Keebler, 1989; Dickinson & Smith, 1994; Hindman et al., 2012; Zucker et al., 2010) suggest that early childhood educators typically pose simple questions that require a minimal amount of cognitive effort to respond and that elicit one-word responses (Beck & McKeown, 2001).

In addition, another challenge that arises in research is when to ask the questions and whether the questions themselves may divert the child from the storyline. Wasik et al. (2006) undertook a language and literacy intervention with 16 teachers and 207 children,

in which the teachers were trained to use certain discussion techniques during their storybook reading. The scaffolding techniques were based on the research of Dickinson and Smith (1994) and Snow (1983), which demonstrated the following: engaging and encouraging active listening, demonstrating rich language, and proving feedback. Wasik et al. (2006) found a correlation between the timing of questions and vocabulary development, with questions asked before and after reading having a stronger influence on receptive and expressive language skills than questions asked while reading. However, Strasser et al. (2013) in their study with 72 children showed that explaining new vocabulary and unfamiliar words during storytelling did not hinder children's story comprehension. Therefore, further investigation is required to determine whether or not this association exists. Although the importance of scaffolding during storybook reading has been established (Van de Pol et al., 2010; Zucker et al., 2020), one area that requires further consideration is how different reading styles influence children's vocabulary and story comprehension; thus, researchers contend that how educators read to children is as important (Flack et al., 2018; Morrow et al., 1990; Senechal, 1997; Teale, 2003). Therefore, it is essential to determine which teaching style is deemed more beneficial for the language and literacy development of young children, and specifically in the context of e-books.

2.5 Interactive and Performance Teaching Styles

Although shared storybook reading is a well-researched practice with young children, few studies have examined the scaffolding styles that teachers may use in combination with different types of media. The second study intends to examine the impact of e-books versus print books on children learning English as an additional language in terms of vocabulary and story comprehension. Study 2 also investigates the reading styles of teachers: the *interactive style* which engages children in discussion of the story as they read, and the *performance style* characterised by few interruptions during the storytelling session, accompanied by extended pre- and post-story discussion. The timing of discourse before, during, or after reading a storybook might be significant, especially for young children with limited language skills who may benefit from adult pauses while reading to scaffold learning (Gathercole & Baddeley, 2014; Reese & Cox, 1999).

The two teaching styles (a) interactive style and (b) performance style are described as follows in the literature: the *interactive reading style* is considered a common approach (e.g., Baker et al., 2020; Baker et al., 2015; Flack et al., 2018; Walch, 2016). It is a

technique in which the reader frequently stops the story to engage the audience in the reading process (Arnold & Whitehurst, 1994; Whitehurst et al., 1998). Reading that involves such intensive interaction with children has been demonstrated to be particularly beneficial for vocabulary acquisition (Whitehurst et al., 1994; for reviews see Flack et al., 2018; Mol et al., 2008; Noble et al., 2019). In interactive storybook reading, an adult reads aloud a story, explains its content and vocabulary, and includes open-ended questions in a process of feedback and validation (Nevo & Vaknin-Nusbaum, 2018). It has been established that children learn about their environment by active participation in retelling and discussing the storyline, characters, and the vocabulary involved in the story (Pollard-Durodola et al., 2016). This type of teaching style employs the support of adults to cater to the higher-level cognitive requirements of its young audience by drawing parallels between the story and the children's own experiences and emotions (Zeece, 2007). Consequently, it should not come as a surprise that reading aloud from books with questions and answers is a common and necessary part of preschool literacy intervention programmes designed to foster language and literacy development. When adults employ the *performance teaching* style, they emphasise significant words or expressions and change their voices, facial expressions, and body language to impersonate the characters in the story. However, there are few breaks in the text for questioning like there would be in an uninterrupted performance – instead, there are extended conversations before or after the reading (Dickinson, 2001; Dickinson & Smith, 1994).

Greene Brabham and Lynch-Brown (2002) conducted a study with 117 1st graders (6-7 years old) and 129 3rd graders (8-9 years old) that included three conditions with equivalent level and quantity of comments and questions: (1) minimal extratextual talk – just reading condition, (2) an after reading performance style, and (3) during reading interactive style. Teachers were trained and guided by a scripted strategy to use one of the styles when reading two informational storybooks to pupils. This experimental comparison of reading-aloud styles supports the hypothesis that vocally mediated, interactional, and performance reading-aloud styles are more impactful than an adult reading a story without discussion for vocabulary learning. Interactional reading facilitated vocabulary acquisition more than performance reading. Findings confirm that teacher scaffolding and student discussions significantly aid students' acquisition of vocabulary, as well as their ability to construct meaning from storybook reading. Sociolinguistic theory's emphasis on adults' roles in supporting children in constructing meaning from written texts, as well as the importance

of social interaction and scaffolding in language and literacy development and education, were validated by the significant effects of reading aloud styles with verbal mediation on vocabulary gains (Ninio & Bruner, 1978; Vygotsky, 1986). Morrow et al. (1990) summed up differences between just reading and socially mediated reading by explaining, "Reading to a child is not sufficient for maximum literacy growth. It is the talk about books that surrounds the reading that seems to be the key" (p. 268).

In another study by Van den Brook et al. (2011) in the Netherlands, 40 children aged 2-3 years and 42 3rd grade children aged 8-9 years old were randomly assigned to one of two experimental conditions. Depending on the condition, questions were asked either during or after listening to the story. Stories and conditions were counterbalanced in order to ensure that each story occurred in every condition across all children. Results showed that the children benefited more from causal questioning techniques during reading as compared to after reading, perhaps because of reduced working memory demands (Van den Brook et al., 2011). Dickinson and Smith (1994) examined correlations for 25 preschool *teachers' reading styles* and their students' scores on story retellings and the Peabody Picture Vocabulary Test-Revised (PPVT-R). Analysing one videotaped readaloud session from each classroom, the researchers found that 10 teachers used a performance style, limited discussion during reading, and followed up with extended discussion after the reading performance. Fifteen teachers read with interactional styles but exhibited two different approaches. Dickinson and Smith identified 10 of these teachers as "didactic interactional" because they asked children to respond to questions, repeat factual information, and recite parts of the text in chorus during the reading. The other five interactional teachers were called "co-constructive interactional" by the researchers because they had children predict, analyse, generate word meanings, and draw conclusions as they read. According to Dickinson and Smith, the most beneficial reading style for students' comprehension and vocabulary development was a performance style of reading, in which there were few interruptions during the read-aloud, but more time was spent on discussion afterwards.

Sezer et al. (2021) performed a study in a Turkish language setting to examine three readaloud styles using children's picture books: just reading, performance reading, and interactional reading, on the text reading fluency and reading comprehension of 152 first grade and third-grade students. The just reading approach consists of reading the story without additional analysis. During the performance reading, the teacher and pupils engage in a discourse, albeit a limited one. After the story concluded, the reader engaged the pupils in a discussion regarding the story. By asking questions, the reader encouraged pupils to relate the narrative to their own experiences. Their findings showed that while the first graders' text reading fluency scores were not affected by the read-aloud approaches, their story comprehension scores were affected in favour of the group in the just reading condition. The third-grade students' reading fluency and story comprehension did not vary by the read-aloud method. According to the researchers, these findings do not provide a clear picture of the effectiveness and differences of various read-aloud approaches for children's picture books.

Reese and Cox (1999) suggested that the style of reading should take into consideration the child's initial level of skill. In their experiment, Reese and Cox (1999) investigated the relative advantages of three different teaching styles of extratextual talk during shared reading on children's language development in a six-week intervention. Preschoolers were randomly assigned to one of three intervention groups, in which they engaged in one-onone shared reading with an adult who employed either a describer style (describing illustrations), a comprehender style (focusing on the content of the story), or a performance style (similar to the comprehender style but with discussion occurring after the story had been shared). The children's vocabulary, print awareness, and story comprehension skills were evaluated both before and after the intervention. The researchers found that the describer style was the most advantageous to children's vocabulary and print awareness. However, after the children's initial vocabulary skills were considered, the describer and performance-oriented styles resulted in larger vocabulary improvements. Study 2 attempts to contribute to this line of work on comprehension and vocabulary development by examining the different types of storybook scaffolding offered by a teacher and their effects on the literacy development of children learning English as an additional language.

2.6 Scaffolding and E-books

Research that examines adult–child interactions during shared storybook reading has identified key behaviours that appear to facilitate language learning (Zucker et al., 2021; Korat et al., 2011; Moody et al., 2010; Rvachew et al., 2017). However, adults may be less likely to make use of these facilitative behaviours during experiences with digital storybooks (Parish-Morris et al., 2013). E-books may be suitable as an addition to, rather

than a replacement for, high-quality adult–child interactions around storybooks (Kelley & Kinney, 2017). Scaffolding is also important for e-book reading. It has been found that young children benefited more from language acquisition while reading an e-book with adult scaffolding than when reading the same book independently (Segal-Drori et al., 2010; Strouse et al., 2013).

There are currently a number of experimental studies with young children that test the effect of independent e-book "reading", and these studies suggest that e-books may be a useful tool for fostering language and literacy development (e.g., Savva et al., 2022; Takacs et al., 2015). However, little is known about how children interact with adults when using these devices. Technology like e-books may provide a vast array of narratives and information for children to discover on their own; however, children may be exposed to material that is too complex for them to fully process without the help of an adult. In research comparing book reading in electronic and paper formats, it is common for an adult to read the print book aloud to the participant, while the electronic device generally utilises audio narration to "read" the book to the participant. A study by Strouse and Ganea (2016) suggests that not having the necessary vocabulary to understand the concepts presented in a non-fiction e-book may hinder a child's comprehension in the absence of appropriate scaffolding by an adult. Their study examined whether adult prompting during the reading of an e-book improved children's comprehension of a scientific concept. Ninety-one 4year-olds listened to a non-fiction story in three conditions: (a) e-book, (b) e-book read by a researcher, or (c) print book read by a researcher. Although children were able to transfer the knowledge from the story and apply it to a problem, for children with lower vocabulary levels, performance was higher in the conditions with a researcher reading with them (Strouse & Ganea, 2016). This study shows that adult scaffolding is important for learning to occur in situations where children may be exposed to narratives and information that they may not have the vocabulary to fully grasp. An experimental research examining parent co-use of interactive media with a sample of 40 low-income 4- and 5-year-old children had comparable results. On post-tests, children in the parent-child condition performed significantly better than children who interacted with media independently (Griffith et al., 2022).

Scholars support adult scaffolding during e-book reading as research has demonstrated that young children learned more language from an e-book supported by adult scaffolding than

storybook reading without scaffolding (Savva et al., 2022; Segal-Drori et al., 2010; Strouse et al., 2013). However, traditional adult storytelling scaffolding has inevitably changed over the last ten years due to the inclusion of e-books. Young children today are increasingly reading e-books without adult mediation. For instance, 80% of parents say their child aged 5 to 11 years interact with a tablet computer or smartphone independently (Auxier et al., 2020). E-books provide a diverse array of digital attributes that may resemble the adult-child scaffolding process, therefore rendering e-books a potentially valuable educational tool for young children (Moody, 2010). Thus, the first study of this thesis aims to analyse the impact of these features on children's learning.

The constructivist approach to learning best supports this major change, since it enables children to actively create their knowledge via their own initiative while engaging with ebooks through reading or listening (Haas et al., 2017). Children are able to actively engage and build new knowledge as they combine previous knowledge, experiences, skills and the use of digital features to comprehend the story's content and learn new vocabulary. For example, when engaging with an e-book the computer may narrate and highlight the text to support vocabulary and print awareness; it may provide animation to support comprehension and use vibrant illustrations to engage the young reader both visually and audibly (Kayaoglu & Akbas, 2011). Rvachew et al. (2017) compared shared reading interactions with e-books versus paper books, with an adult reader for the development of emergent literacy skills. Their study examined twenty-eight children from kindergarten participating in an interactive reading style offered by an adult reader versus an animated e-book linking words from the text with animated illustrations. The e-book condition exhibited higher levels of emergent literacy knowledge compared to the print book condition, particularly for children with limited letter knowledge. Therefore, e-books intended to promote supportive adult reading practices may increase emerging literacy skills, particularly in children who enter school with poor letter knowledge. Written text combined with synchronised dynamic graphics and sounds may enhance story content and aid children's mental imaging of the story (Bus et al., 2015; Korat, 2010).

According to Altun (2018), e-books have the potential to enhance information processing, promote learning, and provide children with valuable experiences that contribute to constructive learning. Consequently, e-books enable children to engage in scaffolding, as proposed by Bruner (1966). Ihmeideh (2014) compared the results of an e-book and a print

book with adult support in both conditions in a Jordanian preschool with a total of 92 children in order to measure the children's development in print awareness, vocabulary, alphabetic knowledge, and phonological awareness skills. The results showed that the children in the e-book condition outperformed those in the control condition. Korat et al. (2022a) conducted an intervention in kindergarten with 103 children (aged 5–6 years) from LSES families, using an e-book for vocabulary enrichment. The following three conditions were evaluated: (a) children read the e-book with a dictionary and the teacher's support, (b) children read the e-book with the dictionary independently, and (c) children read the e-book without a dictionary (control). Children who read the e-book with the dictionary and the teacher's support acquired more vocabulary in all three tested measures (receptive understanding, word explanations and word production) and outperformed the children who read the e-book independently with the dictionary, as well as the control group. These achievements were maintained after a month. The researchers support that the distinctive components of the e-book, combined with the adult scaffolding within the classroom context and curriculum, support the learning of new words.

There are limited studies comparing the effects of adult scaffolding in both print and ebook condition within a class context (see Savva et al., 2022). Most studies in which an adult is present during the intervention consist of the adult only reading the narrative to the child (either print or e-book condition); or assisting the child in matters pertaining to how the participant should operate the digital devise (e.g., Korat et al., 2014a) while not employing story-related language to scaffold children's learning (Chiong et al., 2012; Parish-Morris et al., 2013). Children are also reported to discuss more about device-related content rather than story-related content while reading e-books in comparison to print books (Richter & Courage, 2017). As a result, early childhood teachers play a critical role in the integration of technology in literacy instruction. Integrating technology into daily literacy instruction can offer rich, engaging, and creative experiences that expand young children's understanding of what it means to be literate in today's world. A deeper examination is needed to determine how teacher-child talk during shared reading of ebooks versus print books can positively engage and scaffold language and literacy learning in the early years setting.

2.7 Summary

This chapter has explored teacher scaffolding during print book reading and e-book

reading, as well as two scaffolding techniques, the performance and interactive style. Overall, e-books and storybook applications provide teachers with possibilities to promote shared reading among young children. Numerous studies have investigated the educational advantages of youngsters reading digital books. Less is known, however, about how young children engage with and learn from e-books, as well as how teachers facilitate these early reading experiences, particularly for children whose native language is not the one read in the e-book. Therefore, this thesis explored the abovementioned teaching styles while children were listening to an e-book and print book for children learning English as an additional language (EAL) from print books and e-books, which is explored in Study 2 of this thesis.

Chapter 3: Young Children Learning English as an Additional Language (EAL) and Storybooks (Print & E-books)

3.1 Introduction

English language is one of the most widely spoken languages in the world whilst it is the official language of more countries than any other language (Rao, 2019). Learning to speak more than one language is associated with numerous advantages (e.g., executive functioning, analytical thinking) (Bialystok & Craik, 2010) and an increasing proportion of children around the world are growing up learning multiple languages. These children are educated in English at school, yet they have been exposed to a language other than English at home since birth (Department for Education, 2015). Children who enter school with limited English vocabulary are at risk of underachievement as vocabulary skills lay the foundation for reading, which is essential for later subject area learning (Figueras-Daniel & Barnett, 2013) and the root of academic failure (Durham et al., 2007). Education is a long-term endeavour whose results continue to impact life outcomes. The primary objective of the Department for Education (2021) Statutory Framework for the Early Years Foundation Stage is to teach children the core skills upon which they will construct their future education as early school years are the foundation for academic outcomes (Department for Education, 2021). The most essential of these skills are language and literacy development. Consequently, it is not surprising that the majority of research evaluating interventions for children learning English as an additional language (henceforth EAL) have focused on the development of these essential developmental skills in children. It is, therefore, the purpose of this chapter, firstly, to describe the profile of young EAL children in Cyprus (where Study 2 took place), and secondly, to explore reading of print books and e-books, with and without support, and how digital elements contained in e-books might help children progress.

According to the UK Department for Education (2015), a child's first language is the language he or she has been exposed to since infancy. In the literature, learners for whom English is not their first language are referred to by a variety of terms including: Bilingual Student, Second Language Learners (SLL), English as a Second Language (ESL), Limited

English Proficient (LEP), Language Minority Student, English Learner (EL), and Learning English as an Additional Language (EAL). I propose that English as an Additional Language (EAL) may be the most appropriate term for this thesis as participants in the experimental study were taught English at school, although the majority of them spoke more than one language at home. Furthermore, in North America, individuals whose first language is not English are commonly referred to as English as Second Language (ESL) or English Language Learners (ELL). The term English as an Additional Language (EAL) is predominant in the United Kingdom (Siegel, 2016), and will be used throughout this thesis when referring to learners whose first language is not English; for some children, English may be their third or fourth language.

Children learning EAL are not a homogenous group; they reflect a wide range of linguistic, cultural, ethnic, and educational characteristics (Demie, 2018). Reading in English is more difficult for pupils who are simultaneously learning to speak English than for those who are already fluent in the language (Goldenberg, 2020). In addition to mastering English decoding skills, EAL children must develop adequate English language skills to understand what they are reading (Pico & Woods, 2023). Language assistance may be especially useful for this population. Many EAL children may attend school with poor English language skills, notably vocabulary knowledge, because their exposure to English prior to school entry varies (e.g., Mahon & Crutchley, 2006). According to statistics from the UK Department for Education, 20.2% of pupils in UK primary schools are learning English as an additional language (Department for Education, 2019). In the early years of schooling in the United Kingdom, national assessments of language and literacy demonstrate a persistent achievement disparity between EAL children and their monolingual Englishspeaking classmates (Strand et al., 2015). Undoubtedly, there is a need to provide support for EAL children's early language and literacy development starting as early as their first entry at school.

3.2 Characteristics of EAL Learners

Globalisation has brought a rise in cultural and linguistic heterogeneity in current educational settings (Scarino, 2014). Policymakers and educators have a vested interest in the topic of EAL; however, there is a lack of data on how to support this group of children when included in the mainstream classroom who are not native English speakers and who

speak a non-English language at home in English-speaking schools.

If children learning EAL differ from their monolingual peers, it is essential to examine the difficulties they face as well as any advantages they may have internalised as a result of their acquisition of more than one language. Festman (2021) argues that learning a new language is aided by the child's current linguistic repertoire, which includes the child's prior knowledge of language in the form of memorised vocabulary, grammatical structures, sound patterns, etc. Similarities between the new language and a previously taught one has been hypothesised to facilitate learning (Hammarberg, 2001). For children that are learning a third language it has been seen that words and phrases from both other languages have been incorporated into the target language (Rothman et al., 2019). There is also the fact that there seems to be a window of opportunity within which a child can acquire a second language and use it fluently (Lenneberg, 1967), which makes the early years stages quite critical for these young children that enter an English-speaking school but not knowing a word of English. Another factor to consider is that children under the age of eight years are not yet completely developed cognitively, and hence lack key abilities (such as the ability to generalise, abstract, or form inferences) that may be helpful for adults learning a second language (Grosjean, 2010). Cummins (1984) found that EAL learners need over five years to become fully academically proficient in their second language. Determining whether the language difficulties exhibited by children with EAL result from a lack of language exposure or an underlying language impairment is a significant difficulty for practitioners (Bedore & Peña, 2008; De Lamo White & Jin, 2011; Kohnert, 2010). Consequently, there are several factors to consider while educating EAL learners and assisting them in the acquisition of a new language, especially when early years educators may need to utilise a variety of contemporary materials to assist and develop children's literacy and language skills. Less research exists on the component processes that drive the vocabulary development of children learning to speak in English as an additional language (EAL), notably in Cyprus, where English language learners have a heterogeneous linguistic background.

3.3 The Case of Cyprus

Cyprus is an island nation in the Eastern Mediterranean Sea. It is the third largest and most populous island in the region and located east of Greece. Cyprus itself is de facto partitioned into the south, which is effectively under control of the Republic and accounts for 59% of the island, while the northern region is handled by the self-declared Turkish Republic of Northern Cyprus. The North is recognised only by Turkey. The international community considers the northern region as part of Cyprus which is occupied by Turkish forces.

According to the preliminary data released by the Statistical Service of Cyprus regarding the population and housing census of Cyprus conducted on October 1, 2021, the population residing in the Government controlled areas of Cyprus was 923,272 individuals. Among this population, the total count of foreign nationals was determined to be 193,300, constituting approximately 21.1% of the total population. Regarding the population recorded by country of citizenship, the majority of the foreign nationals come from Greece, the UK, Romania, Bulgaria, the Philippines, Russia, Sri Lanka, Vietnam, Syria, Ukraine, India, Poland, Georgia, and Germany.

The right to education guaranteed by Article 20 of the Constitution of Cyprus (1960, April 6) does not apply just to residents of the republic, but also to the children of those who have obtained a permit to stay in Cyprus. Parents may enrol their children at a school within their educational zone, regardless of whether or not they are foreign nationals. Consequently, the right to education is guaranteed by the constitution of the Republic of Cyprus, and all pupils may enrol in public schools. Depending on the preferences of their parents, children may attend either a private or public school. The primary differences between the two schools are as follows: lessons in the public schools are taught in Greek and are provided at no cost. However, in addition to other subjects, children also learn basic English. There are also private schools in Cyprus, and each requires tuition fees. The majority of private schools teach English as their primary language and adhere to the National Curriculum of England. Educators in Cyprus commonly encounter various student populations in the classroom. In many instances, children enrol in preschool in Cyprus with no prior knowledge of English, and they may have considerable language knowledge gaps.

Similarly to the situation in Cyprus, EAL learners in England are a growing and diverse population, comprising little over 21.2% of primary school students (Department for Education, 2019). These children have different levels of English language competency upon arrival to school, ranging from complete fluency to new to English. In the United

States, around 21% of children and adolescents aged 5 to 17 years speak a language other than English at home (U. S. Census Bureau, 2021). According to Jiménez-Castellanos and Garcia (2017), about one in five students in the United States speak English as a second language. Thus, statistics suggest that a considerable proportion of young people nowadays are exposed to more than one language during their childhood (Crystal, 2003). Studying the effects of how to teach young EAL learners has implications for education policy not only in Cyprus but globally. As EAL students are educated alongside their non-EAL English-speaking peers, it is expected that English will be acquired solely through curriculum engagement and interaction with peers (Costley, 2014). However, EAL learners experience considerable challenges in becoming proficient readers and frequently struggle with developing their emerging literacy skills upon entering school, compared to their peers (Fitton et al., 2018). In addition, according to studies conducted in the United Kingdom, EAL learners typically perform worse than their first language English-speaking peers, on measures of expressive and receptive vocabulary, grammar, and story comprehension, even after five years of exposure to English from school entry (Burgoyne et al., 2011; Hutchinson et al., 2003). It is essential to note that the difference in performance on language assessments between monolingual and bilingual children depends on both the duration and frequency of language exposure (Bedore et al., 2012). Indeed, Bedore et al. (2012) found that Spanish-English bilingual children who were exposed to English at least 75% of the time performed comparable to children who were exposed to English 100% of the time on grammar and vocabulary assessments, indicating that more may be done in the school context to meet the educational needs of children learning EAL.

3.4 Storybook Reading and Scaffolding for Children Learning EAL

According to meta-analyses on the effects of storybook reading on language-related outcomes, the current evidence supporting the efficacy of shared reading interventions is far weaker than previously believed (e.g., Noble et al., 2019). Reviews described in Chapter 1 and Chapter 2 regarding storybook reading did not include research involving EAL learners and storybook reading; the studies were confined to those having fluent English speaker participants; in other cases authors made no mention of the language barrier of the participants.

This gap was identified by researchers, and a meta-analysis and two systematic reviews

were published recently (Fitton et al., 2018; Kennedy & McLoughlin, 2023; Pico & Woods, 2023). Fitton et al. (2018) examined in their meta-analysis how shared book reading affects the English language and literacy skills of young children (students up to 12 years of age) learning English as a second language. The final analysis included 54 studies of shared reading conducted in the United States, including both home- and school-based interventions. The primary analyses indicated that shared book reading interventions appear to affect EAL's English language and literacy skills positively. The overall combined effect size in the current study for the impact of shared book reading on EAL learner's outcomes was g = 0.28 with large heterogeneity, $Q_E(53) = 253.78$, p < .001, $I^2 = 79.12$. This meta-analysis, however, only included research conducted in the United States, an English-speaking country.

More recently, two systematic reviews were published relating to young EAL learners and shared storybook reading (i.e., Kennedy & McLoughlin, 2023; Pico & Woods, 2023). Pico and Woods (2023) in their systematic review synthesised findings from studies on the effects of shared book reading with Spanish-speaking children, who are simultaneously developing their English language proficiency in school settings in the United States. The objective of the review was to systematically analyse peer-reviewed experimental studies that investigated the impact of shared book reading on language-related outcomes for Spanish-speaking bilinguals. They identified 17 relevant studies, primarily examining vocabulary outcomes, with mostly medium to large effect sizes found on researcher designed measures. This review found positive effects of shared book reading on language outcomes for Spanish-speaking emergent bilinguals, particularly for vocabulary development.

In terms of scaffolding during shared storybook reading, the systematic review by Kennedy and McLoughlin (2023) demonstrates how scarce the literature is regarding the evaluation of adult scaffolding during storybook reading with young EAL learners as this was the only review found on adults scaffolding (specifically dialogic reading) for EAL learners relating to their learning from storybook reading. The researchers searched for articles published between 1988 to 2020 to include in their review. Six articles met their inclusion criteria investigating the effectiveness of dialogic reading in developing the emergent literacy skills of EAL learners (i.e., Brannon & Dauksas, 2014; Cohen et al., 2012; Correa et al., 2015; Huennekens & Xu, 2010; 2015; and McCabe et al., 2010). During dialogic reading,

children are encouraged to engage in conversation and interact with the reader in matters relating to the story's characters, the content of the story and personal experiences (Cohen et al., 2012). Overall, the studies indicated that storybook reading interventions are effective for developing EAL learner's vocabulary skills.

However, how a shared storybook intervention is carried out is crucial, especially for children where their first language is not the language narrated in the storybook. Simply exposing children to words by reading aloud a story is probably insufficient to encourage language development (Pico & Woods, 2023). During storybook reading, Chlapana and Tafa (2014) studied the effects of direct and interactive instruction on the vocabulary development of immigrant kindergarten children living in Crete. A total of 87 children, ages 4 to 6 years, were recruited from 12 public kindergarten classrooms. These children were learning Greek as a second language. The children were randomly assigned to two experimental groups and one control group. In the first experimental group, children were presented with brief explanations of target words through direct instruction. In the second experimental group, children actively engaged in discussions pertaining to target words, employing interactive instructional strategies. The findings of the study indicated that interactive instruction was more effective in facilitating the acquisition of target vocabulary compared to direct instruction. Collins (2010) investigated the effects of rich explanations and sophisticated vocabulary learning from storybook reading. Rich definitions consisted of pointing at the image of the target word, offering a broad meaning of the term, providing a synonym, making a gesture of the word when suitable, and utilising the word in a context other than that being described in the story. The author assigned 80 preschoolers to experimental (stories with rich explanations) and control (stories without explanations) conditions and found that rich explanations contributed significantly to EAL learners word acquisition from stories. Given the language-specific nature of vocabulary acquisition, it is crucial to determine the most effective ways for developing EAL leaners skills during storybook reading for vocabulary development as well as story comprehension.

A few studies have examined the effects of an adult offering translations and explanations in a different language than the story. The results of these studies, however, are mixed since they have often only examined results in a single language (e.g., Spanish: Fitton et al., 2018; Leacox & Jackson, 2014; Lugo-Neris et al., 2010; Pico & Woods, 2023; Ulanoff & Pucci, 1999). Furthermore, in many classrooms, as is the situation in Cyprus, students who attend public or private schools come from diverse backgrounds, while their teacher does not speak their native language. Moreover, the majority of children that enter the preschools have little to no English skills, and their vocabulary is quite limited. Therefore, the educator should explore other scaffolding approaches that may be implemented within every class, regardless of the languages spoken by the children.

3.5 E-books and Young Children Learning EAL

E-books have potential for facilitating children's engagement (Richter & Courage, 2017) and fostering emergent literacy skills (Shamir & Korat, 2007), phonological awareness (Chera & Wood, 2003), as well as comprehension and vocabulary skills (Smeets & Bus, 2015). Well-designed e-books may support EAL learners, as e-books show enormous potential as they may stimulate readers' visual, aural, and even kinesthetic senses to absorb and comprehend a story with the support of the multimedia and interactive features included in e-books, such as hotspots, sounds and animation (De Jong & Bus, 2002, 2004; Neuman, 1997; Verhallen et al., 2006).

E-books have been suggested as highly efficient for enhancing vocabulary acquisition among children (Silverman, 2007, Verhallen & Bus, 2010; Verhallen et al., 2006), and they provide opportunities for teachers to support young children's shared reading experiences (Neumann & Merchant, 2022). Sun et al. (2019) evaluated an animated e-book with 102 preschoolers in Singapore learning Mandarin between the ages of 4 and 5 years. The researchers employed four conditions: (a) animated e-book, (b) static e-book with sound, (c) static e-book only, and (d) a control condition where children played a mathematics game on an iPad. The study examined the impact of the intervention conditions on children's target word acquisition and story comprehension. The findings of the study indicated that children who were exposed to the animated condition had superior performance in terms of target word production and storytelling compared to the rest of the conditions. No significant variations were seen between the two static settings. There were no consistent differences between the two static conditions. The authors stated that their study focuses on Mandarin, which is one of the "notorious" languages that is difficult to learn as a second language, and their positive findings from animated e-books suggests that such a reading format may be beneficial for EAL learners to acquire this difficult language. Silverman and Hines (2009) conducted a study that examined a multimedia-enhanced story on vocabulary outcomes with English-language learners and non-English-language learners in pre-kindergarten through second grade. While non-English-language learners had no added benefits from media enhancements, there were positive vocabulary gains for English-language learners in both target word and general vocabulary knowledge. Verhallen and Bus (2010) conducted a study on low-income immigrant children living in the Netherlands and learning Dutch as a foreign language. In their experiment, they examined the effects of animated e-books on vocabulary acquisition of 5-year-old children. The story was presented with either static or video images. Reading the story aloud, with the addition of video, was found to be especially effective for vocabulary acquisition (g=0.36). In a similar study, Verhallen et al. (2006) investigated the effectiveness of animated features embedded in e-books and whether these had positive effects on students. Sixty kindergarten children participated in the study; all learning Dutch as a foreign language. They all read a traditional book and an e-book featuring music, narration, and sounds. The study showed that the vocabulary skills of the children were improved at the post-reading phase of the e-book (g=0.62). The study also suggests that the addition of the electronic features does not distract children's attention during reading. Instead, the results showed a connection between the electronic features and improvement of children's attention. Therefore, these studies demonstrate the potential benefits e-books may offer to young EAL learners. The effectiveness of animated e-books in relation to children's vocabulary development is recognised by these studies, as confirmed by the meta-analysis reported in Chapters 4 and 6 of this thesis.

3.6 E-books and Adult Scaffolding for Young Children Learning EAL

Prior studies have mostly focused on the advantages and disadvantages of e-books for young children, without adult scaffolding. Two studies have evaluated e-books with scaffolding support from adults; for native English speakers (Neumann & Merchant, 2022) and for bilingual children (Yang et al., 2022). A recent case study by Neumann and Merchant (2022) explored how one teacher scaffolded a young child's learning and interactions during shared reading of an e-book in her early years' classroom. The teacher used a wide range of words, repetition, and questioning strategies to engage the child with the story content and maintain her interest which provided a positive experience for the child. In the study, the teacher's comments, and questions offered during the shared reading with the story book app were useful in scaffolding the child's interactions with the story and helped prompt the child to respond. Shared reading of digital books, in which an adult

actively scaffolds the young learner, can provide positive opportunities for children to experience language within stories. Using teacher scaffolding to engage and keep the interest of young children through the interactive use of questions, new words, and repetition of words has the ability to engage the child in this activity (Neumann & Merchant, 2022). Although e-book features may be advantageous for native English speakers, this finding may not be generalisable to those who learn English as an additional language.

Yang et al. (2022) carried out a randomised controlled trial to investigate the impact of bilingual discussion prompts along with feedback in a multimedia interactive e-book on parent-child shared reading for young English language learners aged 3–7 years in China. In the treatment condition, a total of sixty-four parent-child pairs read a multimedia English storybook with bilingual discussion prompts. In comparison, forty-three parent-child pairs participated in the identical activity of reading the multimedia storybook without the inclusion of discussion prompts. After reading the storybook twice, it was observed that the children that participated in the discussion-prompt group had superior performance in terms of story comprehension and retelling measures when compared to the control group. This notable difference in outcomes suggests that engaging in discussions facilitated by prompts positively impacted children's comprehension and ability to retell the story. However, children in both groups showed comparable gains in English vocabulary. There is a dearth of research on teacher scaffolding and e-book versus print book reading for young children learning English as an additional language. The present thesis (Study 2; Chapter 5) attempts to contribute to this line of work on story comprehension and vocabulary development by examining the different types of storybook scaffolding by a teacher and their effects on children learning English as a second language. Designers, educators, and policymakers need such evidence and information to guide innovations, design, and policymaking regarding educational technologies in English education in English learning countries. English language is one of the most widely spoken languages in the world whilst it is the official language of more countries than any other language. Consequently, it is of vital importance to examine the effectiveness of e-books in terms of improving the vocabulary skills of children learning English as an additional language.

Furthermore, the limited research on the topic poses a significant problem to Cypriot teachers. Indeed, teachers in Cyprus are frequently faced with a diverse group of children

within their class context. Quite a few immigrants come every year to live and work in Cyprus from all over the world, especially Russia, Israel, and Spain. Immigrant children often enter preschool in Cyprus without any prior knowledge of the English language and in some instances have severe lags in their knowledge. Therefore, it is vital to carry out further research on the topic in order to reach some solid conclusions. This potential literature gap could present a new research quest for academia. To the best of my knowledge, no studies have investigated the effects of different storybook media (e-book and print) on young EAL children's vocabulary and story comprehension while the teacher scaffolds children in discussion of text as they read, and the performance style, which presents stories in a skilful storytelling manner with few breaks in the text. Thus, the current study aims to examine and investigate these two factors: teaching style (interactive versus performance) and book medium (e-book versus print book).

3.7 Summary

Chapter 3 has examined additional language acquisition through e-book and print book reading, which is explored in Study 2. There is a paucity of studies evaluating e-books for young EAL learners, particularly those that include teacher scaffolding. Therefore, Study 2 aimed at investigating two different book mediums—the electronic book and the print book—using two different scaffolding techniques —the interactive teaching style and the performance teaching style, for EAL learners in terms of vocabulary development and story comprehension.

Chapter 4 presents the first of the two studies which makes up this thesis, with Study 1 serving as a meta-analysis which aimed to analyse the efficacy of e-books in facilitating young children's language and literacy development when compared to more traditional presentations of stories, such as narrating a story aloud or reading a print storybook. An overview of the thesis's research questions is now presented.

3.8 An Overview of the Thesis's Research Questions

This section provides an overview of the research questions this thesis addresses. As outlined in the introduction, this PhD comprises of two, distinct, but complementary studies:

Study 1 is a meta-analysis with the purpose of examining the efficacy of e-books in facilitating young children's language and literacy development when compared to more traditional presentations of stories, such as narrating a story aloud or reading a print storybook. Study 2, as described in Chapter 5, evaluates e-books and print books with the use of two scaffolding techniques for young EAL learners. The two studies, therefore, work in unison, with the meta-analysis complementing the experimental study of this thesis in order to answer the broader research question of the PhD thesis; specifically, can e-books support children's language and literacy development?

The meta-analysis attempts to answer the following three research questions:

RQ1: Do e-books foster language and literacy development compared to traditional printlike story presentations?

RQ2: Which specific language and literacy outcomes are best developed from e-book reading with and without adult scaffolding?

RQ3: Are interactive or multimedia features helpful in an e-book for the development of language and literacy skills?

Study 2 addressed EAL learners and adult scaffolding with the use of e-books and print books. The purpose of the study was to compare the impact of e-books versus print books on children learning English as an additional language in terms of vocabulary and story comprehension. The study also compared two reading styles alongside the two media (print and digital) to assess whether word learning and comprehension were enhanced. The following research questions were tested:

RQ1: Which type of book media (i.e., e-book versus print book) produces better results in the development of vocabulary and story comprehension for children learning English as an additional language (EAL)?

RQ2: Which reading style (i.e., interactive versus performance) better facilitates young

children's (learning English as an additional language) vocabulary and story comprehension development?

Chapter 4 presents the first of the two studies which makes up this thesis, with Study 1 serving as a meta-analysis which can place itself within the field of e-books and print books.

Chapter 4: Study 1 – Meta-Analysis

The first and utmost goal of this meta-analysis is to examine current e-book studies and evaluate the efficiency of e-books in facilitating language and literacy development in young children. Thus, the first question raised in this study is whether e-books offer any educational advantages when compared to print books and hence should be used to develop young children's language and literacy skills. More specifically, the question is whether e-books can replace the benefits of an adult reading a story to young children for language (e.g., story comprehension, story recall, vocabulary learning) and literacy development (e.g., alphabet knowledge, phonological awareness, print awareness, word reading and writing). Their ability shall be assessed in terms of increasing a child's language and literacy skills with and without the support of an adult.

A second goal of this meta-analysis is to evaluate the presence of scaffolding support in either the print condition, the e-book condition, or both conditions and its effects on language and literacy development, and whether outcome measures are better developed with the support of digital features in e-books in comparison to print book reading with and without adult support.

A third aim of this meta-analysis is to analyse experimental studies involving interventions with e-books in an effort to evaluate the specific features embedded in e-books that may simulate extraneous support (e.g., adult scaffolding) provided to children while listening/reading a story for language and literacy development. To that end, the present meta-analysis investigates the embedded features within e-books, which may resemble adult scaffolding, and to what extent these digital features can promote learning when compared to traditional reading of print books.

Chapter 4 presents a published research paper that the author has written during her research studies. Durham University standards permit the inclusion of a published article in its original form within a student's research thesis, provided that the document was published during the course of their study. The published paper can be cited as follows: Savva, M., Higgins, S., & Beckmann, N. (2022). Meta-analysis examining the effects of electronic storybooks on language and literacy outcomes for children in grades Pre-K to grade 2. *Journal of Computer Assisted Learning*, *38*(2), 526-564. https://doi-org.ezphost.dur.ac.uk/10.1111/jcal.12623

4.1 Research Questions

As explained earlier, the aim of this meta-analysis is to analyse the efficacy of e-books in facilitating young children's language and literacy development when compared to more traditional presentations of stories, such as narrating a story aloud or reading a print storybook. Thus, only studies that compare e-book story presentations to more traditional presentations of the same or a similar story were included in the meta-analysis. The meta-analysis attempts to answer three research questions:

1. Do e-books foster language and literacy development compared to traditional printlike story presentations?

2. Which specific language and literacy outcomes are best developed from e-book reading with and without adult scaffolding?

3. Are interactive or multimedia features helpful in an e-book for the development of language and literacy skills?

4.2 Previous Meta-Analyses of the Effects of E-books Compared to Print Books

There are three meta-analyses currently available comparing e-books to print books. First, Zucker et al. (2009) focused on experimental studies published from January 1997 to January 2007 using e-books targeted at children aged 2 to 12 years old. Overall, seven experimental studies met their inclusion criteria. Findings explored the effectiveness of e-books in decoding and comprehension. For decoding (k = 2), an average, weighted effect of d = 0.09, (95% CI = [-0.35, 0.53]) was reported, which was statistically insignificant. For story comprehension (k = 7), a small positive significant effect of d =0.31, (95% CI = [0.06, 0.55]) associated with e-books was found. Additional research in this area is required to determine to what extent e-books effectively support children in language and literacy development. The small number of experimental studies available at the time, and the fact that most included studies dated back more than a decade, suggests adding studies with newer types of digital e-books using current technology in a new meta-analysis is justified. Second, Takacs et al. (2014) evaluated multimedia ebooks, television shows and films, in comparison to traditional print-like stories, for the years of 1980-2014, for comprehension and vocabulary development. Their analysis included 29 experimental and quasi-experimental studies. Multimedia-enhanced digital

stories were found to be more beneficial than traditional print books for story comprehension (g = 0.40; 95% CI = [0.22, 0.58]) and vocabulary development (g = 0.30; 95% CI = [0.07, 0.53]). These findings are partially based on digital stories presented as animated films, children's television series, and video clips, which were not within the scope of this meta-analysis. Indeed, we excluded studies with video and television presentations given their technological differences to e-books. The current meta-analysis sought to evaluate the specific features included in e-books which may resemble adult scaffolding while reading a print book. There is more emphasis on the text itself in e-books compared to video-based story presentations as the text is foregrounded; there is no emphasis on text or interactivity in video-based stories. Finally, in a third meta-analysis, Takacs et al. (2015) were interested in evaluating e-books with at-risk students. The participants consisted of a broad range of disadvantaged children with different risk factors, such as struggling readers, children with special needs, children with learning disabilities, severe language impairments and developmental delays. The present meta-analysis focuses on typically developing readers.

An important stage in any meta-analysis is the decision as to which types of studies are eligible for inclusion. A significant distinction between the abovementioned metaanalyses and the current study is the exclusion of studies which generally have a greater risk of allocation bias, i.e., non-random allocation to groups (Coolican, 2014). Risk of bias depends on how studies are carried out, including their design and their conduct (Higgins et al., 2013). Employing experimental designs with random allocation of participants into groups minimises allocation bias and controls for differences between participants (Sullivan, 2011). When quasi-experimental studies are introduced in a metaanalysis, there is a risk of introducing allocation bias as quasi-experimental studies involve non-random assignment of participants to conditions or orders of conditions (Coolican, 2014). Thus, to avoid any design bias, the current meta-analysis included only experimental studies. The quality or rigour of the research methodology used in each study was assessed based on the quality codes proposed by Troia (1999). The methodological strengths and limitations of each study included in the meta-analysis were identified using conventional evaluative criteria for quantitative research. The set of criteria for evaluating these studies with respect to internal and external validity are presented in Table 4.3.

Furthermore, findings from above mentioned research syntheses are not necessarily applicable to e-books available on the market today, as most of the primary studies included for analysis were published before 2008 and included e-books that are likely outdated today. Given the changes in technology, today's e-books are not easily comparable with the e-books available 20 years ago. For example, the study by De Jong and Bus (2004) used electronic compact discs read-only memory (CD-ROMs) (From Het Spectrum Electronic Publishing) which are no longer available to purchase today. Due to the fact that CD-ROMs were not as technologically evolved as e-books are today, it is important to continue the analysis of e-books which are being used by young children today. Modern e-books include various features, such as dictionaries, hotspots, puzzles and questions, which need to be examined in comparison to traditional reading of print books.

4.3 Method

4.3.1 Inclusion and Exclusion Criteria

Several inclusion criteria were defined for systematically collecting and reviewing primary studies (see Table 4.1). First, studies with experimental designs were included in the meta-analysis because experimental designs provide causal evidence of the effects of using e-books with children, and because there are fewer threats to validity in experimental compared to quasi-experimental designs (Shadish et al., 2002). For a study to be included its experimental design had to include a comparison condition in which the same or a similar story was presented using a print book. The comparison condition could also include presentations of static illustrations with audio narration on an e-book.

Second, we also limited our search to studies that used e-books with children between the ages of 3 and 8 years, in a peer-reviewed journal or unpublished dissertation - conducted in any country, published in English, from January 2008 to January 2021. Given the rapid pace at which technology changes, we were interested in studies within the last decade as older technologies are likely obsolete.

Third, we included studies that evaluated the e-books' efficacy in facilitating language and literacy development. The studies included at least one outcome measure, such as the child's language and/or literacy skills including story comprehension and vocabulary, and code related literacy skills, such as phonological awareness, letter knowledge, concepts of print, spelling, word reading, or general reading skills.

Fourth, the meta-analysis included studies with regular developing children and children from low/medium and high socioeconomic status (SES) families (e.g., Korat et al., 2014b). These children might have smaller vocabularies (Takacs et al., 2015) but in any given mainstream classroom, teachers are more likely to encounter children from various backgrounds as well as low/medium socioeconomic status. The current meta-analysis was undertaken with the aim to give insight to educators and school administrators into the use of e-books in a class environment, where teachers read a story to the children with language and literacy objectives. Thus, e-book efficiency is being evaluated with high/medium and low socioeconomic status families.

Hence, the following criteria were used to define the set of studies to be *excluded* from the meta-analysis:

1. Non-experimental studies (e.g., Kendeou et al., 2008) and quasi-experimental studies (e.g., Altun, 2018; Phadung et al., 2016).

2. Studies investigating children at risk for learning difficulties (e.g., Shamir et al., 2012), children with learning problems, such as struggling readers and/or special needs (e.g., Shamir et al., 2011), children with developmental delays or children with severe learning impairments (Smeets et al., 2014).

3. Studies undertaken in a home environment (e.g., Chiong et al., 2012; Korat & Shneor, 2019; Noel, 2013). These studies were excluded, because parents play a unique role in a child's life compared to teachers, and hence findings cannot easily be generalised to adult-facilitated learning in school.

4. Studies with a control group that followed the regular school programme rather than reading or listening to a print book. These studies were not included (e.g., Korat, 2010; Korat & Blau, 2010; Korat et al., 2017) to ensure a clearly defined counterfactual condition.

5. Studies with films, video, and television shows were excluded (e.g., Neuman et al., 2020; Verhallen & Bus, 2009; 2010).

| Included | Excluded | | | | | |
|-----------------------------------------------------------------------------|--------------------------------------------------------|--|--|--|--|--|
| Design | | | | | | |
| Experimental random allocation studies | Quasi-experimental studies | | | | | |
| - | Qualitative or observational studies | | | | | |
| Independent Variables | | | | | | |
| Treatment condition: | Treatment condition: | | | | | |
| a digital presentation of a narrative story or informational | Digital media such as video, television presentations, | | | | | |
| text (tablet, desktop computer, personal computer or | cartoons | | | | | |
| telephone) that: | | | | | | |
| • Simulates a print book (e.g., pages that "turn"); | | | | | | |
| • Has written text on page; | | | | | | |
| • Includes either some form of hypermedia (e.g., | | | | | | |
| images, sounds, music, animation) or is a static e-book | | | | | | |
| (only highlighted text and oral narration). | | | | | | |
| Control condition: | Control condition: | | | | | |
| Drint hools which has the same or similar contant to | Following the regular school timetable | | | | | |
| Find book which has the same of similar content to treatment condition: | | | | | | |
| Static-illustrations with audio parration on an e-book | | | | | | |
| Dependent Variables | | | | | | |
| Story comprehension | | | | | | |
| Vocabulary | | | | | | |
| Decoding-related skills | | | | | | |
| Participants | | | | | | |
| Grades Pre-K through 2 or Children ages 3- to 8-years | Children at risk for learning difficulties | | | | | |
| Typical developing from high/medium/low SES | C | | | | | |
| Educational context | | | | | | |
| School environment | Home environment | | | | | |
| Publication | | | | | | |
| Peer-reviewed journals | | | | | | |
| Unpublished dissertations | | | | | | |
| (conducted in any country - printed in English) | | | | | | |
| Publication dates January 2008 to January 2021 | | | | | | |

 Table 4.1 Inclusion and Exclusion Criteria

4.3.2 Literature Search

We identified potential studies using electronic database searches, hand searches of key journals, and by searching the reference sections of identified studies. First, we searched for all articles involving e-books in the three major databases in the education and psychology fields: PsycINFO, ERIC, and Web of Science available from January 2008 to January 2021. The search strategy included these key words and the Boolean operator "AND" for these combinations of words: *e-books* AND *literacy, e-books* AND *language, electronic book, computer, tablet, e-storybook, digital book, multimedia book, media book, interactive book, interactive storybook, e-literature, talking book, living book, living book, living storybook, emergent literacy, reading, vocabulary, word learning, story comprehension, reading, spelling, adult scaffolding, adult support, dialogic reading, adult mediation, children, kindergartner, preschooler, elementary.* Next, we hand-

searched the following relevant journals in order to obtain additional studies: *Journal of Computer Assisted Learning, Education and Information Technologies, Journal of Research on Technology in Education, Journal of Educational Psychology, Early Childhood Education Journal,* and *Reading Research Quarterly* for the dates January 2008 to January 2021. Furthermore, we searched the reference sections of review articles and the included articles that met our selection criteria. Finally, we searched for dissertations and theses reporting data that might be suitable for the present meta-analysis.

The database search produced 1,235 reports, which were scanned based on the titles and the abstracts, of which 206 studies were reviewed in full and several were excluded due to study design (e.g., quasi-experimental) or sample characteristics that differed from the characteristics specified for this meta-analysis (e.g., participants' age range, atypically developing children) (see Appendix A for PRISMA diagram of the literature search). Finally, 29 studies met the inclusion criteria. Of these, 18 articles reported on single studies, whereas eleven contained multiple treatments/experimental groups. The 29 studies that were submitted to the meta-analysis are marked with an asterisk (*) in the references section.

4.3.3 Study Coding

We selected the following research variables to examine more closely by assigning codes for the following aspects of each study: publication year, type of design (experimental between-subject design/within-subject design), characteristics of participants (e.g., age, ethnicity, SES, primary language), literacy outcomes measured (e.g., decoding, vocabulary or comprehension skills), software features (e.g., genre, game options embedded, hotspots, dictionary option, embedded prompts/lessons) and the length of the intervention. We also coded adult scaffolding/mediation: whether children received adult mediation while reading the paper book and/or the digital book. There were studies in which an adult guided the reading of both books; an adult guided the paper book reading but not the digital book and vice versa. All coding was completed by the first author and quality checked by the second and third authors (see Table 4.2 for a summary).

| Study and Year | Design of the study | Country - Language/ Ethnicity | SES | Age (years) | E-book Intervention (Device) | Material | Interactive features | Multimedia features | Comparison condition | Adult Scaffolding /Mediation – By whom | Number of Reading sessions | Outcome Measures |
|--------------------------------------|-------------------------------------------------------|---------------------------------------------------------------------------------------------------------|--------------|----------------|------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------|------------------------------------------------------------------------------------|---------------------------------------|-------------------------------------------------------------|---------------------------------------|------------------------------------------------------------------------------------------------------------|
| Altun (2021) | Experimental Matched comparison group design | Turkey- Turkish | | 5 | E-book (n = 97) (iPad) | Kırmızı Kanatlı Baykuş ("Red-Winged Owl") Who Stole the Moon? | - | Some animation, background music, character movements. | Print book (n = 99) | No | One reading of each story | Story Comprehension (1 measure) Vocabulary (Receptive) (1 measure) |
| Broemmel et al. (2015) | Experimental Between – subjects | United States English, Spanish and Chinese African- American, Hispanic, Chinese | High/ Low | 4-5 | E-books plus traditional picture books (n = 10) (Computer) | The Alphabet Song Book (2006) Icky Sticky Frog (2000) Coral Reef Hide and Seek (2005) | - | High quality animation, rich narration, sound effects and music. | Print book (n = 14) | Yes- By researcher (E-book +Print condition) | Three readings of each book. | Story Comprehension (1 measure) |
| Critelli (thesis) (2011) | Experimental Between – subjects | Northern New Jersey | | 4–6 | E-book (n = 5) (Computer) | Bubbles CD-ROM story by Diane Foushee | - | - | Print book (n = 5) | No | 3 sessions | Phonological awareness (1 measure) Word reading (1 measure) Story Comprehension (1 measure) |
| Eng et al. (2020) Experiment 1 | Within subject design (randomised order) | United States | | 3-5 | Contingent E-book (n = 17) (Tablet) | Cat's Pajamas and Zoom City | The continger story included that activated on the child's | nt responsivity l animations contingently vocalisations. | Board Book (n = 18) | No | One reading of each story | Story Recall (1 measure) |
| Eng et al. (2020) Experiment 2 | Within subject design (randomised order) | United States | | 3-5 | Contingent E-book (n = 16) (Tablet) | Cat's Pajamas and Zoom City | The continger story included that activated on the child's | nt responsivity l animations contingently vocalisations. | Static E-book (n = 17) (Tablet) | No | One reading of each story | Story Recall (1 measure) |
| | | | | | | | | | | | | (Continued) |

| Study and Year | Design of the study | Country - Language/ Ethnicity | SES | Age (years) | E-book Intervention (Device) | Material | Interactive features | Multimedia features | Comparison condition | Adult Mediation/Sc affolding – By whom | Number of Reading sessions | Outcome Measures |
|-----------------------------------------|---------------------------------------|-------------------------------------|-----|----------------|----------------------------------------------------------------------|---------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------|------------------------------------------|-------------------------------------------------|--------------------------------------------------|------------------------------------------------------------------------|
| Gong and Levy (2009) Comparison 1 | Experimental Between – subjects | Canada | | 4-5 | E-book with 'Bouncing Ball' (n = 24) (Laptop) | 6 stories including familiar topics. | - | A small ball bounced above each word in synchrony with the reader's voice. | Static e-book (n = 24) (Laptop) | No | 10 days- 6 different storybook sessions | Print Knowledge (1 measure) |
| Gong and Levy (2009) Comparison 2 | Experimental Between – subjects | Canada | | 4-5 | E-book with 'Violation' (n = 24) (Laptop) | 6 stories including familiar topics, such as animals, families, and friendship. | - | Bouncing ball condition + addition of two unreadable items per page of each book. | Static e-book (n = 24) (Laptop) | No | 10 days | Print Knowledge (1 measure) |
| Gong and Levy (2009) Comparison 3 | Experimental Between – subjects | Canada | | 4-5 | E-book with 'Action' (n = 24) (Laptop) | 6 stories including familiar topics, such as animals, families, and friendship. | Violation condition + child clicked on the violated item using the computer mouse. | | Static e-book (n = 24) (Laptop) | No | 10 days | Print Knowledge (1 measure) |
| Homer et al. (2014) Comparison 1 | Experimental Between- subjects | USA- English | | 5-7 | Kinect with activities (n = 12) (Microsoft's Xbox) | Children Make Terrible Pets by Microsoft Games Studio for the Kinect | Children were required to use movements to interact with the book. | On-screen animated character. | Book reading (n = 14) | Yes-By experimenter (Print condition) | 1 reading | Reading Words (1 measure) Vocabulary (Expressive) (1 measure) |
| Homer et al. (2014) Comparison 2 | Experimental Between- subjects | USA- English | | 5-7 | Kinect without activities (n = 13) (Microsoft's Xbox) | Children Make Terrible Pets by Microsoft Games Studio for the Kinect | Kinect sensor detected the participant's image and displayed a live video- feed of the child on- screen. | On-screen animated character. | Book reading (n = 14) | Yes-By experimenter (Print condition) | 1 reading | Reading Words (1 measure) Vocabulary (Expressive) (1 measure) |

(Continued)

| Study and Year | Design of the study | Country - Language /Ethnicity | SES | Age (years) | E-book Intervention (Device) | Material | Interactive features | Multimedia features | Comparison condition | Adult Mediation/ Scaffolding – By whom | Number of Reading sessions | Outcome Measures |
|--------------------------------------------|------------------------------------------|-------------------------------------|-----------------------|----------------|-------------------------------------------------------------------|------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------|------------------------------------------------------------------------------|-------------------------------------------------|-------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------|
| Ihmeideh (2014) | Experime ntal Between- subjects | Jordan Amman Arabic | Middl e and Low | 4-5 | E-book (n = 48) (Computer) | Arabic books: The Conceited Ibex, Rami, The Dreams' King, The Three Goats | - | Animated illustrations | Print Book (n = 44) | Yes- By teacher (Both conditions) | 15 min each day for eight weeks. | Vocabulary (Expressive) (1 measure) Print awareness (1 measure) Phonological awareness (1 measure) |
| Karemaker et al. (2017) Comparison 1 | Experime ntal Between- subjects | Oxfordshi re – England | Mix | 5.6-7 | E-book with E-friend button (n = 32) | The Parachute | E-friend button opens up a window with story questions. | - | 'Flat e-book' (n = 30) (Computer) | No | 1 reading session | Vocabulary (Expressive) (1 measure) Word Reading (1 measure) Story Comprehension |
| Karemaker et al. (2017) Comparison 2 | Experime ntal Between- subjects | Oxfordshi re – England | Divers e SES | 5.6-7 | (Computer) E-book with Dictionary (n = 28) (Computer) | The Parachute | Dictionary button highlighted challenging words. By clicking on a word children could listen to pronunciation and definition. | - | 'Flat e-book' (n = 30) (Computer) | No | 1 reading session | (2 measures) Vocabulary (Expressive) (1 measure) Word Reading (1 measure) Story Comprehension (2 measures) |
| Kelley and Kinney (2017) | Experime ntal Between- subjects | US | Divers e SES | 3-5 | Interactive e-book (Learn) (n = 15) (Tablet) | Hansel and Gretel by Mindshapes Ltd | Six embedded questions related to the story and children can tap pictures to respond to the questions. | Animated characters. | E-book without interactive features (Watch) (n = 15) (Tablet) | No | Three sessions | Vocabulary (Expressive) (2 measures) Vocabulary (Receptive) (1 measure) Vocabulary (1 measure) Story comprehension (1 measure) |

(Continued)

| Study and Year | Design of the study | Country - Language /Ethnicity | SES | Age (years) | E-book Intervention (Device) | Material | Interactive features | Multimedia features | Comparison condition | Adult Mediation/ Scaffolding – By whom | Number of Reading sessions | Outcome Measures |
|-------------------------------------------------------------------|------------------------------------------|-------------------------------------|--------|----------------|------------------------------------------------------------------------------------------------|----------------------------|-------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------|-------------------------------------------------|-------------------------------------|--------------------------------------------------------------------------------------------------------------|
| Korat, Levin, Atishkin, Turgeman (2014a) Comparison 1 | Experime ntal Between- subjects | Israel Hebrew | Middle | 4-6 | E-book with dynamic visual vocabulary support (Dictionary) (n = 38) | The Empty Pot (1990) | - | Animated representations of target verbs in vocabulary support. | E-book without dictionary support (n = 37) (Computer) | No | Reading 3 times | Vocabulary (Expressive) (1 measure) Vocabulary (Receptive) (1 measure) Vocabulary (1 measure) |
| Korat, Levin, Atishkin, Turgeman (2014a) Comparison 2 | Experime ntal Between- subjects | Israel Hebrew | Middle | 4-6 | (Computer) E-book with static visuals dictionary support (n = 37) (Computer) | The Empty Pot (1990) | - | At the end of the narrators' reading of the screen, a large bubble appears with the target written form of the word and a figurative static presentation of a flower, with the narrator concomitantly saying a short explanation of the word | E-book without dictionary support (n = 37) (Computer) | No | Reading 3 times | Vocabulary (Expressive) (1 measure) Vocabulary (Receptive) (1 measure) Vocabulary (1 measure) |
| Korat, Levin, Atishkin, Turgeman (2014a) Comparison 3 | Experime ntal Between- subjects | Israel Hebrew | Middle | 4-6 | E-book with adults vocabulary support (n = 37) (Computer) | The Empty Pot (1990) | - | At the end of the narrators' reading of the screen, a large bubble appears with the target written form of the word and a figurative static presentation of a flower, with the narrator concomitantly saying | E-book without dictionary support (n = 37) (Computer) | Yes-By researchers (E-book condition) | Reading 3 times | Vocabulary (Expressive) (1 measure) Vocabulary (Receptive) (1 measure) Vocabulary (1 measure) |
a short explanation of the word

(Continued)

| Study and Year | Design of the study | Country - Language /Ethnicity | SES | Age (years) | E-book Intervention (Device) | Material | Interactive features | Multimedia features | Comparison condition | Adult Mediation/ Scaffolding | Number of Reading | Outcome Measures |
|---------------------------------------------------------------------------------|------------------------------------------|-------------------------------------|-----|----------------|----------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------|-------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------|------------------------------------|------------------------------------------|------------------------------------------------------------------------------------------------|
| Korat, Levin, Ben-Shabt, Shneor and Bokovza (2014b) Comparison 1 | Experiment al Between- subjects | Israel Hebrew | Low | 7-8 | E-book with dictionary dynamic visuals (DVs) with the printed word (n = 41) (Computer) | <i>The Empty</i> <i>Pot</i> written and illustrated by Demi (1990) | - | Animated representations of target verbs in vocabulary support and printed words shown on screen. | Static E-book without a dictionary (control; n = 44) (Computer) | <u> </u> | 4 repetitions of e-book reading | Vocabulary (Expressive) (1 measure) Vocabulary (1 measure) Spelling (1 measure) |
| Korat, Levin, Ben-Shabt, Shneor and Bokovza (2014b) Comparison 2 | Experiment al Between- subjects | Israel Hebrew | Low | 7-8 | E-book with dictionary dynamic visuals (DVs) without printed words (n = 42) (Computer) | The Empty Pot (1990) | - | Animated representations of target words in the form of pictures. | Static E-book without a dictionary (control; n = 44) (Computer) | No | 4 repetitions of e-book reading | Vocabulary (Expressive) (1 measure) Vocabulary (1 measure) Spelling (1 measure) |
| Korat, Levin, Ben-Shabt, Shneor and Bokovza (2014b) Comparison 3 | Experiment al Between- subjects | Israel Hebrew | Low | 7-8 | E-book with dictionary static visuals (SVs) with printed words (n = 43) (Computer) | The Empty Pot (1990) | - | Static representations of target verbs in vocabulary support e.g. a large bubble appears with the target written form of the word and a figurative static presentation of a flower. | Static E-book without a dictionary (control; n = 44) (Computer) | No | 4 repetitions of e-book reading | Vocabulary (Expressive) (1 measure) Vocabulary (1 measure) Spelling (1 measure) |
| Korat, Levin, Ben-Shabt, Shneor and Bokovza (2014b) | Experiment al Between- subjects | Israel Hebrew | Low | 7-8 | E-book with dictionary static visuals (SVs) without printed words | The Empty Pot (1990) | - | Static representations of target verbs in vocabulary support e.g. a large bubble | Static E-book without a dictionary (control; n = 44) | No | 4 repetitions of e-book reading | Vocabulary (Expressive) (1 measure) Vocabulary (1 measure) Spelling |

| | (n = 45) | appears with a static | (1 measure) |
|--------------|----------|-----------------------|-------------|
| Comparison 4 | | presentation of a | |
| | Computer | flower. | |
| | | | (Continued) |

| Study and Year | Design of the study | Country - Language/ Ethnicity | SES | Age (years) | E-book Intervention (Device) | Material | Interactive features | Multimedia features | Comparison condition | Adult Scaffolding – By whom | Number of Reading sessions | Outcome Measures |
|-----------------------------------------|---------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------|-----|----------------|---------------------------------------------------------------------------------------------|----------------------------------------------|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------|----------------------------------------------------|-------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------|
| Kozminsky and Asher- Sadon (2013) | Experimental Between- subjects | Israel Hebrew | Mix | 5-6 | E-book (n = 25) (Computer) | Pochzani m | - | - | Print book (n = 25) | Yes- By experimenter (Print condition) | 5 reading sessions | Story Comprehension (1 measure) Print Knowledge (1 measure) Vocabulary (1 measure) Phonological Awareness (1 measure) Spelling |
| Lee (2020) | Within subject design (randomised order) | US 54 Anglo, 35 African American, 1 Middle Eastern, 5 Hispanic, and 5 multiracial students | Low | 6-7 | E-book with recorded word explanation (n = 50) (iPad) | Frederick and Swimmy | - | Recorded word explanations | Static e-book (n = 50) (iPad) | No | 2 readings | (1 measure) Vocabulary (Expressive) (1 measure) |
| Liao et al. (2020) Comparison 1 | Experimental Between- subjects | Taiwan Mandarin Chinese | | 4-5 | E-book highlight synchronisation (implicit instruction) (n = 19) (Laptop) | Six original Chinese storybook s | - | Individual Chinese characters were highlighted in red when they were being pronounced, thus implicitly teaching the | Static Read- only E-book (n = 20) | No | 1 reading of each book | Print Awareness (1 measure) |

children which Chinese characters were associated with which sounds. (Continued)

| Study and Year | Design of the study | Country - Language/Ethn icity | SES | Age (years) | E-book Intervention (Device) | Material | Interactive features | Multimedia features | Comparison condition | Adult Scaffolding /Mediation – By whom | Number of Reading sessions | Outcome Measures |
|-------------------------------------------|---------------------------------------------------|---------------------------------------------------------------------------------------|-----------------|----------------|---------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------|-------------------------------------------------|----------------------------------|-------------------------------------------------------------------------------|
| Liao et al. (2020) Comparis on 2 | Experimental Between- subjects | Taiwan Mandarin Chinese | | 4-5 | E-book print discussion (explicit instruction) (n = 20) (Laptop) | Six original Chinese storybooks | - | Verbal cues to or discussions of the print (references to the print) were explicitly added to the narration of the story. Also visual cues to synchronise with the verbal cues. | Static Read- only E-book (n = 20) | No | 1 reading of each book | Print Awareness (1 measure) |
| Neuman et al. (2017) | Within subject design (randomised order) | US African American; Hispanic; 2% were of European descent | Low | 3-4 | E-book (n = 19) (iPad) | Ish, Sid the Science Kid, Superkids: A sticky situation, The Valentine all from Speakaboos | - | Animated characters | Print Book (n = 19) | No | 2 sessions in each condition | Vocabulary (Expressive) (1 measure) Story Comprehension (2 measures) |
| O'Toole (2015) (thesis) | Experimental Between- subjects | Chicago 70 children were White; 2 Black; 4 Hispanic; 6 Asian; 14 children | Middl e/High | 4 | E-book read aloud by a live adult and narrated by an | Just in Passing by Susan Bonners | - | - | Print book read aloud by a live adult and narrated | No | 1 session | Vocabulary (Receptive) (1 measure) Story Comprehension (1 measure) |

| were biracial; and 4 parents did not respond | audio device (1989) (n = 50) | by an audio device (n = 50) |
|----------------------------------------------------|---------------------------------|-----------------------------------|
| | (Tablet) | |

(Continued)

| Study and Year | Design of the study | Country - Language/ Ethnicity | SES | Age (years) | E-book Intervention (Device) | Material | Interactive features | Multimedia features | Comparison condition | Adult Scaffolding /Mediation – By whom | Number of Reading sessions | Outcome Measures |
|--------------------------------------------------|---------------------------------------------------|-----------------------------------------------------------------------|--------------------|----------------|-----------------------------------------------------|------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------|---------------------------------------|---------------------------------------------|-------------------------------------------------|-------------------------------------|-------------------------------------------------------------------------------|
| O'Toole and Kannass (2018) Comparison 1 | Experimental Between- subjects | US 70 White, 2 Black, 4 Hispanic, 6 Asian, 14 biracial | Middle and high | 4-5 | Live E-book (n = 25) (Tablet) | n/a | - | - | Live Print Book (n = 25) | No | 1 session | Vocabulary (Receptive) (1 measure) Story Comprehension (1 measure) |
| O'Toole and Kannass (2018) Comparison 2 | Experimental Between- subjects | US 70 White, 2 Black, 4 Hispanic, 6 Asian, 14 biracial | Middle and high | 4-5 | Audio narrated e-book (n = 25) (Tablet) | n/a | - | - | Audio narrated Print book (n = 25) | No | 1 session | Vocabulary (Receptive) (1 measure) Story Comprehension (1 measure) |
| Pearman (2008) | Within subject design (randomised order) | United States 32 white; 1 black; 21 Hispanic | | 7-8 | E-book (n = 27) (Computer – CD-ROM) | Heather Hits Her First Home Run and A Long Hard Day on the Ranch | Hotspots for word pronunciatio ns, graphics, sound effects, object labels, and definitions | Animation | Print book (n = 27) | No | 20 days- two sessions | Story Comprehension (1 measure) |
| Reich et al. (2019) | Experimental design Between | USA - Southern California | Middle and high | 3-5 | E-book (n = 100) (iPad) | Chris P. Bacon: My Life so Far | 6 hotspots per when tapped r word or provi- animation. | page, that epeated the ded some | Print book (n = 100) | No | 1 reading | Story Comprehension (3 measures) Vocabulary (Expressive) (1 measure) |

| Study and Year | Design of the study | Country - Language /Ethnicity | SES | Age (years) | E-book Intervention (Device) | Material | Interactive features | Multimedia features | Comparison condition | Adult Scaffolding /Mediation – By whom | Number of Reading sessions | Outcome Measures |
|----------------------------------|------------------------------------------------------|-------------------------------------|--------|----------------|--------------------------------------------------|-----------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------|-------------------------------------------------|-------------------------------------|-----------------------------------------------------------------------------|
| Richter and Courage (2017) | Within subject design (randomised order) | Canada | Middle | 3-5 | E-book (n = 39) (iPad) | Leo the Lightning Bug and A Frog Thing | Hotspots to reactivate the multimedia features (e.g., thunder and lightning) or to produce word repetitions. | Multimedia features accompany the narration and provide additional visual or auditory information (e.g., sounds, animations) | Print Book (n = 40) | No | 1 reading | Story Comprehension (1 measure) |
| Rvachew et al. (2017) | Within subject design (randomised order) | Canada (English) | Low | 5-6 | E-book (n = 14) (iPad) | Caillou: What's That Funny Noise? and Caillou: My First Play | - | Living words link animated text to animated illustrations. A prompt bar for the adult reader which suggests comments and questions. | Print Book (n = 14) | Yes-By teacher (Both conditions) | 1 reading | Story Comprehension (1 measure) Phonological Awareness (1 measure) |
| Sapsaglam et al. (2020) | Experiment al Between- subjects | Turkey | | 5-6 | E-book (n = 10) (Computer to projector) | Hungry Caterpillar, Brave Firfir, Tiny Seed, The | - | - | Print book (n = 10) | No | 1 reading of each story | Story Comprehension (1 measure) |

| Most |
|-------------|
| Ordinary |
| Tree of the |
| Forest, The |
| Dinosaur |
| Came Out |
| of My Seed |
| (Continued) |
| (Continued) |

| Study and Year | Design of the study | Country - Language /Ethnicity | SES | Age (years) | E-book Intervention (Device) | Material | Interactive features | Multimedia features | Comparison condition | Adult Instruction /Scaffoldin g – By whom | Number of Reading sessions | Outcome Measures |
|----------------------------------------------|--------------------------------------|-------------------------------------|-------------|----------------|---------------------------------------------------------------------------------------------|--------------------------------------------------------------|-------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------|-------------------------------------------------------|-------------------------------------|-----------------------------------------------------------------------------------------------------------------------|
| Sari et al. (2019) | Experimental Between- subjects | Turkey Turkish | Divers e | 4-6 | E-book with animated illustrations with and without music/sounds (n = 41) | Bear Is in Love with Butterfly, Little Kangaroo | - | Animated characters and objects, sound effects, music | E-book with static illustrations with and without music/sounds (n = 42) | No | 2 readings of each story | Story Comprehension (1 measure) Vocabulary (Receptive) (1 measure) Vocabulary (Expressive) (1 measure) |
| Segal-Drori et al. (2010) Comparison 1 | Experimental Between- subjects | Israel Hebrew | Low | 5-6 | E-book without adult instruction (EB) (n = 32) (Computer) | Confused Yuval and The Tractor in the Sandbox | Hotspots | Automatic dynamic visuals that dramatise story details, sections and the complete story scene as well as extra music and film effects that may "bring the story content to life" | (Laptop) Print book with adult instruction (PBI) (n = 32) (Computer) | Yes – By researchers (Print condition) | Four sessions | Phonological Awareness (1 measure) Reading Words (1 measure) Print Knowledge (1 measure) |
| Segal-Drori et al. (2010) Comparison 2 | Experimental Between- subjects | Israel Hebrew | Low | 5-6 | E-book with adult instruction (EBI) (n = | Confused Yuval and The Tractor in | Hotspots | Automatic dynamic visuals that dramatize story details, | Print book with adult instruction | Yes – By researchers (Both | Four sessions | Phonological Awareness (1 measure) Reading Words (1 measure) |

| 32) | the | sections and the | (PBI) (n = 32) | conditions) | Print Knowledge |
|------------|---------|-------------------|----------------|-------------|-----------------|
| | Sandbox | complete story | | | (1 measure) |
| (Computer) | | scene as well as | (Computer) | | |
| | | extra music and | | | |
| | | film effects that | | | |
| | | may "bring the | | | |
| | | story content to | | | |
| | | life'' | | | |
| | | | | | (Continued) |

| Study and Year | Design of the study | Country - Language /Ethnicity | SES | Age (years) | E-book Intervention (Device) | Material | Interactiv e features | Multimedia features | Comparison condition | Adult Instruction /Scaffoldin g – By whom | Number of Reading sessions | Outcome Measures |
|------------------------------------------|---------------------------------------------------|-------------------------------------|------------|----------------|--------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------|----------------------------------------------------------------|--------------------------------------------|-------------------------------------------------------|---------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------|
| Smeets and Bus (2015) Comparison 1 | Experimental Between- subjects | Netherlan ds 50% Dutch L1 | | 4-6 | Animated e-books (n = 36) (Computer) | Pete on the Pavement, Bear Is in Love with Butterfly, Rokko the Crocodile, Bolder and the Boat, Cycling With Grandpa | - | Animated characters and objects, sound effects, music | Static e-book (n=34) (Computer) | No | 4 repetition s of each story | Vocabulary (Expressive) (1 measure) Vocabulary (Receptive) (1 measure) Vocabulary (1 measure) Story Comprehension (1 measure) |
| Smeets and Bus (2015) Comparison 2 | Experimental Between- subjects | Netherlan ds 50% Dutch L1 | | 4-6 | Interactive animated e-books (n = 33) (Computer) | Pete on the Pavement, Bear Is in Love with Butterfly, Rokko the Crocodile, Bolder and the Boat, Cycling With Grandpa | Hotspots shape as a magnifyin g glass to search for unknown words | Animated characters and objects, sound effects, music | Static e-book (n = 34) (Computer) | No | 4 repetition s of each story | Vocabulary (Expressive) (1 measure) Vocabulary (Receptive) (1 measure) Vocabulary (1 measure) Story Comprehension (1 measure) |
| Takacs and Bus (2016) | Within Subject design (randomised order) | Netherlan ds | Middl e | 4-6 | Animated e-book (n = 20) (Computer) | The Little Kangaroo, Imitators | - | Animated scenes | Static e-book (n = 19) (Computer) | No | 3 sessions | Vocabulary (Receptive) (1 measure) Story comprehension (1 measure) |

(Continued)

| Study and Year | Design of the study | Country - Language/Ethnicity | SES | Age (years) | E-book Intervention (Device) | Material | Interactive features | Multimedia features | Comparison condition | Adult Instruction/ Scaffolding – By whom | Number of Reading sessions | Outcome Measures |
|------------------------------|------------------------------------------|-------------------------------------------|-----|----------------|-------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|--------------------------------------------|------------------------|--------------------------------|---------------------------------------------------|-------------------------------------|-------------------------------------------|
| Willoughby et al. (2015) | Experime ntal Between- subjects | Ontario English | | 3-4 | Alphabet e-books (n = 33) (iPad) | A to Z Alphabet book, Letters A to Z, Interactive Alphabet, This place is a zoo, Alphabet Zoo, Z is for Zebra, ABC Magic, ABC Magic 2, ABC Funnimals, Animal ABC, Animal Alphabet | Interactive games, audio hotspots | Animated hotspots | ABC Paper books (n = 30) | No | 8 weeks – 16 sessions | Phonological Awareness (3 measures) |
| Zhou and | Experime | United States | | 4-5 | Multimedia | The Polar Bear | Hotspots to | Animated | Print book | No | 2 reading | Vocabulary |
| Yadav (2017) Comparison 1 | ntal Between- subjects | All the participants speak English as the | | | e-book (n = 18) | Horizon | see and hear word labels when | characters | (n = 18) | | sessions | (Receptive) (1 measure) |
| companion i | | first language. | | | (iPad) | | touching illustrations | | | | | Story Comprehension (1 measure) |
| Zhou and | Experime | United States | | 4-5 | Multimedia e- | The Polar Bear | Hotspots to | Animated | Print Book | Yes- By | 2 reading | Vocabulary |
| Yadav (2017) | ntal | | | | book with | Horizon | see and hear | characters | with questions | researcher | sessions | (Receptive) |
| Composions 2 | Between- | All the participants | | | questioning $(n - 18)$ | | word labels | | (n = 18) | (Deth | | (1 measure) |
| Comparison 2 | subjects | speak English as the | | | (n = 18) | | wnen | | | (DOIN | | |

| | | first language. | | | (iPad) | | touching illustrations | | conditions) | | Story Comprehension (1 measure) |
|--------------------------|-------------------------------------------------------|----------------------------------------------------------------------------------------------------|----------------|-----|----------------------------------|--------------------------------------------|---------------------------------------------------------|------------------------|-------------------------------------------|---------------------------------------|--------------------------------------------------------------------|
| Zipke (2017) (Exp. 1) | Within subject design (randomis ed order) | New England State 15 Caucasian, 4 African American, 2 Latin American, 4 Asian American | Middle/ Low | 4-5 | E-book (n = 13) (Computer) | Tacky Goes to Camp, Tacky in Trouble | Hotspots on - pictures to hear the object name | Print book (n = 12) | Yes By teacher (Print condition) | Two separate 30-min sessions | Story Comprehension (1 measure) Vocabulary (1 measure) |

4.3.4 Methodological Rigour of Studies

The quality codes for establishing the methodological rigour of the studies included a selection of characteristics examined by Troia (1999) which can be found in Table 4.3. Twenty-one (72%) experimental studies met standards for high quality; the other eight studies (28%) were moderate in quality. All experimental studies used random assignment.

Participant characteristics were not adequately described in half of the studies as children's ethnicity or primary language were typically not mentioned. Treatment conditions were not described in adequate detail to permit replication attempts in 23 out of 29 studies. Of course, word-count limitations in journals often dictate how much information can be provided by authors. Nonetheless, conditions need to be adequately described for replication purposes as well as enabling researchers and educators to employ innovative treatments with their students. In regard to treatment fidelity, according to Troia (1999), all research studies need to ensure that a procedure is used to ensure that treatment conditions are being implemented faithfully. As seen in Table 4.3, fidelity of treatment was not reported in eight studies, so it is unclear how reliably reported treatment results can be attributed to the intervention. Only five experiments reported effect sizes. The provision of effect sizes would facilitate an evaluation of the relative effectiveness of the experimental treatments used in these intervention studies.

| | Study (alphabetically) | Random assignment | Control group intervention | Sufficient participant description | Treatment conditions explicitly described | Operationalised measures | Reliability of measures reported | Treatment fidelity ensured | Effect size reported | Quality Rating |
|-----|----------------------------------------------------------------------------|----------------------|-------------------------------|------------------------------------------|----------------------------------------------|-----------------------------|----------------------------------------|----------------------------------|-------------------------|-------------------|
| 1 | Altun (2021) | <u>(3)</u> Y | (3) Y | (3) Y | (2) Y | (3) Y | (2) Y | (3) Y | N | High |
| 2 | Broemmel et al. (2015) | Y | Y | Y | Y | Y | N | Y | N | High |
| 3. | Critelli (2011) | Y | Y | N | Y | Y | N | N | N | Moderate |
| 4. | Eng, Tomasic & Thiessen (2020) | Y | Y | N | Y | Y | Y | Y | Y | High |
| 5. | Gong and Levy (2009) | Y | Y | N | Y | Y | Y | Ν | N | Moderate |
| 6. | Homer et al. (2014) | Y | Y | Ν | Y | Y | Ν | Y | Ν | Moderate |
| 7. | Ihmeideh (2014) | Y | Y | Ν | Y | Y | Y | Ν | Ν | Moderate |
| 8. | Karemaker et al. (2017) | Y | Y | Ν | Ν | Y | Y | Y | Y | High |
| 9. | Kelley & Kinney (2017) | Y | Y | Ν | Ν | Y | Y | Y | N | Moderate |
| 10. | Korat, Levin, Atishkin and Turgeman (2014) | Y | Y | Y | Y | Y | Y | Ν | N | High |
| 11. | Korat, Levin, Ben-Shabt, Shneor and Bokovza (2014) | Y | Y | Y | Y | Y | Y | N | N | High |
| 12. | Kozminsky & Asher-Sadon (2013) | Y | Y | Ν | Y | Y | Y | Y | Ν | High |
| 13. | . Lee (2020) | Y | Y | Y | Y | Y | Y | Y | Y | High |
| 14. | . Liao et al. (2020) | Y | Y | Ν | Y | Y | Y | Y | Ν | High |
| 15. | Neuman et al. (2017) | Y | Y | Y | Y | Y | Y | Y | Ν | High |
| 16 | O'Toole (2015) | Y | Y | Y | Ν | Y | Y | Y | Ν | High |
| 17. | O'Toole & Kannass (2018) | Y | Y | Y | Ν | Y | Y | Y | Ν | High |
| 18. | Pearman (2008) | Y | Y | Y | Y | Y | Y | Y | Ν | High |
| 19. | Reich et al. (2019) | Y | Y | Y | Y | Y | Y | Y | N | High |
| | | | | | | | | | (0 | Continued) |

Table 4.3 Assessment of Methodological Quality Based on Selected Internal and External Validity Criteria from Troia (1999)

| Study (alphabetically) | Random assignment (3) † | Control group intervention (3) | Sufficient participant description (3) | Treatment conditions explicitly described (2) | Operationalised measures (3) | Reliability of measures reported (2) | Treatment fidelity ensured (3) | Effect size reported (1) | Quality Rating |
|---------------------------------------|-------------------------------|--------------------------------------|-------------------------------------------------|-----------------------------------------------------|------------------------------------|-----------------------------------------------|-----------------------------------------|--------------------------------|-------------------|
| 20. Richter and Courage (2017) | Y | Y | N | Y | Y | Y | Y | N | High |
| 21. Rvachew et al. (2017) | Y | Y | Y | Y | Y | Ν | Y | Ν | High |
| 22. Sapsaglam et al. (2020) | Y | Y | Ν | Y | Y | Ν | Ν | Ν | Moderate |
| 23. Sarı, Basal, Takacs, Bus (2019) | Y | Y | Y | Y | Y | Y | Y | Y | High |
| 24. Segal-Drori et al. (2010) | Y | Y | Y | Y | Y | Y | Y | Ν | High |
| 25. Smeets and Bus (2015) | Y | Y | Ν | Y | Y | Y | Ν | Ν | Moderate |
| 26. Takacs and Bus (2016) | Y | Y | Ν | N | Y | Y | Y | Ν | Moderate |
| 27. Willoughby et al. (2015) | Y | Y | Y | N | Y | Y | Y | Ν | High |
| 28. Zhou and Yadav (2017) | Y | Y | Y | Y | Y | Y | Y | Ν | High |
| 29. Zipke (2017) | Y | Y | Y | Y | Y | Y | Ν | Y | High |
| Ratio of studies meeting criterion | 29/29 | 29/29 | 16/29 | 23/29 | 29/29 | 24/29 | 21/29 | 5/29 | |
| Percentage | 100% | 100% | 55% | 79% | 100% | 82% | 72% | 17% | |

Table 4.3 Assessment of Methodological Quality Based on Selected Internal and External Validity Criteria from Troia (1999)

Note: †Each quality variable is weighted based on its importance for ensuring internal or external validity (adapted from Troia, 1999). Quality variables weighted: 3-indicate strong importance and it is considered a fatal flaw if one of these variables are missing, 2-indicate moderate importance, and 1-indicate some importance. A score was assigned of the full weight if all aspects of the variable were present in the study and a 0 if any single aspect of the variable was missing. Based on this adapted system, a sum score of 15-20 and no fatal flaws indicates High Quality, 8-14 indicates Moderate Quality, 0-7 indicates Low Quality.

4.3.5 Data Analysis

The index used to calculate effect sizes was Hedges's g statistic to correct for small sample size (Hedges & Olkin, 2014). Effect size is a standardised statistic that shows the direction (positive or negative) and magnitude of the intervention's effect. This effect size can be found by calculating the mean of the e-book intervention group minus the mean of the comparison group divided by the pooled standard deviation. The formula is:

$$g = \underline{y1} - \underline{y2}$$
sp

with y1, y2, and sp denoting the mean of sample 1, the mean of sample 2, and the pooled standard deviation, respectively (Borenstein et al., 2009). To interpret this effect size (Hedges' g and Cohen's d are interpreted in the same way). Cohen recommended the following rule of thumb: a value of d = 0.00 represents a zero effect, d = 0.20 represents a small effect, d = 0.50 represents a moderate effect, and d = 0.80 represents a large effect (Cohen, 1988). We report Hedges' g as an indicator of effect size, which is provided by the statistical software package Comprehensive Meta-Analysis (Version 3.3) (Borenstein et al., 2013).

In order for all groups to be considered independent, Borenstein et al. (2009) suggests when dealing with multiple comparisons within a study to combine the data of the experimental groups and then compute the overall effect size for the control group versus the merged experimental groups. If studies included more than one treatment or more than one experimental group, the experimental groups were merged, however; they were treated as separate studies when examining outcome measures and interactive/multimedia e-books.

Moderator analyses were performed, using a random effects model, to contrast different categorical study variables: digital book features (interactive, multimedia, static, combinations); adult scaffolding (e-book condition, print book condition or both, or neither); study design (between/within study design); sample characteristics (SES, age range of participants); digital device (tablet versus computer); repetition of story reading (1 or 2 readings, more than two).

4.4 Results

4.4.1 Descriptive Statistics

We found 29 studies with 44 contrasts eligible for inclusion in this meta-analysis. Of the 29 studies, 27 appeared as journal articles, and two were published in the form of a dissertation. The majority of studies were conducted in a school setting (n = 27) and two in a university laboratory. The majority of studies were conducted in English-speaking countries, USA (13), UK (1), and Canada (4). Four studies were conducted in Israel (Hebrew), three in Turkey (Turkish), two in The Netherlands (Dutch), one in Taiwan (Mandarin) and one in Jordan (Arabic). Four studies were published in 2020, two in 2019, seven studies were published in 2017, four in 2015 and another four in 2014, whilst the remaining publications were evenly spread across the other years (2021, 2018, 2016, 2013, 2011, 2010, 2009, 2008). Twenty-four studies were carried out with participants from preschool and kindergarten and five studies with first grade (6-7 years old) and second grade (7-8 years old) children. Ten studies used a standardised test to assess the learning outcomes, whereas 19 studies used new tests that were developed by the authors. Duration of the treatment varied from one to eight weeks, whereas the number of sessions varied from one to 16 with average session duration varying from 6 to 60 minutes. In total 2,317 children participated across all studies, 1,450 as experimental participants and 867 as control participants. All studies had an experimental design (in line with the inclusion criteria). The majority of studies focused on story comprehension yielding 20 effect sizes. Eighteen studies were concerned with vocabulary learning (i.e., receptive, expressive), six studies with phonological awareness, five studies with print awareness, four studies with reading words, and another two studies were focused on spelling. The majority of studies included e-books with various digital features. Thirteen studies used e-books with multimedia features, two dealt with interactive e-books and eleven used interactive and multimedia e-books. Five studies dealt with basic 'unembellished' e-books which included oral narration.

4.4.2 Overall Effects

The associations between language and literacy outcomes and e-books in comparison to print books were examined. Estimates of the overall effect as well as five moderators can be found in Tables 4.4 and 4.5. The available statistics were entered in the Comprehensive Meta-Analysis (Version 3.3) software (Borenstein et al., 2013) to calculate Hedges' *g* for each comparison to correct for small sample size studies. For all included comparisons, an effect size of g = 0.25 was found, which represents a small positive significant effect favouring the e-book condition (k = 29; SE = 0.08; 95% CI = [0.09, 0.42]; p = 0.002). This effect was heterogeneous, Q (28) = 92.88, p = 0.00, $I^2 = 70$ (see Table 4.4).

4.4.3 Heterogeneity

The I-squared statistic (I²) depicts the percentage of the total variability in effect estimates due to heterogeneity (between-study variability) (Borenstein et al., 2009). A high percentage in the I-squared index suggests high heterogeneity. The forest plot (Appendix B) confirmed the high degree of heterogeneity. The I-squared value indicated that 70% of variation in effect size was due to heterogeneity (between-study variability) rather than chance (sampling error).

4.4.4 Publication Bias

Meta-analysis, like all research, has limitations which must be addressed. One of the main limitations of meta-analysis is publication bias. It should be noted that publication bias is not only a limitation to meta-analysis, but to all research in general (Borenstein et al., 2009). In a meta-analysis, a researcher searches the literature for studies dealing with the topic in question. After the meta-analysis is performed the researcher comes to conclusions dependent on the studies included in the meta-analysis. If the studies available for synthesis are not representative or exhaustive, the validity of the conclusions is threatened (Vevea & Woods, 2005). Given the potentially serious implications of publication bias, a number of researchers have suggested statistical methods for detecting and correcting it in the context of meta-analysis

Table 4.4 Statistics for Each Study

| Study | Outcome(s)† | Hedges's g | Standard | 95% confidence | Z-Value | p-value | Study Design | Publication | SES | Device | Adult Support |
|------------------|---------------------|------------|----------|----------------|---------|---------|--------------|--------------|-----------------|---------------------|------------------|
| | | | error | interval | | | | Status | | | |
| Altun (2021) | SC, RV | -0.00 | 0.14 | [-0.28, 0.27] | -0.02 | 0.980 | Between | Journal | - | iPad | |
| Broemmel (2015) | SC | -0.53 | 0.40 | [-1.32, 0.26] | -1.30 | 0.192 | Between | Journal | High & Low | Computer | Both conditions |
| Critelli (2011) | SC, PA, RW | -0.06 | 0.57 | [-1.19, 1.05] | -0.11 | 0.907 | Between | Dissertation | - | Computer | |
| Eng (2020) | SC | 1.00 | 0.25 | [0.50, 1.50] | 3.95 | 0.000 | Within | Journal | - | Tablet | |
| Gong (2009) | РК | 0.42 | 0.23 | [-0.03, 0.88] | 1.79 | 0.072 | Between | Journal | - | Laptop | |
| Homer (2014) | RW, EV | 0.08 | 0.32 | [-0.56, 0.72] | 0.24 | 0.804 | Between | Journal | - | Microsoft's Xbox | Print condition |
| Ihmeideh (2014) | EV, PA, PK | 1.53 | 0.24 | [1.05, 2.02] | 6.26 | 0.000 | Between | Journal | Middle & Low | Computer | Both conditions |
| Karemaker (2017) | SC, EV, RW | 0.20 | 0.22 | [-0.24, 0.65] | 0.90 | 0.367 | Between | Journal | Diverse SES | Computer | |
| Kelley (2017) | SC, V, RV, EV | 0.06 | 0.35 | [-0.63, 0.77] | 0.19 | 0.848 | Between | Journal | Diverse SES | Tablet | |
| Korat (2014a) | EV, RV, V | 0.58 | 0.19 | [0.20, 0.96] | 3.04 | 0.002 | Between | Journal | Middle | Computer | E-book condition |
| Korat (2014b) | EV, V, S | 0.43 | 0.17 | [0.10, 0.77] | 2.57 | 0.010 | Between | Journal | Low | Computer | |
| Kozminsky (2013) | SC, V, PA, PK, S | -0.44 | 0.28 | [-1.00, 0.10] | -1.58 | 0.113 | Between | Journal | - | Computer | Print condition |
| Lee (2020) | EV | 0.42 | 0.20 | [0.03, 0.81] | 2.11 | 0.034 | Within | Journal | Low | iPad | |
| Liao (2020) | РК | 1.01 | 0.28 | [0.45, 1.58] | 3.54 | 0.000 | Between | Journal | - | Laptop | |
| Neuman (2017) | SC, V | -0.04 | 0.31 | [-0.66, 0.58] | -0.12 | 0.900 | Within | Journal | Low | iPad | |
| O'Toole (2015) | SC, RV | 0.28 | 0.20 | [-0.11, 0.67] | 1.40 | 0.159 | Between | Dissertation | - | Tablet | |
| O'Toole (2018) | SC, RV | 0.07 | 0.19 | [-0.31, 0.46] | 0.35 | 0.723 | Between | Journal | - | Tablet | |
| Pearman (2008) | SC | 0.42 | 0.27 | [-0.11, 0.95] | 1.55 | 0.12 | Within | Journal | - | Computer | |
| Reich (2019) | SC, EV | -0.18 | 0.14 | [-0.46, 0.09] | -1.31 | 0.189 | Between | Journal | High | iPad | |
| Richter (2017) | SC | 0.04 | 0.22 | [-0.39, 0.48] | 0.20 | 0.837 | Within | Journal | Middle | iPad | |

(Continued)

Table 4.4 Statistics for Each Study

| Study | Outcome(s)† | Hedges's g | Standard | 95% confidence | Z-Value | p-value | Study Design | Publication | SES | Device | Adult Support |
|--------------------|-------------|------------|----------|----------------|---------|---------|--------------|-------------|-------------|-------------|-----------------|
| | | | error | interval | | | | Status | | | |
| Rvachew (2017) | SC, PA | 0.60 | 0.37 | [-0.13, 1.35] | 1.60 | 0.108 | Within | Journal | Low | iPad | Both conditions |
| Sapsaglam (2020) | SC | -1.09 | 0.46 | [-2.01, -0.17] | -2.32 | 0.020 | Between | Journal | - | Computer to | |
| | | | | | | | | | | projector | |
| Sari (2019) | SC, EV, RV | 0.16 | 0.21 | [-0.26, 0.59] | 0.74 | 0.457 | Between | Journal | Diverse SES | Laptop | |
| Segal-Drori (2010) | PA, RW, PK | 0.72 | 0.22 | [0.29, 1.16] | 3.26 | 0.001 | Between | Journal | Low | Computer | Both conditions |
| Smeets (2015) | SC, RV, EV, | -0.00 | 0.20 | [-0.41, 0.40] | -0.01 | 0.988 | Between | Journal | - | Computer | |
| | V | | | | | | | | | | |
| Takacs (2016) | SC, RV | 0.21 | 0.31 | [-0.40, 0.83] | 0.67 | 0.498 | Within | Journal | Middle | Computer | |
| Willoughby | PA | -0.09 | 0.24 | [-0.58, 0.39] | -0.38 | 0.700 | Between | Journal | - | iPad | |
| (2015) | | | | | | | | | | | |
| Zhou (2017) | SC, RV | 0.36 | 0.23 | [-0.09, 0.82] | 1.55 | 0.121 | Between | Journal | - | iPad | Both conditions |
| Zipke (2017) | SC, V | 0.36 | 0.40 | [-0.42, 1.16] | 0.90 | 0.364 | Within | Journal | Middle and | Computer | Print condition |
| | | | | | | | | | Low | | |
| Random | | 0.25 | 0.08 | [0.09, 0.42] | 3.11 | 0.002 | | | | | |

Note: †SC= story comprehension, V=Vocabulary, RV= receptive vocabulary, EV=expressive vocabulary, PA=phonological awareness, PK=print knowledge, RW=reading words, S=spelling.

Table 4.5 Overall Results of the 29 Studies and Moderator Analyses†

| | | Effec | t Size and 95 | 5% confidenc | e interval | Test of (2-1 | of null Fail) | Heterog | geneity |
|------------------------------------------|----|-------|---------------|----------------|-------------|--------------|------------------|---------|---------|
| | k | g | SE | Lower Limit | Upper Limit | Z-value | P-value | Q | I^2 |
| Overall | 29 | 0.259 | 0.083 | 0.096 | 0.422 | 3.119 | 0.00 | 92.88 | 70 |
| High/Middle SES | 18 | 0.17 | 0.09 | -0.00 | 0.36 | 1.86 | 0.06 | 49.58 | 65 |
| Low or diverse SES | 11 | 0.39 | 0.14 | 0.11 | 0.68 | 2.72 | 0.00 | 33.61 | 70 |
| Tablet/iPad studies | 12 | 0.18 | 0.09 | -0.00 | 0.37 | 1.90 | 0.05 | 24.43 | 55 |
| Computer/Laptop | 17 | 0.29 | 0.12 | 0.04 | 0.54 | 2.34 | 0.01 | 60.98 | 73 |
| High quality rating studies | 21 | 0.26 | 0.08 | 0.98 | 0.42 | 3.15 | 0.00 | 54.14 | 63 |
| Moderate quality rating studies | 8 | 0.20 | 0.25 | -0.29 | 0.70 | 0.79 | 0.42 | 37.82 | 81 |
| 1 or 2 repetitions of story reading | 18 | 0.24 | 0.08 | 0.06 | 0.41 | 2.74 | 0.00 | 41.66 | 59 |
| More than 3 repetitions of story reading | 11 | 0.26 | 0.17 | -0.07 | 0.60 | 1.53 | 0.12 | 48.10 | 79 |
| Studies with children aged 3-6 years | 24 | 0.24 | 0.10 | 0.04 | 0.43 | 2.38 | 0.01 | 89.90 | 74 |
| Studies with children over 6 years old | 5 | 0.35 | 0.09 | 0.16 | 0.55 | 3.58 | 0.00 | 1.54 | 00 |

† Moderator analyses for digital features and adult support are described on separate tables.

4.4.4.1 Funnel Plot. For the investigation of publication and other biases in meta-analyses researchers may use funnel plots as a primary visual tool (Sterne et al., 2005). This can be done by visually inspecting the funnel plot to see if effect sizes are symmetrically distributed in a funnel shape.

Asymmetrical funnel plots may indicate publication bias. As seen in Figure 4.1, small scale studies appear towards the top of the plot and show a higher concentration on the left side of the mean than on the other. However, publication bias is only one of the possible reasons behind asymmetrical plots; there may be other reasons explaining funnel plot asymmetry. Poor methodological design in smaller studies may yield exaggerated estimates of intervention effects (Egger et al., 2003). Funnel plots can be useful for detecting bias, but their interpretation can be tricky, and they leave open the question of how to proceed if publication bias is suspected (Vevea & Woods, 2005). In the case of a small meta-analysis, such as the current one, Lipsey and Wilson (2001) argue that plots may be hard to interpret and caution reviewers against overanalysing funnel plots.

4.4.4.2 The Trim and Fill Method. The "trim and fill" method developed by Duval and Tweedie (2000), after observing an asymmetrical funnel plot, aims at both identifying publication bias and adjusting results for it (Duval, 2005). More specifically, it is used to estimate an average effect size that is corrected for publication bias (Vevea & Woods, 2005).

The number of studies missing from a biased funnel plot are estimated by reflecting on the negative side of the funnel plot, the largest effects, so studies with small size are added to the set (Vevea & Woods, 2005) (Figure 4.2). A new mean effect size with the new small size studies is added in. The final estimate is interpreted as the average (weighted) effect size estimate corrected for publication bias. Duval (2005) suggests that this approach should be seen as a sensitivity analysis of the potential effect that missing studies have had on the observed result.





Figure 4.2 Trim and Fill Funnel Plot Hedges's g and Standard Error



4.4.5 Effect Sizes for Each Outcome

To answer the first and second research questions, effect sizes were inspected to evaluate the differences between e-books and traditional print book presentations in terms of children's language and literacy outcomes (Table 4.6, Table 4.7 and Figure 4.3).

4.4.5.1 Story Comprehension. We found 24 contrasts in 20 articles/reports in which story comprehension was measured from a paper book in comparison to an e-book. The 20 studies assessing story comprehension were based on measures of children's retelling of the story, ten used questions, seven asked the children to retell the story and three utilised a mix of the two measures. For story comprehension, a zero effect of g = 0.05 was found when comparing e-books to traditional print books (k = 20; SE = 0.08; 95% CI = [-0.11, 0.21]; p = 0.54). This effect was heterogeneous, Q(19) = 41.62, p = 0.002, $I^2 = 54$.

4.4.5.2 Vocabulary Learning. All comparisons assessing vocabulary focused on storybased word knowledge were analysed from 18 studies and 28 comparisons. Of the 18 studies interested in vocabulary development, 11 studies assessed expressive vocabulary and nine measured receptive vocabulary. For overall vocabulary learning, a positive significant effect of moderate size was found (g = 0.40; k = 18; SE = 0.14; 95% CI = [0.10, 0.69]; p = 0.00). This effect was heterogeneous, Q (17) = 135.82, p = 0.00, $I^2 = 87$. Specifically, a moderate positive and significant effect was found for expressive vocabulary (g = 0.54; k = 11; SE = 0.23; 95% CI = [0.08, 1.00]; p = 0.02; $I^2 = 91$), and a small positive and significant effect for receptive vocabulary (g = 0.20; k = 9; SE = 0.09; 95% CI = [0.01, 0.39]; p = 0.03; $I^2 = 45$).

| Outcome | Number of | Effect size | Standard | 95% | р |
|---------------------------|-----------|--------------|----------|---------------|-----|
| Measure | included† | (Heages's g) | error | interval | |
| Story Comprehension | 20 | 0.05 | 0.08 | [-0.11, 0.21] | .54 |
| All Vocabulary | 18 | 0.40 | 0.14 | [0.10, 0.69] | .00 |
| Expressive Vocabulary | 11 | 0.54 | 0.23 | [0.08, 1.00] | .02 |
| Receptive Vocabulary | 9 | 0.20 | 0.09 | [0.01, 0.39] | .03 |
| All code related | 11 | 0.28 | 0.16 | [-0.03, 0.59] | .08 |
| Phonological Awareness | 6 | 0.28 | 0.19 | [-0.08, 0.66] | .12 |
| Print Knowledge | 5 | 0.63 | 0.30 | [0.02, 1.23] | .04 |
| Reading Words | 4 | 0.02 | 0.28 | [-0.54, 0.59] | .93 |
| Spelling | 2 | -0.02 | 0.25 | [-0.52, 0.47] | .93 |

Table 4.6 Overall Results for Outcome Measures

[†]Combined e-book interventions against print books in same studies.





Note: SC= story comprehension, RV= receptive vocabulary, EV=expressive vocabulary, PA=phonological awareness, PK=print knowledge, RW=reading words, S=spelling.

| Study | Age | N _{total} (ⁿ e-book + ⁿ print) | Intervention Group | Comparison Group | Measured Outcomes | Effect size for treatments | 95% CI for treatment |
|----------------------------------|-------------|-------------------------------------------------------------------|-------------------------------|---------------------------------------------------|-------------------------------------------|----------------------------|----------------------------|
| Comprehension-Relate | ed Outcomes | | | | | | |
| Altun (2021) | 5 | 196 (97 + 99) | E-book | Print Book | Five story comprehension questions | 0.09 | [-0.18, 0.37] |
| Broemmel (2015) | 4-5 | 24 (10 + 14) | E-book and print book | Print Book | Audiotaped book retellings | -0.53 | [-1.32, 0.26] |
| Critelli (2011) | 4-6 | 10 (5 + 5) | E-book | Print Book | 10-item comprehension subtest | 0.00 | [-1.12, 1.12] |
| Eng (2020) Exper. 1 | 3-5 | 35 | Contingent E-book | Board Book | 10 Story recall questions | 0.70 | [0.03, 1.36] |
| Eng (2020) Exper. 2 | 3-5 | 33 | Contingent E-book | Static E-book | 10 Story recall questions | 1.46 | [0.71, 2.21] |
| Karemaker (2017) Comparison 1 | 5.6-7 | 47 (32 + 15) | E-friend group | Flat E-book | Story comprehension tasks and recall test | 0.07 | [-0.53, 0.67] |
| Karemaker (2017) Comparison 2 | 5.6-7 | 43 (28 + 15) | Dictionary group | Flat E-book | Story comprehension tasks and recall test | -0.32 | [-0.94, 0.29] |
| Kelley (2017) | 3-5 | 30 (15 + 15) | Interactive E-book (Learn) | E-book without interactive features (Watch) | Story retell measure | -0.10 | [-0.80, 0.58] |
| Kozminsky (2013) | 5-6 | 50 (25 + 25) | E-book | Print Book | Plot understanding test | -0.71 | [-1.27, -0.14] |
| Neuman (2017) | 3-4 | 38 | E-book | Print Book | Free recall and Story sequencing | -0.07 | [-0.69, 0.55] |
| O'Toole (2015) | 4 | 100 (50 + 50) | E-book | Print Book | Story comprehension test | 0.14 | [-0.24, 0.53] |
| O'Toole (2018) Comparison 1 | 4 | 50 (25 + 25) | Live E-book | Live Print Book | 7 Open-ended 'wh' questions | 0.03 | [-0.51, 0.57] |
| O'Toole (2018) Comparison 2 | 4 | 50 (25 + 25) | Audio narrated e-book | Audio narrated print book | 7 Open-ended 'wh' questions | -0.13 | [-0.68, 0.41] |
| | | | | | | | (Continued) |

| Table 4.7a Statistics of Each Study for | Comprehension and Outcome | Measures Descriptions |
|-------------------------------------------------|---------------------------|-----------------------|
| | | |

| Study | Age | N _{total} | Intervention Group | Comparison Group | Measured Outcomes | Effect size for treatments | 95% CI for |
|----------------------|------------|---------------------------------------------|------------------------------------------------------------------------|-------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------|-------------------------------|----------------|
| | | (ⁿ e-book + ⁿ print) | 1 | | | | treatment |
| Comprehension-Relate | d Outcomes | | | | | | |
| Pearman (2008) | 7-8 | 54 | E-book | Print Book | Oral retelling | 0.42 | [-0.11, 0.95] |
| Reich (2019) | 3-5 | 200 (100 + 100) | E-book | Print Book | 14 Free recall questions Story Sequence 8 questions on story events and characters | -0.10 | [-0.38, 0.16] |
| Richter (2017) | 3-5 | 79 | E-book | Print book | 9 Recall questions | 0.04 | [-0.39, 0.48] |
| Rvachew (2017) | 5-6 | 28 | E-book | Print Book | Recall of story | 0.35 | [-0.37, 1.07] |
| Sapsaglam (2020) | 5-6 | 20 (10 + 10) | E-book | Print book | 5 Story comprehension questions | -1.09 | [-2.01, -0.17] |
| Sari (2019) | 4-6 | 83 (41 + 42) | E-book with animated illustrations with and without music/sounds | E-book with static illustrations with and without music/sounds | Story comprehension questions | 0.52 | [0.08, 0.95] |
| Smeets (2015) | 4-6 | 53 (36 + 17) | Animated | Static | Retelling story | 0.12 | [-0.44, 0.68] |
| Comparison 1 | | | e-book | e-book | | | |
| Smeets (2015) | 4-6 | 50 (33 + 17) | Interactive | Static | Retelling story | | [-0.52, 0.63] |
| Comparison 2 | | | e-book | e-book | | 0.05 | |
| Takacs (2016) | 4-6 | 39 | Animated e-book | Static e-book | Story retelling | 0.24 | [-0.37, 0.86] |
| Zhou (2017) | 5 | 36 (18 + 18) | Multimedia | Print book | Story comprehension test | 0.06 | [-0.57, 0.70] |
| Comparison 1 | | | e-book | | | | |
| Zhou (2017) | 5 | 36 (18 + 18) | Multimedia e-book | Print book with questions | Story comprehension test | 0.44 | [-0.20, 1.08] |
| Comparison 2 | | | with questioning | | | | |
| Zipke (2017) | 4-5 | 25 | E-book | Print Book | Story comprehension test | -0.42 | [-1.19, 0.34] |

Table 4.7a Statistics of Each Study for Comprehension and Outcome Measures Descriptions

| Study | Age | N _{total} (ⁿ e-book + ⁿ print) | Intervention Group | Comparison Group | Measured Outcomes | Effect size for treatments | 95% CI for treatment |
|----------------------------------|------------|-------------------------------------------------------------------|---------------------------------------|------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------|-------------------------|
| Vocabulary-Relate | d Outcomes | | | | | | |
| Altun (2021) | 5 | 196 (97 + 99) | E-book | Print Book | The Turkish Receptive Language Test | -0.10 | [-0.38, 0.17] |
| Homer (2014) Comparison 1 | 5-7 | 19 (12 + 7) | Kinect with activities | Book reading | Expressive vocabulary: 10 Words from Active Decoding Words | 0.28 | [-0.59, 1.17] |
| Homer (2014) Comparison 2 | 5-7 | 20 (13 + 7) | Kinect without activities | Book reading | Expressive vocabulary: 10 Words from Active Decoding Words | 0.37 | [-0.52, 1.27] |
| Ihmeideh (2014) | 4.7-5.2 | 92 (48 + 44) | E-book | Print Book | Target vocabulary: Ten words were selected from the e-books' text Children were shown ten pictures of objects and were asked to say the word of the object. | n 2.82 | [2.25, 3.40] |
| Karemaker (2017) Comparison 1 | 5.6-7 | 47 (32 + 15) | E-friend E-book | Flat e-book | Expressive vocabulary: Target definitions | -0.43 | [-1.04, 0.17] |
| Karemaker (2017) Comparison 2 | 5.6-7 | 43 (28 + 15) | Dictionary E-book | Flat e-book | Expressive vocabulary: Target definitions | -0.23 | [-0.85, 0.37] |
| | | | | E hash with sut | Expressive vocabulary: Definitional expressive vocabulary test | 0.40 | [-0.29, 1.10] |
| Kelley (2017) | 3-5 | 30 (15 + 15) | Interactive e-book (Learn) | interactive features | Decontextual expressive vocabulary test Receptive vocabulary test | 0.31 | [-0.38, 1.01] |
| | | | | (watch) | Target vocabulary test Target vocabulary used in story retell | -0.33 | [-1.03, 0.36] |
| | | | | E-book without | Expressive vocabulary: Word meaning expressive test | 0.60 | [-0.04, 1.25] |
| Korat (2014a) Comparison 1 | 4-6 | 50 (38 + 12) | E-book with dynamic visual vocabulary | dictionary | Receptive vocabulary test Target words test Target vocabulary used in | 0.56 | [-0.08, 1.20] |
| | | | | | story retell | 0.38 | [-0.25, 1.02] |
| | | | | | Expressive vocabulary: Word meaning expressive test | 0.47 | [-0.17, 1.12] |
| Korat (2014a) Comparison 2 | 4-6 | 49 (37 + 12) | E-book with static visual vocabulary | E-book without dictionary | Receptive vocabulary test Target words test Target vocabulary used in | 0.36 | [-0.27, 1.01] |
| | | | | | story retell | 0.00 | [-0.64, 0.64] |
| | | | | | | | (Continued) |

(Continued)

| Study | Age | N _{total} (ⁿ e-book + ⁿ print) | Intervention Group | Comparison Group | Measured Outcomes | Effect size for treatments | 95% CI for treatment |
|-------------------------------|----------|-------------------------------------------------------------------|--------------------------------------------------------|------------------------------|---------------------------------------------------------------------------------------------------------------------|----------------------------------|-------------------------|
| Vocabulary-Related | Outcomes | | | | | | |
| Korat (2014a) Comparison 3 | 4-6 | 49 (37 + 12) | E-book with adult vocabulary support | E-book without dictionary | Expressive vocabulary: Word meaning expressive test Receptive vocabulary test | 1.26 0.85 | [0.57, 1.95] |
| | | | 5 11 | | Target words test Target vocabulary used in story retell | 0.72 | [0.06, 1.38] |
| Korat (2014b) Comparison 1 | 7-8 | 52 (41 + 11) | E-book with dictionary dynamic visuals (DVs) with | E-book without | Expressive vocabulary: Word Meaning: Expressive test | 1.02 | [0.34, 1.71] |
| - | | | the printed word | dictionary | Target words test Target vocabulary used in story retell | 0.65 | [-0.01, 1.32] |
| Korat (2014b) Comparison 2 | 7-8 | 53 (42 + 11) | E-book with dictionary dynamic visuals (DVs) | E-book without | Expressive vocabulary: Word Meaning: Expressive test | 0.55 | [-0.11, 1.21] |
| Comparison 2 | | | without printed words | dictionary | Target words test Target vocabulary used in story retell | 0.34 | [-0.30, 1.00] |
| Korat (2014b) | 7-8 | 54 (43 + 11) | E-book with dictionary static visuals (SVs) with | E-book without | Expressive vocabulary: Word Meaning: Expressive test | 0.69 | [0.03, 1.36] |
| Comparison 3 | | | printed words | dictionary | Target words test Target vocabulary used in story retell | 0.37 | [-0.28, 1.03] |
| Korat (2014b) | 7-8 | 56 (45 + 11) | E-book with dictionary static visuals (SVs) without | E-book without | Expressive vocabulary: Word Meaning: Expressive test | 0.67 | [0.00, 1.33] |
| Comparison 4 | 7-0 | 50 (45 + 11) | printed words | dictionary | Target words test Target vocabulary used in story retell | 0.19 | [-0.45, 0.84] |
| Kozminsky (2013) | 5-6 | 50 (25 + 25) | E-book | Print Book | Target Vocabulary test Target vocabulary used in story retell | -0.51 | [-1.07, 0.04] |
| Lee (2020) | 6-7 | 100 | E-book with recorded word explanation | Static e-book | Target Vocabulary Express meaning test | 0.42 | [0.03, 0.81] |
| Neuman (2017) | 3-4 | 38 | E-book | Print Book | Vocabulary target test | -0.00 | [-0.63, 0.61] |
| O'Toole (2015) | 4 | 100 (50 + 50) | E-book | Print Book | The Peabody Picture Receptive Vocabulary Test (PPVT-4) (Dunn & Dunn, 2007) | 0.41 | [0.02, 0.80] |
| O'Toole (2018) 1 | 4 | 50 (25 + 25) | Live E-book | Live Print Book | Receptive word learning test | 0.24 | [-0.30, 0.79] |
| O'Toole (2018) 2 | 4 | 50 (25 + 25) | Audio narrated e-book | Audio narrated print | Receptive word learning test | 0.17 | [-0.37, 0.71] |

Table 4.7b Statistics of Each Study for Vocabulary and Outcome Measures Descriptions

| | | | | book | | | |
|--------------|-----|-----------------|--------|------------|----------------------------------|-------|---------------|
| Reich (2019) | 3-5 | 200 (100 + 100) | E-book | Print Book | The vocabulary related questions | -0.26 | [-0.54, 0.01] |
| | | | | | | | |

(Continued)

Table 4.7b Statistics of Each Study for Vocabulary and Outcome Measures Descriptions

| Study | Age | N _{total} (ⁿ e-book + ⁿ print) | Intervention Group | Comparison Measured Outcomes Group | | Effect size for treatments | 95% CI for treatment |
|-----------------------------|------------|-------------------------------------------------------------------|------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------|----------------------------------|--------------------------------|
| Vocabulary-Relate | d Outcomes | | | | | | |
| Sari (2019) | 4-6 | 83 (41 + 42) | E-book with animated illustrations with music/sounds | E-book with static illustrations with music/sounds | Receptive target vocabulary test Expressive target vocabulary test | 0.05 -0.08 | [-0.37, 0.48] [-0.51, 0.33] |
| Smeets (2015) | | | | Static | Expressive vocabulary: TvK Expressive vocabulary test | -0.37 | [-0.94, 0.19] |
| Comparison 1 | 4-6 | 53 (36 + 17) | Animated E-book | e-book | Receptive: PPVT standardised test Target vocabulary test sentence completion task | on -0.23 0.06 | [-0.80, 0.33] [-0.50, 0.63] |
| Smeets (2015) | | | Interactive | Static | Expressive vocabulary: TvK Expressive vocabulary test | -0.08 | [-0.66, 0.49] |
| Comparison 2 | 4-6 | 50 (33 + 17) e-book e-book | e-book | Receptive: PPVT standardised test Target vocabulary test sentence completion task | on -0.07 0.55 | [-0.65, 0.49] [-0.03, 1.13] | |
| Takacs (2016) | 4-6 | 39 | Animated e-book | Static e-book | Vocabulary receptive test | 0.18 | [-0.43, 0.80] |
| Zhou (2017) | 5 | 36 (18 + 18) | Multimedia-book | Print book | Target Receptive vocabulary test | 0.65 | [-0.00, 1.30] |
| Zhou (2017) Comparison 2 | 5 | 36 (18 + 18) | Multimedia e-book with questioning | Print book with questions | Target Receptive vocabulary test | 0.30 | [-0.34, 0.94] |
| Zipke (2017) | 4-5 | 25 | E-book | Print Book | Word Recognition test | 1.16 | [0.33, 1.98] |

| Study | Age | N _{total} (ⁿ e-book + ⁿ print) | Intervention Group | Comparison Group | Measured Outcomes | Effect sizes for treatments | 95% CI for treatment |
|-------------------------------------|---------|-------------------------------------------------------------------|-----------------------------------------------------------------------------|---------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------|--------------------------------|
| Code-Related Ou | itcomes | | | | | | |
| Critelli (2011) | 4-6 | 10 (5 + 5) | E-book | Print book | 10 questions for Phonological awareness 10 reading words | -0.33 0.13 | [-1.46, 0.79] [-0.98, 1.25] |
| Gong (2009) Comparison 1 | 4-4.9 | 32 (24 + 8) | E-book with 'Bouncing Ball' | Static e-book | Print discrimination task - Reading subtest of Wilkinson's (1993) the WRAT-3 | 0.61 | [-0.17, 1.41] |
| Gong (2009) Comparison 2 | 4-4.9 | 32 (24 + 8) | E-book with 'Violation' | Static e-book | Print discrimination task - Reading subtest of Wilkinson's (1993) the WRAT-3 | 0.26 | [-0.52, 1.04] |
| Gong (2009) Comparison 3 | 4-4.9 | 32 (24 + 8) | E-book with 'Action' | Static e-book | Print discrimination task - Reading subtest of Wilkinson's (1993) the WRAT-3 | 0.39 | [-0.39, 1.17] |
| Homer (2014) Comparison 1 | 5-7 | 19 (12 + 7) | Kinect with activities | Book reading | Reading words: 20 Sight Words Active Decoding Words High Frequency Words | -0.25 | [-1.14, 0.64] |
| Homer (2014) Comparison 2 | 5-7 | 20 (13 + 7) | Kinect without activities | Book reading | Reading words: 20 Sight Words Active Decoding Words High Frequency Words | -0.14 | [-1.02, 0.73] |
| Ihmeideh (2014) | 4.7-5.2 | 92 (48 + 44) | E-book | Print Book | Ten questions were developed to measure phonological awareness skills Print Awareness: Clay's (1979) test | 0.78 1.00 | [0.36, 1.20] |
| Karemaker (2017) Comparison 1 | 5.6-7 | 47 (32 + 15) | E-friend E-book | Flat e-book | Target word reading test | -0.47 | [-1.08, 0.13] |
| Karemaker (2017) Comparison 2 | 5.6-7 | 43 (28 + 15) | Dictionary E-book | Flat e-book | Target word reading test | -0.35 | [-0.97, 0.26] |
| Korat (2014b) Comparison 1 | 7-8 | 52 (41 + 11) | E-book with dictionary dynamic visuals (DVs) with the printed word | E-book without dictionary | Word Spelling test | 0.19 | [-0.46, 0.85] |
| Korat (2014b) Comparison 2 | 7-8 | 53 (42 + 11) | E-book with dictionary dynamic visuals (DVs) without printed words | E-book without dictionary | Word Spelling test | 0.17 | [-0.48, 0.82] |

Table 4.7c Statistics of Each Study for Decoding and Outcome Measures Descriptions

(Continued)

| Study | Age | N _{total} (ⁿ e-book + ⁿ print) | Intervention Group | Comparison Group | Measured Outcomes | Effect sizes for treatments | 95% CI for treatment |
|---------------------------------------|--------|-------------------------------------------------------------------|----------------------------------------------------------------------------|-----------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------|--------------------------------------------------|
| Code-Related Ou | tcomes | | | | | | |
| Korat (2014b) Comparison 3 | 7-8 | 54 (43 + 11) | E-book with dictionary static visuals (SVs) with printed words | E-book without dictionary | Word Spelling test | 0.28 | [-0.37, 0.93] |
| Korat (2014b) Comparison 4 | 7-8 | 56 (45 + 11) | E-book with dictionary static visuals (SVs) without printed words | E-book without dictionary | Word Spelling test | 0.10 | [-0.54, 0.75] |
| Kozminsky (2013) | 5-6 | 50 (25 + 25) | E-book | Print Book | Phonological Awareness test Print Knowledge test Orthographic Awareness test All based on existing tests (Blum, 2001; Drori, 1998; Tuvaland Zeiler, 1995) | -0.10 -0.58 -0.33 | [-0.64, 0.44] [-1.14, -0.02] [-0.87, 0.22] |
| Liao (2020) Comparison 1 | 4-5 | 29 (19 + 10) | E-book highlight synchronization | Static Read- only E-book | Preschool Word and Print Awareness (PWPA) test | 0.43 | [-0.32, 1.18] |
| Liao (2020) Comparison 2 | 4-5 | 30 (20 + 10) | E-book print discussion | Static Read- only E-book | Preschool Word and Print Awareness (PWPA) test | 1.91 | [1.03, 2.79] |
| Rvachew (2017) | 5-6 | 28 | E-book | Print Book | Letter sound Phonological tests: matching a spoken sound to the appropriate printed letter, matching pictures of words that share a common rime or onset, identifying a frequently occurring word from a list of three, and identifying a new word on the basis of a familiar first letter. | 0.86 | [0.10, 1.62] |
| Segal-Drori (2010) Comparison 1 | 5-6 | 48 (32 + 16) | E-book without adult instruction | Print Book with adult instruction | Phonological Awareness test Reading 9 words 16 questions dealing with print awareness (Shatil, 2001) | -0.05 -0.64 0.55 | [-0.64, 0.53] [-1.24, -0.04] [-0.04, 1.15] |
| Segal-Drori (2010) Comparison 2 | 5-6 | 48 (32 + 16) | E-book with adult instruction | Print Book with adult instruction | Phonological Awareness test Reading 9 words 16 questions dealing with print awareness (Shatil, 2001) | 0.95 2.76 3.04 | [0.33, 1.57] [1.95, 3.57] [2.19, 3.88] |
| Willoughby (2015) | 3-4 | 63 (33 + 30) | Alphabet e-books | ABC Paper books | Letter-naming task Letter-sound task Test of phonological awareness kindergarten version (TOPA-K; Torgesen & Bryant, 1994) | -0.09 | [-0.58, 0.39] |

Table 4.7c Statistics of Each Study for Decoding and Outcome Measures Descriptions

4.4.5.3 Code-related Literacy Skills. Overall, of the 20 contrasts in 11 studies with coderelated literacy skills as outcome measure, six studies targeted phonological awareness, five print awareness, four word reading skills, and two spelling. The overall effect for the 11 studies measuring the additional effect of technology was positive but small and nonsignificant (g = 0.28; k = 11; SE = 0.16; 95% CI = [-0.03, 0.59]; p = 0.08). The effect was heterogeneous, Q(10) = 41.18, p = 0.00; $I^2 = 75$. Results of code-related skills are presented separately as follows. Most effects were statistically non-significant suggesting that children in the e-book condition performed as well as children in the print condition. Specifically, six studies compared the effects of e-books to print books for *phonological awareness* with a small positive though non-significant effect (g = 0.28; k = 6; SE = 0.19; 95% CI = [-0.08, 0.66]; p = 0.12; $I^2 = 61$). Five studies including nine contrasts were under the category of *print knowledge* with a moderate positive significant effect: (g = 0.63; k =5; SE = 0.30; 95% CI = [0.02, 1.23]; p = 0.04; $I^2 = 86$). Studies investigating the ability of children to *read words* following their intervention showed a non-significant effect (g =0.02; k = 4; SE = 0.28; 95% CI = [-0.54, 0.59]; p = 0.93; $I^2 = 73$). Of the 29 studies, only two studies (with five contrasts) investigated *spelling* revealing a negative effect (g = -0.02; k = 2; SE = 0.25; 95% CI = [-0.52, 0.47]; p = 0.93; $I^2 = 60$).

4.4.6 Multimedia versus Interactive E-books versus "Non-Embellished" E-books

To answer the third research question, the effects of multimedia and interactive features were compared to traditional print book reading. For a summary of the findings, see Table 4.8. As the effects of the different e-book features (e.g., hot-spots, dictionary, animation) on outcome measures were heterogeneous, the differences among stories including only multimedia, only interactive features, and those with both multimedia and interactive features were tested. There were 20 experimental interventions from 13 studies which investigated the effects of e-books with multimedia features (overall g = 0.38; SE = 0.13; 95% CI = [0.12, 0.63]; p = 0.00; $I^2 = 75$), two studies with interactive e-books (g = 0.24; SE = 0.19; 95% CI = [-0.14, 0.63]; p = 0.21; $I^2 = 00$), and eleven studies with 15 contrasts with e-books including both interactive and multimedia features (overall g = 0.26; SE = 0.12; 95% CI = [0.02, 0.49]; p = 0.03; $I^2 = 62$). Overall, the effects for e-books with multimedia features (overall g = 0.26; SE = 0.12; 95% CI = [0.02, 0.49]; p = 0.03; $I^2 = 62$). Overall, the effects for e-books with multimedia features were found to be more beneficial when compared to print book reading. The other two categories were statistically non-significant as indicated by the confidence intervals. The meta-analysis also coded studies with non-embellished, basic e-

books without any multimedia and interactive features (k = 5). For all included contrasts, an effect size of g = -0.15 was found, which represents a small negative but non-significant effect (SE = 0.21; 95% CI = [-0.56, 0.26]; p = 0.47; $I^2 = 59$).

4.4.6.1 Story Comprehension. Stories including both multimedia and interactive features (k = 7) had a small positive but non-significant effect on story comprehension compared to print books, g = 0.21, (95% CI = [-0.07, 0.50]). The same non-significant effect for story comprehension was found in multimedia stories as well. Interactive stories (k = 2) and basic non-embellished stories (k = 5) yielded significant negative effects for story comprehension favouring the print condition.

4.4.6.2 Vocabulary Learning. Multimedia-only stories for expressive and receptive vocabulary yielded a positive moderate but non-significant effect of g = 0.46 (k = 9; SE = 0.24; 95% CI = [-0.01, 0.94]; p = 0.06; $I^2 = 91$). For e-books including both multimedia and interactive features a small positive but non-significant effect of g = 0.11 (k = 5; SE = 0.16; 95% CI = [-0.20, 0.43]; p = 0.47) was observed. Specifically, for expressive vocabulary, which had the most contrasts in both groups, multimedia-only e-books yielded a moderate positive and significant effect of g = 0.60 (k = 7; 95% CI = [-0.03, 1.24]). An effectively zero effect was observed for multimedia-interactive e-books (g = -0.01; k = 4; 95% CI = [-0.32, 0.30]).

4.4.6.3 Code-related Literacy Skills. Multimedia-only stories showed an additional positive significant advantage for e-books over print books when code-related skills were concerned (specifically print knowledge k = 3, spelling k = 1, phonological awareness k = 2), g = 0.63, SE = 0.18, 95% CI = [0.28, 0.99], p = .00, $I^2 = 62$.

However, it was not possible to test the difference between interactive-only stories and multimedia-only stories, as there was only one contrast for interactive-only stories. E-book stories with both multimedia and interactive features yielded positive non-significant effects in all outcomes when compared to print books.

| | Number of | Effect | | 95% | |
|----------------------------|-------------|--------|----------|----------------|-----|
| | comparisons | size | Standard | confidence | |
| Type of technology story | included | (g) | error | interval | р |
| Multimedia E-books only | 13† | 0.38 | 0.13 | [0.12, 0.63] | .00 |
| Story Comprehension | 7 | 0.15 | 0.09 | [-0.03, 0.35] | .11 |
| Vocabulary | 9† | 0.46 | 0.24 | [-0.01, 0.94] | .06 |
| Receptive vocabulary | 5 | 0.09 | 0.15 | [-0.20, 0.39] | .54 |
| Expressive vocabulary | 7 | 0.60 | 0.32 | [-0.03, 1.24] | .06 |
| Code related | 5† | 0.63 | 0.18 | [0.28, 0.99] | .00 |
| Phonological Awareness | 2 | 0.80 | 0.18 | [0.43, 1.17] | .00 |
| Print Knowledge | 3 | 0.81 | 0.19 | [0.44, 1.19] | .00 |
| Reading Words | 0 | - | - | - | - |
| Spelling | 1 | 0.18 | 0.16 | [-0.14, 0.52] | .26 |
| E-books with both | 11† | 0.26 | 0.12 | [0.02, 0.49] | .03 |
| Interactive and Multimedia | | | | | |
| features | 7 | 0.21 | 0.14 | [-0.07, 0.50] | .14 |
| Story Comprehension | 5† | 0.11 | 0.16 | [-0.20, 0.43] | .47 |
| Vocabulary | 3 | 0.22 | 0.18 | [-0.13, 0.58] | .21 |
| Receptive vocabulary | 4 | -0.01 | 0.16 | [-0.32, 0.30] | .94 |
| Expressive vocabulary | 4 | 0.23 | 0.22 | [-0.20, 0.67] | .29 |
| Code related | 2 | 0.14 | 0.22 | [-0.29, 0.57] | .51 |
| Phonological Awareness | 2 | 0.83 | 0.42 | [0.00, 1.66] | .04 |
| Print Knowledge | 2 | 0.24 | 0.38 | [-0.50, 0.98] | .52 |
| Reading Words | 0 | - | - | - | - |
| Spelling | | | | | |
| Interactive E-books only | 2† | 0.24 | 0.19 | [-0.14, 0.63] | .21 |
| Story Comprehension | 2 | -0.19 | 0.19 | [-0.57, 0.18] | .31 |
| Receptive Vocabulary | 0 | - | - | - | - |
| Expressive Vocabulary | 1 | 1.16 | 0.23 | [0.70, 1.63] | .00 |
| Phonological | 0 | - | - | - | - |
| Awareness | 0 | - | - | - | - |
| Print Knowledge | 1 | -0.43 | 0.22 | [-0.87, 0.00] | .05 |
| Reading Words | 0 | - | - | - | - |
| Spelling | | | | | |
| E-books without | 5† | -0.15 | 0.21 | [-0.56, 0.26] | .47 |
| Multimedia and Interactive | | | | | |
| features (static/basic) | 5 | -0.27 | 0.21 | [-0.69, 0.14] | .19 |
| Story Comprehension | 2 | 0.31 | 0.14 | [0.03, 0.58] | .02 |
| Receptive Vocabulary | 0 | - | - | - | - |
| Expressive Vocabulary | 2 | -0.14 | 0.25 | [-0.63, 0.34] | .56 |
| Phonological Awareness | 1 | -0.58 | 0.28 | [-1.14, -0.02] | .03 |
| Print Knowledge | 1 | 0.13 | 0.57 | [-0.98, 1.25] | .81 |
| Reading Words | 1 | -0.33 | 0.28 | [-0.87, 0.22] | .24 |
| Spelling | | | | | |

Table 4.8 E-books With and Without Multimedia and Interactive Features on Language

 and Literacy Outcomes

 Spening

 †Combined e-book interventions against print books in same studies with same (or without) technological features.

4.4.7 Animation and Hotspots

When comparing studies with only animation as an e-book feature from the list of studies with various multimedia features, ten studies evaluated the effects of animation in comparison to traditional print books. The overall effect was small in size (g = 0.32) and significant (95% CI = [0.01, 0.63]; k = 11; SE = 0.15; p = 0.04; $I^2 = 76$). In regard to vocabulary development (k = 9), when animation in e-books was compared to a print-like condition, a positive non-significant effect was found (g = 0.49; 95% CI = [-0.04, 1.03]; SE = 0.27; p = 0.07; $I^2 = 91$). Animation did not make a significant effect (g = 0.15; 95% CI = [-0.03, 0.35]; k = 7).

Only two studies evaluated hotspots in e-books without other digital features (Karemaker et al., 2017; Zipke, 2017), thus we were not able to evaluate the results of hotspots as a feature. However, we were able to evaluate ten studies which included hotspots combined with animation in their e-book condition and found an overall significant positive effect on language and literacy development (k = 10; g = 0.20; SE = 0.10; 95% CI = [0.00, 0.40]; p = 0.04; $I^2 = 45$). For separate outcome measures, the results were small and non-significant: story comprehension (g = 0.01; 95% CI = [-0.15, 0.17]; k = 7); vocabulary development (g = 0.49; 95% CI = [-0.10, 1.08]; k = 5). For code-related skills, only four studies assessed the effects of hotspots in combination with animation (g = 0.14; 95% CI = [-0.39, 0.68]).

4.4.8 Adult Scaffolding/Mediation Effects

We calculated the effects of adult support in either the print condition, the e-book condition or both conditions in order to find out the effects of adult scaffolding during storybook reading (see Table 4.9). The calculated effect of adult scaffolding in any condition produced a positive non-significant effect (g = 0.44; k = 9; SE = 0.23; 95% CI = [-0.01, 0.89]; p = 0.05; $I^2 = 81$). In five out of 29 studies, e-books were assessed for language and literacy outcomes when adult support was present in both the paper and e-book condition. In this small set of studies, we found a large positive effect favouring the e-book condition. However, the confidence interval crosses zero, showing that this effect was not statistically different from zero, indicating that children in the e-book condition performed as well as children in the print book condition (g = 0.86, 95% CI = [-0.01,

1.73]). When comparing the effects of adult scaffolding in the print book condition with the independent use of e-books with various features, we found a negative effect favouring the print book condition (g = -0.07; k=4; SE = 0.16; 95% CI = [-0.40, 0.24]; p = 0.63; $I^2 = 88$), meaning that adult scaffolding in print books outperformed the digital features offered in e-books.

| | | Effect | | 95% | |
|-----------------------------------------------|------------|-------------|----------|----------------|-----------|
| | Number | size | Standard | confidence | |
| | of studies | (g) | error | interval | р |
| Studios with adult support aithor in a book | 0.4 | 0.44 | 0.22 | [0 01 0 201 | 05 |
| Studies with adult support either in e-book | 94 | 0.44 | 0.25 | [-0.01, 0.89] | .05 |
| condition, or in print condition or both | | | | | |
| | ~ . | 0.06 | 0.44 | F 0 01 1 701 | 0.7 |
| Studies with adult support in both conditions | ЪŢ | 0.86 | 0.44 | [-0.01, 1./3] | .05 |
| (e-book and print are supported) | 2 | 0.10 | 0.00 | | 60 |
| Story Comprehension | 3 | 0.12 | 0.29 | [-0.45, 0.70] | .68 |
| Vocabulary | 2 | 1.57 | 1.26 | [-0.90, 4.06] | .21 |
| Code related | 3 | 1.30 | 0.42 | [0.48, 2.13] | .00 |
| Studies with adult support in print condition | 4† | -0.07 | 0.16 | [-0.40, 0.24] | .63 |
| only | 2 | -0.61 | 0.23 | [-1.06, -0.15] | .00 |
| Story Comprehension | 3 | 0.29 | 0.47 | [-0.63, 1.21] | .54 |
| Vocabulary | 3 | -0.19 | 0.17 | [-0.54, 0.14] | .25 |
| Code related | | | | | |
| Studies with adult support in e-book | 1 | 1.00 | 0.24 | [0.52, 1.49] | .00 |
| condition only (not in print book) | | | | | |
| Studies without adult support in e-book | 26† | 0.18 | 0.06 | [0.04, 0.31] | .00 |
| condition (independent use) and all included | | | | | |
| comparisons in print condition (with and | | | | | |
| without support print) | 18† | 0.04 | 0.08 | [-0.12, 0.22] | .59 |
| Story Comprehension | 17† | 0.25 | 0.10 | [0.04, 0.45] | .01 |
| Vocabulary | 9† | 0.06 | 0.14 | [-0.23, 0.35] | .68 |
| Code related | | | | | |
| Studies without adult support in both | 22† | 0.21 | 0.07 | [0.06, 0.35] | .00 |
| conditions (independent use of e-book versus | | | | | |
| only narration of print book – no | | | | | |
| scaffolding) | 16† | 0.10 | 0.08 | [-0.05, 0.27] | .19 |
| Story Comprehension | 14† | 0.25 | 0.10 | [0.04, 0.46] | .01 |
| Vocabulary | 6† | 0.17 | 0.20 | [-0.21, 0.57] | .37 |
| Code related | | | | | |

| Table 4.9 Studies | With and Wi | ithout Adult | Scaffolding , | Mediation | in Print and | d E-books |
|--------------------------|-------------|--------------|---------------|-----------|--------------|-----------|
| | | | | | | |

[†]Combined e-book interventions against print books in same studies.

4.4.9 Adult Scaffolding for Multimedia and Interactive E-books

Due to the small number of studies, there was no way to control for adult scaffolding (multimedia e-books and adult support in both conditions k = 3, interactive e-books and adult support k = 1, interactive and multimedia e-book conditions and adult support k = 3).

4.5 Discussion

The aims of this meta-analysis were, (1) to evaluate the effectiveness of e-books in comparison to traditional print books when they are being read by an adult either with scaffolding or handled independently; (2) to investigate whether outcome measures are better developed with the support of digital features in e-books in comparison to print book reading with and without adult support; and (3) to explore which type of digital features offer valuable and educational support to our young learners. The overall experimental data surrounding the efficiency of e-books are rather controversial, and there is no general agreement regarding their use for the development of language and literacy skills. Therefore, this meta-analysis offers the most recent data in an effort to better understand and evaluate e-books used by young children today. It has been hypothesised, that the specific features in e-books may mirror or resemble teacher/adult scaffolding when the adult is reading a traditional print book in an effort to develop young children's story comprehension, vocabulary knowledge and code-related skills. The overall impact of the 29 primary studies published between 2008 and 2021 appear modest as the results produced a small positive overall effect (g = 0.25; 95% CI = [0.09, 0.42]) favouring the ebook condition, however effect sizes vary across studies and subgroups.

The results differ when they are based on separate effects of language or literacy outcomes. It is apparent that various cognitive factors and content elements can influence how well young children learn from storybooks in any format (Nueman et al., 2021; Richter & Courage, 2017). The variation among the procedures and outcome measures that have been reported in the literature further complicate the evaluation of effectiveness (Takacs et al., 2015). Therefore, the effect of story comprehension in e-books versus adult narration was investigated in print conditions and this result offers a unique contribution. When comparing these findings with those of the two existing meta-analyses (Takacs et al., 2014; Zucker et al., 2009), the most striking difference relates to the effect on story comprehension. Both prior meta-analyses found positive and significant effects on story comprehension (g = 0.05; 95% CI = [-0.11, 0.21]), indicating that children in the e-book conditions performed as well as children in the print conditions. This finding is consistent with that of Liman Kaban and Karadeniz (2021) who evaluated the use of different digital reading media in 96 students. Despite the use of diverse reading media in

the experimental and control groups, there was no significant difference in their reading comprehension levels.

Vocabulary development results are in line with results reported in Takacs et al.'s (2014) meta-analysis as well as recent experimental studies (e.g., Korat et al., 2022a; Korat et al., 2017; Lee, 2020) supporting that digital books are more effective than print books especially for expressive vocabulary. Expressive vocabulary is one of the most difficult types of vocabulary for a child to develop. Children need to express the meaning of new words using definitions and by explaining the full meaning of a word. If an e-book is able to achieve this based on the digital features of storybooks, then this result is incredibly promising. Indeed, half of the studies with expressive vocabulary as an outcome had a medium to large positive effect size favouring the e-book condition. This might be explained by the type of features included within e-books. For example, the study by Korat et al. (2014b) produced large positive effect sizes in all four comparisons (see Table 4.7b) as their e-book included a dynamic and static dictionary. Regardless of digital design effects (static versus dynamic), the dictionary offered the opportunity to the students to observe unknown words in the form of an image by giving them an immediate answer to an unknown word. This example contradicts the effectiveness of audio narration of definitions and pronunciations as used in the study of Karemaker et al. (2017), which included an auditory dictionary, resulting in negative effects on expressive vocabulary (g = -0.23; 95% CI = [-0.85, 0.37]). The study by Korat et al. (2014b) is in line with Mayer's (2005) cognitive theory of multimedia learning which argues that multimedia narrations paired with images generate verbal and visual mental representations that are elaborated in working memory and can be combined with prior information to form new knowledge, such as expressive vocabulary development.

When comparing multimedia additions in e-books our findings revealed a small positive effect favouring the e-book condition, which showed that e-books are more beneficial than encounters with traditional print books with or without adult scaffolding. This effect was significant for code-related skills (g = 0.63; 95% CI = [0.28, 0.99]. However, previous experimental literature evaluating multimedia e-books in relation to the development of code-related skills (e.g., Arslan-Ari & Ari, 2022; Yow & Priyashri, 2019), supported that e-book reading may not be an activity that develops phonological awareness, print knowledge and reading words as children tend to pay more attention to the pictures rather
than the text. A possible explanation for this might be that teacher mediation was evident in the studies included in this meta-analysis, indicating that teacher scaffolding with the use of multimedia e-books has a positive impact on the acquisition of code-related skills.

In regard to interactive features added to e-books, only two studies tested the difference between an interactive story and a non-interactive (print or static) story (Karemaker et al., 2017; Zipke, 2017). These studies showed no significant differences between interactive and non-interactive stories. A similar non-significant effect was also identified in e-book studies that included both interactive and multimedia features (k = 11). However, when analysing the results of e-books without any features, such as static e-books with basic oral narration, compared to children listening to print book narrations, results showed a small negative though non-significant effect (g = -0.15; 95% CI = [-0.56, 0.26]). This finding, although preliminary due to the small number of studies with "unembellished" e-books (k = 5), might suggest that interactive and multimedia components, either separately or added together in an e-book, might serve as a scaffolding tool for young children, thus suggesting that e-books may have the potential to scaffold children's literacy development.

When analysing the effects of digital features as scaffolding tools in comparison to print books without adult support, the results reveal a small positive, non-significant effect indicating that unsupported e-books offer the same results as unsupported print book reading. However, in accordance with an earlier meta-analysis (Takacs et al., 2014), the same non-significant result was expected to be found when adult mediation was present during print book reading in comparison to independent use of e-books. This meta-analysis revealed that adult mediation during print book reading outperformed digital features when e-books were used independently. This result is based on only four studies; the positive effects of adult-child interactions during storytelling sessions cannot be replaced by digital features. In light of this finding, e-books with support were compared with print books with support, revealing that adult scaffolding combined with digital features offered in ebooks facilitated children's learning. It has been argued in past studies that adults may be less likely to scaffold children's understanding and elicit dialogic reading during e-book narrations in contrast to print book storytelling sessions (Eggleston et al., 2022; Munzer et al., 2019; Parish-Morris et al., 2013). However, the results of the current study are in agreement with Neumann's (2020a) findings which showed that the teacher discussed more and read the e-book for a longer period of time than the print book. The current

findings suggest that the combination of digital features and adult content-related discussions is significantly more effective in comparison to adult support offered during print book reading. This study supports evidence from recent observations (e.g., Korat et al., 2022b) which also found a considerable advantage for children who read the e-book with adult scaffolding in terms of story comprehension. Overall, the findings suggest that adult scaffolding in combination with e-books' digital features may be important factors to consider during adult-child storybook reading.

4.5.1 Limitations

The main limitations of this study are the relatively small number of primary studies that were available and the range of outcomes that these studies included, as this means that the influence of a number of key features could not be evaluated robustly. First, the analysis was unable to test whether the quality of scaffolding affects learning and comprehension. It would be helpful to evaluate different types of scaffolding and review any differing effects. Second, discrete effects from different kinds of multimedia (such as animation, music and sound effects) as well as particular interactive features (such as games, hotspots, and the availability of a dictionary function) could not be assessed. The embedded features included in the e-books of the meta-analysed studies did not address a particular feature in more than one or two studies but were categorised as more broadly as multimedia and/or interactive. Third, a number of studies investigated the effects of ebooks on reading words, phonological awareness or spelling, in contrast to other studies which focused on story comprehension and vocabulary learning, even though the acquisition of code-related skills are equally important to acquire at this sensitive period of a child's educational development. Finally, adult scaffolding in both e-books and print books was limited. In the current study digital features in combination with adult scaffolding outperformed adult mediated print books; future studies could further explore this new finding in order to consolidate our findings. A further limitation is the relatively small samples in some studies which compounds the issues listed above. The meta-analysis may also be susceptible to publication bias, as, whilst every effort was made to identify relevant studies systematically, the scale of these studies may have further compounded publication bias, as smaller studies would need larger effect sizes to reach a threshold for statistical significance. Finally, a number of the included studies have limitations in terms of their methodological rigour or in the details of their reporting which may also have

affected the findings from the overall synthesis.

4.5.2 Practical Implications

In this meta-analysis e-books and print books were examined, the digital features of ebooks as well as the support given by adults while reading a story for the promotion of language and literacy development. The findings of this study offer a number of important implications for future practice and the construction of e-books. Story comprehension is a major skill which young children need to acquire. Reading an e-book and expecting children to understand the content and meaning of the story independently might be a little misguided. Most e-book studies in this meta-analysis involved e-books with the incorporation of hotspots. This means that while the children are listening to the story, they have the opportunity to place their finger on the screen and listen to words shown on the specific page of the story. This feature could be assumed as an advantage as words are repeated, and the child has the opportunity to listen to the pronunciation of the word and connect the word to the picture. However, this action might also be a source of interference while the narrator is reading the story and as a result, hotspots direct the child's attention towards the hotspots, and the child is unable to absorb important and necessary information in order to understand the content of the story. The distracting hotspots placed on the page of an e-book are considered extraneous information, which may have a negative impact on children's learning processes resulting in cognitive load. The child's natural urge to press each hotspot numerous times might come from the notion that children expect interactivity because they are accustomed to game-like activities, which diverts their attention away from the storyline. There is, therefore, a definite need for digital book designers to assess the specifics incorporated in e-books and the timing of appearance of these features in order to maximise children's ability to learn.

Furthermore, our meta-analysis highlights the fact that when children are using an e-book without adult scaffolding their learning and development depends on the variety of features embedded in e-books which, more often than not, are contrasting one another. Each feature, depending on the frequency and timing included in an e-book, may harm or support children's learning. Looking closely at the studies included in this meta-analysis, the studies that offered multiple interactive and multimedia support did not benefit children as well as the print book condition with adult support. In addition, children in the e-book conditions were responsible for their own learning whereas children that were supported

in the print conditions were guided by an educated and experienced adult using instructional scaffolding to reach his/her educational goals. Teachers are extremely important in promoting children's e-book reading experiences (Christ et al., 2018; Reich et al., 2016). The teacher knows and understands the objectives of the educational activity and uses storytelling and scaffolding techniques to get the most out of every story. Most commercially available e-books do not include features that resemble external support (e.g., adult scaffolding) and as a result e-books may not be able to replace adult scaffolding. In order to reduce cognitive load designers should take into consideration Mayer's multimedia principles. Designing with the coherence principle in mind entails the exclusion of simultaneous presentation of words, sounds and animation, as text and animation together may overload the visual channel (Mayer, 2005). An implication of these findings is that both digital features and adult scaffolding should be used to expand on existing designs of digital storybook apps for educational purposes.

4.5.3 Implications for Future Research

It is recommended that future interventions investigate in more detail the effects of specific interactive features. The past decade has seen very few studies concerning interactivity in e-books. As technology evolves, interactive features evolve as well. In addition, future studies should include a wider spectrum of outcome variables. Most studies have focused on vocabulary development and story comprehension; thus, it is recommended that future studies include a wider range of outcome variables and could include network metaanalysis approaches. Indeed, outcomes, such as reading, spelling, phonological awareness, and print awareness, are equally important to children's literacy development. This metaanalysis is the first that included code-related skills in the analysis showing that coderelated outcomes have the potential to be developed through e-book reading, even though the number of studies included was small (especially for spelling and reading words). It is furthermore suggested that future studies include follow-up testing, by re-administering their outcome measures, to investigate whether the effects of e-book and print book reading interventions extend beyond the period of the intervention and may have measurable long-term effects. It should be noted that none of the studies included in this meta-analysis performed a re-administration of post-tests to evaluate retention. It is recommended that future studies investigate the role of scaffolding and the type of scaffolding offered to young children during e-book reading sessions. It is further recommended the investigation of separate digital effects with adult support in either the print condition, the e-book condition or both. It is important to evaluate each feature separately and to evaluate and make clear which of these features, if any, could lead to long-term benefits to learning.

4.6 Conclusion

In the present research synthesis, which included 29 studies and 2,317 young children, evidence was found to suggest that e-book story telling sessions are able to support language and literacy learning equally well as traditional print book storytelling sessions. For the development of children's vocabulary, e-books are more effective than print books, especially in expressive vocabulary. Small and non-significant effects were found for story comprehension. A key concern about e-books for preschoolers is that their multimedia and interactive features could be a distraction during reading and might interfere with children's learning. However, a small positive significant effect was found for multimedia e-books in contrast to e-books with minimal to no digital features. Our results confirmed that comprehension, vocabulary and code-related skills are facilitated as a result of e-book reading and were not affected negatively by e-books with both interactive and multimedia features.

The meta-analysis was also able to assess the effect of e-book reading in comparison to print book reading with and without the presence of an adult. Adult support was either in the print condition, the e-book condition or both conditions. Adult support was not instructional in the sense of providing guidance on how to use a tablet/computer or to keep a child on task. Instead, the adult asked questions regarding the story either during or after teaching, explained unknown words, discussed the content of the story based on children's own personal experiences and played games relating to comprehension of the story content, vocabulary and code related skills. The independent use of e-books showed a small positive non-significant effect in comparison to print book reading with and without the presence of an adult. The findings suggest that the specific digital features offered by the studies included in the meta-analysis are equal to adult scaffolding and appear to have the ability to scaffold children's learning. Moreover, adult support combined with e-book features outperformed all comparison conditions. The evidence from this study suggests

that activities such as storybook reading accompanied with adult-child dialogic interactions offer a unique experience and play an important role in language and literacy development - regardless of book type.

The next chapter of this thesis introduces the second study, which is an experimental investigation that assesses the effectiveness of e-books and print books for young English as an Additional Language (EAL) learners. The findings of the meta-analysis presented in Chapter 4 indicated that providing adult support during e-book reading, such as asking questions and engaging in discussions about story content, has demonstrated significant advantages for children's language and literacy development. Therefore, the purpose of Study 2 was to evaluate the effectiveness of two scaffolding techniques, namely interactive and performance teaching approaches, in facilitating the learning process of non-native English-speaking children. The primary objective was to determine whether or not these techniques are effective when children are exposed to either an electronic book or a print book during shared storybook reading. Study 2 aimed to assess the efficacy of two scaffolding techniques, namely interactive and performance teaching styles, in facilitating the learning process of children learning English as an additional language. The primary focus was to determine if these teaching styles are effective when children are exposed to either an e-book or a print book during shared storybook reading.

Chapter 5: Experimental Study

The purpose of this study was to investigate two different book mediums—the electronic book and the print book—while reading two different stories twice using two different teaching styles—the interactive teaching style and the performance teaching style. The children (n=60) who participated in the study were learning English as an additional language, from diverse backgrounds who resided in Cyprus and attended a private English school. Both the book medium and the teaching styles were assessed on measures linked to the development of vocabulary (expressive and receptive vocabulary) through the use of target words featured in the stories as well as story comprehension (implicit and explicit).

The meta-analysis presented in Chapter 4 (i.e., Savva et al., 2022) examined the specific impact of e-book reading on language and literacy development of young children when compared to traditional reading of print books with or without adult scaffolding in a structured and controlled environment. Within the meta-analysis, the authors evaluated specific e-book features and their effect in terms of supporting children's learning progress. Some aspects of e-books were considered more beneficial than others when looking at language and literacy outcomes. The effect sizes of the studies included in the meta-analysis, as well as the overall literature review that this study is based upon, offer some conclusions as to what constitutes a "perfect" e-book. Perfect in this sense describes the specific features included in e-books which were found to promote language and literacy development for typical developing children. In this study, these "perfect" e-books were put to the test for children learning English as an additional language.

5.1 The "Perfect" E-book

Today, children use new devices, from smartphones to tablets to personal computers, and they are exposed to digital literacy environments from an early age (Bers et al., 2014; Marsh et al., 2005; Parette et al., 2010). They experience digital storybooks along with printed ones (Altun, 2017; Rideout, 2011). Digital storybooks provide multimedia enhancements: animated pictures, background sounds, music, sound effects and video as well as interactive features such as hotspots, games, and dictionaries (Bus et al., 2015; Korat, 2010; Verhallen et al., 2006). A key concern about e-books for preschoolers is that

their multimedia and interactive features could be a distraction during reading and might interfere with children's learning. However, apart from being vibrant, many children's digital books offer a rich source for learning as is confirmed by meta-analytic findings (e.g., Savva et al., 2022; Takacs et al., 2014; Zucker et al., 2009). The findings of the metaanalysis by Savva et al. (2022), whose objective and methods are explained in the preceding chapter (Chapter 4), are summarised below in order to understand the concept of the "perfect" e-book.

After reviewing all meta-analyses on e-books for young children and evaluating the effect sizes of included studies, conclusions can be drawn as to the quality of e-books and the quantity of embedded features (see Table 5.1 and Table 5.2). Table 5.1 presents the studies included in the meta-analysis by Savva et al. (2022) that exhibited a positive effect size favouring the e-book condition for vocabulary and story compression. Table 5.2 presents the studies that reported a negative effect size favouring the print condition for vocabulary and story compression. There are some variables to consider when evaluating the e-books included in the meta-analysis. The first variable is the comparison condition. For a study to be included in the meta-analysis its experimental design had to include a comparison condition in which the same or a similar story was presented using a print book format or an e-book with static-illustrations without multimedia or interactive features. The second variable had to do with multimedia and interactive features. The e-books included in the meta-analysis had various features and the e-books were eventually categorised in four categories: (a) e-books with multimedia features, (b) e-books with interactive features, (c) e-books that had both interactive and multimedia features, and (d) e-books with no features versus a traditional print book. The third variable that could have an effect on the results was relating to the presence of adult scaffolding. Adult support was either in the print condition, the e-book condition or both conditions. Adult support was not instructional in the sense of providing guidance on how to use a tablet/computer or to keep a child on task. Instead, the adult asked questions regarding the story either during or after the reading session, explained unknown words, discussed the content of the story based on children's own personal experiences and played games relating to comprehension of the story content, for the development of vocabulary, story comprehension and code related skills.

| Study and Year | Sample Size | E-book Intervention | Features included in e-book | Outcome measures | Effect Size Hedges's g | 95% CI for treatment |
|-------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------|----------------------------------------------------------------------------------|-------------------------------------------------------------------|---------------------------|-------------------------|
| | Sile | | e book | | incuges 5 g | CI for treatment |
| Eng, Tomasic & Thiessen (2020) Experiment 1 | % 35 Contingent E-book Animations that activated conting on the child's vocalisations | | Animations that activated contingently on the child's vocalisations | Story Comprehension | 0.70 | [0.03, 1.36] |
| Eng, Tomasic & Thiessen (2020) Experiment 2 | Tomasic &33Contingssen (2020)E-boocriment 2 | | Animations that activated contingently on the child's vocalisations | Story Comprehension | 1.46 | [0.71, 2.21] |
| Ihmeideh (2014) | 92 | E-book | Animated illustrations | Target vocabulary | 2.82 | [2.25, 3.40] |
| Korat, Levin, Atishkin & Turgeman (2014) | 49 | E-book with adult vocabulary support | Dictionary of target words | Expressive vocabulary | 1.26 0.85 | [0.57, 1.95] |
| Turgeman (2014) | | | | Receptive vocabulary Target vocabulary used in story retell | 0.72 | [0.19, 1.51] |
| Korat, Levin, Ben- Shabt, Shneor & Bokovza (2014) Comparison 1 | 52 | E-book with dictionary dynamic visuals with the printed word | Animated representations of target verbs and printed words shown on screen | Expressive vocabulary | 1.02 | [0.34, 1.71] |
| Korat, Levin, Ben- Shabt, Shneor & Bokovza (2014) Comparison 3 | 54 | E-book with dictionary static visuals with printed words | Static representations of target verbs and printed words shown on screen | Expressive vocabulary | 0.69 | [0.03, 1.36] |
| Korat, Levin, Ben- Shabt, Shneor & Bokovza (2014) Comparison 4 | 56 | E-book with dictionary static visuals without printed words | Static representations of target verbs | Expressive vocabulary | 0.67 | [0.00, 1.33 |
| Lee (2020) | 100 | E-book with recorded word explanation | Multimedia future: Recorded word explanations | Expressive vocabulary | 0.42 | [0.03, 0.81] |
| O'Toole (2015) | 100 | E-book | - - | Receptive vocabulary | 0.41 | [0.02, 0.80] |
| Zipke (2017) | 7) 25 E-book Interactive hotspots on pictures to hear the object name | | Receptive vocabulary | 1.16 | [0.33, 1.98] | |

Table 5.1 Vocabulary Development and Story Comprehension - What Worked?

| Study and Year | Sample Size | E-book Intervention | Features included in e-book | Outcome measures | Effect Size | 95% |
|----------------------------|-------------|------------------------------------|-------------------------------------------------------|-----------------------|-------------|------------------|
| | | | | | Hedges's g | CI for treatment |
| Altun (2021) | 196 | E-book | Some animation, background | Receptive Vocabulary | -0.10 | [-0.38, 0.17] |
| | | | music, character movements. | | | |
| Broemmel et al. (2015) | 24 | E-book | Animation, rich narration, sound effects and music | Story Comprehension | -0.53 | [-1.32, 0.26] |
| Karemaker (2017) | 58 | Interactive E-book (n=28) | Dictionary button highlighted challenging words. By | Expressive Vocabulary | -0.23 | [-0.85, 0.37] |
| | | | clicking on a word children | Story Comprehension | -0.32 | |
| | | | could listen to pronunciation and definition. | | | [-0.94, 0.29] |
| Kelley and Kinney (2017) | 30 | Interactive and Animated | Six embedded questions | Target vocabulary | -0.33 | [-1.03, 0.36] |
| | | E-book | related to the story and | | 0.10 | |
| | | | children can tap pictures to | Story Comprehension | -0.10 | [0 90 0 59] |
| | | | Animated characters. | | | [-0.80, 0.58] |
| Kozminsky (2013) | 50 | E-book | - | Story Comprehension | -0.71 | [-1.27, -0.14] |
| | | | | Target Vocabulary | -0.51 | [1 07 0 04] |
| O'Toole (2018) | 50 | Audio narrated e-book | - | Story Comprehension | -0.13 | [-0.68, 0.41] |
| | | | | | | [,] |
| Reich et al. (2019) | 200 | Interactive and Animated E-book | Interactive: 6 hotspots per | Story Comprehension | -0.10 | [-0.38, 0.16] |
| | | E book | repeated the word or provided | Receptive Vocabulary | -0.26 | |
| | | | some animation | | | [-0.54, 0.01] |
| Sari et al. 2019 | 81 | E-book with music and sounds | Animated characters and objects, sound effects, music | Receptive Vocabulary | -0.77 | [-1.23, -0.32] |
| | | | | Expressive Vocabulary | -0.31 | [-0.75, 0.12] |
| | | | | Story Comprehension | -0.10 | [-0.53, 0.32] |
| | | | | | | (Continued) |

Table 5.2 Vocabulary Development and Story Comprehension - What Did Not Work?

117

| Study and Year | Sample Size | E-book Intervention | Features included in e-book | Outcome measures | Effect Size | 95% |
|-------------------------------------|-------------|------------------------------------|------------------------------------------------|-----------------------|-------------|------------------|
| | | | | | Hedges's g | CI for treatment |
| Smeets & Bus (2015) Comparison 2 | 50 | Interactive and Animated E-book | Hotspots shape as a magnifying glass to search | Expressive Vocabulary | -0.08 | [-0.66, 0.49] |
| | | | for unknown words | Receptive Vocabulary | | |
| | | | | | -0.07 | [-0.65, 0.49] |
| | | | Animated characters and | | | |
| | | | objects, sound effects, music | | | |
| Smeets & Bus (2015) | 53 | Animated | Animated characters and | Expressive Vocabulary | -0.37 | [-0.94, 0.19] |
| Comparison 1 | | E-book | objects, sound effects, music | | | |
| | | | | Receptive Vocabulary | | |
| | | | | | -0.23 | [-0.80, 0.33] |
| Zipke (2017) | 25 | E-book | Interactive hotspots on | Story comprehension | -0.42 | [-1.19, 0.34] |
| | | | pictures to hear the object | | | |
| | | | name | | | |

Table 5.2 Vocabulary Development and Story Comprehension - What Did Not Work?

It is suggested that the "perfect e-book" could include a story with animated illustrations and/or characters which highlight the objectives of the study. For example, if a study is seeking to develop a child's vocabulary, then it is suggested to include animated representations of target words for vocabulary development and story comprehension. The animated features should be closely related and helpful towards achieving the study objectives. For example, the study by Eng et al. (2020) used a within-subject design with 3 experiments with ninety 3- to 5-year-old children. Children were presented with a contingent e-book (animations that activated contingently on the child's vocalisations) and two non-contingent control conditions: a print book and a static e-book. The results revealed that children's recall scores were significantly higher in the contingent book condition compared to the other two conditions for children with less developed attention regulation for story comprehension. Therefore, the first requirement an e-book must meet in order to be deemed effective is that animations need to be congruent to the text as it enhances children's story recall by orienting their attention to nonverbal information that matches the story narrative. This result was also confirmed by the study of Korat et al. (2014b) that utilised different functions on an e-book (i.e., a dictionary with static visuals with and without printed focal words and dynamic visuals with and without printed focal words), for 250 second graders from low- socioeconomic status families. All experimental conditions produced positive effect sizes, favouring the e-book conditions with the dictionary in comparison to the control condition with an e-book without a dictionary. The comparison with the highest effect size favouring the e-book condition was the e-book with animation (dynamic visuals with the printed words; g=1.02, CI=[0.34, 1.71) which shows that animation combined with a dictionary improves children's expressive vocabulary skills.

It should be noted, however, that the popular expression by Ludwig Mies van der Rohe *less is more* may apply to an e-book's effectiveness as well, according to the results of the studies presentenced in Tables 5.1 and 5.2. Studies that attempted to evaluate the effectiveness of animation combined with background music and sounds in comparison to print or static storybooks were found to have negative effects favouring the print book condition (e.g., Altun, 2021; Broemmel et al., 2015; Sari et al., 2019; Smeets & Bus, 2015 [comparison 1]). It appears that the combination of animation, music, and sounds may interfere with the narrative, preventing the development of children's vocabulary and comprehension skills. Sari et al.'s (2019) study included 99 typically developing children

aged 4–6 years from two public kindergartens in Turkey. Their primary objective was to distinguish between the impacts of visual and auditory enhancements in digital books. The music and sounds featured in the e-books matched the story events. The music, which was present the majority of the time and took a new form in each scene, conveyed the mood of the main characters or emphasised dramatic events. Their results revealed that animated pictures supported children's story comprehension (g=0.52, CI=[0.08, 0.95]) while music and background sounds did not (g=-0.10, CI=[-0.53, 0.32]). They argued that music and background features may require additional operations that surpass a child's capability for working memory. When additional auditory information is presented, this uses the auditory channel just as the narrative, thereby competing with the narration for limited processing capacity in the auditory channel (Mayer, 2009). In accordance with Mayer's (2009) coherence model, when processing capacity is used to process the music and sounds, there may be less capacity available for paying attention to the narration, making inferences, and linking the narration with the visual information, which could result in poorer performance on story comprehension tests (Moreno & Mayer, 2000). Background music and sounds cause superfluous processing that uses limited working memory capacity in the auditory channel, consequently diminishing children's ability to deduce the meaning of new words from their context (Kirschner, 2002).

Another feature highlighted in the meta-analysis as non-effective are interactive features such as hotspots (e.g., Karemaker et al., 2017; Kelley & Kinney, 2017; Reich et al., 2019; Smeets & Bus, 2015 [comparison 2]; Zipke, 2017). In Karemaker et al. (2017), the researchers included an e-book with dictionary button with challenging words - by clicking on a word children could listen to the pronunciation and definition. This did not improve children's expressive vocabulary and story comprehension as opposed to the comparison condition which was an e-book without any hotspots (named Flat e-book by the authors). One reason might be that when children press on hotspots and digital buttons, it may interfere with their ability to concentrate. The study of Smeets and Bus (2015) included an interactive animated e-book with hotspots shaped as a magnifying glass for children to search for unknown words. Authors noted that the interactivity included in the e-books were highly controlled: the researchers inserted only one hotspot per interactive moment, children were allowed to play for a maximum of 30 seconds with a hotspot, and the number of hotspot moments were restricted to four per session. Even though the interaction with the hotspots was monitored and supervised by adults, the results for expressive and

receptive vocabulary favoured the control condition (the static e-book). The inclusion of animation and interaction in an e-book may also account for the fact that the aforementioned study's results did not result in the development of children's skills. Other studies that included both interactive and animated features within an e-book produced similar results (e.g., Kelley & Kinney, 2017; Reich et al., 2019). Overall, every feature included in an e-book should have an educational goal.

Moving on to another important variable that may interfere with the results is adult scaffolding. Korat et al. (2014a) included in their study an e-book with a dictionary. The dictionary was displayed on four distinct channels, each of which provided the same oral reading of the text and oral explanations of target verbs, which automatically appeared on the screen once the full page had been read by the narrator. The four channels of the dictionary were as follows: reading the book (1) with adults' support for target verbs; (2) with a dynamic visuals dictionary of target verbs; (3) with a static visuals dictionary of target verbs; (4) with no dictionary support. The condition that outperformed all the other conditions was the *E-book with adult vocabulary support*. This suggests that regardless of the features offered in an e-book, it cannot outperform an adult reading the e-book with the child. The meta-analysis presented in this thesis revealed that adult mediation during print book reading outperformed digital features when e-books were used independently. However, when e-books with adult support were compared with print books with support, the combination of digital features and adult content-related discussions were significantly more effective in comparison to adult support offered during print book reading. Therefore, another requirement for achieving the best outcomes when reading an e-book is that adult scaffolding is essential in the e-book condition for both vocabulary development and story comprehension. Based on the above findings, the current study sought to search for an online e-book to be included in the experimental study that is considered the "perfect" e-book.

The features which are considered beneficial in this context are:

• Narration and highlighted text. This is a typical feature included in e-books, but it is very important that the language used is pronounced clearly and words are highlighted while appearing on the screen.

• Story with animated illustrations and/or characters and/or some animated scenes which highlight the objective of the study (e.g., if the study is seeking to evaluate the learning of a specific word such as "jump" then the character of the story might be seen jumping while saying the word. This type of animation is clearly connected with the objective of the study).

• Embedded questions regarding the story content (as a digital feature or adults asking the questions). This will help children with story comprehension and also make them think about various events in the story.

• All included features should be selected and restricted to exactly what a study is seeking to answer. For instance, the e-book should not include 10 hotspots on any page, but rather every hotspot should be carefully selected and should be aligned with the goals of the study.

5.2 Research Questions

The purpose of the study was to compare the impact of e-books versus print books on children learning English as an additional language in terms of vocabulary and story comprehension. The study also compared two reading styles alongside the two media (print and digital) to assess whether word learning and comprehension were enhanced. The following research questions were tested:

1. Which type of book medium (i.e., e-book versus print book) produces better results in the development of vocabulary and story comprehension for children learning English as an additional language (EAL)?

2. Which teaching style (i.e., interactive versus performance) better facilitates young children's (learning English as an additional language) vocabulary and story comprehension development?

5.3 Method

5.3.1 Participants

Participants in this study were 60 typically developing children (32 boys and 28 girls) aged 3-7 years (M = 5.26 years, SD = 1.16) from different socioeconomic status families residing in Larnaca, Cyprus. Children were recruited from four classrooms in one private English school attending the following classes: Pre-Kindergarten (3-4 years old), Kindergarten (4-5 years old), Pre-Primary (5-6 years old) and First Grade (7 years old). In order for children to be eligible for participation it was important for children not to speak English as their first language. Thus, children learning English as an additional language, with no apparent developmental delays, were eligible for participation.

Ethics approval was obtained before recruiting participants. To obtain written informed consent for eligible children, the school principal was provided with written information about the study explaining the scientific goals and the opportunity for the children to listen to children's stories. The activities were deemed as being part of common classroom practice and therefore no additional parental consent was sought. Only when the school principal gave written consent for the children's participation were the children included in the study. All procedures were approved by the ethical review board of Durham University.

An a priori power analysis was conducted using G*Power version 3.1.9.7 (Faul et al., 2007) to determine the minimum sample size required to answer the research questions of the study. Results indicated the required sample size to achieve 80% power for detecting a medium effect, at a significance criterion of $\alpha = .05$, was N = 60 for ANOVA analyses. Thus, the obtained sample size of N = 60 is adequate to test the study's research questions. Therefore, the study included 60 children recruited from the four classes mentioned above and the main languages spoken by participants were: 17 children spoke Greek, 14 children spoke Russian, eight spoke Spanish, five children spoke Hebrew, three children spoke Arabic, five French, two Indian, one German, one Turkish, one Bulgarian, two Czech, and one Slovak. The children attended the school from 0 to 2 years, with the majority of children entering the school the year that the intervention took place (60%).

5.3.2 Covid-19 Pandemic

The initial plan was to acquire the necessary data for this study from schools in Cyprus between September 2019 and June 2020. Throughout the first few months of the school year, I met with school principals and teachers to brief them on the research project. However, I was unable to continue due to Covid-19, as Cyprus was under full lockdown after the New Year. The Ministry of Education instructed school closures due to an unknown virus whose duration was unknown. We believed that the school would be closed for two weeks. Schools did not reopen for the remainder of the school year. When the new school year began, after receiving ethical approval from the University and consent from the school principal (see Appendix C), I was still concerned whether or not I should begin collecting data, since I knew it would take time to complete the research and the situation with Covid-19 was still unpredictable. In November of 2020, a second lockdown was implemented because so many Cypriots had contracted the virus; however, schools for younger children (preschools and nurseries) remained open with numerous safety measures in place. In December 2020, I began testing the study tests on a group of children whose first language was English, in order to evaluate the instructions and other issues that may arise during official administration (for more information, see Section 5.3.4.7 Testing the measures). I concluded my pilot testing at the end of 2020 and was prepared to begin administering the pre-tests as soon as the school reopened after the Christmas break. When schools reopened in January 2021, I started administering the standardised test in addition to the receptive and expressive vocabulary tests. Although high school students were not required to attend school, the government insisted that children of lower ages do so since it was difficult for parents to work from home while caring for their children. Preschools functioned as normal as possible as a result of the implementation of several protocols and the frequent testing of all staff members. This provided me the opportunity to start gathering my data at the school that I was working at, however I was unable to visit other schools in the area in order to gather a larger sample size.

By March, I gathered all of the pre-tests and then began the intervention. At the beginning of April 2021, the intervention sessions were completed, and I began conducting the post-tests (i.e., expressive and receptive tests, story comprehension tests). During that time, I was infected with the virus as well. In May-June 2021, I re-administered the vocabulary

tests alongside the transfer task. The school year ended with the collection of all post-tests. Nonetheless, I wanted to conduct the standardised test again. Therefore, I waited until September 2021 for the school to begin. Fortunately for the study, due to the pandemic, few of the children of study participants' families elected to migrate overseas. The majority of families remained in Cyprus, allowing me to collect data from 49 of 60 children. It is also important to note that some participants left Cyprus before May 2021, which prevented me from collecting their second post-tests.

Moreover, in this study, I had a dual role as both a researcher and practitioner/assessor. As a result of Covid-19 restrictions, there was a lack of a second assessor present throughout the administration of the research tools. Nevertheless, during the development of the research tools for this study, I was mindful of the fact that it was conducted during the midst of the Covid-19 pandemic. Given the uncertainty of when the situation would return to normal, I took great care to ensure that my research tools and the procedure I implemented were both accurate and straightforward. The research tools of the study may be found in *Section 5.3.4* and Appendix N, which provides information on the process of selecting research tools.

Overall, it was a challenging academic year for me as well as for the participants of the study. There were several protocols in place, and things were not operating as normally or as easily as before. I was required to wear a mask in order to speak and read to the children, and it was imperative that they not be in close proximity. In addition, during the post-test administration, some children were absent due to the virus or being in close contact of an infected family member, making it difficult to gather all data on the same day or week as the rest of the participants in the condition. Moreover, during the design phase of the project, I intended to visit more schools in the Larnaca region to recruit additional data from additional participants to increase the size of my sample. However, strict protocols prevented me from entering other schools, so I could only collect data from the pupils at the school where I worked. This is a limitation of the study, as it was intended to have at least twice as many children in each condition.

5.3.3 Materials

5.3.3.1 The Storybooks Included in the Study (Print and E-book). During the story selection process, the following actions took place: a) the stories had to be age-appropriate for all age groups; b) the storybook needed to be available in an e-book format; c) the content of the e-book had to include a set of criteria established as predictors of language and literacy learning (Savva et al., 2022). Two stories were selected: The Kissing Hand (Audrey Penn, 1993) and Elmer and Rose (David McKee, 2005). The Kissing Hand is a story about a young raccoon named Chester who was anxious about leaving home and going to school. So, his mother decided to share a "family secret" with him: the "kissing hand". She gave him a kiss on his paw and told him to press the kiss to his cheek whenever he missed her. The Elmer and Rose story is part of a series involving Elmer the elephant. The synopsis of the story is that Grandpa Eldo asks Elmer and Wilbur (Elmer's cousin) to help a young elephant (i.e., Rose) find her way back to her herd - and they get a shock when they see she is pink! No wonder she is called Rose. But there is an even greater surprise in store when they reach her herd - because everyone single one of them is pink.

The print storybooks were available as e-books by Oceanhouse Media (http://www.oceanhousemedia.com). The Oceanhouse Media e-book apps are much like the corresponding print books: the text and illustrations are preserved, and there is little animation. An oral reading is available, with sound effects, and text highlighting that follows the narration. After the text on a page has been read, the e-book reader can tap on individual pictures in the text to listen to the name of the object or action pop up and hear the word spoken again. No other hotspots, animation, or games appear in the app. In sum, the reader is in control of this e-book and all of the media effects are integral to the storyline. Children were asked if they knew the stories and only one child knew The Kissing Hand and none of them listened to Elmer and Rose before, although Elmer is a beloved character for children of this age group (3-7 years old) as the series includes more than 30 different stories with Elmer the elephant.

5.3.3.2 Teaching Styles: Interactive and Performance. Two different teaching styles were selected for each story condition, the interactive style which engages children in discussion during the narration of the story, and the performance style which presents stories with questions and discussion at the beginning and end of the story. More specifically, the interactive style entails that before reading the book, necessary

information about the book was provided, which focused on the features of the book. The book was read with an emphasis on discussion, the images were shown to the children, and the children's responses to the text and illustrations were discussed. Before, during, and after reading, the degree of interaction between the teacher and the children increased about the book by employing extratextual talk practices. Thus, the participants were given a range of opportunity to generate their own interpretations of the text. After finishing the story, the participants' reactions to the book's plot were discussed. Regarding the performance style, before reading the story, the children were asked questions and made predictions about its content. The relevant details regarding the book and its author(s) were given. The story was read without any discussions, and the illustrations were presented to the pupils. After the reading, the teacher and the children discussed the story's plot.

For each story a "plan" was created with the use of questions and discussion points to be asked for each page of the story. Identical questions were posed to children regardless of book medium. The children were asked the same questions, at the same time point, while being exposed to either an e-book or a print book. During the interactive style these questions were asked during appropriate points. For example, when the front cover was shown, children were asked the following questions: "*Can you read the title? The title is The kissing hand. What animals do you see on the front cover? What are the animals doing?*". During the performance teaching style, the children were asked these questions at the beginning and at end of the story. Detailed plans of each story can be found in Appendix D.

In the intervention sessions each condition listened to the two stories twice. Condition 1 listened to the e-book with interactive style adult scaffolding (n=15; 7 boys and 8 girls); condition 2 listened to the e-book with performance style adult scaffolding (n=15; 9 boys and 6 girls); condition 3 listened to the print book with interactive style adult scaffolding (n=15; 8 boys and 7 girls), and condition 4 listened to the print book with performance style adult scaffolding (n=15; 8 boys and 7 girls). All participants had initial experience with computers individually and in small groups as part of the curriculum. All children were speaking English as an additional language, who according to the teachers' records had normal learning development without any recognised impairments.

5.3.4 Research Tools

This study focused on children's receptive and expressive vocabulary knowledge, as well as story comprehension by applying the following research tools:

5.3.4.1 English Language Proficiency Test. The following standardised assessment was administered to the children before and after the intervention. The standardised test was the PLS-5 UK Preschool Language Scale - Fifth Edition (https://www.pearsonclinical.co.uk/store/ukassessments/en/Store/Professional-

Assessments/Speech-%26-Language/Preschool-Language-Scale---Fifth-

Edition/p/P100009263.html). The PLS-5 UK offers a comprehensive developmental language assessment, with items that range from pre-verbal, interaction-based skills to emerging language and early literacy. The test tests children's auditory comprehension and expressive communication. The standardised test consists of 67 items, and administration begins at a point specified by the child's age at the time of administration. The administration of the test is halted after the child made five consecutive errors. Then the raw scores are calculated by recording the last item administered and subtracting the zero scores from this item number. The difference is the raw score for either auditory comprehension or expressive communication. The total language score is calculated by adding these scores together. The standard score of each child was used in the analysis. The results obtained from the PLS-5 English Language Proficiency Test supported the separation of the participants into different groups, ensuring that the groups were comparable in terms of their language competency. In order to assure the rigour of the study, it was crucial to employ a standardised test to separate the participants into groups, as opposed to a test created by the researcher.

5.3.4.2 English Language 5-Point Scale. The class teachers of the children who participated in the study were provided with five language levels, ranging from Low Ability to High Ability, and asked to indicate where they considered their pupil fell regarding children's use of the English language (see Appendix E). The five language levels were the following:

- Low Ability: Student uses gestures more often than words to communicate.
- Upper Low Ability: Student communicates using one word (sometimes with pronouns: I, he, she).

• Medium Ability: Student produces sentences using 2-3 words with pronouns and nouns or verb. Grammar/vocabulary might not be correct.

• Upper Medium Ability: Student uses 4 words to communicate combined with pronouns, verbs and nouns. Able to describe situations. Grammar/vocabulary might not be correct.

• High Ability: Student speaks fluently using correct grammar and vocabulary.

Each item was scored on a scale from 1 to 5 (i.e., Low Ability=1; Upper Low Ability=2; Medium Ability=3; Upper Medium Ability=4; High Ability=5).

5.3.4.3 Receptive and Expressive Vocabulary Tests of Words Included in the Stories.

The following tests were developed for this study and were administered to the children before and after the intervention: (a) Receptive target vocabulary test. To assess children's receptive knowledge of word meanings, children were asked to select the target word out of four pictures. The correct image was presented among three distractors, all of which were selected from the same storybook. The experimenter scored the receptive word learning test (1 point = correct; 0 points = incorrect). Children aged 3-4 years answered 20 words and 4–7-year-olds 30 words. Internal reliability assessed by Cronbach's alpha was .86 for the pretest and .86 for the posttest. (b) Expressive target vocabulary test. This test assessed children's knowledge of the meaning of target words. The children had to explain in their own words the meaning of words. The child was asked to explain the meanings of 15 words for ages 4-7 years and 8 words for ages 3-4 years that appeared in the two ebooks. Definitions were scored on a scale of 0 to 4: 0 was given if the child said, "I don't know"; 1 was given if the child provided a definition for a word that sounded like the target word or rhymed with the target word; 2 was given when the child used the target word within a sentence; 3 was given when the child used a word similar or close to the target word; 4 was given when the child provided a precise definition for the word (the receptive and expressive vocabulary tests can be found in Appendix F and Appendix G, respectively). That is, a general definition that is not related to the specific context of the story. Internal reliability assessed by Cronbach's alpha was .85 for the pretest and .87 for the posttest.

5.3.4.4 Implicit Story Comprehension Test (ISC). The Implicit Story Comprehension test (ISC) includes making inferences about the feelings of the story protagonists, causal relations, predictions, and theme. ISC requires deeper story comprehension skills. Six

questions were asked for implicit comprehension (feelings, causal inference, dialogue, prediction, and theme). Each question was scored on a scale from 0 to 2 (0: No answer / I don't know; 1: Partially correct answer. Only one part of the question answered; 2: Complete correct answer) (the test is presented in Appendix H). Internal reliability assessed by Cronbach's alpha was 0.75.

5.3.4.5 Explicit Story Comprehension (ESC) Recall with Prompts Test. The Explicit Story Comprehension test (ESC) is related to the identification of the story elements (characters, setting, initiating event, problem, and solution). Children were to freely recall the answers. If they could not recall the answers correctly, the researcher read a list of multiple-choice options to select from. Scoring: children were asked 7 questions to freely recall the answer. If they were not able to answer the open-ended question, the children were offered three multiple-choice options. Considering free recall as a higher level of memory, the researcher calculated a total score with recall based on accuracy and whether the child freely recalled events or needed prompting. For this, correct answers that are freely recalled were given 2 points, correct answers derived from the multiple-choice options were given 1 point and incorrect answers received 0 points (the test is presented in Appendix I). The explicit story comprehension questions were intended to assess surface story comprehension, and the implicit questions addressed deeper comprehension. Internal reliability assessed by Cronbach's alpha was 0.80. The rationale for choosing these research tools which were designed by the experimenter, may be found in the Appendix N.

5.3.4.6 Transfer Task. The children listened to a simple and short story created by the author in order to assess their ability to transfer some of the knowledge acquired during the interventions to other story listening sessions (information regarding the Lexile level of the text are presented in Appendix J). With this task, children were put in a position where they needed to identify important target words and story-related characteristics (e.g., setting, main characters, problem-solution) in order to determine if their experience with the other two stories led to the development of skills that can be observed when reading other stories. Therefore, children listened to a short and simple story with one sentence on each page (12 pages). The story was created by the experimenter and the illustrations by students not involved in the study. After listening to the story, in print or e-book format depending on their condition, the children were asked seven questions, and the

experimenter recorded their answers (the test is presented in Appendix J). Each question was scored on a scale from 0 to 2 (0: No answer / I don't know / random answer; 1: Partially correct answer. Only one part of the question answered; 2: Complete correct answer). Internal reliability assessed by Cronbach's alpha was 0.82.

5.3.4.7 Testing the Measures. A pilot session was conducted to ensure that the research measures utilised for this study were suitable to children of this age learning English as an additional language. The measures were administered to a group of children (n=7; ages: 4-6 years) whose first language was English in order to determine if the measurements and questions asked were understandable to English-speaking children. In fact, children scored quite well on all study instruments and obtained high scores for the majority of the items included in the tests. Some modifications were made to the wording in order to simplify instructions for EAL children.

5.3.5 Design

5.3.5.1 Research Design. This study is a 2x2x2 factorial design (a between-within design); 2 (e-book, print book) by 2 (teaching style 1/teaching style 2) is between-person and 2 repeated measures (pre-test, post-test 1), where participants were allocated to conditions, taking into account a balanced representation of age, gender, and language proficiency. Prior to the reading sessions, the participants received the Preschool Language Scale - Fifth Edition (PLS-5 UK), which was used to check if the participants had similar language knowledge among the four conditions. Furthermore, the student's teacher answered a language 5-point scale test for each of their students'. The scale included five language related statements starting from low ability to high ability. After analysing the results of these two tests, the children were allocated to the four experimental conditions. The participants were randomly allocated to conditions, in a stratified way to control for the following key variables, i.e., English level based on the standardised test, gender, age and teachers' 5-point language scale level.

One week after the pre-tests, participants in all four conditions received two reading sessions of two storybooks over four-weeks. Variations in reading procedures can affect the quantity of words children learn. For instance, children acquire more vocabulary from

a book that has been read aloud multiple times as opposed to only once (McLeod & McDade, 2011; Sénéchal, 1997). Biemiller (2004) and Penno et al. (2002) expressed concern that children become bored after three repeats of a story, which is the number of repetitions that is most frequently suggested as effective. In addition, children are not exposed to new contexts for the target words during repeated readings. Therefore, this study engaged participants in two reading sessions, a procedure that has been used in previous research on young children's learning from story reading and found to be sufficient for generating observable learning outcomes (Verhallen et al., 2006). At the end of the second reading session, children received the post-tests. See tables following (Table 5.3 and Table 5.4) for the description of each condition.

| Condition | Description |
|--------------------------------------------------|---------------------------------------------------------|
| "Perfect" e-book with teaching style 1 | Children listen to the e-book while teacher scaffolds |
| (<i>n</i> =15) | using teaching style 1 (Performance style) |
| "Perfect" e-book with teaching style 2 | Children listen to the e-book while teacher scaffolds |
| (<i>n</i> =15) | using teaching style 2 (Interactive style) |
| Print book with teaching style 1 (<i>n</i> =15) | Children listen to the teacher reading print book while |
| | teacher scaffolds using teaching style 1 (Performance |
| | style) |
| Print book with teaching style 2 (<i>n</i> =15) | Children listen to the teacher reading print book while |
| | teacher scaffolds using teaching style 2 (Interactive |
| | style) |

 Table 5.3 Description of Conditions

5.3.6 Procedure

5.3.6.1 Pretest Phase. During the pretest phase the children received the Preschool Language Scale - Fifth Edition (PLS-5 UK) which is a standardised test in order to evaluate participants' language skills at baseline. The test lasted around 20-40 minutes depending on children's age and language capabilities. Then the children received the receptive and expressive tests for target vocabulary which took around 10 minutes to administer. All participants were tested individually in a quiet classroom prior to the intervention. This study was conducted within the participant's school setting, which may have increased the children's focus on the adult reader due to the demanding nature of the setting and situation. The experimenter was responsible for reading the stories to the children during the print storybook intervention, observing the e-book conditions, providing adult scaffolding (i.e., interactive and performance style), and administering the research tools.

5.3.6.2 Intervention

E-book Condition. After the pre-test data were collected, four sessions followed in which children listened to the two stories presented on a tablet. During small group sessions with two children, they heard one story per session totalling to four sessions (each story was heard twice). The decision to have the children work in pairs was based on research indicating that learning in small groups had a significant impact on improving young children's literacy (Karweit & Wasik, 1996; Leslie & Allen, 1999), including research on e-books (Shamir, 2009; Shamir & Korat, 2007; Shamir et al., 2008). The reason that the sessions were spread over four days (one story a day) was because each comparison condition included children as young as 3 years old and it is hard for them to concentrate for long periods, considering that each story session included scaffolding.

All children who participated in the study had initial experience with tablets individually and in small groups as part of the curriculum. The children's activity with the tablet took place in a quiet classroom, working two at a time. Each child was shown how the application operates and was given the following instructions: "You will listen to a story on the tablet. You will have to touch the screen to change the page or go back and you may also touch the characters". The children were given technical support if needed. During the performance teaching style, the experimenter would ask questions relating to the story before and after the story was read. During the interactive style, the experimenter asked questions during the narration of the story and making conversation after a page or a sentence was complete. Each reading session lasted about 20-25 minutes in total. Children were instructed not to talk with each other while they listened to the narration, and they rarely did.

Print-book Condition (Control). Children were in small groups of 6-8 children each time. The experimenter sat in front of them holding the print-story book narrating the story while showing the pictures at the same time. Depending on the experimental condition (performance teaching style versus interactive teaching style), the experimenter asked questions during the storytelling session, or at the beginning and end, respectively. Each reading session lasted about 20-25 minutes in total. The print book condition is designated as the control group as print storybook reading is considered a standard and common

procedure in early years settings, and it is a language and literacy development activity conducted on a daily basis in classes with young children aged 3-7 with the aim to develop listening and communication skills, vocabulary, grammar, improve concentration, memory, imagination and creativity.

5.3.6.3 Posttest Phase. After the intervention sessions, two sessions took place for posttesting which included the (a) story comprehension test (implicit and explicit tests), and (b) the expressive and receptive knowledge of the target words. The expressive vocabulary test was administered first to avoid exposing children to the target words beforehand and applied the receptive test second. The posttests took place 2 days after completing the two reading sessions of each story. Each posttest (vocabulary tests and story comprehension) lasted about 15-30 minutes in all. After one month the experimenter administered the receptive and expressive tests again. Finally, the transfer task was administered in order to evaluate children's learnable skills developed during the interventions. After six months children received the Preschool Language Scale - Fifth Edition (PLS-5 UK) again. The procedure is summarised in Table 5.4.

| Pretest Phase | Standardised test PLS-5 UK |
|--------------------------|------------------------------------------------------------|
| | Receptive and Expressive Vocabulary tests |
| Story reading Phase | Day 1 |
| | Listening to The Kissing Hand |
| | Day 2 |
| | Listening to Elmer and Rose |
| | Day 3 |
| | Listening to The Kissing Hand |
| | Day 4 |
| | Listening to Elmer and Rose |
| Posttest Phase | Day 5-6 |
| | Receptive and Expressive Vocabulary tests |
| | Story Comprehension tests |
| 2 nd Posttest | After 1 month |
| | Receptive and Expressive Vocabulary tests |
| Transfer Task | When posttests were completed |
| Standardised test | After 6-7 months from first administration of the PLS-5 UK |
| PLS-5 UK | |

Table 5.4 An Example of the Schedule of the Experiment

5.3.7 Statistical Analyses

The IBM SPSS Statistics Version 28 software (IBM Corp., Armonk, NY, USA) was used for performing the statistical analyses using the results from the measures reported above. The analysis was performed on the following dependent variables on condition:

- Vocabulary:
 - Receptive Vocabulary
 - Expressive Vocabulary
- Story Comprehension:
 - Implicit Story Comprehension
 - Explicit Story Comprehension
 - Both tests combined to report overall comprehension of the story
- Transfer Task
- PLS-5 UK

To test the effects of the intervention in order to answer the research questions of the study, two contrasts were created: (a) the two conditions with print book format (control) versus the two conditions with e-book format (comparison of book mediums) (RQ1); and (b) the two conditions with performance teaching style versus the two conditions with interactive teaching style (comparison of teaching styles) (RQ2). Therefore, this entails a 2 (teaching styles) x 2 (book mediums) x 2 (pre, post) between-within subjects factorial design.

In order to answer the first research question: Which type of book medium (i.e., e-book versus print book) produces better results in the development of vocabulary and story comprehension for children learning English as an additional language (EAL)?, the e-book and print book (control) conditions were compared for each dependent variable. The data were analysed with the utilisation of repeated measures ANOVA for the dependent variables receptive, expressive vocabulary and the standardised language test PLS-5 UK.

Similarly, to answer the second research question: Which teaching style (i.e., interactive versus performance) better facilitates young children's (learning English as an additional language) vocabulary and story comprehension development, the interactive and

performance conditions were compared for each dependent variable. The data were analysed with repeated measures ANOVA for the dependent variables receptive, expressive vocabulary and the standardised language test PLS-5 UK.

The results section begins with a check for baseline differences. The first research question is then analysed, beginning with the receptive and expressive vocabulary results, story comprehension, transfer task, and concluding with the standardised test results. Then the second research question is analysed.

5.4 Results

5.4.1 Preliminary Analyses

Sixty typically developing children (28 girls and 32 boys) aged 3–7 years (M = 5.26 years, SD = 1.16) were enlisted from four classrooms from one private English school. The children attended the following grades: Pre-Kindergarten (ages 3–4; n=7; m=4, f=3), Kindergarten (ages 4-5.4; n=19; m=8, f=11), Pre-Primary (5.5–7; n=18; m=12, f=6), and First Grade (6-7.5 years old; n=16; m=8, f=8). Each class had one teacher. The participants were randomly allocated to conditions, in a stratified way taking into account a balanced representation of age, gender, language proficiency English level based on the Preschool Language Scale - Fifth Edition (PLS-5) standardised test, and their teachers' evaluation on a 5-point language scale level.

A preliminary analysis was performed to assess pre-intervention differences across conditions regarding children's language skills with the Preschool Language Scale - Fifth Edition (PLS-5). Using Univariate Analysis of Variance (ANOVA), **no significant differences were found among children in the four conditions at baseline** based on the means of the PLS-5 standardised test with an effect size of $0.00 \ (F(3, 56)=0.02, p=0.99)$, suggesting participants had similar general vocabulary knowledge. Furthermore, in order to establish that the conditions were indeed well-matched, each class's teachers also completed a 5-point English language scale. Table 5.5 presents an overview of descriptive statistics for all conditions. According to Table 5.5, the conditions were evenly balanced at the beginning in terms of the participants' language level as measured by the PLS-5 standardised the test.

| Teaching style | | Interact | tive Style | Perform | nance Style | Total | S | tatistic | es |
|---------------------|----|----------|------------|---------|-------------|---------|------|----------|------------|
| Book medium | | E- | Print | E- | Print book | | F | р | η_p^2 |
| | | book | book | book | | | | | |
| Ν | | 15 | 15 | 15 | 15 | 60 | | | |
| Age | М | 5.27 | 5.80 | 5.00 | 5.27 | 5.27 | | | |
| | SD | (1.38) | (0.94) | (0.92) | (0.96) | (1.16) | | | |
| Boys/Girls | | 7/8 | 8/7 | 9/6 | 8/7 | 32/28 | | | |
| PLS-5 UK | М | 67.47 | 67.33 | 66.27 | 67.13 | 67.05 | 0.02 | 0.99 | 0.00 |
| | SD | (13.98) | (17.02) | (13.59) | (13.23) | (14.17) | | | |
| | | 2.67 | 0.70 | 0.70 | 0.47 | 2.65 | | | |
| English Language 5- | Μ | 3.67 | 3.73 | 3.73 | 3.47 | 3.65 | | | |
| point scale | SD | (1.23) | (1.33) | (1.28) | (1.24) | (1.24) | | | |

Table 5.5 Means (and standard deviations) of Basic Characteristics per Condition

Due to the Covid-19 pandemic, there were 11 missing cases on the standardised post-test PLS-5 UK. For the first story, *The Kissing Hand*, there were five missing cases on story comprehension and 11 missing cases for the receptive and expressive vocabulary (2nd post-test). For the second story, *Elmer and Rose*, there were 11 missing cases for both first and second post administration of the receptive vocabulary tests and nine for the expressive vocabulary tests (first and second posttests), and 13 missing cases for the story comprehension test. Table 5.6 provides an overview of the number of participants with available data across research tools.

| Condition/ Measures | Rece Vocał | ptive oulary | Expr Vocał | essive oulary | Implici Compre | t Story chension | Explicit Story comprehension | | Transfer Task | PLS- 5 UK |
|------------------------------------|---------------|-----------------|---------------|------------------|-------------------|---------------------|---------------------------------|----|------------------|--------------|
| | | | | | Pre test | t | | | | |
| Stories* | KH | ER | KH | ER | KS | ER | KS | ER | | |
| E-book Interactive style | 15 | 15 | 15 | 15 | | | | | | 15 |
| E-book Performance style | 15 | 15 | 15 | 15 | | | | | | 15 |
| Print-book Interactive style | 15 | 15 | 15 | 15 | | | | | | 15 |
| Print-book Performance style | 15 | 15 | 15 | 15 | | | | | | 15 |
| | | | | 1 st | Post test | | | | | |
| E-book Interactive style | 15 | 15 | 15 | 15 | 15 | 13 | 15 | 13 | 11 | 14 |
| E-book Performance style | 15 | 12 | 15 | 12 | 15 | 11 | 15 | 11 | 11 | 11 |
| Print-book Interactive style | 15 | 11 | 15 | 11 | 13 | 10 | 13 | 10 | 11 | 12 |
| Print-book Performance style | 15 | 13 | 15 | 13 | 15 | 13 | 15 | 13 | 13 | 12 |
| | | | | 2 nd | Post Test | | | | | |
| E-book Interactive style | 15 | 14 | 15 | 15 | | | | | | |
| E-book Performance style | 12 | 12 | 12 | 12 | | | | | | |
| Print-book Interactive style | 11 | 10 | 11 | 11 | | | | | | |
| Print-book Performance style | 13 | 13 | 13 | 13 | | | | | | |

Table 5.6 The Number of Participants Who Responded to Each Measure for Each Story

*KS= The story "Kissing Hand", ER=The story "Elmer and Rose"

Comparing the four conditions from pre to post, each outcome was examined separately and described in different sections below. The dependent variables were also analysed based on teaching style (i.e., interactive versus performance) and book medium (print versus e-book). If a second post-test was administered, the findings are reported and compared to the first (pre) and second (post 1) test administrations. To assess story comprehension, participants were evaluated at posttest only.

5.4.2 Research Question 1 - Comparison of Book Mediums: E-book versus Print Format

5.4.2.1 Receptive and Expressive Vocabulary - E-book versus Print Format.

A repeated measures analysis of variance (ANOVA) test was performed to compare the effect of book medium on receptive (Table 5.7) and expressive vocabulary (Table 5.8) from pre-test to post-test 1. Due to covid-related issues, the second post-test that took place one month after the intervention restricted the collection of data from all participants; therefore, the repeated measures analyses below analysed the results obtained from the pre-test and the first post-test where the sample in both conditions was present (results for the second posttest are found in Appendix M). Therefore, the effect of time (pretest, first posttest) as a within-person factor, and condition (e-book, print book) a between-person factor were assessed.

For *receptive* vocabulary there was a statistically significant effect for time with a large effect size (F(1, 58) = 127.936, p < .001, $\eta_p^2 = 0.68$). There was no interaction effect (F(1, 58) = 0.177, p=0.67, $\eta_p^2 = 0.00$), i.e., the two conditions performed equally well over time. Furthermore, statistically significant results were found for condition (F(1, 58) = 4.597, p=0.03, $\eta_p^2 = 0.07$), which reveals that participants differ on receptive vocabulary depending on their condition, such that the print book outperformed the e-book (see Figure 5.1). In addition, the effect sizes for the two conditions revealed a large effect for print book with d=0.93 and a medium effect for e-book with d=0.68. However, it is important to exercise caution when interpreting the main effect finding, given the baseline results differ between conditions. The significant result found for main effects, as in Table 5.8, may be driven by this variation rather than the intervention itself. The study's conclusions will thus be derived from the interaction effect (condition x time) of the repeated measures ANOVA analyses presented from this point onwards.

| | | | | Tir | ne | | | |
|---------------------|----|----------------|---------------|--------------|-------|----------------|---------------|--------------|
| | | | Pre | | | | | |
| Condition | Ν | Mean (SD) | Std. Error | 95% CI | Ν | Mean (SD) | Std. Error | 95% CI |
| E-book | 30 | 0.47 (0.19) | 0.03 | [0.41, 0.53] | 30 | 0.60 (0.19) | 0.03 | [0.53, 0.66] |
| Print Book | 30 | 0.56 (0.15) | 0.03 | [0.50, 0.62] | 30 | 0.70 (0.15) | 0.03 | [0.63, 0.76] |
| Total | 60 | 0.51 (0.17) | 0.02 | [0.47, 0.56] | 60 | 0.65 (0.18) | 0.02 | [0.60, 0.69] |
| ANOVA results | | | F | | р | | i | η_p^2 |
| Time effect | | 12 | 7.936 | | <.001 | | 0 | .68 |
| Condition effect | | 4. | 597 | | 0.036 | | 0 | .07 |
| Condition x Time | | 0. | 177 | | 0.676 | | 0 | .00 |

Table 5.7 Repeated Measures ANOVA Results for Receptive Vocabulary Based on BookMedium (pre-test versus post-test 1)

Figure 5.1 Repeated Measures for Receptive Vocabulary Based on Book Medium



For expressive vocabulary (Table 5.8, Figure 5.2) the effect of time was again statistically significant with a moderate effect size (F(1, 58) = 31.044, p < .001, $\eta_p^2 = 0.34$), however, the condition effect and the condition by time effect were not (p > 0.05), which shows that

everyone improved from pre-test to post-test in expressive vocabulary (DV) regardless of condition. In addition, it was observed that e-books had a medium effect size (d = 0.35), which was found to be greater than the medium effect size observed for print books (d = 0.19).

Table 5.8 Repeated Measures ANOVA Results for Expressive Vocabulary Based on BookMedium (pre-test versus post-test 1)

| | | | | Ti | me | | | |
|------------------|----|----------------|---------------|--------------|-------|----------------|---------------|--------------|
| | | | Pre | | | | Post 1 | |
| Condition | N | Mean (SD) | Std. Error | 95% CI | N | Mean (SD) | Std. Error | 95% CI |
| E-book | 30 | 0.61 (0.58) | 0.11 | [0.39, 0.84] | 30 | 0.83 (0.66) | 0.12 | [0.58, 1.08] |
| Print Book | 30 | 0.83 (0.64) | 0.11 | [0.61, 1.06] | 30 | 0.96 (0.72) | 0.12 | [0.70, 1.21] |
| Total | 60 | 0.72 (0.62) | 0.08 | [0.56, 0.88] | 60 | 0.89 (0.69) | 0.09 | [0.71, 1.07] |
| ANOVA results | | | F | | р | | η | 2 p |
| Time effect | | | 31.044 | | <.001 | | 0 | .34 |
| Condition effect | | | 1.095 | | 0.30 | | 0 | .01 |
| Condition x Time | | | 2.490 | | 0.12 | | 0 | .04 |

Figure 5.2 Repeated Measures for Expressive Vocabulary Based on Book Medium



Time

5.4.2.2 Implicit and Explicit Story Comprehension Results - E-book versus Print Format. One-way ANOVA was performed to analyse the effect of book medium – e-book versus print book – on implicit, explicit and overall story comprehension of the story at post-test 1. These results were obtained after the intervention and three one-way ANOVAs were conducted in order to obtain the results presented in Table 5.9. The results revealed that for the *Implicit* story comprehension test there was no statistically significant main effect for medium (e-book versus print book conditions, F(1, 56)=3.22, p=0.07, $\eta_p^2=0.05$). The effect size for e-books versus print books for the Implicit story comprehension test was d=-0.48 favouring the print book condition.

The results analysed for *Explicit* story comprehension and both tests combined revealed that there was a statistically significant effect (p<0.05) with the print book condition outperforming the e-book condition on both measures (explicit story comprehension: F(1, 56)=4.36, p=0.04, $\eta_p^2=0.07$; Both tests combined: F(1, 56)=4.32, p=0.04, $\eta_p^2=0.07$). The effect size for e-books versus print books for the Explicit story comprehension test was d=-0.56 favouring the print book condition.

Table 5.9 One-way ANOVA Results for Implicit and Explicit Story Comprehension (posttest 1)

| | E-book condition | | | | | Print book condition | | | | | | |
|---------------------------------|------------------|----------------|---------------|-----------------|----|----------------------|---------------|-----------------|------|------|------------|-----------|
| Dependent Variables | N | Mean (SD) | Std. Error | 95% CI | N | Mean (SD) | Std. Error | 95% CI | F | р | η_p^2 | Cohen's d |
| Implicit Story Comprehension | 30 | 0.85 (0.44) | 0.08 | [0.69, 1.01] | 28 | 1.06 (0.43) | 0.08 | [0.89, 1.22] | 3.22 | 0.07 | 0.05 | -0.48 |
| Explicit Story Comprehension | 30 | 0.78 (0.41) | 0.07 | [0.63, 0.92] | 28 | 1.00 (0.37) | 0.07 | [0.85, 1.15] | 4.36 | 0.04 | 0.07 | -0.56 |
| Both Tests combined | 30 | 0.81 (0.40) | 0.07 | [0.67, 0.96] | 28 | 1.03 (0.37) | 0.07 | [0.88, 1.17] | 4.32 | 0.04 | 0.07 | -0.57 |

5.4.2.3 PLS-5 Results - E-book versus Print Format. Two one-way ANOVAs were performed to analyse the effect of book medium – e-book versus print book - on the standardised language test PLS-5 UK at pre-test and post-test. The results revealed that there was no statistically significant main effect for book medium at either time point (Pre: F(1,58)=0.01, p=0.92, $\eta_p^2=0.00$; Post: F(1,47)=0.64, p=0.42, $\eta_p^2=0.01$). The effect size for e-book versus print book for the standardised post-test revealed a small effect size of d=0.22, $\eta_p^2=0.01$, favouring the e-book condition (Table 5.10).

Table 5.10 One-way ANOVA Results of PLS-5 Based on Book Medium (pre-test and post-
test)

| | | E-book condition | | | Print Book condition | | | | | | | |
|-----------------------|----|------------------|-------|---------------|----------------------|-------|-------|---------------|------|------|------------|-----------|
| Dependent Variable | N | Mean | SD | Std. Error | Ν | Mean | SD | Std. Error | F | р | η_p^2 | Cohen's d |
| PLS-5 Pre | 30 | 66.87 | 13.56 | 2.61 | 30 | 67.23 | 14.98 | 2.61 | 0.01 | 0.92 | 0.00 | -0.02 |
| PLS-5 Post | 25 | 67.84 | 13.72 | 2.72 | 24 | 64.71 | 13.58 | 2.78 | 0.64 | 0.42 | 0.01 | 0.22 |

Moreover, a repeated measures analysis was conducted to assess time, condition, and condition x time effects for the standardised language test PLS-5 from pre-test to post-test. The results of this test were collected before the intervention as a baseline measure and six months later after the intervention. The results were statistically non-significant for all effects (Table 5.11). The effect size for the e-book and print book conditions was small, with d=0.02 and d=0.04 for each condition, respectively.

| | | | | Tim | <u></u> | | | | | |
|---------------------|----|------------------|---------------|----------------|---------|------------------|---------------|----------------|--|--|
| | | | | IIm | e | | | | | |
| | | | Pre | | Post 1 | | | | | |
| Condition | Ν | Mean (SD) | Std. Error | 95% CI | Ν | Mean (SD) | Std. Error | 95% CI | | |
| E-book | 30 | 68.20 (14.19) | 2.73 | [62.69, 73.70] | 25 | 67.84 (13.72) | 2.72 | [62.35, 73.32] | | |
| Print Book | 30 | 65.29 (13.10) | 2.79 | [59.67, 70.90] | 24 | 64.71 (13.52) | 2.78 | [59.11, 70.30] | | |
| Total | 60 | 66.78 (13.61) | 1.95 | [62.81, 70.67] | 49 | 66.31 (13.57) | 1.94 | [62.35, 70.19] | | |
| ANOVA results | | F | | | р | | | η_p^2 | | |
| Time effect | | 0.28 | 7 | | 0.595 | | | 0.00 | | |
| Condition effect | | 0.63 | 1 | | 0.431 | | | 0.01 | | |
| Condition x Time | | 0.010 | 5 | | 0.900 | | | 0.00 | | |

Table 5.11 Repeated Measures ANOVA Results for PLS-5 Based on Book Medium (pre-
test versus post-test)
5.4.3 Research Question 2 - Interactive versus Performance Teaching Style

5.4.3.1 Receptive and Expressive Vocabulary - Interactive versus Performance Teaching Style.

A repeated measures ANOVA was performed to compare the effect of the two teaching styles on receptive vocabulary (Table 5.12) and expressive vocabulary (Table 5.13) over time (pre-test versus post-test 1). Therefore, the effect of time (pretest and first posttest) as a within-person factor, and condition (interactive teaching style, performance teaching style) a between-person factor were assessed. The second post-test was not included in this repeated measures analysis as a number of participants did not complete the second post test due to covid-related issues (results for the second post-test can be found in Appendix M).

There was a significant main effect for time on receptive vocabulary with a large effect size (F(1,58)=127.661, p=<.001, $\eta_p^2=0.68$). Effects for teaching style were also statistically significant, as differences between conditions were found at both time points, (F(1,58)=5.823, p=0.019, $\eta_p^2=0.09$). However, there was a non-significant interaction effect between time and condition (F(1,58)=0.052, p=0.820, $\eta_p^2=0.00$), suggesting that everyone improved from pre-test to post-test in receptive vocabulary regardless of condition (see Figure 5.3). Regarding effect size, the study observed Cohen's *d* of 0.81 and 0.74 for interactive style and performance style, respectively, indicating a large effect. Nevertheless, it is critical to exercise caution when interpreting the finding of the main effect, considering that the receptive vocabulary outcomes differed between groups at pretest. Rather than the intervention per se, this variation might have contributed to the significant outcome.

Table 5.12 Repeated Measures ANOVA Results for Receptive Vocabulary Based on

 Teaching Style (pre-test versus post-test 1)

| | Time | | | | | | | | |
|----------------------|------|----------------|---------------|--------------|----|----------------|---------------|--------------|--|
| | | | Pre | | | | Post 1 | | |
| Condition | Ν | Mean (SD) | Std. Error | 95% CI | Ν | Mean (SD) | Std. Error | 95% CI | |
| Interactive style | 30 | 0.57 (0.16) | 0.03 | [0.51, 0.63] | 30 | 0.70 (0.16) | 0.03 | [0.63, 0.76] | |
| Performance style | 30 | 0.46 (0.17) | 0.03 | [0.40, 0.52] | 30 | 0.59 (0.18) | 0.03 | [0.53, 0.66] | |
| Total | 60 | 0.51 (0.17) | 0.02 | [0.47, 0.56] | 60 | 0.65 (0.18) | 0.02 | [0.60, 0.69] | |
| ANOVA results | | F | | р | | | | η_p^2 | |
| Time effect | | 127.661 | | <.00 |)1 | | C |).68 | |
| Condition effect | | 5.823 | | 0.0 | 1 | | C |).09 | |
| Condition x Time | | 0.052 | | 0.82 | | | 0.00 | | |

Figure 5.3 Repeated Measures for Receptive Vocabulary Based on Teaching Style



The same analysis was run for expressive vocabulary (Table 5.13). The presentation of the repeated measures ANOVA results includes subtotals across groups for each time point. The effect of time was again significant with p<.001 and a small positive effect size

 $\eta_p^2=0.22$ illustrating that the two teaching styles support expressive vocabulary development. In terms of Cohen's *d*, a medium effect size was found for the interactive style (*d*=0.34) and a small effect size was found for the performance style (*d*=0.18). Furthermore, statistically significant results were found for the time by condition interaction effect (*F*(1, 58) = 5.014, *p*=0.02, η_p^2 =0.08), which reveals that the interactive versus performance teaching style condition made a difference over time, whereas condition effect was non-significant (*F*(1, 58) = 2.157, *p*=0.14, η_p^2 =0.03). Specifically, participants improved at different rates over time in terms of their expressive vocabulary (see Figure 5.4), i.e., children participating in the interactive style condition.

Table 5.13 Repeated Measures ANOVA Results for Expressive Vocabulary Based onTeaching Style (pre-test versus post-test 1)

| | Time | | | | | | | | |
|-------------------|------|--------|------------|--------------|-------|--------|------------|--------------|--|
| | | | Pre | | | | Post 1 | | |
| Condition | Ν | Mean | Std. Error | 95% CI | Ν | Mean | Std. Error | 95% CI | |
| | | (SD) | | | | (SD) | | | |
| Interactive style | 30 | 0.81 | 0.11 | [0.58, 1.04] | 30 | 1.05 | 0.12 | [0.80, 1.30] | |
| | | (0.65) | | | | (0.73) | | | |
| Performance style | 30 | 0.63 | 0.11 | [0.41, 0.86] | 30 | 0.74 | 0.12 | [0.49, 0.99] | |
| - | | (0.57) | | | | (0.61) | | | |
| Total | 60 | 0.72 | 0.08 | [0.56, 0.88] | 60 | 0.89 | 0.08 | [0.72, 1.07] | |
| | | (0.62) | | | | (0.69) | | | |
| ANOVA results | | | F | | р | | η_p^2 | | |
| Time effect | | | 32.339 | | <.001 | | 0.3 | 5 | |
| Condition effect | | | 2.157 | | 0.14 | | 0.0 | 3 | |
| Condition x Time | | | 5.014 | 0.02 | | | 0.08 | | |

Figure 5.4 Repeated Measures for Expressive Vocabulary Based on Teaching Style



ine

5.4.3.2 Implicit, Explicit and Overall Story Comprehension Results - Interactive versus Performance Teaching Style. One-way ANOVA was performed to compare the effect of teaching style on story comprehension separately for the three story comprehension test scores (implicit, explicit, and combined, see Table 5.14) at post-test 1. The *implicit* story comprehension analysis revealed that there was a statistically significant effect of teaching style on Implicit story comprehension (F(1, 56)=4.47, p=0.03, $\eta_p^2=0.07$) favouring the Interactive style. There was no statistical significant difference between the Interactive and Performance teaching styles for *explicit* comprehension (F(1, 56)=2.76, p=0.10, $\eta_p^2=0.04$). When these two test results were combined and analysed in order to evaluate an overall score for comprehension the results were statistically significant (F(1, 56)=4.15, p=0.04, $\eta_p^2=0.06$). Overall, the three study instruments analysed for story comprehension demonstrated that interactive scaffolding delivered superior outcomes compared to performance scaffolding.

Table 5.14 One-way ANOVA Results of Implicit, Explicit and Overall StoryComprehension Results Based on Teaching Style (post-test 1)

| | Interactive Teaching Style | | | | | Performance Teaching Style | | | | | | |
|---------------------------------|----------------------------|----------------|---------------|-----------------|----|-------------------------------|---------------|-----------------|------|------|------------|-----------|
| Dependent Variables | Ν | Mean (SD) | Std. Error | 95% CI | Ν | Mean (SD) | Std. Error | 95% CI | F | р | η_p^2 | Cohen's d |
| Implicit Story Comprehension | 28 | 1.07 (0.46) | 0.08 | [0.91, 1.24] | 30 | 0.83 (0.41) | 0.08 | [0.67, 0.99] | 4.47 | 0.03 | 0.07 | 0.55 |
| Explicit Story Comprehension | 28 | 0.98 (0.39) | 0.07 | [0.82, 1.13] | 30 | 0.80 (0.41) | 0.07 | [0.65, 0.95] | 2.76 | 0.10 | 0.04 | 0.44 |
| Both Tests combined | 28 | 1.02 (0.40) | 0.07 | [0.88, 1.17] | 30 | 0.81 (0.37) | 0.07 | [0.67, 0.96] | 4.15 | 0.04 | 0.06 | 0.54 |

Using Cohen's d, the effect size calculation for teaching styles for the Implicit story comprehension test revealed a medium effect size of d=0.55 and for Explicit story comprehension test revealed a moderate effect size of d=0.44, favouring the Interactive teaching style.

5.4.3.3 PLS-5 Results - Interactive versus Performance Teaching Style. One-way ANOVA was performed to analyse the effect of the interactive and performance teaching style conditions on the standardised language test PLS-5 UK at pre-test and post-test (Table 5.15). The results revealed that there were no statistically significant main effects for teaching style (interactive versus performance) at pre or post-test (Pre: F(1, 58)=0.03,

p=0.85, $\eta_p^2=0.00$; Post: F(1, 47)=0.12, p=0.72, $\eta_p^2=0.00$).

| | Int | eractive | Teachin | ng Style | P | erforma | nce Tea | ching | | | | |
|------------------|-----|----------|---------|---------------|----|---------|---------|---------------|------|------|------------|-----------|
| Research Tool | N | Mean | SD | Std. Error | N | Mean | SD | Std. Error | F | р | η_p^2 | Cohen's d |
| PLS-5 Pre | 30 | 67.40 | 15.31 | 2.60 | 30 | 66.70 | 13.19 | 2.60 | 0.03 | 0.85 | 0.00 | 0.04 |
| PLS-5 Post | 26 | 66.96 | 14.04 | 2.68 | 23 | 65.57 | 13.31 | 2.85 | 0.12 | 0.72 | 0.00 | 0.10 |

Table 5.15 One-way ANOVA Results of PLS-5 Based on Teaching Style (pre-test and post-
test)

Moreover, a repeated measures analysis was conducted to assess time effects (see Table 5.16), condition effects, and condition x time for the standardised language test PLS-5 from pre-test to post-test based on teaching styles. The results were statistically nonsignificant for all effects (p>0.05). In addition, the effect sizes for the two conditions revealed small effects for both teaching styles, with d=0.04 for the interactive style and d=0.12 for the performance style.

Table 5.16 Repeated Measures ANOVA Results for PLS-5 Based on Teaching Style (pre-
test versus post-test)

| | | | | T | ime | | | | | |
|---------------------|----|------------------|---------------|----------------|-------|------------------|---------------|----------------|--|--|
| | | | Pre | | | | Post 1 | | | |
| Condition | N | Mean (SD) | Std. Error | 95% CI | N | Mean (SD) | Std. Error | 95% CI | | |
| Interactive style | 26 | 66.35 (13.65) | 2.69 | [60.92, 71.76] | 26 | 66.96 (14.04) | 2.68 | [61.55, 72.36] | | |
| Performance style | 23 | 67.26 (13.85) | 2.86 | [61.49, 73.02] | 26 | 65.57 (13.31) | 2.85 | [59381, 71.31] | | |
| Total | 49 | 66.78 (13.61) | 1.96 | [62.84, 70.76] | 49 | 66.31 (13.57) | 1.96 | [62.31, 70.20] | | |
| ANOVA results | | | F | | р | | | η_p^2 | | |
| Time effect | | 0.389 | | | 0.536 | | | 0.00 | | |
| Condition effect | | 0.004 | | | 0.950 | | | 0.00 | | |
| Condition x Time | | 1 | 0.18 | 39 | | 0.00 | | | | |

5.4.4 Transfer Task

The interaction effect of book medium by teaching style for the Transfer Task was not significant, F(1, 42) = 0.725, p = .399, $\eta_p^2 = 0.01$, nor were the main effects of book medium and teaching style. Therefore, these results suggest that there is no transfer effect (Table 5.17).

| Book | Teaching | Ν | Mean | Std. Error | 95% CI |
|-------------|-------------|----|---------------|------------|--------------|
| Medium | Style | | (SD) | | |
| E-book | Interactive | 11 | 0.66 | 0.16 | [0.32, 0.99] |
| | style | | (0.62) | | |
| | Performance | 11 | 0.90 | 0.16 | [0.57, 1.24] |
| | style | | (0.52) | | |
| Print Book | Interactive | 11 | 0.76 | 0.16 | [0.43, 1.10] |
| | style | | (0.56) | | |
| | Performance | 13 | 0.73 | 0.15 | [0.42, 1.04] |
| | style | | (0.49) | | |
| ANOVA | | | F | р | η_p^2 |
| results | | | | | - F |
| Book Mediu | ım effect | 0. | 045 | 0.833 | 0.00 |
| Teaching St | vle | 0 | 445 | 0 508 | 0.01 |
| Effect | J 10 | 0. | | 0.500 | 0.01 |
| Book Mediu | ım x | 0. | 725 | 0.399 | 0.01 |
| Teaching St | yle | | | | |

Table 5.17 *Two-way ANOVA Results for Transfer Task of Book Medium by Teaching Style* (*post-test 1*)

5.4.5 Bonferroni Correction

Within the scope of my study, four crucial analyses were performed using repeated measures ANOVA.

- **Comparison of Book Mediums (RQ1):** This set of analysis include comparing the effects of e-books and print books on each dependent variable, namely vocabulary and story comprehension. I conducted two analyses for vocabulary: one for receptive skills and one for expressive skills. To enhance understanding of the story, I conducted two types of analyses: implicit and explicit. Given that there are four dependent variables, there will be four comparisons within this family.
- Comparison of Teaching Styles (RQ2): This family of analyses involves comparing interactive versus performance teaching conditions for each dependent variable (vocabulary and story comprehension). In the same manner, I conducted

two analyses for vocabulary: one for receptive skills and one for expressive skills. For story comprehension, I conducted two analyses: implicit and explicit. Given that there are four analyses, there will be four comparisons within this family.

When using the Bonferroni adjustment, the adjusted significance level becomes 0.012, which is obtained by dividing 0.05 by 4. Based on the new significance level the story comprehension outcomes do not survive the adjustment.

Prior to adjusting the significance level, the analysis for RQ1 (e-books versus print books) using the explicit story comprehension test (refer to Table 5.9) demonstrated a statistically significant outcome with a p-value of 0.04. Due to the revised significance level, this outcome does not meet the criteria for statistical significance after adjustment, thereby rendering the results as non-significant. In relation to RQ2, which examines the difference between interactive and performance teaching styles, the implicit story comprehension test (refer to Table 5.16) yielded a p-value of 0.03 before applying the Bonferroni correction. Given the modified p-value, this finding is not statistically significant. Hence, the aforementioned story comprehension effects are being examined with caution. It is imperative to underscore the consistent inclusion of confidence intervals (CIs) and effect sizes. The evaluation of the results should not be predicated solely on the p-value and the statistical significance of the findings.

5.5 Discussion

The current study demonstrates the importance of selecting appropriate book mediums and teaching styles when teaching young children whose first language is not the language taught within the stories. The aims of this experimental study were: (1) to evaluate which type of media produces better results in the development of vocabulary and story comprehension for children learning English as an additional language; and (2) to investigate which reading style better facilitates young children's (learning English as an additional language) vocabulary and story comprehension development. Thus, the study assessed the effects of e-book and print book reading with two different teaching styles on vocabulary development and story comprehension. The study recruited 60 participants from ages three to seven years, residing in Larnaca, Cyprus, and learning English as an additional language (EAL).

In line with our expectations, all experimental conditions progressed from pre-intervention to post-intervention as evidenced by increased mean scores. An initial objective of the study was to identify which book medium could better develop children's vocabulary skills and story comprehension abilities, the "perfect" e-book or the print book. While the participants' mean score differences before and after the intervention increased, there was no interaction effect observed for receptive vocabulary and expressive vocabulary. This indicates that everyone showed improvement from the pre-test to the post-test, regardless of the condition they were in. The second research question of this research was to determine which teaching style could potentially support children learning EAL, the interactive style or the performance style. In all dependent variables, the interactive teaching style enhanced children's learning more effectively than the performance teaching style. One of the study's strengths is that the findings are based on an intervention that comprised two different stories rather than simply one. When children are learning a new language, repetition or repeated exposure to essential information promotes a better comprehension of the material (Takanishi & Le Menestrel, 2017). These findings are discussed below in detail.

5.5.1 E-book versus Print Book

5.5.1.1 Vocabulary Growth

Two different levels of vocabulary development were measured in this study, receptive vocabulary and expressive vocabulary. It is important to note that there were differences between conditions at pre-test, as indicated by the results of the receptive and expressive vocabulary tests. Consequently, the use of repeated measures analysis was imperative in order to control for the pre-test variations.

In both conditions (e-book versus print book), children improved their receptive vocabulary. When comparing the participants' pre-test to post-test receptive vocabulary growth, the print book condition produced better results (see Table 5.8 and Figure 5.1), however results were non-significant (p>0.05). To the best of my knowledge, there is no research comparing the efficacy of e-books versus print books, with adult support, for children learning EAL; consequently, these results cannot be compared to those of earlier studies. However, when comparing the results of this study with previous studies concerning the use of digital media and bilingual children, this finding is contrary to previous studies which have suggested that animated stories (i.e., videos) supported children learning a new language more than static e-books (i.e., Silverman & Hines, 2009; Verhallen et al., 2006).

Expressive vocabulary was statistically non-significant in terms of book medium (F(1, 58) = 1.095, p > 0.05, $\eta_p^2 = 0.01$). Compared to the findings of the meta-analysis in Chapter 4 of the thesis, where expressive vocabulary was found to have a medium positive effect size in favour of the e-book condition (g = 0.54; 95% CI = [0.08, 1.00]), this result demonstrates that typically developing children and children learning EAL absorb information differently. Digital media with various technological enhanced features may be effective for children acquiring vocabulary in their native language but should not be assumed to be equally effective for children whose native language is not the language being taught during storybook reading. In addition, EAL learners typically find it simpler to employ receptive language skills than expressive language abilities (Uchikoshi, 2014). Although the studies by Silverman and Hines (2009) and Verhallen et al. (2006) may demonstrate a favourable trend in the use of multimedia stories to increase vocabulary development among bilinguals, Silverman's study (2013) produces different results. Silverman performed research on the function of video in vocabulary development in kindergarten

classes with a mix of monolingual and bilingual pupils in each class. Silverman investigated the effects of video viewing and book reading on vocabulary learning (*n*=78). There was no significant difference in vocabulary learning between the two groups (F(1,77) = 0.06, p = .81, d=0.00), contrary to the findings of Verhallen et al. (2006; *F*(1, 53)=26.63, *p* <.001, η_p^2 =0.33). In light of these results, Silverman qualifies her findings with a few limitations, notably the limited sample size and quasi-experimental nature of the study. The common limitation of this study with Silverman's study (2013) is its small sample size.

5.5.1.2 Improvement in Story Comprehension

While analysing the results retrieved from book medium analyses, post-scores revealed a statistically significant result for explicit story comprehension (p=0.04, $\eta_p^2=0.07$) and nonsignificant results for implicit story comprehension (p=0.07, $\eta_p^2=0.05$; see Table 5.10). Implicit story comprehension includes making inferences about the feelings of the story protagonists, causal relations, predictions, and theme; it requires deeper story comprehension skills. Explicit story comprehension is related to the identification of the story elements (characters, setting, initiating event, problem, and solution). The significant outcome regarding explicit story comprehension could potentially be explained by the assumption that children who do not speak the language of the narrative as their primary language may experience greater ease in comprehending basic story components, such as the setting and main characters, as opposed to grasping and articulating the emotions and feelings of said characters. The utilisation of illustrations holds significant value in this context, as children are able to visually comprehend the various story elements through the accompanying images. While the first findings suggested that print books might enhance story comprehension, it is important to highlight that these results lost their statistical significance after applying the Bonferroni adjustment.

Furthermore, when comparing the two book mediums, the condition with the highest mean-score in both implicit and explicit story comprehension tests (the two test combined) was the print book (M= 1.03, SD = 0.37, F(1, 27)=210.56, p=<.001; 95% CI = [0.88, 1.17]; $\eta_p^2=0.07$, d=-0.57). This result is closely linked with the result found by the meta-analysis (Study 1) which found a small positive but non-significant effect size on story comprehension, favouring the print book condition (g = 0.05; 95% CI = [-0.11, 0.21]). The

high overall scores found for the print book condition might be explained by the metaanalysis performed by Hadley et al. (2022), in which they investigated how teacher language practices differed according to activity setting and group size. They found that shared book-reading increased the use of language activities, but a small group featured less positive language practices compared to other contexts. In this study, the number of students that took part every session of the e-book condition was two, and for the print book condition were seven. Therefore, group size may also impact literacy development. Recent research by Korat et al. (2022a) stands in sharp contrast with the results found in this study, which found a considerable advantage for children who read the e-book with adult scaffolding in terms of story comprehension. Their study examined 103 children (aged 5–6 years) from LSES families. The participants were separated in three conditions: (a), the children read the e-book with a dictionary and the teacher's support; (b), the children read the e-book with the dictionary independently; and (c), the children read the e-book without a dictionary - static presentation of story pictures (control). The findings suggest that children who received support from both a dictionary and a teacher while reading an e-book demonstrated a greater acquisition of vocabulary compared to those who relied solely on the dictionary for assistance. Additionally, the results indicate that this group outperformed the control group, which listened to an e-book that closely resembled a print book with static illustrations.

5.5.2 Interactive Teaching Style versus Performance Teaching Style

5.5.2.1 Vocabulary Growth

The current study found that in terms of expressive vocabulary development (receptive vocabulary results were non-significant, p>0.05) the interactive teaching style produced better results in comparison to the performance teaching style (M = 0.70, SD = 0.16, F(1, 58) = 5.042, p = 0.02, $\eta_p^2 = 0.08$, d=0.64). The results support previous evidence from Thomas et al. (2020). In their study, they examined the influence of interactive reading on children from low socioeconomic status (SES) families whose home language was different from their school language. The 194 children in the interactive reading condition had accomplished greater vocabulary gains than those in the control condition. This emphasises the importance of children who are learning multiple languages participating during interactive reading sessions in order for vocabulary development (Chlapana & Tafa, 2014). Furthermore, these findings reinforce the notion that explaining word definitions

during the reading session is more effective than exposing learners to unfamiliar terms while reading a story without explaining any words (Schatz & Baldwin, 1986), particularly for children whose first language is not the language spoken within the story.

The findings highlight the advantage of employing the Dual Coding Theory (Paivio, 2008), which claims that children learn better when information is offered to them concurrently through both the visual and verbal channels, as opposed to just one channel. In addition, the results confirm Vygotsky's (1978) Cognitive-Social Theory, which emphasises the significance of adult mediation for increasing children's learning. This finding may imply that, when reading a narrative to young children, it is the teaching approach, not the medium, that makes the difference, as children in the interactive style condition made a much greater improvement from pre-test to post-test in terms of expressive vocabulary acquisition than those in the performance style condition (see Table 5.15 and Figure 5.4). According to Sun et al. (2020) many teachers utilise the performance teaching style, which entails little to no breaks during storytelling sessions. In their study, 37 kindergarten teachers and 440 kindergartners aged 4-5 years in Singapore were observed during story time in order to determine the relationship between teachers' instructional approaches and children's language development. Their findings revealed that teacher's language strategies varied considerably, with the majority not using scaffolding techniques during the reading session. Sun et al. (2020) indicate that medium-level instruction was the most beneficial approach, which is consistent with the aforementioned finding when comparing interactive and performance teaching styles.

5.5.2.2 Improvement in Story Comprehension

Comparing the two teaching styles with regard to implicit and explicit story comprehension, results revealed that implicit story comprehension provided statistically significant outcomes (p=0.03, $\eta_p^2=0.07$, d=0.55), favouring the interactive teaching style condition, whereas explicit story comprehension yielded non-significant results (p=0.10). Although the results obtained from the interactive teaching condition and their improvement of implicit story comprehension, it is crucial to note that these improvements lost their statistical significance when the Bonferroni correction was implemented. Nonetheless, contrary to the results obtained for story comprehension based on the book medium, where explicit story comprehension was statistically significant in comparison to implicit story comprehension, this result indicates that the type of adult scaffolding used

to support EAL learners may indeed have a significant impact in advancing children's story comprehension in more demanding skills. Implicit story comprehension requires children whose first language is not the language used in the narrative to form inferences about the emotions of the story's protagonists, causal relationships, predictions, and theme, which the Interactive teaching style seems to be able to develop in young EAL learners.

5.5.3 Limitations of the Study

Some limitations of the current study should be taken into account in future studies. The small number of participants in each learning intervention group restricts findings' generalisability across different learning contexts. Additionally, as a consequence of COVID-related issues, certain analyses were limited by missing data, thus necessitating careful interpretation of those findings. A larger sample size is required, in order to achieve broader generalisation of the findings. Future research should investigate if reading group sizes matter in this type of investigation, as the way of teaching in terms of group size was different in the two media types (print book and e-book). Second, an e-book narrator that includes more interactive digital support of the kind provided here by adults can be developed. This option can be implemented regarding vocabulary support as well for other language aspects and might be highly productive. Third, since the intervention (e-book conditions) was administered with pairs of children working together, potential influence of the peer context should be taken into account when considering the intervention effects. Fourth, the study acknowledges that the absence of a second assessor during the delivery of research tools is a limitation. In order to address these constraints in future research, I suggest integrating inter-rater reliability assessments, which would include the participation of other assessors. Finally, since only one teacher implemented the experimental study, caution should be used in assuming that the results can be generalised to teachers from other backgrounds, including implementation of this study in a larger sample of teachers.

5.5.4 Theoretical Implications

The "perfect" e-book utilised in this study had minimal or no sound effects, no background music, and animations relevant to the story's content without distracting pupils from the narrative; thus, positive results were expected, which would have been consistent with Mayer's multimedia learning (Mayer, 2005), Paivio's dual coding theory (Paivio, 1986),

and Sweller's cognitive load theory (Sweller, 1988; Sweller et al., 2011). Specifically, it was expected that animated e-books are able to support children's vocabulary and story comprehension development, as evident by the results in Study 1 (i.e., Chapter 4). However, according to Boers et al. (2017), contrary to previous studies, this may not be the case for EAL learners when reading/listening to e-books. Boers et al. (2017) noted that the effect of multimedia support in the acquisition of new words differs for first language learners and EAL learners, as first language learners already have a mental association between the images and the words. Specifically, the benefits of multimedia support in vocabulary learning may not necessarily reflect how visual cues may strengthen words' concreteness or elicit a mental image (Paivio et al., 1968), and it is quite possible that native speakers already preserve these words in memory. For EAL learners, such a relationship between a shown image and a mental word does not exist; hence, the superiority of multimedia e-books over print books cannot be based only on Paivio's dual coding theory (Boers et al., 2017).

5.5.5 Practical Implications

One of the most common early years practice, storybook reading, has been found to have positive effects on children's language and literacy development (Bus et al., 1995; Mol & Bus, 2011; Hindman et al., 2014). Although the benefits of adult-child shared reading for children whose first language is the language of the story have been well-documented, shared reading in second language or multilingual contexts has received less attention. In reality, more than half of the global population is multilingual and can read in many languages (McBride, 2015). Alkaaby and Mavriqi (2021) suggest that in order for children to acquire a new language, it is essential for them to be exposed to it through listening. According to Chou (2014), when teachers read stories aloud to their pupils, the pupils' vocabulary expands and develops significantly. Storybooks have been investigated as a tool for teaching vocabulary, with studies indicating a retention rate of 20–30% for new terms learned (Biemiller & Boote, 2006). When reading aloud to children it allows them to build a foundation of knowledge that prepares them for the next stage of their education, which is to practise their ability to communicate verbally (Alkaaby & Mavriqi, 2021). As a result, research strongly supports adult guidance while listening to a storybook, as well as the importance of adult scaffolding during storytelling in developing young children's literacy skills (Homer et al., 2014; Korat et al., 2011; Moody et al., 2010; Rvachew et al.,

2017; Neumann & Neumann, 2014; Yelland & Masters, 2007; Zucker et al., 2020). Participants of this study progressed from pre-test to post-test, in all conditions. This result emphasises that children learning EAL are able to develop their vocabulary and comprehension skills through storybook reading combined with adult scaffolding.

Participants receptive vocabulary skills were developed and progressed similarly in both book mediums and teaching styles. This may be due to the fact that receptive vocabulary requires the learner to concentrate on listening to the word read by the teacher or narrated by the e-book and developing a mental image of what the item may look like. For instance, children were asked to identify which image corresponded to the word "raccoon". During the teacher's scaffolding intervention, the receptive vocabulary test words were either stated, discussed, or illustrated scenes from the storybook were shown, in order to engage pupils' attention. This technique, therefore, supported children's receptive vocabulary development either before, during or after the storytelling session. *Expressive vocabulary* (results were non-significant) is somewhat more difficult to acquire than receptive vocabulary, especially for children learning EAL. The development of an expressive vocabulary demands the pupil to comprehend the meaning of a term and attempt to explain it by providing a definition. The children in the study had limited vocabulary, making it more difficult for them to define a term in English. During the storytelling session, participants were able to engage in discussion and ask questions, which may have aided them in acquiring the meaning of an unknown term. With the performance teaching style, the children were required to wait until the end of the story to ask questions. Due to their young age and unfamiliarity with the language, it is likely that by that time the majority of participants will have forgotten their question. Therefore, the interactive teaching style delivers immediate support, and additionally it generates possibilities for dialogue, which enhances children's learning.

For *story comprehension*, however, the print book condition outperformed the e-book condition in overall story comprehension scores favouring the print book condition (F(1, 56)=4.32, p=0.04, $\eta_p^2=0.07$, d=-0.57). These results come as a stark contrast to the two other studies found on e-books and EAL learners (Sun et al., 2019; Yang et al., 2022). Sun et al. (2019) evaluated an animated e-book with 102 preschoolers in Singapore learning Mandarin between the ages of 4 and 5 years. The researchers employed four conditions: (a) animated e-book, (b) static e-book with sound, (c) static e-book only, and (d) a control

condition where children played a math game on an iPad. Sun et al. (2019) found that the animated e-book condition produced a positive effect on children's story comprehension (F(3,151,78) = 4.29, p = .006). Furthermore, Yang et al. (2022) conducted a randomised controlled trial to investigate the impact of bilingual discussion prompts accompanied by feedback within a multimedia interactive e-book on parent-child shared storybook reading. The study included 64 young English language learners, aged 3 to 7 years, residing in China. After reading the storybook twice, children in the discussion-prompt group outperformed the control group (listened to a multimedia interactive e-book without bilingual discussion prompts) on story comprehension and retelling measures. The current study differs from the previous two in that they did not include a print book condition. Therefore, it could be concluded that print books combined with interactive scaffolding may support children's story compression.

5.5.6 Implications for Future Research

Despite these promising results on book mediums and teaching styles for EAL learners, questions remain. There is limited research concerning e-books and EAL learners (i.e., Silverman, 2007; Silverman & Hines, 2009; Sun et al., 2019; Verhallen & Bus, 2011; 2010; Verhallen et al., 2006) and e-books combined with scaffolding techniques for EAL learners (i.e., Neumann & Merchant, 2022; Yang et al., 2022), plus most of these were published more than ten years ago. Therefore, future studies on the current topic are recommended. It is furthermore suggested that future studies include follow-up testing after six months, by re-administering their outcome measures, to investigate whether the effects of book medium and teaching style extend beyond the period of the intervention and may have measurable long-term effects, as transfer effects were not evident within this study. In addition, to develop a full picture of book mediums and teaching styles for young EAL learners, additional studies will need to be conducted to assess literacy related outcomes such as phonological awareness, print awareness, word reading and writing, alphabet knowledge.

5.5.7 Conclusion

The aim of this study was to examine some of the mechanisms underpinning young EAL learners' vocabulary acquisition and story compression development via different book formats. Therefore, this study evaluated two book mediums and two teaching styles for children learning English as an additional language. The findings highlight that children learning EAL may expand their vocabulary and improve their understanding through storybook reading with Interactive adult scaffolding. The factors that determine differences in scaffolding across book formats requires further investigation; however, it is possible that the benefits of shared reading may lie not in the form of the book, but rather in how they are employed by a teacher. In conclusion, by enhancing teachers' understanding of EAL vocabulary and story development, several generations of EAL children are likely to benefit.

Chapter 6: Discussion and Conclusion

The chapter discusses the findings from Study 1 (Meta-Analysis) and Study 2 (Experimental Study) regarding traditional print book and e-book reading and their use for young children, in relation to their language and literacy development. The overall findings are analysed, and the insights gained from the research are examined. The final section presents concluding remarks for both studies. Study 1 compared children's language and literacy development from narrative e-books to more traditional presentation of print stories with and without the support of an adult. As outcome measures, the emphasis was primarily on language development (vocabulary, story comprehension, speaking) and literacy development (phonological awareness, print awareness, word reading and writing, alphabet knowledge). Study 2 examined the effects of e-books and print books, as well as two teaching styles (performance and interactive) on children (aged 3-7 years) learning English as an additional language for story comprehension and vocabulary development.

6.1 Returning to the Purpose of the Thesis: Connecting the Two Studies

The main question raised in this thesis is whether e-books have any educational value for young children and whether they can be used to develop young children's language and literacy skills. There is no doubt that stories read to young children are one of the most important sources of language and literacy development (Bus et al., 1995; Hindman et al., 2014; Mol & Bus, 2011). Storybook reading can be a uniquely valuable and motivating window into the literate world of young children. As mentioned in Chapter 1, print book reading is well established in the literature, and it is regarded as a leading avenue for building early academic skills (Bus et al., 2000; Scarborough & Dobrich, 1994). By engaging in the activity of listening to stories, children expand their story comprehension skills and acquire sophisticated language. Additionally, this type of activity aids in the development of code-related skills, including phonological awareness and concepts of print. As a result, these findings were studied with the use of e-books in order to compare and contrast different book formats, namely children's e-books and print books. Most of the e-books aimed at young children are equipped with interactive and multimedia features, e.g., written text, oral reading, background music and sounds, hotspots activating animations, video, on-demand word definitions, synonyms, and other functions that may

support language and literacy development in unique ways. Another potential benefit of ebooks is their ability to serve as a source of motivation, particularly for those who exhibit a reluctance towards reading, as well as for children with learning difficulties. However, many e-books may contain distracting digital features unrelated to the story, which prevent the reader from comprehending the story and acquiring new vocabulary. There is an urgent need to deepen our understanding of the effectiveness of e-books which can either support or hinder language and literacy development in young children.

Increasingly, policy and learning standards require teachers to use technology to support literacy instruction, and e-books are a popular approach for meeting these standards and supporting children's language and literacy development. However, there is a lack of consensus concerning e-books with regard to:

- a. the extent to which e-books actually support different aspects of children's language and literacy development;
- b. which learners may benefit from use of this technology (e.g., typical developing, EAL learners);
- c. which digital features within e-books improve language and literacy outcomes; and
- d. what kind of adult scaffolding may support children's language and literacy development when using e-books.

We need to continue our e-book analyses and make recommendations on what comprises beneficial e-books for young children, in order to better serve e-book designers, parents, educators and children. The initial investigation of this thesis, specifically Chapter 4 which involves a *meta-analysis*, examined experimental research that employed e-books versus printed books for children in the age range of 3 to 8 years old, between 2008 and 2021. The study aimed to evaluate the impact of these two formats on the language and literacy development of young children. The meta-analysis yielded insights into the types of features and added components integrated into e-books that can facilitate children's educational attainment. The e-book in question was identified as the optimal type in Study 2, specifically in Chapter 5. Study 2 (Chapter 5) referred to this type of e-book as the "perfect" e-book.

The "perfect" e-book is believed to have minimal or no sound effects, no background music, and animations that are relevant to the story's content without distracting pupils

from the narrative. The present thesis utilised a particular form of e-book in its second phase (Study 2). The findings derived from the subsequent meta-analysis indicate that they align with established theories in the field of multimedia learning, such as Mayer's (2005) theory, Paivio's (1986) dual coding theory, and Sweller's (1988; Sweller et al., 2011) cognitive load theory.

Paivio's dual coding theory proposes that the information-processing system is assumed to contain two channels-an auditory/verbal channel and a visual/pictorial channelthereby enabling simultaneous processing of visual and narrative information in short-term memory (Paivio, 2008). Cross-channel links between the mental representations of the narrative and visual elements may potentially enhance the comprehension of the text, even if there are delays in vocabulary acquisition, which can contribute to the development of vocabulary skills despite complex language in narrations (Sari et al., 2019). The integration and connection between verbal (text) and visual (illustration) elements of digital storybooks could help organise information from stories and lead to successful subsequent recall as suggested by the cognitive theory of multimedia learning (Mayer, 2005). Welldesigned animated e-books hold good promise for children's vocabulary skills, as such books can stimulate readers' visual and auditory senses to comprehend unfamiliar language via the congruence between non-verbal sources (motion pictures, images) and the narration (Sun et al., 2019), as predicted by the dual coding theory of learning. The "enhanced" message may facilitate learners' acquisition of the target information and establishment of a coherent mental representation (Sun & Yin, 2020). These theories have been used as supporting pieces of evidence to explain the positive findings of Study 1 regarding the effectiveness of e-books with multimedia features (Bus et al., 2015; Kelley & Kinney, 2017; Takacs & Bus, 2016).

Building upon the findings of Study 1, Study 2 aimed to assess the efficacy of e-books that incorporate multimedia elements. Specifically, the study examined the concept of a "perfect" e-book, characterised by the absence of extraneous sound effects and background music, as well as the inclusion of animations that are pertinent to the narrative without detracting from the reader's focus in comparison to a print book. Furthermore, the effectiveness of scaffolding techniques was assessed and analysed (interactive style versus performance style). The results obtained from Study 1 indicated that the utilisation of adult scaffolding in conjunction with e-book features yielded superior outcomes when compared

to print book conditions. The aforementioned variables were subjected to testing among a sample of children who did not have English as their primary language, but rather a second and third language. Since Paivio's dual coding theory (Paivio, 1986; Sadoski & Paivio, 2013) suggests that animation embedded in an e-book may be more effective in conveying the meaning of a word than a static illustration, this theory may also hold true for children who are learning English as an Additional Language (EAL). Therefore, Study 2 hypothesised that educational media may provide dynamic visual and auditory forms of input that function as scaffolds for learners learning English as an additional language.

Despite the fact that all participants exhibited improved performance after the intervention, in the current study it was observed that the print book condition was more beneficial to enhancing the performance of EAL learners in vocabulary acquisition and story comprehension. Study 2 results come as a stark contrast to the study by Wong and Samudra (2021), in which they investigated whether language learners benefit from media content that is both audibly and visually presented, using English language proficiency as a significant contextual component that may influence vocabulary learning on screens. Their within-subject design study included 43 preschool-aged children from predominantly native Spanish-speaking households. In this study, children viewed six Sesame Street video segments on an eye-tracking Tobii device. Three of the videos taught English vocabulary words with visual-auditory congruence (dual-coding condition), in which a visual representation of the vocabulary word appeared concurrently with an auditory label. In the other condition – the visual-auditory incongruence (non dual-coding) condition children watched three clips where the visual and auditory labels occurred at different points. Researchers reported a main effect for dual-coding as visual-auditory congruence better supporting vocabulary learning in language learners than visual-auditory incongruence, F(1, 38) = 6.07, p = .018. They concluded that when children have the opportunity to simultaneously view and hear, they form cohesive mental representations of on-screen material, which facilitates in-depth language acquisition (Wong & Samudra, 2021). As previously stated, Study 2 demonstrated progress in both e-book and print book conditions from pre-tests to post-tests. However, it is important to note that Study 2 differs from Wong and Samudra's (2021) study in that it compared the e-book condition to a print book condition. However, despite the overall differing outcomes and study designs, the two studies shared a common feature. Wong and Samudra (2021) conducted a study on the efficacy of video with visual-auditory congruence in enhancing children's learning

outcomes. Their findings suggest that their multimedia educational medium (dual-coding condition) shares similarities with the "perfect" e-book described in Study 2. Consequently, it is plausible that an e-book meeting the criteria of a "perfect" e-book could potentially facilitate the development of vocabulary skills among EAL learners.

According to Vygotsky's theory of the Zone of Proximal Development, scaffolds play a pivotal role in facilitating language development by aiding students in comprehending new words and guiding them towards a more profound understanding of the new vocabulary (Vygotsky, 1980). As mentioned earlier, a second variable under investigation in Study 2 was the use of scaffolding techniques during e-book and print book reading. It is important to note that the zones of proximal development for children do not uniformly fall within a consistent range. A number of children may possess more robust EAL vocabularies and/or have increased opportunities for EAL exposure in their home environment, potentially enhancing their ability to acquire knowledge from electronic media. Therefore, Study 2 included two scaffolding techniques in order to support all children during the intervention. The *interactive teaching style* outperformed the *performance teaching style* in all comparisons. This finding was also evident in studies comparing the two teaching styles with young children (Dickinson, 2001; Dickinson & Smith, 1994; Greene Brabham & Lynch-Brown, 2002; Reese & Cox, 1999).

6.2 The Outcomes of the Study for EAL Learners in Cyprus

The findings of the study conducted in Cyprus during the 2020-2021 academic year provide valuable insights into the language learning experiences of children attending a private English school in Cyprus. Despite English being taught as the main language of instruction in this setting, it is important to recognize that for the participants, English was considered an additional language. This distinction is crucial as it underscores the diverse linguistic backgrounds and varying levels of English proficiency within the participant population. Cyprus, situated in the Eastern Mediterranean, has a complex linguistic landscape shaped by its historical and geopolitical contexts. Although Greek and Turkish are recognised as the official languages of Cyprus, English retains considerable significance on account of Cyprus's colonial history during British rule and the official position of English as a language of the European Union. As a result, English language education is pervasive in the Cypriot schooling system, with many private schools offering

English-medium instruction. The population of the study consisted of sixty children between the ages of 3 and 7, reflecting the early childhood years when language acquisition and development are particularly salient. The age range of the participants is noteworthy as it corresponds to the critical period for language learning, during which children are highly receptive to linguistic input and environmental stimuli.

The intervention took place within the school setting, providing a familiar and conducive environment for the participants to engage in the language learning activities. The use of storybooks, both in print and electronic formats, served as a medium for language exposure and skill development. By comparing the efficacy of print books versus e-books, the study addressed contemporary debates surrounding the integration of technology in education and its impact on learning outcomes. Furthermore, the study examined the influence of teaching styles on language acquisition, specifically comparing an interactive style with a performance style. This aspect of the research is particularly relevant as it speaks to pedagogical practices tailored to the needs of EAL learners. The emphasis on adult scaffolding and active engagement in the interactive teaching style aligns with best practices in language education, emphasizing the importance of supportive learning environments and meaningful interactions for language development.

These findings have implications for language education policies and practices in Cyprus, as they highlight the importance of incorporating interactive teaching methods in language learning curricula. The study's findings contribute to the understanding of the impact of interactive teaching styles on language learning by providing evidence for the effectiveness of using interactive methods in English language education in Cyprus. The findings suggest that incorporating interactive teaching methods, such as simulation games, workshops, brainstorming, and translanguaging, can enhance language learning for children who are acquiring English as an additional language in Cyprus. These findings have implications for language education policies and practices in Cyprus, as they highlight the importance of incorporating interactive teaching methods in language learning curricula.

Moreover, these findings have important implications for educational practices in Cyprus, where the need to support the acquisition of English as an additional language is a pressing concern. The results suggest that interactive storybook reading, whether in print or digital format, can be an effective approach to enhancing the vocabulary and comprehension skills of children who are learning English as an additional language. By considering the unique linguistic and cultural context of Cyprus, educators and policymakers can develop targeted interventions and instructional strategies that leverage the potential of both print and digital resources to support the language development of this specific population of learners.

6.2 Lessons Learned from the Studies Reported in this Thesis

The significance of e-book quality and how e-books are used has been highlighted in both studies of this thesis. Study 1 (Savva et al., 2022) found that adult scaffolding combined with e-book features outperformed the print book conditions. Although there are limited studies comparing the effects of adult scaffolding in both print and e-book conditions within a class context with young children, Strouse et al. (2023) published a recent study on parent-child dialogic conversations during e-book reading (with built-in, researcher-designed prompts) and print book reading. Their findings (n=67) showed that there was no indication that reading the digital version of the book or listening to the narration while it was automatically playing reduced the language or conversation quality compared to reading the print version. This study suggests that skilfully designed digital books, especially those with narration, offer prospects for high-quality shared reading that are comparable to those of print books. It should be noted that their e-book included built-in prompts, which, according to the researchers, could be a potent aid for engaging conversation.

However, when it comes to children whose first language is not the language narrated within the e-book, Study 2 revealed that EAL learners performed better with the print book condition in both vocabulary development and story comprehension. Previous studies (with L1) have reported quality differences between print and digital books, such as a higher frequency of on-topic talk in print books (Lauricella et al., 2014; Munzer et al., 2019; Parish-Morris et al., 2013). Study 2 employed a lesson plan that was descriptive in nature, incorporating targeted questions and discussion prompts on specific book pages based on the teaching style that was being assessed (performance versus interactive) to mitigate the aforementioned concerns. Therefore, identical questions were posed to children regardless of book medium. It is important to acknowledge that the sample size for each condition in Study 2 was relatively small, consisting of only 30 participants. This

limitation emphasises the need for further research on EAL learners and studies similar to Study 2, with a larger sample size. Nonetheless, revisiting the recent research conducted by Strouse et al. (2023), wherein an e-book was utilised that included built-in, researcher-designed prompts, it is plausible that the outcomes may vary for EAL learners (as opposed to the results of Study 2) if the aforementioned e-book is complemented with the qualities of a "perfect" e-book.

An essential factor to contemplate when teaching young EAL learners through e-books for their development of vocabulary and story comprehension is the level of proficiency of teachers in utilising e-books as a pedagogical tool. The pedagogical expertise of educators in utilising e-books as a teaching tool is a crucial factor, particularly when teaching EAL learners. Bay (2022) conducted a study where 43 pre-school teachers were asked regarding their views on using digital technology in early childhood learning settings in Turkey. The study revealed that educators had access to web-based technologies, with a predominant usage of smartphone applications (97%). Conversely, a lesser proportion of educators incorporated digital technology within their instructional practises, accounting for 48.8%. In relation to the perspectives of educators, the advantages of digital technology include enhanced job performance (46%), improved educational learning (39%), streamlined knowledge transfer (41%), and improved communication with families (34%). The study reveals that a significant proportion of respondents had negative perceptions regarding the impact of digital technology on children and teachers. It is believed that educators may harbour apprehensions stemming from their limited proficiency and knowledge in utilising digital technology. Specifically, 16% of respondents believed that digital technology causes distraction, while 11% asserted that it leads to addiction in children. Additionally, 7% contended that it causes teachers to lose time in searching for information, health problems (4%), and developmental problems (2%) in children (Bay, 2022). According to Koehler et al. (2007), the cultivation of technology literate individuals necessitates the prior attainment of technological literacy by teachers, who must then apply their technological expertise in classroom practices in a manner that is both meaningful and appropriate. Balçın and Ergün (2016) argue that the integration of technology in the educational process is of paramount importance in fostering individuals who are proficient in the effective utilisation of technology. The literature indicates that teachers play a crucial role in determining children's technology usage (Barron et al., 2011; Puerling, 2012), yet keeping up with the swift pace of technological advancements can pose a

challenge for educators (Ozbay & Ugurelli, 2023).

6.3 Contributions to the research literature

This thesis aimed to evaluate the efficacy of e-books compared to the traditional practice of reading stories using printed books in the early years' classroom. Currently, more teachers are increasingly incorporating e-books into their lessons. While the advantages of print books and their impact on vocabulary and story comprehension development have been well studied, there is a lack of comprehensive research on the effects of e-books, a relatively new medium, on the development of these skills in young children.

The meta-analysis conducted in this thesis revealed significant findings that were not previously reported in other meta-analyses. Consensus has been reached among several meta-analyses and other experimental research that the outcomes have had a modest nevertheless beneficial overall impact in favour of the e-book condition. The findings regarding vocabulary development align with the results reported in previous studies which provide support for the notion that digital books are more effective than print books. An additional advantage of e-books, according to the meta-analysis reported in the thesis, is their capacity to facilitate the growth of expressive vocabulary. When comparing multimedia enhancements in e-books, the results of the meta-analysis indicated a marginally positive effect in favour of the e-book condition, indicating that e-books are more advantageous than traditional print book encounters, regardless of adult scaffolding in the print book condition. For code-related skills, this effect was statistically significant (g = 0.63; 95% CI = [0.28, 0.99]). On the contrary, prior empirical research that examined the impact of multimedia e-books on code-related skill development (e.g., Arslan-Ari & Ari, 2022; Yow & Priyashri, 2019) suggested that e-book reading might not be a productive way for children to enhance their phonological awareness, print knowledge, and word reading abilities. This is because children tended to focus more on the illustrations rather than the textual content. One plausible hypothesis for this phenomenon is that the studies incorporated in this meta-analysis demonstrated teacher mediation, suggesting that the implementation of multimedia e-books by teachers to facilitate learning in code-related skills is beneficial.

Regarding the comprehension of the story, previous meta-analyses demonstrated that story

comprehension was positively and significantly affected by e-books in comparison to print books. However, this meta-analysis revealed a small positive effect on story comprehension, which was not statistically significant (g = 0.05; 95% CI = [-0.11, 0.21]). This suggests that children in the e-book conditions performed equally well compared to children in the print conditions. This conclusion aligns with the study conducted by Liman Kaban and Karadeniz (2021), where they examined the utilisation of various digital reading platforms among a sample of 96 students. Although various reading mediums were utilised in both the experimental and control groups, there was no notable disparity in their story comprehension scores. Hence, it was crucial to examine this finding in the experimental investigation of the thesis to see whether story comprehension might be enhanced by utilising e-books as opposed to print books, together with the addition of scaffolding support.

Another key discovery within the meta-analysis, that propelled Study 2 of the thesis was the comparison between the impact of digital features as scaffolding tools and print books without adult support. The results indicated a minor positive effect, although not statistically significant, suggesting that reading unsupported e-books yields similar outcomes as reading unsupported print books. However, based on a previous meta-analysis (Takacs et al., 2014), it was anticipated that a similar lack of significant findings would be observed when adults mediated print book reading compared to autonomous usage of ebooks. The findings of this meta-analysis indicated that when adults offered scaffolding support during the reading of print books, it yielded superior outcomes compared to reading e-books in isolation. Nevertheless, the presence of both an e-book and adult scaffolding resulted in superior performance compared to reading a print book with or without adult scaffolding. Hence, the meta-analysis enhanced the existing body of research by emphasising that the incorporation of digital components and discussion pertaining to the narrative is significantly more influential than the support offered during conventional print book reading. In summary, the findings suggest that the participation of adults in assisting children with reading e-books is an important factor to take into consideration during adult-child storybook reading.

Considering the aforementioned findings and the significant role of adult scaffolding in assisting learners, Study 2 examined the impact of using e-books compared to print books with adult scaffolding. In the literature, there a very few research that explored scaffolding

support with the usage of e-books. As a result, an additional measure was devised to assess the effectiveness of two distinct teaching techniques: interactive and performance teaching styles. The majority of research in the literature have mostly concentrated on assessing the impacts of a single teaching approach employed by adults during instruction, whether it be teaching a specific subject or engaging in storytelling. Both teaching techniques, interactive and performance teaching styles, discussed in the literature appear to provide favourable outcomes. Hence, it was crucial to compare these two teaching styles in the context of reading e-books and print books, as no research have previously compared these two methods using different media.

Furthermore, throughout my examination of the literature, it became evident that only a small number of studies have incorporated children who do not speak English as their primary language. The majority of research comparing e-books to paper books focused on typically developing children or children with special educational needs. It was very significant for me to incorporate children whose first language is not English, as the study was conducted in Cyprus, a country that attracts many immigrants each year whose first language is not English. However, these students were being educated in English at private schools, which is often where they enrol.

After completing a thorough search of the literature, only one research was discovered that assessed the use of an e-book in bilingual children and provided scaffolding support (Yang et al., 2022). Several studies have assessed the use of e-books with bilingual children, L2 versus L1 children, or EAL learners, but they did not include adult support in discussing the story. Comparing the utilisation of e-books with adult support to that of print books with adult support, with EAL leaners, was crucial to this investigation so as to advance beyond the conclusions drawn from the meta-analysis. Yang et al. (2022) conducted a randomised controlled study to examine the effects of bilingual conversation prompts and feedback in a multimedia interactive e-book on parent-child shared reading for young English language learners aged 3-7 years in China. Within the treatment condition, a grand total of sixty-four pairs consisting of parents and children engaged in the activity of reading a multimedia English storybook while utilising multilingual conversation prompts. By contrast, a total of forty-three sets of parents and children took part in the same activity of reading the multimedia storybook, but without any conversation prompts included. Upon reading the storybook twice, it was noted that the children who took part in the discussion-

prompt group had higher performance in terms of their understanding of the story and ability to repeat it, as compared to the control group. The significant disparity in results indicates that participating in guided conversations enhanced children's understanding and capacity to recount the narrative. Nevertheless, children in both groups shown similar improvements in their English vocabulary. There is a lack of study on the use of teacher scaffolding and the comparison between e-books and print books for young children who are learning English as an additional language. Hence, the study aimed to determine if an e-book is as effective in facilitating English language acquisition for children with additional language needs compared to typically developing children. Which scaffolding approach, interactive or performance, would be more effective in supporting this target population? Can both forms of media, namely e-books and print books, benefit EAL learners?

The findings of Study 2 indicated that both e-books and print books were beneficial for EAL learners, since the average scores for all dependent variables (i.e., vocabulary and story comprehension) rose from the pre-test to the post-test. However, it is important to note that these results were not statistically significant (p>0.05). Therefore, both media have the capacity to support EAL learners. Furthermore, when it comes to providing support through scaffolding, the interactive teaching style produced superior outcomes compared to the performance style, particularly in relation to the development of expressive vocabulary. This finding is significant, particularly for the specific group being studied, since it demonstrates that teachers may employ an interactive teaching approach to assist their students in improving their vocabulary and understanding of stories. This can be achieved by utilising many forms of media, such as e-books or print books. The findings of the thesis enhance our comprehension of the influence of an interactive teaching style on language acquisition by presenting evidence of the efficacy of utilising interactive approaches in English language instruction for children who are learning English as an additional language. The study discovered that children who are learning English as an additional language may improve their vocabulary skills by participating in storybook reading with interactive adult scaffolding. Interactive adult scaffolding refers to the process in which learners engage in active participation and interactive conversations to create information (Kim & Wilkinson, 2019; Wells, 2015). Both investigations conducted in this thesis have demonstrated that facilitating students' engagement in interactive scaffolding optimises the process of learning. This method entails engaging in

interactive dialogues where students engage in the exchange of ideas through logical reasoning, discussion, critical analysis, and expressing their thoughts (Yıldırım & Uzun, 2021). Both learners and teachers cooperate to exchange perspectives and ideas, participating in dialogic discussions to attain favourable learning outcomes. Reznitskaya (2012) states that interactive instruction comprises several essential elements. These strategies encompass asking open-ended questions to stimulate innovative thinking, distributing responsibility between educators and students, offering constructive feedback to facilitate understanding and stimulate critical thinking, engaging in metacognitive reflection for clarification, and collectively building knowledge in a social setting through interaction with peers. Lefstein (2005) describes the use of interactive teaching styles in education as democratic because it involves equal participation, self-reflection, openness, and challenging power dynamics via joint efforts to create meaning. This scaffolding technique is student-centered and child-friendly. It requires learners to actively participate in order to achieve the lesson objectives and promote their educational development, specifically in vocabulary and story comprehension.

The findings of the thesis underscore the importance of considering instructional strategies alongside the choice of educational materials. While e-books offer interactive features that can enhance engagement, traditional print books remain viable resources, particularly when supplemented with effective scaffolding techniques. These findings have several implications for educators and curriculum developers. Firstly, they underscore the importance of considering the pedagogical approaches embedded within instructional materials, whether digital or traditional. Secondly, the results highlight the potential of e-books with scaffolding support as a viable alternative to traditional print books, particularly in catering to the diverse needs of EAL learners. Additionally, the study emphasizes the significance of interactive teaching strategies in fostering vocabulary development and overall learning outcomes.

In conclusion, the thesis aimed to assess the efficacy of utilising e-books with scaffolding help as opposed to typical teaching methods with print books. There is a limited number of research that have directly evaluated the impact of scaffolding in both print and e-book forms (i.e., Broemmel et al., 2015; Ihmeideh, 2014; Rvachew et al., 2017; Segal-Drori et al., 2010; and Zhou & Yadav, 2017). Furthermore, there are even fewer studies that have examined the impacts of scaffolding among EAL learners, such as the study conducted by

Yang et al. (2022). The effectiveness of e-books, especially when combined with scaffolding support, has not been thoroughly studied in educational research, particularly in relation to EAL learners. This study sought to address this disparity by examining the effects of e-books with scaffolding assistance vs traditional print books on vocabulary acquisition and story comprehension among EAL learners in a classroom setting. The results showed that participants in both the e-book and print book groups made improvements in vocabulary from the pre-test to the post-test. This suggests that both mediums are successful in helping EAL learners expand their vocabulary. Nevertheless, a noteworthy insight arose concerning the significance of interactive teaching methods. The use of interactive teaching methods, whether through e-books or print books, helped students expand their expressive vocabulary. These findings indicate that the method of instruction, rather than the medium used, had a significant impact on enhancing learning results.

This study contributes to the growing body of literature on the use of technology in education, particularly concerning the effectiveness of e-books with scaffolding support compared to traditional print books. While both mediums demonstrated efficacy in supporting vocabulary development among EAL learners, the study emphasizes the pivotal role of interactive scaffolding in facilitating learning outcomes. Moving forward, educators and curriculum developers should consider a balanced approach that leverages both instructional strategies and technological resources to meet the diverse needs of learners in today's digital age.

6.4 Conclusion

The 21st century has witnessed significant advancements in technology, leading to rapid changes in our world. Consequently, children are growing up in an environment where their lives are significantly influenced by digital technologies (Flewitt & Cowan, 2019). Based on the findings of the two studies included in this thesis, e-books may appear to be an effective educational resource for young children. However, educators and parents may need to take certain factors into account when selecting appropriate e-books. The first study has demonstrated which characteristics and elements can support children's learning, particularly for those whose native language is the book's language of narration. Moreover, Study 1 has suggested that the combination of adult scaffolding and suitable e-books is

more effective in facilitating children's learning compared to children engaging with ebooks independently. However, there are limited studies comparing the effects of adult scaffolding in both print and e-book conditions within a class context with young children (see Savva et al., 2022). Consequently, Study 2 was formulated with the inclusion of an additional element, which involved the participation of EAL learners in the process of listening to an e-book narrated in the English language. As far as existing research is concerned, there is a dearth of studies that have examined the impact of different book mediums (namely, e-books and print) on the vocabulary and story comprehension of young EAL learners. Additionally, it is worth noting that the teacher's role in facilitating the learning process through two distinct reading styles - interactive and performance - has not been explored in this context. For children whose first language is not the language of the e-book, the teacher's input and scaffolding support have proven to be of great value for their language development. In the context of EAL learners, adult scaffolding support plays a vital role in facilitating progress in language development, whether it be through electronic or print media. This was demonstrated in Study 2, where participants showed improvement in both book formats. However, the print book intervention outperformed the "perfect" e-book intervention, suggesting that for EAL learners' other factors may play a role. It is plausible that the advantages of shared reading may not be attributed to the book's format, but rather to the scaffolding technique employed by an educator. In summary, enhancing teachers' comprehension of EAL vocabulary and narrative development is likely to benefit multiple generations of EAL students.

References

Adams, M. J. (1990). Beginning to read: Thinking and learning. MIT Press.

- Alkaaby, F., & Mavriqi, D. (2021). The effectiveness of reading aloud on vocabulary acquisition for young learners. [Independent thesis, Malmö University]. <u>https://www.diva-portal.org/smash/get/diva2:1525946/FULLTEXT02</u>
- Altun, D. (2017). Young children's literacy habits in digital world regarding digital equality perspective. Paper presented at the 69th OMEP World Assembly and International Conference, Opatija, Croatia.
- Altun, D. (2018). The efficacy of multimedia stories in preschoolers' explicit and implicit story comprehension. *Early Childhood Education Journal*, 46(6), 629-642. <u>https://doi.org/10.1007/s10643-018-0916-8</u>
- *Altun, D. (2021). Twice upon a mind: Preschoolers' narrative processing of electronic and printed stories. *Early Childhood Education Journal*, 49(2), 349-359. <u>https://doi.org/10.1007/s10643-020-01079-9</u>
- Anuradha, K. T., & Usha, H. S. (2006). E-books access models: an analytical comparative study. *The Electronic Library*, 24(5), 662-679. <u>https://doi.org/10.1108/02640470610707268</u>
- Apple. (2013). Apple awarded \$30 million iPad deal from LA Unified School District [Press release]. Retrieved from <u>www.apple.com/pr/library/2013/06/19Apple-</u> <u>Awarded- 30-Million-iPad-Deal-From-LA-Unified-School-District.html</u>
- Aram, D. (2006). Early literacy interventions: The relative roles of storybook reading, alphabetic activities, and their combination. *Reading and Writing*, 19(5), 489-515. <u>https://doi.org/10.1007/s11145-006-9005-2</u>
- Aram, D., & Biron, S. (2004). Joint storybook reading and joint writing interventions among low SES preschoolers: Differential contributions to early literacy. *Early Childhood Research Quarterly*, *19*, 588-610. https://doi-org.ezphost.dur.ac.uk/10.1016/j.ecresq.2004.10.003
- Arnold, D. S., & Whitehurst, G. J. (1994). Accelerating language development through picture book reading: A summary of dialogic reading and its effect. In D. K. Dickinson (Ed.), *Bridges to Literacy: Children, families, and schools* (pp. 103– 128). Blackwell Publishing.

Arslan-Ari, I., & Ari, F. (2022). The effect of visual cues in e-books on pre-K children's

visual attention, word recognition, and comprehension: An eye tracking study. *Journal of Research on Technology in Education*, *54*(5), 800-814. <u>https://doi.org/10.1080/15391523.2021.1938763</u>

- Association of American Publishers. (2021, January 25). AAP November 2020 statshot report: publishing industry up 24.5% for month; up 0.8% year to date. The Association of American Publishers (AAP). <u>https://publishers.org/news/aapnovember-2020-statshot-report-publishing-industry-up-24-5-for-month-up-0-8year-to-date/</u>
- Auxier, B., Anderson, M., Perrin, A., & Turner, E. (2020, July 28). Parenting children in the age of screens. Pew Research Center. <u>https://www.pewresearch.org/internet/2020/07/28/parenting-children-in-the-ageof-screens/</u>
- Baker, D. L., Santoro, L., Biancarosa, G., & Baker, S. K. (2015). Effects of quality of instruction on student vocabulary and comprehension during read alouds. Paper Presented at the Spring Conference for the Society of Research and Educational Effectiveness (SREE).
- Baker, D. L., Santoro, L., Biancarosa, G., Baker, S. K., Fien, H., & Otterstedt, J. (2020). Effects of a read aloud intervention on first grade student vocabulary, listening comprehension, and language proficiency. *Reading and Writing*, 33(10), 2697-2724.

https://doi.org/10.1007/s11145-020-10060-2

- Balçin, M. D., & Ergün, A. (2016). Technological pedagogical content knowledge (TPCK) self-efficacy scale for pre-service science teachers on material development: development, reliability and validity study. *Turkish Journal of Education*, 5(3), 130-143. <u>https://doi.org/10.19128/turje.48236</u>
- Barron, B., Cayton-Hodges, G., Bofferding, L., Copple, C., Darling-Hammond, L., & Levine, M. H. (2011). *Take a giant step: A blueprint for teaching young children in a digital age*. Joan Ganz Cooney Center at Sesame Workshop. Retrieved from <u>https://joanganzcooneycenter.org</u>
- Bay, D. N. (2022). The perspective of preschool teachers on the use of digital technology. *Southeast Asia Early Childhood Journal*, 11(2), 87-111. https://doi.org/10.37134/saecj.vol11.2.6.2022
- Beck, I. L., & McKeown, M. G. (2007). Different ways for different goals, but keep your eye on the higher verbal goals. *Vocabulary Acquisition: Implications for Reading*

Comprehension, 182-204.

- Beck, I. L., & McKeown, M. G. (2001). Text talk: Capturing the benefits of read-aloud experiences for young children. *The Reading Teacher*, 55(1), 10-20. <u>http://www.jstor.org/stable/20205005</u>
- Beck, I. L., McKeown, M. G., & Kucan, L. (2013). *Bringing words to life: Robust vocabulary instruction*. Guilford Press.
- Bedore, L. M., & Peña, E. D. (2008). Assessment of bilingual children for identification of language impairment: Current findings and implications for practice. *International Journal of Bilingual Education and Bilingualism*, 11(1), 1-29. <u>https://doi-org.ezphost.dur.ac.uk/10.2167/beb392.0</u>
- Bedore, L. M., Peña, E. D., Summers, C. L., Boerger, K. M., Resendiz, M. D., Greene, K., ... & Gillam, R. B. (2012). The measure matters: Language dominance profiles across measures in Spanish–English bilingual children. *Bilingualism: Language and Cognition*, 15(3), 616-629. https://doi.org/10.1017/S1366728912000090
- Bers, M. U., Flannery, L., Kazakoff, E. R., & Sullivan, A. (2014). Computational thinking and tinkering: Exploration of an early childhood robotics curriculum. *Computers & Education*, 72, 145–157. https://doi-org.ezphost.dur.ac.uk/10.1016/j.compedu.2013.10.020
- Beschorner, B. & Hutchison, A. (2013). iPads as a literacy teaching tool in early childhood. *International Journal of Education in Mathematics, Science and*

Technology, *1*(1), 16–24.

- Bialystok, E. (2018). Bilingual education for young children: Review of the effects and consequences. *International Journal of Bilingual Education and Bilingualism*, 21(6), 666-679.
 https://doi-org.ezphost.dur.ac.uk/10.1080/13670050.2016.1203859
- Bialystok, E., & Craik, F. I. (2010). Cognitive and linguistic processing in the bilingual mind. *Current Directions in Psychological Science*, 19(1), 19-23. <u>https://doi.org/10.1177/0963721409358571</u>
- Biemiller, A. (2004). Teaching vocabulary in the primary grades: Vocabulary instruction needed. In J. F. Baumann & E. J. Kame'enui (Eds.), *Vocabulary instruction: Research to practice* (pp. 28–40). Guilford.
- Biemiller, A., & Boote, C. (2006). An effective method for building meaning vocabulary in primary grades. *Journal of Educational Psychology*, 98(1), 44–62.
https://psycnet.apa.org/doi/10.1037/0022-0663.98.1.44

- Blok, H. (1999). Reading to young children in educational settings: A meta-analysis of recent research. *Language Learning*, 49(2), 343-371. <u>https://doi-org.ezphost.dur.ac.uk/10.1111/0023-8333.00091</u>
- Boers, F., Warren, P., Grimshaw, G., & Siyanova-Chanturia, A. (2017). On the benefits of multimodal annotations for vocabulary uptake from reading. *Computer Assisted Language Learning*, 30(7), 709–725.
 https://doi.org/10.1080/09588221.2017.1356335
- Borenstein, M., Hedges, L. V., Higgins, J. P. T., & Rothstein, H. R. (2009). *Introduction* to meta-analysis. John Wiley & Sons Ltd.
- Borenstein, M., Hedges, L., Higgins, J., & Rothstein, H. (2013). *Comprehensive metaanalysis* (Version 3.3.070) [Software]. <u>https://www.meta-analysis.com</u>
- Brannon, D., & Dauksas, L. (2014). The effectiveness of dialogic reading in increasing English language learning preschool children's expressive language. *International Research in Early Childhood Education*, 5(1), 1–10.
- *Broemmel, A. D., Moran, M. J., & Wooten, D. A. (2015). The Impact of Animated Books on the Vocabulary and Language Development of Preschool-Aged Children in Two School Settings. *Early Childhood Research & Practice*, 17(1), n1. <u>https://files.eric.ed.gov/fulltext/EJ1072346.pdf</u>

Bruner, J. S. (1966). Toward a theory of instruction (Vol. 59). Harvard University Press.

- Burchinal, M. R., Peisner-Feinberg, E., Pianta, R., & Howes, C. (2002). Development of academic skills from preschool through second grade: Family and classroom predictors of developmental trajectories. *Journal of School Psychology*, 40(5), 415-436. <u>https://doi.org/10.1016/S0022-4405(02)00107-3</u>
- Burgess, S. R., Hecht, S. A., & Lonigan, C. J. (2002). Relations of the home literacy environment (HLE) to the development of reading-related abilities: A one-year longitudinal study. *Reading Research Quarterly*, 37(4), 408–426. <u>https://doi.org/10.1598/rrq.37.4.4</u>
- Burgoyne, K., Whiteley, H.E. & Hutchinson, J.M. (2011). The development of comprehension and reading-related skills in children learning English as an additional language and their monolingual, English-speaking peers. *British Journal of Educational Psychology*, 81, 344-354. https://doi.org/10.1348/000709910X504122

Bus, A. G., & Van Ijzendoorn, M. H. (1988). Mother-child interactions, attachment, and

emergent literacy: A cross-sectional study. *Child Development*, *59*(5), 1262-1272. https://doi.org/10.2307/1130489

- Bus, A. G., Belsky, J., Ijzendoorn, M., & Crnic, K. (1997). Attachment and book reading patterns: A study of mothers, fathers, and their toddlers. *Early Childhood Research Quarterly*, *12*(1), 81–98.
 https://doi.org/10.1016/S0885-2006(97)90044-2
- Bus, A. G., Leseman, P. P., & Keultjes, P. (2000). Joint book reading across cultures: A comparison of Surinamese-Dutch, Turkish-Dutch, and Dutch parent-child dyads. *Journal of Literacy Research*, 32(1), 53-76. https://doi.org/10.1080/10862960009548064
- Bus, A. G., Neuman, S. B., & Roskos, K. (2020). Screens, apps, and digital books for young children: The promise of multimedia. AERA Open, 6(1). <u>https://doi.org/10.1177/2332858420901494</u>
- Bus, A. G., Takacs, Z. K., & Kegel, C. A. (2015). Affordances and limitations of electronic storybooks for young children's emergent literacy. *Developmental Review*, 35, 79-97. <u>https://doi.org/10.1016/j.dr.2014.12.004</u>
- Bus, A. G., Van Ijzendoorn, M. H., & Pellegrini, A. D. (1995). Joint book reading makes for success in learning to read: A meta-analysis on intergenerational transmission of literacy. *Review of Educational Research*, 65(1), 1-21. <u>https://doi.org/10.3102/00346543065001001</u>
- Cain, K., Oakhill, J., & Bryant, P. (2004). Children's reading comprehension ability: Concurrent prediction by working memory, verbal ability, and component skills. *Journal of Educational Psychology*, 96(1), 31. <u>https://psycnet.apa.org/doi/10.1037/0022-0663.96.1.31</u>
- Chen, M. P., Wang, L. C., Zou, D., Lin, S. Y., Xie, H., & Tsai, C. C. (2022). Effects of captions and English proficiency on learning effectiveness, motivation and attitude in augmented-reality-enhanced theme-based contextualized EFL learning. *Computer Assisted Language Learning*, 35(3), 381-411. <u>https://doi-org.ezphost.dur.ac.uk/10.1080/09588221.2019.1704787</u>
- Chera, P., & Wood, C. (2003). Animated multimedia 'talking books' can promote phonological awareness in children beginning to read. *Learning and Instruction*, 13, 33-52. <u>https://doi-org.ezphost.dur.ac.uk/10.1016/S0959-4752(01)00035-4</u>
- Chiong, C., Ree, J., Takeuchi, L., & Erickson, I. (2012, June 11). Print books vs. ebooks: Comparing parent-child co-reading on print, basic, and enhanced e-book

platforms. Joan Ganz Cooney Center.

http://www.joanganzcooneycenter.org/upload_kits/jgcc_ebooks_quickreport.pdf

- Chlapana, E., & Tafa, E. (2014). Effective practices to enhance immigrant kindergarteners' second language vocabulary learning through storybook reading. *Reading and Writing*, 27(9), 1619-1640. <u>https://doi.org/10.1007/s11145-014-9510-7</u>
- Chomsky, C. (1972). Stages in language development and reading exposure. *Harvard Educational Review*, 42(1), 1-33.
- Chomsky, C. (1979). Language and reading. In R. E. Shafer (Ed.) *Applied Linguistics and Reading* (pp. 112-128). Newark, DE: International Reading Association.
- Chou, M. (2014). Assessing English vocabulary and enhancing young English as a foreign language (EFL) learners' motivation through games, songs, and stories. *Education 1–13, 42* (3), 284–297.

https://doi.org.ezphost.dur.ac.uk/10.1080/03004279.2012.680899

- Chow, B. W. Y., McBride-Chang, C., Cheung, H., & Chow, C. S. L. (2008). Dialogic reading and morphology training in Chinese children: effects on language and literacy. *Developmental Psychology*, 44(1), 233–244. <u>https://doi.org/10.1037/0012-1649.44.1.233</u>
- Christ, T., Wang, X. C., & Erdemir, E. (2018). Young children's buddy reading with multimodal app books: Reading patterns and characteristics of readers, texts, and contexts. *Early Child Development and Care*, 188(8), 1012-1030. <u>https://doi.org/10.1080/03004430.2016.1241776</u>
- Clark, H. H. (1978). Review of linguistic theory: What can it say about reading? *PsycCRITIQUES*, 23. <u>https://doi.org/10.1037/017441</u>
- Clark, K. F., & Graves, M. F. (2005). Scaffolding students' comprehension of text. *The Reading Teacher*, 58(6), 570-580. <u>https://doi.org/10.1598/RT.58.6.6</u>
- Clarke, L., & Abbott, L. (2016). Young pupils', their teacher's and classroom assistants' experiences of iPads in a Northern Ireland school: "Four and five years old, who would have thought they could do that?". *British Journal of Educational Technology*, 47(6), 1051-1064. <u>https://doi.org/10.1111/bjet.12266</u>
- Clay, M. M. (1966). *Emergent reading behaviour*. [Unpublished doctoral dissertation]. University of Auckland.
- Clearinghouse, W. W. (2014). What Works Clearinghouse procedures and standards handbook (Version 3.0). *Washington, DC: Institute of Education Sciences, US*

Department of Education.

Clemens, L. F., & Kegel, C. A. (2021). Unique contribution of shared book reading on adult-child language interaction. *Journal of Child Language*, 48(2), 373-386. <u>https://doi.org/10.1017/S0305000920000331</u>

Cohen, J. (1988). Statistical power analysis for the behavioral sciences. Academic press.

- Cohen, L. E., Kramer-Vida, L., & Frye, N. (2012). Using dialogic reading as professional development to improve students' English and Spanish vocabulary. *NHSA Dialog*, 15(1), 59–80. <u>https://doi-org.ezphost.dur.ac.uk/10.1080/15240754.2011.636490</u>
- Cole, R. W. (2008). Educating everybody's children: Diverse teaching strategies for diverse learners. ASCD.
- Collins, M. F. (2010). ELL preschoolers' English vocabulary acquisition from storybook reading. *Early Childhood Research Quarterly*, 25(1), 84-97. <u>https://doi-org.ezphost.dur.ac.uk/10.1016/j.ecresq.2009.07.009</u>
- Constitution of Cyprus. (1960, April 6). *Cyprus 1960 (rev. 2013)*. <u>https://www.constituteproject.org/constitution/Cyprus_2013</u>
- Coolican, H. (2014). *Research methods and statistics in psychology*. (6th ed.). Psychology Press.
- Correa, V. I., Lo, Y., Godfrey-Hurrell, K., Swart, K., & Baker, D. L. (2015). Effects of adapted dialogic reading on oral language and vocabulary knowledge of Latino preschoolers at risk for English language delays. *Multiple Voices for Ethnically Diverse Exceptional Learners*, 15(2), 3–21. https://doi.org/10.56829/2158-396X.15.2.3
- Costley, K. C. (2014). *The Positive Effects of Technology on Teaching and Student Learning*. Russellville: Associate Professor of Curriculum & Instruction.
- Crescenzi, L., Jewitt, C., & Price, S. (2014). The role of touch in preschool children's learning using iPad versus paper interaction. *The Australian Journal of Language* and Literacy, 37(2), 86-95. <u>https://doi.org/10.1007/BF03651936</u>
- *Critelli, K. (2011). The effect of multimedia e-books on the acquisition of early literacy skills (Publication No. 1500577) [Doctoral dissertation, The William Paterson University of New Jersey]. ProQuest Dissertations Publishing.

Crystal, D. (2003). English as a global language. Cambridge University Press.

Cummins, J. (1984). *Bilingualism and special education: Issues in assessment and pedagogy*. Multilingual Matters.

- Dalton, B., & Grisham, D. L. (2011). eVoc strategies: 10 ways to use technology to build vocabulary. *The Reading Teacher*, 64(5), 306-317. https://doi.org/10.1598/RT.64.5.1
- Davidse, N. J., de Jong, M. T., Bus, A. G., Huijbregts, S. C. J., & Swaab, H. (2011). Cognitive and environmental predictors of early literacy skills. *Reading and Writing*, 24, 395-412. <u>https://doi.org/10.1007/s11145-010-9233-3</u>
- Davidson, C., Danby, S., Ekberg, S., & Thorpe, K. (2021). The interactional achievement of reading aloud by young children and parents during digital technology use. *Journal of Early Childhood Literacy*, 21(4), 475-498. <u>https://doi-org.ezphost.dur.ac.uk/10.1177/1468798419896040</u>
- De Houwer, A. (1999). Environmental factors in early bilingual development: The role of parental beliefs and attitudes. *Bilingualism and Migration*, 1999, 75-96. <u>https://doi.org/10.1515/9783110807820.75</u>
- De Jong, M. T., & Bus, A. G. (2002). Quality of book-reading matters for emergent readers: An experiment with the same book in a regular or electronic format. *Journal of Educational Psychology*, 94(1), 145– 155. <u>https://doi.org/10.1037/0022-0663.94.1.145</u>
- De Jong, M. T., & Bus, A. G. (2003). How well suited are electronic books to supporting literacy?. *Journal of Early Childhood Literacy*, 3(2), 147-164. <u>https://doi.org/10.1177/14687984030032002</u>
- De Jong, M. T., & Bus, A. G. (2004). The efficacy of electronic books in fostering kindergarten children's emergent story understanding. *Reading Research Quarterly*, 39(4), 378-393. <u>https://doi.org/10.1598/RRQ.39.4.2</u>
- De Lamo White, C., & Jin, L. (2011). Evaluation of speech and language assessment approaches with bilingual children. *International Journal of Language & Communication Disorders*, 46(6), 613-627. https://doi-org.ezphost.dur.ac.uk/10.1111/j.1460-6984.2011.00049.x
- Demie, F. (2018). English as an additional language and attainment in primary schools in England. Journal of Multilingual and Multicultural Development, 39(3), 210-223. <u>https://doi-org.ezphost.dur.ac.uk/10.1080/01434632.2017.1348508</u>
- Department for Education. (2015). Schools, pupils and their characteristics: January 2015 (No. SFR 16/2015). Available at: <u>https://www.gov.uk/government/statistics/schools-pupils-and-their-</u> <u>characteristics-january-2015</u>

Department for Education. (2019). Schools, pupils and their characteristics: January 2019. Available at:

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attac hment_data/file/812539/Schools_Pupils_and_their_Characteristics_2019_Main_ Text.pdf (Accessed: 22 July 2019).

- Department for Education. (2021). *Statutory framework for the early years foundation stage: September 2021*. Available at: <u>https://www.gov.uk/government/publications/early-years-foundation-stage-</u> <u>framework--2</u> (Accessed: 22 November 2021).
- Dewey, J. (1923). *Democracy and education: An introduction to the philosophy of education*. Macmillan.
- Dickinson, D. K. (2001). Book reading in preschool classrooms: Is recommended practice common? In D. K. Dickinson & P. O. Tabors (Eds.), *Beginning literacy with language: Young children learning at home and school* (pp. 175–204). Brookes.
- Dickinson, D. K., & McCabe, A. (2001). Bringing it all together: The multiple origins, skills, and environmental supports of early literacy. *Learning Disabilities Research and Practice*, 16, 186–202. <u>https://doi.org/10.1111/0938-8982.00019</u>
- Dickinson, D. K., & Morse, A. B. (2019). *Connecting through talk: Nurturing children's development with language*. Paul H. Brookes Publishing.
- Dickinson, D. K., & Smith, M. W. (1994). Long-term effects of preschool teachers' book readings on low-income children's vocabulary and story comprehension. *Reading Research Quarterly*, 29, 104–122. <u>https://doi.org/10.2307/747807</u>
- Dickinson, D. K., & Snow, C. E. (1987). Interrelationships among prereading and oral language skills in kindergartners from two social classes. *Early Childhood Research Quarterly*, 2(1), 1-25. https://doi.org/10.1016/0885-2006(87)90010-X
- Dickinson, D. K., & Tabors, P. O. (2001). *Beginning literacy with language: Young children learning at home and school*. Paul H Brookes Publishing.
- Dickinson, D. K., De Temple, J. M., Hirschler, J. A., & Smith, M. W.(1992). Book reading with preschoolers: Co-construction of text at home and at school. *Early Childhood Research Quarterly*, 7(3), 323–346. https://doi.org/10.1016/0885-2006(92)90025-T
- Dickinson, D. K., Hofer, K. G., Barnes, E. M., & Grifenhagen, J. F. (2014). Examining teachers' language in Head Start classrooms from a Systemic Linguistics

Approach. *Early Childhood Research Quarterly*, 29(3), 231-244. https://doi.org/10.1016/j.ecresq.2014.02.006

- Dickinson, D., & Keebler, R. (1989). Variation in preschool teachers' styles of reading books. *Discourse Processes*, 12(3), 353-375. https://doi-org.ezphost.dur.ac.uk/10.1080/01638538909544735
- Dore, R. A., Hassinger-Das, B., Brezack, N., Valladares, T. L., Paller, A., Vu, L., Golinkoff, R. M., & Hirsh-Pasek, K. (2018). The parent advantage in fostering children's e-book comprehension. *Early Childhood Research Quarterly*, 44, 24-33. <u>https://doi.org/10.1016/j.ecresq.2018.02.002</u>
- Dowdall, N., Melendez-Torres, G. J., Murray, L., Gardner, F., Hartford, L., & Cooper, P. J. (2020). Shared picture book reading interventions for child language development: A systematic review and meta-analysis. *Child Development*, 91(2), e383-e399. <u>https://doi-org.ezphost.dur.ac.uk/10.1111/cdev.13225</u>
- Dunn, L. M., & Dunn, L. M. (1981). Peabody picture vocabulary test. San Antonio: Pearson Publishers.
- Durham, R. E., Farkas, G., Hammer, C. S., Tomblin, J. B., & Catts, H. W. (2007). Kindergarten oral language skill: A key variable in the intergenerational transmission of socioeconomic status. *Research in Social Stratification and Mobility*, 25(4), 294–305. <u>https://doi.org/10.1016/j.rssm.2007.03.001</u>
- Durkin, D. (1966). *Children who read early*. New York: Teachers College Press. Hindman, A. H., Skibbe, L. E., & Foster, T. D. (2014). Exploring the variety of parental talk during shared book reading and its contributions to preschool language and literacy: evidence from the early childhood longitudinal study-birth cohort. *Reading and Writing*, 27, 287-313. <u>https://doi.org/10.1007/s11145-013-9445-4</u>
- Duval, S. (2005). The trim and fill method. In H.R. Rothstein, A. J. Sutton & M. Borenstein (Eds.), *Publication bias in meta-analysis: Prevention, assessment and adjustments*, (pp. 127-144). Wiley. <u>https://doi.org/10.1002/0470870168</u>
- Duval, S., & Tweedie, R. (2000). Trim and fill: a simple funnel-plot–based method of testing and adjusting for publication bias in meta-analysis. *Biometrics*, 56(2), 455-463. <u>https://doi.org/10.1111/j.0006-341X.2000.00455.x</u>
- Egger, M., Juni, P., Bartlett, C., Holenstein, F., & Sterne, J. (2003). How important are comprehensive literature searches and the assessment of trial quality in systematic reviews? Empirical study. *Health Technology Assessment*, 7(1), 1-76. Retrieved

from

https://pdfs.semanticscholar.org/18da/e5c7d340138dba23fcfe35754b09d1dd89c8. pdf

- Eggleston, C., Wang, X. C., & Choi, Y. (2022). Mother-toddler shared reading with electronic versus print books: Mothers' language use and perspectives. *Early Education and Development*, 33(7), 1152-1174. https://doi.org/10.1080/10409289.2021.1943638
- Ehri, L. C. (1989). The development of spelling knowledge and its role in reading acquisition and reading disability. *Journal of Learning Disabilities*, 22(6), 356-365. <u>https://doi-org.ezphost.dur.ac.uk/10.1177/002221948902200606</u>
- Ehri, L. C., & Sweet, J. (1991). Fingerpoint-reading of memorized text: What enables beginners to process the print?. *Reading Research Quarterly*, 442-462. <u>https://doi.org/10.2307/747897</u>
- *Eng, C. M., Tomasic, A. S., & Thiessen, E. D. (2020). Contingent responsivity in Ebooks modeled from quality adult-child interactions: Effects on children's learning and attention. *Developmental Psychology*, 56(2), 285–297. <u>https://doi.org/10.1037/dev0000869</u>
- Estefani, T., & Queiroz, J. (2018). Children's picturebook goes digital: implications on cognition. MATLIT: Materialidades da Literatura, 6(2), 115-127. https://doi.org/10.14195/2182-8830_6-2_9
- Farrant, B. M., & Zubrick, S. R. (2012). Early vocabulary development: The importance of joint attention and parent-child book reading. *First Language*, 32(3), 343-364. <u>https://doi.org/10.1177/0142723711422626</u>
- Farver, J. A. M., Lonigan, C. J., & Eppe, S. (2009). Effective early literacy skill development for young Spanish-speaking English language learners: An experimental study of two methods. *Child Development*, 80(3), 703-719. <u>https://doi-org.ezphost.dur.ac.uk/10.1111/j.1467-8624.2009.01292.x</u>
- Faul, F., Erdfelder, E., Lang, A. G., & Buchner, A. (2007). G* Power 3: A flexible statistical power analysis program for the social, behavioral, and biomedical sciences. *Behavior Research Methods*, 39(2), 175-191. https://doi.org/10.3758/BF03193146
- Festman, J. (2021). Learning and processing multiple languages: The more the easier?. *Language Learning*, 71(S1), 121-162. <u>https://doi-org.ezphost.dur.ac.uk/10.1111/lang.12437</u>

- Figueras-Daniel, A., & Barnett, W. S. (2013). *Preparing young Hispanic dual language learners for a knowledge economy*. National Institute for Early Education.
- Fisher, D., Flood, J., Lapp, D., & Frey, N. (2004). Interactive read-alouds: Is there a common set of implementation practices? *The Reading Teacher*, 58(1), 8-17. <u>https://doi.org/10.1598/RT.58.1.1</u>
- Fitton, L., McIlraith, A. L., & Wood, C. L. (2018). Shared book reading interventions with English learners: A meta-analysis. *Review of Educational Research*, 88(5), 712-751. <u>https://doi-org.ezphost.dur.ac.uk/10.3102/0034654318790909</u>
- Flack, Z. M., Field, A. P., & Horst, J. S. (2018). The effects of shared storybook reading on word learning: A meta-analysis. *Developmental Psychology*, 54(7), 1334– 1346. <u>https://doi.org/10.1037/dev0000512</u>
- Fletcher, K. L., & Reese, E. (2005). Picture book reading with young children: A conceptual framework. *Developmental Review*, 25(1), 64-103. <u>https://doi-org.ezphost.dur.ac.uk/10.1016/j.dr.2004.08.009</u>
- Flewitt, R., & Cowan, K. (2019). Valuing young children's signs of learning:Observation and digital documentation of play in early years classrooms. *Final Report*. The Froebel Trust.
- Florit, E., Roch, M., & Levorato, M. C. (2011). Listening text comprehension of explicit and implicit information in preschoolers: The role of verbal and inferential skills. *Discourse Processes*, 48(2), 119-138. <u>https://doiorg.ezphost.dur.ac.uk/10.1080/0163853X.2010.494244</u>
- Gathercole, S. E., & Baddeley, A. D. (2014). *Working memory and language*. Psychology Press.
- Gerde, H. K., & Powell, D. R. (2009). Teacher education, book-reading practices, and children's language growth across one year of Head Start. *Early Education and Development*, 20(2), 211-237.
 https://doi-org.ezphost.dur.ac.uk/10.1080/10409280802595417
- Gilkerson, J., Richards, J. A., & Topping, K. J. (2017). The impact of book reading in the early years on parent–child language interaction. *Journal of Early Childhood Literacy*, 17(1), 92-110. <u>https://doi.org/10.1177/1468798415608907</u>
- Giroir, S., Grimaldo, L. R., Vaughn, S., & Roberts, G. (2015). Interactive read-alouds for English learners in the elementary grades. *The Reading Teacher*, 68(8), 639-648. <u>https://doi.org/10.1002/trtr.1354</u>

Goldenberg, C. (2020). Reading wars, reading science, and English learners. Reading

Research Quarterly, 55, S131-S144. https://doi.org/10.1002/rrq.340

- *Gong, Z., & Levy, B. A. (2009). Four year old children's acquisition of print knowledge during electronic storybook reading. *Reading and Writing*, 22(8), 889-905. <u>https://doi.org/10.1007/s11145-008-9130-1</u>
- Gonzalez, J. E., Pollard-Durodola, S., Simmons, D. C., Taylor, A. B., Davis, M. J.,
 Fogarty, M., & Simmons, L. (2014). Enhancing preschool children's vocabulary:
 Effects of teacher talk before, during and after shared reading. *Early Childhood Research Quarterly*, 29(2), 214-226. https://doi.org/10.1016/j.ecresq.2013.11.001
- Greene Brabham, E., & Lynch-Brown, C. (2002). Effects of teachers' reading-aloud styles on vocabulary acquisition and comprehension of students in the early elementary grades. *Journal of Educational Psychology*, 94(3), 465–473. https://doi.org/10.1037/0022-0663.94.3.465
- Griffith, S. F., Hart, K. C., Mavrakis, A. A., & Bagner, D. M. (2022). Making the best of app use: The impact of parent-child co-use of interactive media on children's learning in the US. *Journal of Children and Media*, 16(2), 271-287. <u>https://doi-org.ezphost.dur.ac.uk/10.1080/17482798.2021.1970599</u>
- Grolig, L., Cohrdes, C., Tiffin-Richards, S. P., & Schroeder, S. (2019). Effects of preschoolers' storybook exposure and literacy environments on lower level and higher level language skills. *Reading and Writing*, 32(4), 1061-1084. <u>https://doi.org/10.1007/s11145-018-9901-2</u>
- Grosjean, F. (2010). Bilingual. Life and Reality. Harvard University Press, MA.

Haas, L.E., Lasley, E., Sosebee, T., & Cox, D. (2017). Increasing motivation of Struggling Readers: Can e-readers, Apps, and Support Features Help?. *Journal of Literacy and Technology*, 18(2), 90-120.
<u>http://www.literacyandtechnology.org/uploads/1/3/6/8/136889/jlt_v18_2_haas_la</u> sley_sosebee_cox.pdf

- Hammarberg, B. (2001). Roles of L1 and L2 in L3 production and acquisition. In J.
 Cenoz, B. Hufeisen, & U. Jessner (Eds.), *Cross-linguistic influence in third language acquisition. Psycholinguistic perspectives* (pp. 21–41). Multilingual Matters.
- Hargrave, A. C., & Sénéchal, M. (2000). A book reading intervention with preschool children who have limited vocabularies: The benefits of regular reading and dialogic reading. *Early Childhood Research Quarterly*, 15, 75–90. <u>https://doi.org/10.1016/S08852006(99)00038-1</u>

- Hassinger-Das, B., Brennan, S., Dore, R. A., Golinkoff, R. M., & Hirsh-Pasek, K. (2020). Children and Screens. *Annual Review of Developmental Psychology*, 2, 69–92. <u>https://doi.org/10.1146/annurev-devpsych-060320-095612</u>
- Hedges, L. V., & Olkin, I. (2014). Statistical methods for meta-analysis. Academic press.
- Higgins, J. P., Ramsay, C., Reeves, B. C., Deeks, J. J., Shea, B., Valentine, J. C., Tugwell, P., & Wells, G. (2013). Issues relating to study design and risk of bias when including non-randomized studies in systematic reviews on the effects of interventions. *Research Synthesis Methods*, 4(1), 12-25. https://doi.org/10.1002/jrsm.1056
- Hindman, A. H., & Wasik, B. A. (2008). Head Start teachers' beliefs about language and literacy instruction. *Early Childhood Research Quarterly*, 23(4), 479-492. <u>https://doi-org.ezphost.dur.ac.uk/10.1016/j.ecresq.2008.06.002</u>
- Hindman, A. H., Connor, C. M., Jewkes, A. M., & Morrison, F. J. (2008). Untangling the effects of shared book reading: Multiple factors and their associations with preschool literacy outcomes. *Early Childhood Research Quarterly*, 23(3), 330-350. <u>https://doi-org.ezphost.dur.ac.uk/10.1016/j.ecresq.2008.01.005</u>
- Hindman, A. H., Wasik, B, A., & Erhart, A. C. (2012). Shared book reading and head start preschoolers' vocabulary learning: The role of book-related discussion and curricular connections. *Early Education and Development*, 23, 451-474. <u>https://doi.org/10.1080/10409289.2010.537250</u>
- Hiniker, A., Radesky, J. S., Livingstone, S., & Blum-Ross, A. (2019). Moving Beyond "The Great Screen Time Debate" in the Design of Technology for Children. In Extended Abstracts of the 2019 CHI Conference on Human Factors in Computing Systems (pp. 1-6). <u>https://doi.org/10.1145/3290607.3311745</u>
- Hirsh-Pasek, K., Zosh, J. M., Golinkoff, R. M., Gray, J. H., Robb, M. B., & Kaufman, J. (2015). Putting education in "educational" apps: Lessons from the science of learning. *Psychological Science in the Public Interest*, 16(1), 3-34. <u>https://doi.org/10.1177/1529100615569721</u>
- Hisrich, K. E., & McCaffrey, M. (2021). Planning and Preparing for Read-Alouds. *IllInoIs Reading Council Journal*, 49(2), 12. <u>https://doi.org/10.33600/IRCJ.49.2.2021.12</u>
- Hoel, T., Stangeland, E. B., & Schulz-Heidorf, K. (2020). What Happens Before Book Reading Starts? an Analysis of Teacher–Child Behaviours With Print and Digital Books. *Frontiers in Psychology*, 11, 3092.

https://doi.org/10.3389/fpsyg.2020.570652

Hoffman, J. L., & Paciga, K. A. (2014). Click, swipe, and read: Sharing e-books with toddlers and preschoolers. *Early Childhood Education Journal*, 42(6), 379-388. <u>https://doi.org/10.1007/s10643-013-0622-5</u>

Holdaway, D. (1979). The foundations of literacy (Vol. 138). Ashton Scholastic.

- *Homer, B. D., Kinzer, C. K., Plass, J. L., Letourneau, S. M., Hoffman, D., Bromley, M., Hayward, E. O., Turkay, S., & Kornak, Y. (2014). Moved to learn: The effects of interactivity in a Kinect-based literacy game for beginning readers. *Computers & Education*, 74, 37-49. <u>https://doi.org/10.1016/j.compedu.2014.01.007</u>
- Huck, C., Helper, S., & Hickman, J. (1989). *Children's literature in the elementary classroom* (5th ed.). Har- court Brace College Publisher.
- Huennekens, M. E., & Xu, Y. (2010). Effects of a cross-linguistic storybook intervention on the second language development of two preschool English language learners. *Early Childhood Education Journal*, 38(1), 19–26. https://doi.org/10.1007/s10643-010-0385-1
- Huennekens, M. E., & Xu, Y. (2015). Using dialogic reading to enhance emergent literacy skills of young dual language learners. *Early Child Development and Care, 186*(2), 324–340.

https://doi-org.ezphost.dur.ac.uk/10.1080/03004430.2015.1031125

- Hutchinson, J. M., Whiteley, H. E., Smith, C. D., & Connors, L. (2003). The developmental progression of comprehension-related skills in children learning EAL. *Journal of Research in Reading*, 26(1), 19-32. <u>https://doi-org.ezphost.dur.ac.uk/10.1111/1467-9817.261003</u>
- *Ihmeideh, F. M. (2014). The effect of electronic books on enhancing emergent literacy skills of pre-school children. *Computers & Education*, 79, 40-48. <u>https://doi.org/10.1016/j.compedu.2014.07.008</u>
- Ingram, N. B. (2020). The use of e-books, computers, and tablets to enhance reading skills. [Doctoral dissertation, Kennesaw State University]. <u>https://digitalcommons.kennesaw.edu/teachleaddoc_etd/41</u>
- Inoue, T., Georgiou, G. K., Parrila, R., & Kirby, J. R. (2018). Examining an extended home literacy model: The mediating roles of emergent literacy skills and reading fluency. *Scientific Studies of Reading*, 22(4), 273-288. https://doi.org/10.1080/10888438.2018.1435663

Jiménez-Castellanos, O. H., & Garcia, D. (2017). School expenditures and academic

achievement differences between high-ELL-performing and low-ELL-performing high schools. *Bilingual Research Journal*, 40(3), 318-330. <u>https://doi-org.ezphost.dur.ac.uk/10.1080/15235882.2017.1342717</u>

- Justice, L. M., & Ezell, H. K. (2000). Enhancing children's print and word awareness through home-based parent intervention. *American Journal of Speech-Language Pathology*, 9(3), 257-269. <u>https://doi.org/10.1044/1058-0360.0903.257</u>
- Justice, L. M., & Ezell, H. K. (2002). Use of storybook reading to increase print awareness in at-risk children. American Journal of Speech-Language Pathology, 11(1), 17–29. <u>https://doi.org/10.1044/1058-0360(2002/003)</u>
- Justice, L. M., & Ezell, H. K. (2004). Print referencing: An emergent literacy enhancement strategy and its clinical applications. *Language, Speech, and Hearing Services in Schools*, 35(2), 185-193. <u>https://doi.org/10.1044/0161-1461(2004/018)</u>
- Justice, L. M., Meier, J., & Walpole, S. (2005). Learning new words from storybooks: An efficacy study with at-risk kindergartners. *Language, Speech, and Hearing Services in Schools, 36*, 17–32. <u>https://doi.org/10.1044/0161-1461(2005/003)</u>
- Kaderavek, J. N., & Justice, L. M. (2002). Shared storybook reading as an intervention context: Promises and potential pitfalls. *American Journal of Speech-Language Pathology*, 11, 395–406. <u>https://doi.org/10.1044/1058-0360(2002/043)</u>
- Karemaker, A. M., Pitchford, N. J., & O'Malley, C. (2010). Does whole-word multimedia software support literacy acquisition?. *Reading and Writing*, 23(1), 31-51. <u>https://doi.org/10.1007/s11145-008-9148-4</u>
- *Karemaker, A., Jelley, F., Clancy, C., & Sylva, K. (2017). The effects on children's literacy skills of reading e-books with different features: Are 'bells and whistles' over-rated?. *International Journal of Child-Computer Interaction*, 12, 30-36. https://doi.org/10.1016/j.ijcci.2017.01.004
- Karweit, N., & Wasik, B. A. (1996). The effects of story reading programs on literacy and language development of disadvantaged preschoolers. *Journal of Education for Students Placed at Risk*, 1(4), 319-348.
 https://doi-org.ezphost.dur.ac.uk/10.1207/s15327671espr0104_4
- Kayaoglu, M. N., & Akbas, D. (2011). A small scale experimental study: Using animations to learn vocabulary. *Turkish Online Journal of Educational Technology-TOJET*, 10(2), 24-30.
- *Kelley, E. S., & Kinney, K. (2017). Word learning and story comprehension from

digital storybooks: Does interaction make a difference?. *Journal of Educational Computing Research*, 55(3), 410-428. <u>https://doi.org/10.1177/0735633116669811</u>

- Kendeou, P., Bohn-Gettler, C., White, M. J., & Van Den Broek, P. (2008). Children's inference generation across different media. *Journal of Research in Reading*, 31(3), 259-272. <u>https://doi.org/10.1111/j.1467-9817.2008.00370.x</u>
- Kennedy, C., & McLoughlin, A. (2023). Developing the emergent literacy skills of English language learners through dialogic reading: A systematic review. *Early Childhood Education Journal*, 51(2), 317-332. <u>https://doi.org/10.1007/s10643-021-01291-1</u>
- Kim, M. Y., & Wilkinson, I. A. (2019). What is dialogic teaching? Constructing, deconstructing, and reconstructing a pedagogy of classroom talk. *Learning, Culture and Social Interaction*, 21, 70-86. https://doi.org/10.1016/j.lcsi.2019.02.003
- Kirschner, P. A. (2002). Cognitive load theory: Implications of cognitive load theory on the design of learning. *Learning and instruction*, 12(1), 1-10. <u>https://doiorg.ezphost.dur.ac.uk/10.1016/S0959-4752(01)00014-7</u>
- Koehler, M. J., Mishra, P., & Yahya, K. (2007). Tracing the development of teacher knowledge in a design seminar: Integrating content, pedagogy and technology. *Computers & Education*, 49(3), 740-762.
 https://doiorg.ezphost.dur.ac.uk/10.1016/j.compedu.2005.11.012
- Kohnert, K. (2010). Bilingual children with primary language impairment: Issues, evidence and implications for clinical actions. *Journal of Communication Disorders*, 43, 456–473. <u>https://doi.org/10.1016/j.jcomdis.2010.02.002</u>
- Korat, O. (2010). Reading electronic books as a support for vocabulary, story comprehension and word reading in kindergarten and first grade. *Computers & Education*, 55(1), 24-31. <u>https://doi.org/10.1016/j.compedu.2009.11.014</u>
- Korat, O., Atishkin, S., & Segal-Drori, O. (2022a). Vocabulary enrichment using an e-book with and without kindergarten teacher's support among LSES children. *Early Child Development and Care*, *192*(9), 1384-1401. <u>https://doi-org.ezphost.dur.ac.uk/10.1080/03004430.2021.1885391</u>
- Korat, O., & Blau, H. (2010). Repeated reading of CD-ROM storybook as a support for emergent literacy: A developmental perspective in two SES groups. *Journal of Educational Computing Research*, 43(4), 445-466. <u>https://doi.org/10.2190/EC.43.4.b</u>

- Korat, O., & Falk, Y. (2019). Ten years after: Revisiting the question of e-book quality as early language and literacy support. *Journal of Early Childhood Literacy*, 19(2), 206-223. <u>https://doi.org/10.1177/1468798417712105</u>
- Korat, O., Kozlov-Peretz, O., & Segal-Drori, O. (2017). Repeated e-book reading and its contribution to learning new words among kindergartners. *Journal of Education and Training Studies*, 5(7), 60-72. <u>https://doi.org/10.11114/jets.v5i7.2498</u>
- *Korat, O., Levin, I., Atishkin, S., & Turgeman, M. (2014a). E-book as facilitator of vocabulary acquisition: Support of adults, dynamic dictionary and static dictionary. *Reading and Writing*, 27(4), 613-629.
 https://doi.org/10.1007/s11145-013-9474-z
- *Korat, O., Levin, I., Ben-Shabt, A., Shneor, D., & Bokovza, L. (2014b). Dynamic Versus Static Dictionary With and Without Printed Focal Words in E-Book Reading as Facilitator for Word Learning. *Reading Research Quarterly*, 49(4), 371-386. <u>https://doi.org/10.1002/rrq.78</u>
- Korat, O., & Segal-Drori, O. (2016). E-book and printed book reading in different contexts as emergent literacy facilitator. *Early Education and Development*, 27(4), 532–550. <u>https://doi.org/10.1080/10409289.2016.1095613</u>
- Korat, O., & Shamir, A. (2007). Electronic books versus adult readers: Effects on children's emergent literacy as a function of social class. *Journal of Computer Assisted Learning*, 23(3), 248-259.
 https://doi-org.ezphost.dur.ac.uk/10.1111/j.1365-2729.2006.00213.x
- Korat, O., & Shamir, A. (2008). The educational electronic book as a tool for supporting children's emergent literacy in low versus middle SES groups. *Computers & Education*, 50(1), 110-124.

https://doi-org.ezphost.dur.ac.uk/10.1016/j.compedu.2006.04.002

- Korat, O., & Shamir, A. (2012). Direct and indirect teaching: Using e-books for supporting vocabulary, word reading, and story comprehension for young children. *Journal of Educational Computing Research*, 46(2), 135-152. <u>https://doi-org.ezphost.dur.ac.uk/10.2190/EC.46.2.b</u>
- Korat, O., Shamir, A., & Arbiv, L. (2011). E-books as support for emergent writing with and without adult assistance. *Education and Information Technologies*, 16(3), 301-318. <u>https://doi.org/10.1007/s10639-010-9127-7</u>
- Korat, O., & Shneor, D. (2019). Can e-books support low SES parental mediation to enrich children's vocabulary?. *First Language*, 39(3), 344-364.

https://doi.org/10.1177/0142723718822443

- Korat, O., Tourgeman, M., & Segal-Drori, O. (2022b). E-book reading in kindergarten and story comprehension support. *Reading and Writing*, 35(1), 155-175. <u>https://doi.org/10.1007/s11145-021-10175-0</u>
- Kozlowski, M. (2020, August 9). *HarperCollins ebook sales are up 26%*. GoodEReader. https://goodereader.com/blog/business-news/harpercollins-ebook-sales-are-up-26
- Kozlowski, M. (2021, February 5). *Audiobooks and ebooks are 18% of HarperCollins Revenue*. GoodEReader. <u>https://goodereader.com/blog/digital-</u> publishing/audiobooks-and-ebooks-are-18-of-harpercollins-revenue
- *Kozminsky, E., & Asher-Sadon, R. (2013). Media type influences preschooler's literacy development: E-book versus printed book reading. *Interdisciplinary Journal of E-Learning and Learning Objects*, 9(1), 233-247.
 http://www.ijello.org/Volume9/IJELLOv9p233-247Kozminsky0865.pdf
- Kucirkova, N., & Flewitt, R. (2020). The future-gazing potential of digital personalization in young children's reading: views from education professionals and app designers. *Early Child Development and Care*, 190(2), 135-149. https://doi.org/10.1080/03004430.2018.1458718
- Lauricella, A. R., Barr, R., & Calvert, S. L. (2014). Parent-child interactions during traditional and computer storybook reading for children's comprehension:
 Implications for electronic storybook design. *International Journal of Child-Computer Interaction*, 2(1), 17–25. https://doi.org/10.1016/j.ijcci.2014.07.001
- Lantz, J. L., Myers, J., & Wilson, R. (2020). Digital storytelling and young children: Transforming learning through creative use of technology. In *Handbook of research on integrating digital technology with literacy pedagogies* (pp. 212-231). IGI Global. <u>https://doi.org/10.4018/978-1-7998-0246-4.ch010</u>
- Lawrence, A., & Choe, D. E. (2021). Mobile Media and Young Children's Cognitive Skills: A Review. Academic Pediatrics, 21(6), 996-1000. <u>https://doi.org/10.1016/j.acap.2021.01.007</u>
- Leacox, L., & Jackson, C. W. (2014). Spanish vocabulary-bridging technology-enhanced instruction for young English language learners' word learning. *Journal of Early Childhood Literacy*, 14(2), 175–197. <u>https://doi.org/10.1177/1468798412458518</u>
- *Lee, S. H. (2020). Learning vocabulary from e-book reading and recorded word explanation for low-income elementary students with and without reading difficulties. *Reading and Writing*, *33*(3), 691-717.

https://doi.org/10.1007/s11145-019-09983-2

- Lefebvre, P., Trudeau, N., & Sutton, A. (2011). Enhancing vocabulary, print awareness and phonological awareness through shared storybook reading with low-income preschoolers. *Journal of Early Childhood Literacy*, *11*(4), 453-479. https://doi.org/10.1177/1468798411416581
- Lefstein, A. (2005). Thinking about the technical and the personal in teaching. *Cambridge Journal of Education*, *35*(3), 333-356. <u>https://doi.org/10.1080/03057640500319040</u>
- Lenhart, J., Lenhard, W., Vaahtoranta, E., & Suggate, S. (2020). More than words: Narrator engagement during storytelling increases children's word learning, story comprehension, and on-task behavior. *Early Childhood Research Quarterly*, *51*, 338-351. <u>https://doi.org/10.1016/j.ecresq.2019.12.009</u>
- Lenneberg, E. H. (1967). The biological foundations of language. *Hospital Practice*, 2(12), 59-67. <u>https://doi-</u> <u>org.ezphost.dur.ac.uk/10.1080/21548331.1967.11707799</u>
- Leslie, L., & Allen, L. (1999). Factors that predict success in an early literacy intervention project. *Reading Research Quarterly*, 34(4), 404-424. <u>https://doi.org/10.1598/RRQ.34.4.2</u>
- *Liao, C. N., Chang, K. E., Huang, Y. C., & Sung, Y. T. (2020). Electronic storybook design, kindergartners' visual attention, and print awareness: An eye-tracking investigation. *Computers & Education*, 144, 103703. <u>https://doi.org/10.1016/j.compedu.2019.103703</u>
- Liman Kaban, A., & Karadeniz, S. (2021). Children's Reading Comprehension and Motivation on Screen Versus on Paper. SAGE Open, 11(1). https://doi.org/10.1177/2158244020988849
- Lipsey, M. W., & Wilson, D. B. (2001). Practical meta-analysis. Sage.
- Lonigan, C. J., Burgess, S. R., & Anthony, J. L. (2000). Development of emergent literacy and early reading skills in preschool children: evidence from a latentvariable longitudinal study. *Developmental Psychology*, 36(5), 596. <u>https://doi.org/10.1037/0012-1649.36.5.596</u>
- Lugo-Neris, M. J., Jackson, C. W., & Goldstein, H. (2010). Facilitating vocabulary acquisition of young English language learners. *Language, Speech, and Hearing Services in Schools*, 41(3), 314–327. <u>https://doi.org/10.1044/0161-1461(2009/07-0082</u>

- Machado, J. M. (2015). *Early childhood experiences in language arts: Early literacy*. Cengage Learning.
- Mahon, M., & Crutchley, A. (2006). Performance of typically-developing school-age children with English as an additional language on the British Picture Vocabulary Scales II. *Child Language Teaching and Therapy*, 22(3), 333-351. <u>https://doiorg.ezphost.dur.ac.uk/10.1191/0265659006ct311xx</u>
- Marsh, J., Brooks, G., Hughes, J., Ritchie, L., & Roberts, S. (2005). Digital beginnings: Young children's use of popular culture, media and new technologies. University of Sheffield.
- Marulis, L. M., & Neuman, S. B. (2010). The effects of vocabulary intervention on young children's word learning: A meta-analysis. *Review of Educational Research*, 80(3), 300–335. <u>https://doi.org/10.3102/0034654310377087</u>
- Mayer, R. E. (2005). Cognitive theory of multimedia learning. In R. Mayer & R. E.
 Mayer (Eds.), *The Cambridge handbook of multimedia learning*, (pp. 31-48).
 Cambridge University Press.
- Mayer, R. E. (2014). Incorporating motivation into multimedia learning. *Learning and Instruction*, 29, 171-173. <u>https://doi.org/10.1016/j.learninstruc.2013.04.003</u>
- McBride, C. (2015). *Children's literacy development: A cross-cultural perspective on learning to read and write*. Routledge.
- McCabe, A., Boccia, J., Bennett, M. B., Lyman, N., & Hagen, R. (2010). Improving oral language and literacy skills in preschool children from disadvantaged backgrounds: Remembering, writing, reading (RWR). *Imagination, Cognition* and Personality, 29(4), 363–390. <u>https://doi.org/10.2190/IC.29.4.f</u>
- McCaffrey, M., & Hisrich, K. E. (2017). Read-alouds in the classroom: A pilot study of teachers' self-reporting practices. *Reading Improvement*, 54(3), 93-100.
- McCardle, P., Scarborough, H. S., & Catts, H. W. (2001). Predicting, explaining, and preventing children's reading difficulties. *Learning Disabilities Research & Practice*, 16(4), 230-239. <u>https://doi.org/10.1111/0938-8982.00023</u>

McKee, D. (2005). Elmer and Rose (D. McKee, Illus.). Andersen Press.

- McLeod, A. N., & McDade, H. L. (2011). Preschoolers' incidental learning of novel words during storybook reading. *Communication Disorders Quarterly*, 32(4), 256-266. <u>https://doi-org.ezphost.dur.ac.uk/10.1177/1525740109354777</u>
- Mol, S. E., & Bus, A. G. (2011). To read or not to read: a meta-analysis of print exposure from infancy to early adulthood. *Psychological Bulletin*, 137(2), 267-296.

https://doi.org/10.1037/a0021890

- Mol, S. E., Bus, A. G., & De Jong, M. T. (2009). Interactive book reading in early education: A tool to stimulate print knowledge as well as oral language. *Review of Educational Research*, 79(2), 979-1007.
 https://doi.org/10.3102/0034654309332561
- Mol, S. E., Bus, A. G., De Jong, M. T., & Smeets, D. J. (2008). Added value of dialogic parent–child book readings: A meta-analysis. *Early Education and Development*, 19(1), 7-26. <u>https://doi.org/10.1080/10409280701838603</u>
- Montag, J. L., Jones, M. N., & Smith, L. B. (2015). The words children hear: Picture books and the statistics for language learning. *Psychological Science*, 26(9), 1489-1496. <u>https://doi.org/10.1177/0956797615594361</u>
- Moody, A. K. (2010). Using electronic books in the classroom to enhance emergent literacy skills in young children. *Journal of Literacy and Technology*, 11(4), 22-52.
- Moody, A. K., Justice, L. M., & Cabell, S. Q. (2010). Electronic versus traditional storybooks: Relative influence on preschool children's engagement and communication. *Journal of Early Childhood Literacy*, *10*(3), 294-313. https://doi.org/10.1177/1468798410372162
- Moreno, R., & Mayer, R. E. (2000). A coherence effect in multimedia learning: The case for minimizing irrelevant sounds in the design of multimedia instructional messages. *Journal of Educational Psychology*, 92(1), 117–125. <u>https://doi.org/10.1037/0022-0663.92.1.117</u>
- Morrow, L. M. (1988). Young children's responses to one-to-one story readings in school settings. *Reading Research Quarterly*, 23(1), 89-107. <u>https://doi.org/10.2307/747906</u>
- Morrow, L. M., O'Connor, E. M., & Smith, J. K. (1990). Effects of a story reading program on the literacy development of at-risk kindergarten children. *Journal of Reading Behavior*, 22(3), 255-275. https://doi-org.ezphost.dur.ac.uk/10.1080/10862969009547710
- Munzer, T. G., Miller, A. L., Weeks, H. M., Kaciroti, N., & Radesky, J. (2019). Differences in parent-toddler interactions with electronic versus print books. *Pediatrics*, 143(4). <u>https://doi.org/10.1542/peds.2018-2012</u>
- National Research Council (1998). Preventing reading difficulties in young children. C.E. Snow, M. S. Burns, & P. Griffin (Eds.). National Academy Press.

- Neuman, S. B. (1997). Television as a learning environment: A theory of synergy. In J. Flood, S. B. Heath, & D. Lapp (Eds.), *Handbook of research on teaching literacy through the communicative and visual arts* (pp. 15–30). Simon & Schuster.
- Neuman, S. B., & Kaefer, T. (2018). Developing low-income children's vocabulary and content knowledge through a shared book reading program. *Contemporary Educational Psychology*, 52, 15-24.
 https://doi.org/10.1016/j.cedpsych.2017.12.001

Neuman, S. B., Flynn, R., Wong, K., & Kaefer, T. (2020). Quick, incidental word learning in educational media: all contexts are not equal. *Educational Technology Research and Development*, 68(6), 2913-2937. <u>https://doi.org/10.1007/s11423-</u>

<u>020-09815-z</u>

- Neuman, S. B., Samudra, P., & Wong, K. M. (2021). Two may be better than one: Promoting incidental word learning through multiple media. *Journal of Applied Developmental Psychology*, 73, 101252. https://doi.org/10.1016/j.appdev.2021.101252
- *Neuman, S. B., Wong, K. M., & Kaefer, T. (2017). Content not form predicts oral language comprehension: The influence of the medium on preschoolers' story understanding. *Reading and Writing*, 30(8), 1753-1771. <u>https://doi.org/10.1007/s11145-017-9750-4</u>
- Neumann, M. M. (2014). An examination of touch screen tablets and emergent literacy in Australian pre-school children. *Australian Journal of Education*, 58(2), 109-122. <u>https://doi.org/10.1177/0004944114523368</u>
- Neumann, M. M. (2020a). Teacher scaffolding of preschoolers' shared reading with a storybook app and a printed book. *Journal of Research in Childhood Education*, 34(3), 367-384. <u>https://doi.org/10.1080/02568543.2019.1705447</u>
- Neumann, M. M. (2020b). The Impact of Tablets and Apps on Language Development. *Childhood Education*, 96(6), 70-74. <u>https://doi.org/10.1080/00094056.2020.1846394</u>
- Neumann, M. M., & Merchant, G. (2022). "That's a Big Bad Wolf!": Learning through teacher-child talk during shared reading of a story book app. *Early Childhood Education Journal*, 50(3), 515-525. <u>https://doi.org/10.1007/s10643-021-01171-8</u>
- Neumann, M. M., & Neumann, D. L. (2014). Touch screen tablets and emergent literacy. *Early Childhood Education Journal*, 42, 231-239. <u>https://doi.org/10.1007/s10643-013-0608-3</u>

- Nevo, E., & Vaknin-Nusbaum, V. (2018). Enhancing language and print-concept skills by using interactive storybook reading in kindergarten. *Journal of Early Childhood Literacy*, 18(4), 545-569. <u>https://doi.org/10.1177/1468798417694482</u>
- Nicaise, M., & Barnes, D. (1996). The union of technology, constructivism, and teacher education. *Journal of Teacher Education*, 47(3), 205-212. <u>https://doi.org/10.1177/0022487196047003007</u>
- Ninio, A., & Bruner, J. (1978). The achievement and antecedents of labeling. *Journal of Child Language*, 5(1), 1–15. <u>https://doi.org/10.1017/S0305000900001896</u>
- Noble, C., Sala, G., Peter, M., Lingwood, J., Rowland, C., Gobet, F., & Pine, J. (2019). The impact of shared book reading on children's language skills: A metaanalysis. *Educational Research Review*, 28, 100290. <u>https://doi.org/10.1016/j.edurev.2019.100290</u>
- Noel, M. J. (2013). Does medium matter? Increasing preschoolers' vocabulary during shared storybook reading using electronic and print formats. (Publication No. 1542566) [Doctoral dissertation, Western Carolina University]. ProQuest Dissertations Publishing.
- Ofcom. (2020). Children and parents: media use and attitudes report 2019. https://www.ofcom.org.uk/research-and-data/media-literacyresearch/childrens/children-and-parents-media-use-and-attitudes-report-2019
- Ofcom. (2023). Children and Parents: Media Use and Attitudes. https://www.ofcom.org.uk/__data/assets/pdf_file/0027/255852/childrens-mediause-and-attitudes-report-2023.pdf
- *O'Toole, K. J. (2015). Emergent literacy skills in print and electronic contexts: The influence of book type, narration style, and attention. [Doctoral dissertation, Loyola University Chicago]. <u>https://ecommons.luc.edu/luc_diss/1652</u>
- *O'Toole, K. J., & Kannass, K. N. (2018). Emergent literacy in print and electronic contexts: The influence of book type, narration source, and attention. *Journal of Experimental Child Psychology*, 173, 100-115. https://doi.org/10.1016/j.jecp.2018.03.013
- Ozbay, I., & Ugurelli, Y. O. (2023). Changing Children's Literature in the Digital Age: Digital Books. *International Journal of Education and Literacy Studies*, 11(1), 68-85. <u>https://doi.org/10.7575/aiac.ijels.v.11n.1p.68</u>
- Paivio, A. (1986). *Mental representation: A dual coding approach*. Oxford University Press.

Paivio, A. (2008). Dual coding theory. Educating the Other America, 227-242.

- Paivio, A., Yuille, J. C., & Madigan, S. A. (1968). Concreteness, imagery, and meaningfulness values for 925 nouns. *Journal of Experimental Psychology*, 76(1p2), 1. <u>https://psycnet.apa.org/doi/10.1037/h0025327</u>
- Papadakis, S., Kalogiannakis, M., & Zaranis, N. (2018). Educational apps from the Android Google Play for Greek preschoolers: A systematic review. *Computers & Education*, 116, 139-160. <u>https://doi.org/10.1016/j.compedu.2017.09.007</u>
- Parette, H. P., Quesenberry, A. C., & Blum, C. (2010). Missing the boat with technology usage in early childhood settings: A 21st century view of developmentally appropriate practice. *Early Childhood Education Journal*, 37(5), 335-343. <u>https://doi.org/10.1007/s10643-009-0352-x</u>
- Paris, A. H., & Paris, S. G. (2003). Assessing narrative comprehension in young children. *Reading Research Quarterly*, 38(1), 36-76. <u>https://doi.org/10.1598/RRQ.38.1.3</u>
- Parish-Morris, J., Mahajan, N., Hirsh-Pasek, K., Golinkoff, R. M., & Collins, M. F. (2013). Once upon a time: parent–child dialogue and storybook reading in the electronic era. *Mind, Brain, and Education*, 7(3), 200-211. https://doi.org/10.1111/mbe.12028
- Payne, A. C., Whitehurst, G. J., & Angell, A. L. (1994). The role of home literacy environment in the development of language ability in preschool children from low-income families. *Early Childhood Research Quarterly*, 9(3–4), 427–440. <u>https://doi.org/10.1016/0885-2006(94)90018-3</u>
- *Pearman, C. J. (2008). Independent reading of CD-ROM storybooks: Measuring comprehension with oral retellings. *The Reading Teacher*, 61(8), 594-602. <u>https://doi.org/10.1598/RT.61.8.1</u>
- Pellerin, M. (2013). E-inclusion in Early French Immersion classrooms: using digital technologies to support inclusive practices that meet the needs of all learners. *Canadian Journal of Education*, 36(1), 44-70. <u>http://www.jstor.org/stable/canajeducrevucan.36.1.44</u>
- Penn, A. (1993). The Kissing Hand Ruth (E. Harper & N. M. Leak, Illus.). Child Welfare League of America.
- Penno, J. F., Wilkinson, I. A. G., & Moore, D. W. (2002). Vocabulary acquisition from teacher explanation and repeated listening to stories: Do they overcome the Matthew effect? *Journal of Educational Psychology*, 94(1), 23–33.

https://doi.org/10.1037/0022-0663.94.1.23

- Phadung, M., Suksakulchai, S., & Kaewprapan, W. (2016). Interactive whole language estory for early literacy development in ethnic minority children. *Education and Information Technologies*, 21(2), 249-263. https://doi.org/10.1007/s10639-014-9318-8
- Piaget, J. (1973). To understand is to invent: The future of education. Penguin Books.
- Pico, D. L., & Woods, C. (2023). Shared Book Reading for Spanish-Speaking Emergent Bilinguals: A Review of Experimental Studies. *Review of Educational Research*, 93(1), 103-138. <u>https://doi.org/10.3102/00346543221095112</u>
- Pollard-Durodola, S. D., Gonzalez, J. E., Saenz, L., Soares, D., Resendez, N., Kwok, O., ... & Zhu, L. (2016). The effects of content-related shared book reading on the language development of preschool dual language learners. *Early Childhood Research Quarterly*, *36*, 106-121.

https://doi.org.ezphost.dur.ac.uk/10.1016/j.ecresq.2015.12.004

- Puerling, B. (2012). *Teaching in the digital age: Smart tools for age 3 to grade 3*. Redleaf Press.
- Rao, P. S. (2019). The role of English as a global language. *Research Journal of English*, 4(1), 65-79. Available at https://www.rjoe.org.in/Files/vol4issue1/new/OK%20RJOESrinu%20sir(65-79).pdf
- Reese, E., & Cox, A. (1999). Quality of adult book reading affects children's emergent literacy. *Developmental Psychology*, 35(1), 20–28. <u>https://doi.org/10.1037/0012-1649.35.1.20</u>
- Reich, S. M., Yau, J. C., & Warschauer, M. (2016). Tablet-based ebooks for young children: What does the research say?. *Journal of Developmental & Behavioral Pediatrics*, 37(7), 585-591. <u>https://doi.org/10.1097/DBP.00000000000335</u>
- *Reich, S. M., Yau, J. C., Xu, Y., Muskat, T., Uvalle, J., & Cannata, D. (2019). Digital or print? a comparison of preschoolers' comprehension, vocabulary, and engagement from a print book and an e-book. *AERA Open*, 5(3). <u>https://doi.org/10.1177/2332858419878389</u>
- Reid, C. (2016). E-Books and print books can have different effects on literacy comprehension. [Masters Thesis, St. John Fisher College, UK]. Retrieved from <u>https://core.ac.uk/download/pdf/48622296.pdf</u>

Reznitskaya, A. (2012). Dialogic teaching: Rethinking language use during literature

discussions. *The reading teacher*, 65(7), 446-456. https://doi.org/10.1002/TRTR.01066

 *Richter, A., & Courage, M. L. (2017). Comparing electronic and paper storybooks for preschoolers: Attention, engagement, and recall. *Journal of Applied Developmental Psychology*, 48, 92-102. https://doi.org/10.1016/j.appdev.2017.01.002

Rideout, V. (2011). Zero to eight: Children's media use in America. Common Sense Media. <u>https://www.commonsensemedia.org/sites/default/files/research/report/zerotoeigh</u> tfinal2011.pdf

- Rideout, V., & Robb, M. B. (2020). The Common Sense census: Media use by kids age zero to eight. Common Sense Media. <u>https://www.commonsensemedia.org/sites/default/files/uploads/research/2020_ze</u> <u>ro_to_eight_census_final_web.pdf</u>
- Rose, E., Lehrl, S., Ebert, S., & Weinert, S. (2018). Long-term relations between children's language, the home literacy environment, and socioemotional development from ages 3 to 8. *Early Education and Development*, 29(3), 342-356. <u>https://doi.org/10.1080/10409289.2017.1409096</u>
- Roskos, K., Brueck, J., & Widman, S. (2009). Investigating analytic tools for e-book design in early literacy learning. *Journal of Interactive Online Learning*, 8(3). <u>http://www.ncolr.org/jiol/issues/viewarticle.cfm?volID=8&IssueID=27&ArticleI</u> <u>D=138</u>
- Rothman, J., Alonso, J. G., & Puig-Mayenco, E. (2019). *Third language acquisition and linguistic transfer* (Vol. 163). Cambridge University Press.
- *Rvachew, S., Rees, K., Carolan, E., & Nadig, A. (2017). Improving emergent literacy with school-based shared reading: Paper versus ebooks. *International Journal of Child-Computer Interaction*, 12, 24-29. <u>https://doi.org/10.1016/j.ijcci.2017.01.002</u>
- Sadoski, M., & Paivio, A. (2013). *Imagery and text: A dual coding theory of reading and writing*. Routledge.
- *Sapsaglam, Ö., Aydin, D., & Toksoy, N. (2020). Comparisons of Children's Level of Recall: Stories Told through E-Book and Picture Book. *Educational Research* and Reviews, 15(3), 123-128. <u>https://files.eric.ed.gov/fulltext/EJ1250471.pdf</u>
- *Sarı, B., Başal, H. A., Takacs, Z. K., & Bus, A. G. (2019). A randomized controlled

trial to test efficacy of digital enhancements of storybooks in support of narrative comprehension and word learning. *Journal of Experimental Child Psychology*, *179*, 212-226. <u>https://doi.org/10.1016/j.jecp.2018.11.006</u>

- Savva, M., Higgins, S., & Beckmann, N. (2022). Meta-analysis examining the effects of electronic storybooks on language and literacy outcomes for children in grades Pre-K to grade 2. *Journal of Computer Assisted Learning*, 38(2), 526-564. <u>https://doi-org.ezphost.dur.ac.uk/10.1111/jcal.12623</u>
- Scarborough, H. S., & Dobrich, W. (1994). On the efficacy of reading to preschoolers. *Developmental Review*, 14, 245-302. <u>https://doi.org/10.1006/drev.1994.1010</u>
- Scarino, A. (2014). Learning as reciprocal, interpretive meaning-making: A view from collaborative research into the professional learning of teachers of languages. *The Modern Language Journal*, 98(1), 386-401.
 https://doi-org.ezphost.dur.ac.uk/10.1111/j.1540-4781.2014.12068.x
- Schatz, E. K., & Baldwin, R. S. (1986). Context clues are unreliable predictors of word meanings. *Reading Research Quarterly*, 439-453. <u>https://doi.org/10.2307/747615</u>
- Schnotz, W., & Rasch, T. (2005). Enabling, facilitating, and inhibiting effects of animations in multimedia learning: Why reduction of cognitive load can have negative results on learning. *Educational Technology Research and Development*, 53(3), 47-58. <u>https://doi.org/10.1007/BF02504797</u>
- Schugar, H. R., Smith, C. A., & Schugar, J. T. (2013). Teaching with interactive picture e-books in grades K–6. *The Reading Teacher*, 66(8), 615-624. <u>https://doi.org/10.1002/trtr.1168</u>
- *Segal-Drori, O., Korat, O., Shamir, A., & Klein, P. S. (2010). Reading electronic and printed books with and without adult instruction: Effects on emergent reading. *Reading and Writing*, 23(8), 913-930. <u>https://doi.org/10.1007/s11145-</u> 009-9182-x
- Segal-Drori, O., & Shabat, A. B. (2021). Preschoolers' views on integration of digital technologies. *Journal of Childhood, Education & Society*, 2(1), 29-42. <u>https://doi.org/10.37291/2717638X.20212172</u>
- Segers, E., & Verhoeven, L. (2002). Multimedia support of early literacy learning. *Computers & Education*, 39(3), 207-221. <u>https://doi.org/10.1016/S0360-1315(02)00034-9</u>
- Sénéchal, M. (1997). The differential effect of storybook reading on preschoolers' acquisition of expressive and receptive vocabulary. *Journal of Child Language*,

24, 123-138. https://doi.org/10.1017/S0305000996003005

- Sénéchal, M., Thomas, E., & Monker, J. A. (1995). Individual differences in 4-year-old children's acquisition of vocabulary during storybook reading. *Journal of educational Psychology*, 87(2), 218. <u>https://psycnet.apa.org/doi/10.1037/0022-</u> 0663.87.2.218
- Seven, Y., Ferron, J., & Goldstein, H. (2020). Effects of Embedding Decontextualized Language Through Book-Sharing Delivered by Mothers and Fathers in Coparenting Environments. *Journal of Speech, Language, and Hearing Research*, 63(12), 4062-4081. https://doi.org/10.1044/2020_JSLHR-20-00206
- Sezer, B. B., Cetinkaya, F. C., Tosun, D. K., & Yildirim, K. (2021). A comparison of three read-aloud methods with children's picture books in the Turkish language context: Just reading, performance based reading, and interactional reading. *Studies in Educational Evaluation*, 68, 100974. https://doi.org/10.1016/j.stueduc.2020.100974
- Shadish, W. R., Cook, T. D., & Campbell, D. T. (2002). *Experimental and quasiexperimental designs for generalized causal inference*. Houghton Mifflin.
- Shahaeian, A., Wang, C., Tucker-Drob, E., Geiger, V., Bus, A. G., & Harrison, L. J. (2018). Early shared reading, socioeconomic status, and children's cognitive and school competencies: Six years of longitudinal evidence. *Scientific Studies of Reading*, 22(6), 485-502. https://doi.org/10.1080/10888438.2018.1482901
- Shamir, A. (2009). Processes and outcomes of joint activity with e-books for promoting kindergarteners' emergent literacy. *Educational Media International*, 46(1), 81-96. <u>https://doi.org/10.1080/09523980902781295</u>
- Shamir, A., & Korat, O. (2006). How to select CD-ROM storybooks for young children: The teacher's role. *The Reading Teacher*, 59(6), 532-543. <u>https://doi.org/10.1598/RT.59.6.3</u>
- Shamir, A., & Korat, O. (2007). Developing an educational e-book for fostering kindergarten children's emergent literacy. *Computers in the Schools*, 24(1-2), 125-143. <u>https://doi-org.ezphost.dur.ac.uk/10.1300/J025v24n01_09</u>
- Shamir, A., & Korat, O. (2015). Educational electronic books for supporting emergent literacy of kindergarteners at-risk for reading difficulties—What do we know so far?. *Computers in the Schools*, 32(2), 105-121. https://doi.org/10.1080/07380569.2015.1027868

- Shamir, A., Korat, O., & Barbi, N. (2008). The effects of CD-ROM storybook reading on low SES kindergarteners' emergent literacy as a function of learning context. *Computers & Education*, 51(1), 354-367. <u>https://doi.org/10.1016/j.compedu.2007.05.010</u>
- Shamir, A., Korat, O., & Fellah, R. (2012). Promoting vocabulary, phonological awareness and concept about print among children at risk for learning disability: can e-books help?. *Reading and Writing*, 25(1), 45-69. <u>https://doi.org/10.1007/s11145-010-9247-x</u>
- Shamir, A., Korat, O., & Shlafer, I. (2011). The effect of activity with e-book on vocabulary and story comprehension: A comparison between kindergarteners at risk of learning disabilities and typically developing kindergarteners. *European Journal of Special Needs Education*, 26(3), 311-322. <u>https://doi.org/10.1080/08856257.2011.593824</u>
- Siegel, L. (2016). Bilingualism and Dyslexia: The case of children learning English as an additional language. In *Multilingualism, Literacy and Dyslexia* (pp. 137-147). Routledge.
- Silverman, R. (2013). Investigating video as a means to promote vocabulary for at-risk children. *Contemporary Educational Psychology*, 38(3), 170-179. <u>https://doi.org/10.1016/j.cedpsych.2013.03.001</u>
- Silverman, R. D. (2007). Vocabulary development of English-language and English-only learners in kindergarten. *The Elementary School Journal*, 107(4), 365-383. <u>https://doi.org/10.1086/516669</u>
- Silverman, R., & Hines, S. (2009). The effects of multimedia-enhanced instruction on the vocabulary of English-language learners and non-English-language learners in pre- kindergarten through second grade. *Journal of Educational Psychology*, *101*(2), 305–314. <u>https://doi.org/10.1037/a0014217</u>
- Silvern, S. (1985). Parent Involvement and Reading Achievement: A Review of Research and Implications for Practice. *Childhood Education*, 62(1), 44–50. <u>https://doi.org/10.1080/00094056.1985.10520221</u>
- Smeets, D. J., & Bus, A. G. (2012). Interactive electronic storybooks for kindergartners to promote vocabulary growth. *Journal of Experimental Child Psychology*, *112*(1), 36-55. <u>https://doiorg.ezphost.dur.ac.uk/10.1016/j.jecp.2011.12.003</u>

*Smeets, D. J., & Bus, A. G. (2015). The interactive animated e-book as a word learning

device for kindergartners. *Applied Psycholinguistics*, *36*(4), 899-920. https://doi.org/10.1017/S0142716413000556

- Smeets, D. J., Van Dijken, M. J., & Bus, A. G. (2014). Using electronic storybooks to support word learning in children with severe language impairments. *Journal of Learning Disabilities*, 47(5), 435-449. https://doi.org/10.1177/0022219412467069
- Snow, C. E. (1983). Literacy and language: Relationships during the preschool years. *Harvard Educational Review*, 53, 165–189. <u>https://doi.org/10.17763/haer.53.2.t6177w39817w2861</u>
- Snow, C. E., Burns, M. S., & Griffin, P. (1998). Preventing reading difficulties in young children. National Academies Press.
- Sosa, A. V. (2016). Association of the type of toy used during play with the quantity and quality of parent-infant communication. JAMA Pediatrics, 170(2), 132-137. <u>https://doi.org/10.1001/jamapediatrics.2015.3753</u>
- Spencer, E. J., Goldstein, H., & Kaminski, R. (2012). Teaching vocabulary in storybooks: Embedding explicit vocabulary instruction for young children. *Young Exceptional Children*, 15(1), 18-32. <u>https://doi-org.ezphost.dur.ac.uk/10.1177/1096250611435367</u>
- Statistical service of Cyprus. (2021, October 1). Census of Population and Housing 2021, Preliminary Results by District, Municipality/Community. https://www.cystat.gov.cy/en/PressRelease?id=66208
- Sterne, J. A., Becker, B. J., & Egger, M. (2005). The funnel plot. In H.R. Rothstein, A. J. Sutton & M. Borenstein (Eds.), *Publication bias in meta-analysis: Prevention, assessment and adjustments*, (pp. 75-98). Wiley. https://doi.org/10.1002/0470870168
- Stone, S., & Twardosz, S. (2001). Children's books in child care classrooms: Quality, accessibility, and reasons for teachers' choices. *Journal of Research in Childhood Education*, 16(1), 53-69. https://doiorg.ezphost.dur.ac.uk/10.1080/02568540109594974
- Storch, S. A., & Whitehurst, G. J. (2002). Oral language and code-related precursors to reading: Evidence from a longitudinal structural model. *Developmental Psychology*, 38(6), 934–947. <u>https://doi.org/10.1037/0012-1649.38.6.934</u>
- Strand, S., Malmberg, L., & Hall, J. (2015). English as an Additional Language (EAL) and educational achievement in England: An analysis of the National Pupil

Database. Educational Endowment Foundation. https://doi.org/10871/23323

- Strasser, K., Larraín, A., & Lissi, M. R. (2013). Effects of storybook reading style on comprehension: The role of word elaboration and coherence questions. *Early Education & Development*, 24(5), 616-639. https://doi.org/10.1080/10409289.2012.715570
- Strouse, G. A., & Ganea, P. A. (2016). Are prompts provided by electronic books as effective for teaching preschoolers a biological concept as those provided by adults?. *Early Education and Development*, 27(8), 1190-1204. https://doi-org.ezphost.dur.ac.uk/10.1080/10409289.2016.1210457
- Strouse, G. A., & Ganea, P. A. (2017). Parent–toddler behavior and language differ when reading electronic and print picture books. *Frontiers in Psychology*, 8, 677. <u>https://doi.org/10.3389/fpsyg.2017.00677</u>
- Strouse, G. A., O'Doherty, K., & Troseth, G. L. (2013). Effective coviewing:
 Preschoolers' learning from video after a dialogic questioning intervention.
 Developmental Psychology, 49, 2368-2382. <u>https://doi.org/10.1037/a0032463</u>
- Strouse, G. A., Troseth, G. L., & Stuckelman, Z. D. (2023). Page and screen: Storybook features that promote parent-child talk during shared reading. *Journal of Applied Developmental Psychology*, 86, 101522.

https://doi.org/10.1016/j.appdev.2023.101522

- Sullivan, G. M. (2011). Getting off the "gold standard": Randomized controlled trials and education research. *Journal of Graduate Medical Education*, 3(3), 285–289. <u>https://doi.org/10.4300/JGME-D-11-00147.1</u>
- Sulzby, E., & Teale, W. (1996). Emergent literacy. In R. Barr, M. L. Kamil, P. B. Mosenthal, & P. D. Pearson (Eds.), *Handbook of Reading Research* (Vol. 2, pp. 727–757). Lawrence Erlbaum
- Sun, H., Loh, J., & Charles Roberts, A. (2019). Motion and sound in animated storybooks for preschoolers' visual attention and Mandarin language learning: An eye-tracking study with bilingual children. AERA Open, 5(2). <u>https://doi.org/10.1177/2332858419848431</u>
- Sun, H., Toh, W., & Steinkrauss, R. (2020). Instructional strategies and linguistic features of kindergarten teachers' shared book reading: The case of Singapore. *Applied Psycholinguistics*, 41(2), 427-456. <u>https://doi.org/10.1017/S0142716420000053</u>

Sun, H., & Yin, B. (2020). Multimedia input and bilingual children's language

learning. *Frontiers in Psychology*, *11*, 2023. https://doi.org/10.3389/fpsyg.2020.02023

- Sweller, J. (1988). Cognitive load during problem solving: Effects on learning. *Cognitive Science*, 12(2), 257-285. <u>https://doi.org/10.1016/0364-0213(88)90023-7</u>
- Sweller, J. (2016). Working memory, long-term memory, and instructional design. Journal of Applied Research in Memory and Cognition, 5(4), 360-367. <u>https://doi.org/10.1016/j.jarmac.2015.12.002</u>
- Sweller, J., Ayres, P., & Kalyuga, S. (2011). Measuring cognitive load. In J. Sweller, P. Ayres & S. Kalyuga, *Cognitive load theory* (pp. 71-85). Springer.
- Sweller, J., van Merriënboer, J. J., & Paas, F. (2019). Cognitive architecture and instructional design: 20 years later. *Educational Psychology Review*, 31(2), 261-292. <u>https://doi.org/10.1007/s10648-019-09465-5</u>
- *Takacs, Z. K., & Bus, A. G. (2016). Benefits of motion in animated storybooks for children's visual attention and story comprehension. An eye-tracking study. *Frontiers in Psychology*, 7, 1591.
 <u>https://doi.org/10.3389/fpsyg.2016.01591</u>
- Takacs, Z. K., Swart, E. K., & Bus, A. G. (2014). Can the computer replace the adult for storybook reading? A meta-analysis on the effects of multimedia stories as compared to sharing print stories with an adult. *Frontiers in Psychology*, 5, 1366. <u>https://doi.org/10.3389/fpsyg.2014.01366</u>
- Takacs, Z. K., Swart, E. K., & Bus, A. G. (2015). Benefits and pitfalls of multimedia and interactive features in technology-enhanced storybooks: A meta-analysis. *Review of Educational Research*, 85(4), 698-739.
 https://doi.org/10.3102/0034654314566989
- Takanishi, R., & Le Menestrel, S. (2017). Promoting the educational success of children and youth learning English: Promising futures. The National Academies Press. Retrieved October, 26, 2017.
- Teale, W. H. (1987). Emergent literacy: Reading and writing development in early childhood. In *National Reading Conference Yearbook*. National Reading Conference.
- Teale, W. H. (2003). Reading aloud to young children as a classroom instructional activity: Insights from research and practice. In A. van Kleeck, A. A. Stahl, & E. B. Bauer (Eds.), *On reading books to children: Parents and teachers* (pp. 114–139). Lawrence Erlbaum.

- Thomas, N., Colin, C., & Leybaert, J. (2020). Interactive reading to improve language and emergent literacy skills of preschool children from low socioeconomic and language-minority backgrounds. *Early Childhood Education Journal*, 48, 549-560. <u>https://doi.org/10.1007/s10643-020-01022-y</u>
- Tomasello, M., & Farrar, M. J. (1986). Joint Attention and Early Language. Child Development, 57(6), 1454–1463. <u>https://doi.org/10.2307/1130423</u>
- Towson, J. A., Fettig, A., Fleury, V. P., & Abarca, D. L. (2017). Dialogic reading in early childhood settings: A summary of the evidence base. *Topics in Early Childhood Special Education*, 37(3), 132-146. https://doi.org/10.1177/0271121417724875
- Troia, G. A. (1999). Phonological awareness intervention research: A critical review of the experimental methodology. *Reading Research Quarterly*, 34(1), 28-52. <u>https://doi.org/10.1598/RRQ.34.1.3</u>
- U. S. Census Bureau. (2021). Children who speak a language other than English at home in United States. <u>https://datacenter.aecf.org/data/tables/81-children-who-speak-a-language-other-than-english-at-home#detailed/1/any/false/2048,1729,37,871,870,573,869,36,868,867/any/396,39 7</u>
- Uchikoshi, Y. (2014). Development of vocabulary in Spanish-speaking and Cantonesespeaking English language learners. *Applied Psycholinguistics*, 35(1), 119–153. <u>https://doi.org/10.1017/S0142716412000264</u>
- Ulanoff, S. H., & Pucci, S. L. (1999). Learning words from books: The effects of readaloud on second language vocabulary acquisition. *Bilingual Research Journal*, 23, 409–422. <u>https://doi.org/10.1080/15235882.1999.10162743</u>
- Underwood, G., & Underwood, J. D. (1998). Children's interactions and learning outcomes with interactive talking books. *Computers & Education*, 30(1-2), 95-102. <u>https://doi.org/10.1016/S0360-1315(97)00084-5</u>
- Van Daal, V. H., & Sandvik, J. M. (2011). Reading Research And The Promotion Of Reading Outside The School: A Comprehensive Review of Empirical Research. *International Journal About Parents in Education*, 5(2), 110-122.
- Van Daal, V. H., Sandvik, J. M., & Adèr, H. J. (2019). A Meta-analysis of Multimedia Applications: How Effective Are Interventions with e-Books, Computer-Assisted Instruction and TV/Video on Literacy Learning?. In B. Hassinger-Das, *Reading in the Digital Age: Young Children's Experiences with E-books*. J. E. Kim (Ed.)

(pp. 259-296). Springer. https://doi.org/10.1007/978-3-030-20077-0_14

- Van de Pol, J., Volman, M., & Beishuizen, J. (2010). Scaffolding in teacher–student interaction: A decade of research. *Educational Psychology Review*, 22, 271-296. <u>https://doi.org/10.1007/s10648-010-9127-6</u>
- Van den Brook, P., Kendeou, P., Lousberg, S., & Visser, G. (2011). Preparing for reading comprehension: Fostering text comprehension skills in preschool and early elementary school children. *International Electronic Journal of Elementary Education*, 4(1), 259-268. Retrieved from https://www.iejee.com/index.php/IEJEE/article/view/223
- van Kleeck, A. (2008). Providing preschool foundations for later reading comprehension: The importance of and ideas for targeting inferencing in storybook-sharing interventions. *Psychology in the Schools*, 45, 627-643. <u>https://doi.org/10.1002/pits.20314</u>
- Verenikina, I. (2008). Scaffolding and learning: Its role in nurturing new learners. In P.
 Kell, W. Vialle, D. Konza & G. Vogyl (Eds.), *Learning and the learner: Exploring learning for new times* (pp. 161-180). University of Wollongong.
- Verhallen, M. J., Bus, A. G., & de Jong, M. T. (2006). The promise of multimedia stories for kindergarten children at risk. *Journal of Educational Psychology*, 98(2), 410. <u>https://doi.org/10.1037/0022-0663.98.2.410</u>
- Verhallen, M. J., & Bus, A. G. (2009). Video storybook reading as a remedy for vocabulary deficits. Outcomes and processes. *Journal for Educational Research Online*, 1(1), 172-196. <u>https://doi.org/10.25656/01:4561</u>
- Verhallen, M. J., & Bus, A. G. (2010). Low-income immigrant pupils learning vocabulary through digital picture storybooks. *Journal of Educational Psychology*, 102(1), 54. <u>https://doi.org/10.1037/a0017133</u>
- Verhallen, M. J., & Bus, A. G. (2011). Young second language learners' visual attention to illustrations in storybooks. *Journal of Early Childhood Literacy*, 11(4), 480-500. <u>https://doi-org.ezphost.dur.ac.uk/10.1177/1468798411416785</u>
- Vevea, J. L., & Woods, C. M. (2005). Publication bias in research synthesis: sensitivity analysis using a priori weight functions. *Psychological Methods*, 10(4), 428. <u>https://doi.org/10.1037/1082-989X.10.4.428</u>
- Vygotsky, L. S. (1978). Socio-cultural theory. Mind in Society, 6, 52-58.
- Vygotsky, L. S. (1980). *Mind in society: The development of higher psychological processes.* Harvard University Press.

Vygotsky, L. S. (1986). Thought and language. MIT Press.

- Walch, R. L. (2016). *Read alouds and their impact on students' literacy development* [Unpublished master's thesis]. The College at Brockport.
- Walsh, R. L., & Hodge, K. A. (2018). Are we asking the right questions? An analysis of research on the effect of teachers' questioning on children's language during shared book reading with young children. *Journal of Early Childhood Literacy*, 18(2), 264-294. <u>https://doi-</u> org.ezphost.dur.ac.uk/10.1177/1468798416659124
- Wasik, B. A., & Bond, M. A. (2001). Beyond the pages of a book: Interactive book reading and language development in preschool classrooms. *Journal of Educational Psychology*, 93(2), 243. <u>https://doi.org/10.1037/0022-0663.93.2.243</u>
- Wasik, B. A., Bond, M. A., & Hindman, A. (2006). The effects of a language and literacy intervention on Head Start children and teachers. *Journal of Educational Psychology*, 98(1), 63–74. <u>https://doi.org/10.1037/0022-0663.98.1.63</u>
- Wells, G. (2015). Dialogic learning: Talking our way into understanding. *Education as Social Construction*, 62.
- *Willoughby, D., Evans, M. A., & Nowak, S. (2015). Do ABC eBooks boost engagement and learning in preschoolers? An experimental study comparing eBooks with paper ABC and storybook controls. *Computers & Education*, 82, 107-117. <u>https://doi.org/10.1016/j.compedu.2014.11.008</u>
- Whitehurst, G. J. (1992). Dialogic reading: An effective way to read to preschoolers.
- Whitehurst, G. J., & Lonigan, C. J. (1998). Child development and emergent literacy. *Child Development*, 69(3), 848-872. <u>https://doi-org.ezphost.dur.ac.uk/10.1111/j.1467-8624.1998.tb06247.x</u>
- Whitehurst, G. J., & Lonigan, C. J. (2003). Emergent Literacy: Development from Prereaders to Readers. In S. B. Neuman & D. K. Dickinson (Eds.), *Handbook of Early Literacy Research* (Vol. 1) (pp. 281-304). Guilford Press.
- Whitehurst, G. J., Arnold, D. S., Epstein, J. N., Angell, A. L., Smith, M., & Fischel, J. E. (1998). A picture book reading intervention in day care and home for children from low-income families. *Developmental Psychology*, 30(5), 679. <u>https://doi.org/10.1037/0012-1649.30.5.679</u>
- Whitehurst, G. J., Epstein, J. N., Angel, A. L., Payne, A. C., Crone, D. A., & Fischel, J.
 E. (1994). Outcomes of an emergent literacy intervention in Head Start. *Journal* of Educational Psychology, 86, 542–555. <u>https://doi.org/10.1037/0022-</u>

0663.86.4.542

- Whitehurst, G. J., Falco, F. L., Lonigan, C. J., Fischel, J. E., DeBaryshe, B. D., Valdez-Menchaca, M. C., & Caulfield, M. (1988). Accelerating language development through picture book reading. *Developmental Psychology*, 24(4), 552. https://doi.org/10.1037/0012-1649.24.4.552
- Wong, K. M. (2020). The Promise of Educational Media for Dual Language Learners' L1 and L2 Vocabulary Development [Doctoral dissertation, New York University].
- Wong, K. M., & Neuman, S. B. (2019). The power of a story: Reading live and electronic storybooks to young children. In B. Hassinger-Das, *Reading in the Digital Age: Young Children's Experiences with E-books.* J. E. Kim (Ed.) (pp. 157-173). Springer. <u>https://doi.org/10.1007/978-3-030-20077-0_9</u>
- Wong, K. M., & Samudra, P. G. (2021). L2 vocabulary learning from educational media: Extending dual-coding theory to dual-language learners. *Computer Assisted Language Learning*, 34(8), 1182-1204. <u>https://doi.org/10.1080/09588221.2019.1666150</u>
- Worden, P. E., & Boettcher, W. (1990). Young children's acquisition of alphabet knowledge. *Journal of Reading Behavior*, 22(3), 277-295. <u>https://doiorg.ezphost.dur.ac.uk/10.1080/10862969009547711</u>
- Yang, D., Xia, C., Collins, P., & Warschauer, M. (2022). The role of bilingual discussion prompts in shared E-book reading. *Computers & Education*, 190, 104622. <u>https://doi-org.ezphost.dur.ac.uk/10.1016/j.compedu.2022.104622</u>
- Yelland, N., & Masters, J. (2007). Rethinking scaffolding in the information age. Computers & Education, 48(3), 362-382. https://doi.org.ezphost.dur.ac.uk/10.1016/j.compedu.2005.01.010
- Yin, C., & Hwang, G. J. (2018). Roles and strategies of learning analytics in the epublication era. *Knowledge Management & E-Learning: An International Journal*, 10(4), 455-468. <u>https://doi.org/10.34105/j.kmel.2018.10.028</u>
- Yıldırım, Ş., & Uzun, Ş. (2021). An overview of dialogic teaching and its impact on learning. *International Journal of Education, Technology and Science*, 1(2), 135-153.
- Yopp, H. K., & Yopp, R. H. (2000). Supporting phonemic awareness development in the classroom. *The Reading Teacher*, 54(2), 130-143. <u>http://www.jstor.org/stable/20204888</u>

- Yow, W. Q., & Priyashri, S. (2019). Computerized electronic features direct children's attention to print in single-and dual-language e-books. *AERA Open*, 5(3). <u>https://doi.org/10.1177/2332858419878126</u>
- Zeece, P. D. (2007). The Style of Reading and Reading in Style. *Early Childhood Education Journal*, 35(1), 41-45. <u>https://doi-org.ezphost.dur.ac.uk/10.1007/s10643-007-0164-9</u>
- Zevenbergen, A. A., & Whitehurst, G. J. (2003). Dialogic reading: A shared picture book reading intervention for preschoolers. On Reading Books to Children: Parents and Teachers, 177, 200.
- Zhang, R., Zou, D., Xie, H., Au, O. T. S., & Wang, F. L. (2020). A systematic review of research on e-book-based language learning. *Knowledge Management & E-Learning: An International Journal*, 12(1), 106-128. <u>https://doi.org/10.34105/j.kmel.2020.12.006</u>
- *Zhou, N., & Yadav, A. (2017). Effects of multimedia story reading and questioning on preschoolers' vocabulary learning, story comprehension and reading engagement. *Educational Technology Research and Development*, 65(6), 1523-1545. <u>https://doi.org/10.1007/s11423-017-9533-2</u>
- *Zipke, M. (2017). Preschoolers explore interactive storybook apps: The effect on word recognition and story comprehension. *Education and Information Technologies*, 22(4), 1695-1712. <u>https://doi.org/10.1007/s10639-016-9513-x</u>
- Zucker, T. A., Bowles, R., Pentimonti, J., & Tambyraja, S. (2021). Profiles of teacher & child talk during early childhood classroom shared book reading. *Early Childhood Research Quarterly*, 56, 27-40.
 https://doi-org.ezphost.dur.ac.uk/10.1016/j.ecresg.2021.02.006
- Zucker, T. A., Cabell, S. Q., Justice, L. M., Pentimonti, J. M., & Kaderavek, J. N. (2013). The role of frequent, interactive prekindergarten shared reading in the longitudinal development of language and literacy skills. *Developmental Psychology*, 49(8), 1425. <u>https://doi.org/10.1037/a0030347</u>
- Zucker, T. A., Cabell, S. Q., Oh, Y., & Wang, X. (2020). Asking questions is just the first step: Using upward and downward scaffolds. *The Reading Teacher*, 74(3), 275-283. <u>https://doi.org/10.1002/trtr.1943</u>
- Zucker, T. A., Justice, L. M., Piasta, S. B., & Kaderavek, J. N. (2010). Preschool teachers' literal and inferential questions and children's responses during wholeclass shared reading. *Early Childhood Research Quarterly*, 25(1), 65-83.

https://doi-org.ezphost.dur.ac.uk/10.1016/j.ecresq.2009.07.001

Zucker, T. A., Moody, A. K., & McKenna, M. C. (2009). The effects of electronic books on pre-kindergarten-to-grade 5 students' literacy and language outcomes: A research synthesis. *Journal of Educational Computing Research*, 40(1), 47-87. <u>https://doi.org/10.2190/EC.40.1.c</u>
Appendices







217

Appendix C: Consent Form



Shaped by the past, creating the future

Project title: The 'perfect' e-book vs video story vs print: A comparison of different types of media and reading styles of teachers for vocabulary and story comprehension for children learning English as an additional language. Researcher: Marilena Savva Department: School of Education

Dear Mrs. Mouzourou,

The aim of this study is to explore whether different types of storybook media (e-book, video stories and print books) have an effect on young children's vocabulary and story comprehension for children learning English as an additional language. This research project is supervised by Dr Nadin Beckmann (nadin.beckmann@durham.ac.uk) and Professor Steve Higgins (<u>s.e.higgins@durham.ac.uk</u>) from the School of Education at Durham University, UK.

My plan, with your permission, is to undertake the research as outlined in the attached proposal. Neither your school, nor any participants, will be identifiable in my account. My aim is to begin data collection in November and to end by the second week of December.

Please confirm your permission to undertake the outlined data collection activities at your school and also confirm that my plans are considered part of usual school/classroom practice at your school. Your permission and declaration is required by the School of Education Ethics Committee at Durham University as part of my ethics approval process. Your decision will be reported to the Ethics Committee to confirm to them that these conditions have been met (see the required declaration below).

If you consider that the activities proposed do not fall within usual school/classroom practice, then I will employ an active opt-in parental consent process in line with the School of Education's Ethics Code of Conduct.

Thank you for considering this request.

Yours sincerely,

Marilena Savva marilena.savva@durham.ac.uk

October 2020

Declaration (please delete as appropriate):

I give *permission/*de-not give permission for Marilena Savva to carry out the proposed exploration of usual school/classroom practice.

In the case where permission is not given, please provide reasons in the box below.

10

Date

Head Teacher Name (please print)

Head Teacher Signature

Leazes Road Durham University, Durham City, DH1 1TA Telephone +44 (0)191 334 2000 Fax +44 (0)191 334 8311 www.durham.ac.uk Durham University is the trading name of the University of Durham

October 2020

Appendix D: Discussion Points and Questions Asked During the Intervention

During the *Performance Teaching Style*, only the book's cover was presented while the discussion questions are located beneath the title page. After reading the story, the remainder of the discussion points were made while the experimenter revisited some of the illustrations.

While reading the story, the experimenter posed the questions displayed on each page as part of the *Interactive Teaching Style*.

Story 1: The Kissing Hand

The Kissing Hand

Title Page:



Look at the title. Show me the title. Show me the author. Say that this is the cover page. Can you read the title? The title is the kissing hand. What does that mean? What animals do you see on the front cover? What are the animals doing? Do you think they are boys or girls? Where are the animals? What do you think the story is about?

Page la:



What is the name of the animal? What animal is it? Is it a boy or a girl? Describe the animal.

What was Chester's problem? Did he hurt himself or he did not want to go to school? Why was Chester sad?

Page Ib:



Why Chester wanted to stay home?

How old do you think Chester is? How old did you go to school? With who did Chester wanted to stay home? His mother or father? What did he ask his mummy?





Who is Mrs Racoon? What did Mrs Racoon do to Chester? What is nuzzled him on the ear? Say nuzzle. What does gently mean? The story says that his mummy talked to Chester gently.

Page 2b:



What is strange and scary?

What things did mummy refer to when she said about strange and scary things?

Page 3b:



What time did Chester go to school?

Chester's mum said she knows a wonderful secret that will make Chester's night at school warm and cosy. What secret do you think it is? What is warm and cosy? What is wonderful?





Chester wiped away his tears. What did he do? Show me.



Why was the secret old? Where did Chester's mum learn the secret from? What was the secret?

Page 5:



What is the kissing hand?

How can you give your mummy a kissing hand? What is palm? Is it a dog or a hand? Spread open like a fan. Do it with me.

Page 6:



What happened with the kiss? Where did the kiss rush? What is rushed? Where did Chester feel the kiss?

(rushed from his hand up his arm and into his heart)

Page 7a+b:



What Chester needs to do if he feels lonely? (press his hand on his cheek and think mommy loves you) Show how to feel your mommy's love if you had a kissing hand.

How else do we feel our mommy's love? What does lonely mean? Is it jumping up high or being alone? Do you ever feel lonely? What are thoughts?

Page 8a:



Mummy wrapped Chester's fingers around the kiss. How can we do that? Can you wrap your hands? Can you say wrapped? What is wrapped? Why did she do that? (Not to lose the kiss)

Page 8b:



What will happen if Chester washes his hands?

Page 10a+b:



What time did Chester go to school? What colour is the sky? Where was Chester's school? On the tree or under the ground? Is our school like that? Where is the story taking place? Suddenly he turned to his mother and grinned. What does grinned means? Say grinned.

Page IIa+b:



What did Chester do to his mummy's hand? Where did he kiss? (the centre of her hand)

Page 13a:



What does scamper mean? (Chester scamper across the tree limb). Say scamper.

Page 14:



What did mummy do after Chester while Chester was going to school? Did she cry? Or did she put her hand on her cheek? Why did she do that? Show me our cheeks.

Page 15:



What other animals went to school? Who do you think will be Chester's friend? Why?

Page 16:

Would you do the kissing hand to your mommy?

At the end:

Did you like the story? What was your favourite part? Did you hear any music while listening to the story? What kind of music?

Story 2: Elmer and Rose

Elmer and Rose

Title Page:



Look at the title. Show me the title. Show me the author. Say that this is the cover page. Can you read the title? The title is *Elmer and Rose.* Who do you think is Elmer and who is Rose? What animals do you see on the front cover? What are the animals doing? Do you think they are boys or girls? Where are the animals? What do you think the story is about?

Page I:



What is the name if this elephant? Why do you think her name is Rose? What does it mean when the story says that Rose blushes? Show me your head and your toes. Page 2:



Who is Elmer and who is Wilbur? Are they brothers? What does it mean when someone is your cousin? What does the word patchwork mean?



Are the elephants unique? Are Elmer and Wilbur unique? Why? What does a herd of elephants mean? Do they look jolly?



Who is this?

Show me the bush. Make your body a small bush, now a big bush.



Who are they talking about? Who is she? So is Rose a boy or a girl?



What happened to Rose? Why do you think she was hiding? How did she get there?

Page 8:



Why do you think Rose was frightened? What does this word mean?

Page II:



What has happened to Rose here? Why is her name Rose?

Page 16:



Why did the grey elephant looked at Rose awkwardly? Let's role play this page who wants to be Elmer, Rose, Wilbur and the grey elephant?



Why did Rose think that the grey elephant was strange?

Page 20: Between races Wibur played tricks with his voice. He made his voice roar from behind a rock and shout from a treetop.

What kind of game where the elephants playing?

Page 21:



How did Rose react here? Why?

Page 22:



Why do you think Rose wanted to go alone to her herd?



Can you predict why Rose found the grey elephant strange?



Was the grey elephant unique? Are they all unique? Why? Are we unique?

At the end: Did you like the story? What was your favourite part?

| Appendix | E: | English | Language | 5-I | Point | Scale |
|----------|----|---------|----------|-----|-------|-------|
|----------|----|---------|----------|-----|-------|-------|

| | Low Ability | Upper low ability | Medium ability | Upper medium ability | High ability |
|-------------------|-------------------------------------------------------------------------|----------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------|
| Student's Code | Student uses gestures more often than words to communicate. | Student communicates using one word (sometimes with pronouns: I, he, she). | Student produces sentences using 2-3 words with pronouns and nouns or verb. Grammar/vocabulary might not be correct. | Student uses 4 words to communicate combined with pronouns, verbs and nouns. Able to describe situations. Grammar/vocabulary might not be correct. | Student speaks fluently using correct grammar and vocabulary. |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |
| | | | | | |

Appendix F: Receptive Vocabulary Tests

Story 1: The Kissing Hand

The Kissing Hand Receptive Vocabulary Test 3 and 4 years old

| Words | PRE-TEST | POST-TEST 1 | POST-TEST 2 |
|----------|----------|-------------|-------------|
| raccoon | | | |
| forest | | | |
| scary | | | |
| warm | | | |
| cheek | | | |
| palm | | | |
| fan | | | |
| lonely | | | |
| secret | | | |
| press | | | |
| day | | | |
| tears | | | |
| wipe | | | |
| rush | | | |
| fill | | | |
| mother | | | |
| in front | | | |
| give me | | | |
| old | | | |
| watch | | | |

| Words | PRE-TEST | POST-TEST 1 | POST-TEST 2 |
|----------|----------|-------------|-------------|
| | | | |
| raccoon | | | |
| forest | | | |
| swing | | | |
| scary | | | |
| warm | | | |
| cheek | | | |
| palm | | | |
| fan | | | |
| mask | | | |
| jump | | | |
| lonely | | | |
| secret | | | |
| press | | | |
| grin | | | |
| thoughts | | | |
| day | | | |
| wash | | | |
| stood | | | |
| tears | | | |
| wipe | | | |
| centre | | | |
| rush | | | |
| fill | | | |
| mother | | | |
| stick | | | |
| in front | | | |
| turn | | | |
| give me | | | |
| old | | | |
| watch | | | |

The Kissing Hand Receptive Vocabulary Test for 5-7 years old

Story 2: Elmer and Rose

| Elmer and Rose |
|----------------------------------------------------|
| Receptive Vocabulary Test 3 and 4 years old |

| Words | PRE-TEST | POST-TEST 1 | POST-TEST 2 |
|------------|----------|-------------|-------------|
| | | | |
| young | | | |
| toes | | | |
| tail | | | |
| patchwork | | | |
| herd | | | |
| jolly | | | |
| bush | | | |
| under | | | |
| different | | | |
| pinker | | | |
| peeped | | | |
| behind | | | |
| surprise | | | |
| lake | | | |
| between | | | |
| treetop | | | |
| excitement | | | |
| trunk | | | |
| stared | | | |
| blushes | | | |

| Elmer and Rose |
|------------------------------------------------|
| Receptive Vocabulary Test 5-7 years old |

| Words | PRE-TEST | POST-TEST 1 | POST-TEST 2 |
|------------|----------|-------------|-------------|
| | | | |
| young | | | |
| toes | | | |
| tail | | | |
| patchwork | | | |
| herd | | | |
| jolly | | | |
| bush | | | |
| under | | | |
| different | | | |
| pinker | | | |
| frightened | | | |
| peeped | | | |
| behind | | | |
| surprise | | | |
| lake | | | |
| hid | | | |
| between | | | |
| race | | | |
| treetop | | | |
| excitement | | | |
| held | | | |
| trunk | | | |
| grandpa | | | |
| elephant | | | |
| faster | | | |
| staring | | | |
| chuckled | | | |
| blushes | | | |
| journey | | | |
| tricks | | | |

Appendix G: Expressive Vocabulary Tests

| | PRE TES | Т | POST TE | ST 1 | POST TEST | 2 |
|-----------|-------------|-------|-------------|-------|-------------|-------|
| Words | Description | Score | Description | Score | Description | Score |
| wonderful | | | | | | |
| gentle | | | | | | |
| worry | | | | | | |
| night | | | | | | |
| large | | | | | | |
| middle | | | | | | |
| suddenly | | | | | | |
| heart | | | | | | |

The Kissing Hand Expressive vocabulary test 3 and 4 years old

| The Kissing Hand |
|------------------------------------------|
| Expressive vocabulary test 5-7 years old |

| | PRE TES | ST | POST TES | 5T 1 | POST TEST 2 | |
|-----------|-------------|-------|-------------|-------|-------------|-------|
| Words | Description | Score | Description | Score | Description | Score |
| end | | | | | | |
| nuzzle | | | | | | |
| heart | | | | | | |
| wonderful | | | | | | |
| strange | | | | | | |
| gentle | | | | | | |
| cosy | | | | | | |
| worry | | | | | | |

| special | | | |
|----------|--|--|--|
| tiny | | | |
| suddenly | | | |
| large | | | |
| wrap | | | |
| careful | | | |
| middle | | | |

Elmer and Rose Expressive Vocabulary Test 3 and 4 years old

| | PRE TEST | | POST TEST 1 | | POST TEST 2 | |
|---------|-------------|-------|-------------|-------|-------------|-------|
| | | 1 | | 1 | | |
| Words | Description | Score | Description | Score | Description | Score |
| | | | | | | |
| | | | | | | |
| wander | | | | | | |
| | | | | | | |
| | | | | | | |
| chuckle | | | | | | |
| | | | | | | |
| | | | | | | |
| | | | | | | |
| voice | | | | | | |
| | | | | | | |
| | | | | | | |
| cousin | | | | | | |
| | | | | | | |
| | | | | | | |
| 4l- | | | | | | |
| trunk | | | | | | |
| | | | | | | |
| | | | | | | |
| whisper | | | | | | |
| | | | | | | |
| | | | | | | |
| nretty | | | | | | |
| pretty | | | | | | |
| | | | | | | |
| | | | | | | |
| unique | | | | | | |
| | | | | | | |

| Elmer and Rose |
|------------------------------------------|
| Expressive vocabulary test 5-7 years old |

| | PRE TEST | | POST TEST 1 | | POST TEST 2 | |
|----------|-------------|-------|-------------|-------|-------------|-------|
| Words | Description | Score | Description | Score | Description | Score |
| arrive | | | | | | |
| chuckle | | | | | | |
| squealed | | | | | | |
| voice | | | | | | |
| tricks | | | | | | |
| journey | | | | | | |
| awkward | | | | | | |
| cousin | | | | | | |

| treetop | | | |
|---------|--|--|--|
| trunk | | | |
| whisper | | | |
| pretty | | | |
| wander | | | |
| nearby | | | |
| unique | | | |

Appendix H: Implicit Story Comprehension Tests

Story 1: The Kissing Hand

IMPLICIT Story comprehension questions

Experimental Group:

Question 1: How is Chester feeling here? Why?

Answer:

Question 2: Why he did not want to go to school?

Answer:



Question 2: How does the kissing hand work?

Answer:



Question 3: How did Chester feel here? Why did he feel like that?



Question 4a: What is Chester doing here?

Answer:

Question 4b: Why is he doing that?

Answer:



What did Mrs Raccoon do after Chester left for school?

Answer:

Question 5b:

What special words filled her heart? How do you think she felt?

Answer:

Question 6: How do you think Chester is feeling on his first day of school?

Why do you think that?







The Kissing Hand - IMPLICIT Story comprehension questions FOR PRE-KINDERGARDEN ONLY (3-4 years old)

Experimental Group:

Question 1: How is Chester feeling here? Why?

Answer:



Question 2: Why he did not want to go to school?

Answer: _____

Question 3: How did Chester feel here? Why did he feel like that?



Answer:

Question 5b: What special words filled his mum's heart? How do you think she felt?

Answer:

Question 6: How do you think Chester is feeling on his first day of school? Why do you think that?



Story 2: Elmer and Rose

ELMER AND ROSE - IMPLICIT Story comprehension questions

Experimental Group: _____

Question 1: Why was Rose hiding behind the tree?

Answer: _____

Question 2: Why is her name Rose?

Answer:



Answer:

Question 3b: Why does this happens to Rose?

Answer:

Question 4: What did Rose do when she saw the grey elephant?



Question 5: Why did Rose find the grey elephant strange?

Answer:



Question 6: What do you think Elmer and Wilbur think of Rose?

Answer:



Question 7: What colour do you think are the elephants in Wilbur's herd?

Answer:

Question 8: How is Rose different from the other elephants?

ELMER AND ROSE - IMPLICIT Story comprehension questions FOR PRE-KINDERGARDEN ONLY (3-4 years old)

Experimental Group: _

Question 1: Why was Rose hiding behind the tree?

Answer:



Question 3a: What happens to Rose's face?

Answer:

Question 3b: Why does this happens to Rose?

Answer: _____

Question 4: What did Rose do when she saw the grey elephant?

Answer:

Question 5: Why did Rose find the grey elephant strange?

Answer:



Question 5: What was your favourite part of the story?

Appendix I: Explicit Story Comprehension Tests

Story 1: The Kissing Hand

| EXPLIC Recall with prompts story co | I <u>T</u> mprehension measure | | | | |
|------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------|--|--|--|--|
| Experimental Group: | | | | | |
| 1. What is the title of the story? | | | | | |
| Free recall answer: | | | | | |
| 2. What animal was the main character? Where was the main character going? | | | | | |
| Free recall answer: 1) | 2) | | | | |
| 2A. The main character was an animal. Was he a. A monkey? b. A raccoon? c. A mouse? | | | | | |
| 2B. Where was the main character going? a. At the park b. At school c. To play with his friends | | | | | |
| 2. I am going to show you a picture from the story. Tell me the names of the animals in this picture. Who is the second animal? | | | | | |
| Free recall answers: 1) | 2) | | | | |
| 2A. Is it? a. Chester b. Cheetah c. Christmas | | | | | |
| 2B. Name of second animal? a. Mrs Chester b. Mrs Mommy c. Mrs Raccoon | | | | | |
| 2C. The other animal is? a. Chester's dad b. Chester's mum c. Chester's teacher | | | | | |
| 3. Where was Chester's school? What time of the day did he go to school? | | | | | |
| Free recall answer: : 1) | 2) | | | | |
| 3A: Chester's school was a. At the beach. b. On a tree. c. In a cave. | | | | | |
| 3B: What time of the day did Chester go to school? a. Morning b. Afternoon c. Night | | | | | |

4. What was Mrs Raccoon's secret?

Free recall answer:

4A: Mrs Raccoon secret was...

- a. The kissing hand
- b. The kissing toe
- c. The kissing finger

5. What is the Kissing Hand?

Free recall answer:

5A: What is the Kissing Hand?

- a. Mrs Raccoon touched Chester's hand.
- b. Mrs Raccoon kissed Chester's hand.
- c. Mrs Raccoon kissed Chester's cheek.

6. What did Mrs Raccoon say to Chester to do with the kissing hand?

Free recall answer:

6A: She said...

- a. To press his fingers on his cheek.
- b. To press his hand on his heart ..
- c. To press his hand on his cheek.

7. What other animals went to school with Chester?

- -

Free recall answer:

7A:

- a. Possums and foxes.
- b. Ladybugs and butterflies.
- c. Elephants and alligators.

- -

7. I'm going to give you 4 pictures from the story. Put them in order for me. What happened first, second, third, and last?

Proceed by showing the cards and having them put them in order: put what happened first here, what happened next, and what happened last.

- -
Story 2: Elmer and Rose

| ELMER AND ROSE - EXPLICIT Recall with prompts story comprehension measure |
|-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Experimental Group: |
| 1. What is the title of the story? |
| Free recall answer: |
| 2. What's the name of the black and white elephant? |
| Free recall answer: |
| 2A. The name of the black and white elephant is a. Whale b. Wilbur c. William |
| 3. What colour was the old elephant and what is his name? |
| Free recall answers: A) B) |
| 3A. The colour of the old elephant is a. Yellow b. Pink c. Orange |
| 3B. The name of the old elephant is a. Grandpa Eldo b. Grandpa Elmer c. Grandma Elli 4. Where do the elephants in the story live? |
| Free recall answer: |
| 4A: They live a. At the beach. b. In the jungle. c. In a forest. |
| 5. Who was hiding behind a tree? What colour is that elephant? |
| Free recall answers: A) B) |
| 5A. Is was a. Rose b. Pink c. Elmer |
| 5B. Colour of the elephant? a. Green b. Purple c. Pink |

6. What is Rose's problem?

Free recall answer:

6A: Her problem is that...

- a. She got lost from her family/herd.
- b. She lost her ice-cream.
- c. She is afraid of the lion.

7. Who did they see at the lake?

Free recall answer: _____

7A: They saw.. a. A yellow crocodile. b. A grey elephant. c. Nobody.

8. What colour were the elephants in Rose's herd/family?

_ ____

Free recall answer:

8A: They were all...

a. Different colours. b. Pink. c. Grey.

9. I'm going to give you 4 pictures from the story. Put them in order for me. What happened first, second, third, and last?

Proceed by showing the cards and having them put them in order: put what happened first here, what happened next, and what happened last.

_ _

Appendix J: Transfer Task Text Analysis

Given the significance of analysing the Lexile level of a text, I unfortunately found that a considerable number of Lexile level software programmes are not available for free. Consequently, I identified four pieces of free software that I utilised to analyse the text. Each software assessed distinct components of the text. However, a downside of these software is that they analyse text starting from ages 5-6 years old. I could not find a text analyser for a younger audience. Additionally, their emphasis lies on children engaging in the act of reading the text themselves, rather than comprehending the material when it is read aloud by an adult. As a preschool teacher with 12 years of experience, I possess extensive expertise in reading stories to children and collaboratively developing stories with them. Thus, the language included in this specific story comprises terms that children are acquainted with. As a result, the vocabulary employed in this specific narrative comprises terms that that children are acquainted with. Moreover, the particular story that I authored depicted illustrations that my pupils had created two years prior to the experimental study. These were children aged 5 to 6 years old, whose primary language was not English. They created the illustrations that accompanied the written material. I read the story to them in English and they created the illustrations based on their imagination. Hence, this was an additional factor that led me to conclude that the story is suitable for use during the transfer task. The following table depicts the results obtained from each software. From the results obtained, it seems that the text is a simple text and it is easily understood for ages 5-6 years old.

| Readability statistics | s (Readability Stats in Word) | |
|----------------------------|-----------------------------------------|-------------------------------------------------|
| Counts | Words | 88 |
| | Characters | 274 |
| | Sentences | 13 |
| Averages | Sentences per paragraph | 1.0 |
| | Words per sentence | 6.7 |
| | Characters per word | 2.9 |
| Readability | Flesch reading ease | 100.0 |
| | Flesch-Kincaid Grade level | 0.0 |
| | Passive sentences | 0.0% |
| Readability Formula | as: | |
| https://readabilityform | nulas.com/readability-scoring-system | .php#formulaResults |
| Reading Difficulty | Extremely Easy | |
| Lexical Text | Text Score: 54.5% | |
| Density Analysis | Density Range: 50%-59% | |
| | Scale: Above average density | |
| | Style: Texts in the 50%-59% lexica | al density range balance between accessibility |
| | and detail. They cater to a wider au | idience, providing substantial content without |
| | overwhelming readers with specializ | zed language. |
| Lexical Word | Text Score: 48.9% | |
| Diversity Analysis | Diversity Range: 40%-49% | |
| | Scale: Average diversity | |
| | Word Diversity: This writing style sl | howcases a varied vocabulary. Many texts may |
| | fall into this category, striking a | balance between introducing new terms and |
| | reinforcing existing ones. This could | d indicate the author is using a broad range of |
| | terms and expressions, possibly avo | iding repetitiveness. |
| Tests Document Rea | | |
| https://www.online-u | tility.org/english/readability test and | <u>improve.jsp</u> |
| Lexical Density | 54.68 | |
| Gunning Fog index | 2.71 | |
| Flesch Reading Ease | e 107.67 | |
| Coleman Liau index | -2.90 | |
| ARI (Automated Re | adability Index) -4.18 | |
| Readability test: http | os://www.webfx.com/tools/read-able/ | |
| Overall result | | Your direct input has an average reading ease |
| | | of about 107.7 of 100. It should be easily |
| | | understood by 5- to 6-year-olds. |
| Flesch Kincaid read | ling ease: the score is based on a | 107.7 |
| ranking scale of 0-1 | oo, and the higher your score the | |
| Cupping Eag goon | actimates the years of formal | 27 |
| aducation moded t | a comprehend text on the first | 2.1 |
| roading | o comprehend text on the first | |
| Smag index estimate | as the years of education a nerson | 1.8 |
| needs to comprehend | d writing | 1.0 |
| Coleman Liau index | x indicates the U.S. school level a | 11 |
| person needs to be to | understand the text. | |
| | | |

 Table J1 Readability statistics of story used in Transfer Task

Transfer Task AGES 3-4 <u>Story: Fin the Fox and Oli the Ox</u>

1. What was the title of the story?

2. What kind of animals are in the story?

3. What happened to Oli?

4. How was Oli feeling?

5. What objects did Fin get?

6. What happened at the end?

7. What do you think Fin should do next time Oli is stuck in the mud?

Transfer Task AGES 5-7 <u>Fin the Fox and Oli the Ox</u>

1. What was the title of the story?

2. What's the story about?

3. What is Oli's problem?

4. How was Oli feeling?

5. What did Fin do to help Oli?

6. What happened at the end?

7. What do you think Fin should do next time Oli is stuck in the mud?

Appendix K: Intraclass Correlation Coefficient (ICC)

Calculating the Intraclass Correlation Coefficient (ICC) is a crucial step when dealing with nested data, like students within classes. ICC helps quantify the proportion of variance attributable to the clustering structure. If the ICC is high, it indicates that a significant portion of the variability in the outcome can be attributed to the group level (class in this case), which would justify using multilevel models or accounting for the clustering effect in the analysis.

In this study, sixty typically developing children (28 girls and 32 boys) aged 3–7 years (M = 5.26 years, SD = 1.16) were enlisted from four classrooms from one private English school. The children attended the following grades: Pre-Kindergarten (ages 3–4; n=7; m=4, f=3), Kindergarten (ages 4-5.4; n=19; m=8, f=11), Pre-Primary (5.5–7; n=18; m=12, f=6), and First Grade (6-7.5 years old; n=16; m=8, f=8). Each class had one teacher. The participants were randomly allocated to conditions, in a stratified way taking into account a balanced representation of age, gender, language proficiency English level based on the Preschool Language Scale - Fifth Edition (PLS-5) standardised test, and their teachers' evaluation on a 5-point language scale level. The following table presents the means and standard deviations of basic characteristics for each condition at baseline, as mentioned in the thesis.

| Teaching style | | Intera | ctive Style | Perform | ance Style | Total |
|---------------------------|----|---------|-------------|---------|------------|---------|
| Book medium | | E-book | Print book | E-book | Print book | |
| N | | 15 | 15 | 15 | 15 | 60 |
| Age | М | 5.27 | 5.80 | 5.00 | 5.27 | 5.27 |
| | SD | (1.38) | (0.94) | (0.92) | (0.96) | (1.16) |
| Boys/Girls | | 7/8 | 8/7 | 9/6 | 8/7 | 32/28 |
| PLS-5 UK | М | 67.47 | 67.33 | 66.27 | 67.13 | 67.05 |
| | SD | (13.98) | (17.02) | (13.59) | (13.23) | (14.17) |
| English | М | 3.67 | 3.73 | 3.73 | 3.47 | 3.65 |
| Language 5-point scale | SD | (1.23) | (1.33) | (1.28) | (1.24) | (1.24) |

Table K1 Means (and standard deviations) of Basic Characteristics per Condition

In order to understand the extent of the clustering effect, I have run a Multilevel model in SPSS. I computed a multi-level null model using the Linear Mixed Models procedure. This

allowed me to account for the hierarchical structure of the data, where observations are nested within higher-level units (i.e., 60 children within 4 classes across different age groups from pre-kindergarten (3-4 years old) to first grade (6-7.5 years old) from one school).

ICC is:

The formula for

$$ICC = rac{\sigma_b^2}{\sigma^2 + \sigma^2}$$

Where:

- σ_b^2 is the between-group variance.
- σ_w^2 is the within-group variance.

Table K2 ICC Results

| DV | ICC |
|-----------------------------------|-------------------------------------|
| Standardised Test | 0.05 |
| Receptive Vocabulary Test | 0.01 |
| Expressive Vocabulary Test | 0.00 |
| Implicit Story Comprehension Test | 0.23 (includes some missing values) |
| Explicit Story Comprehension Test | 0.30 (includes some missing values) |

The Intraclass Correlation Coefficient (ICC) values I have computed for the different dependent variables (DVs) within the clustered data (children/participants nested within classes) provide insights into the degree of similarity or agreement among measurements within the same class for each DV. The ICC values provide information about the extent to which class membership contributes to the variability in each DV. Higher ICC values indicate that class membership plays a more significant role in explaining differences in test scores among children, whereas lower ICC values suggest that individual differences within classes are more influential. The IES What Works Clearinghouse (2014) estimates an ICC 0.20 for achievement outcomes. It important to note that "class" is confounded with "age", therefore, this factor needs to be considered when interpreting the ICCs (which reflects the nesting of 60 pupils within 4 classes of one school, i.e. the classes differ by age group). Nevertheless, except for one dependent variable, the ICC values derived from this study appear to be close to 0.20, which aligned with the Clearinghouse's (2014) estimate

of an ICC of 0.20 for educational outcomes.

Appendix L: Analysis of Covariance (ANCOVA) Results

Statistical Analyses

The IBM SPSS Statistics Version 29 software (IBM Corp., Armonk, NY, USA) was used for performing the statistical analyses using the results from the measures reported above. The analysis was performed on the following dependent variables on condition:

- Vocabulary:
 - Receptive Vocabulary
 - Expressive Vocabulary
- Story Comprehension:
 - Implicit Story Comprehension
 - Explicit Story Comprehension
 - Both tests combined to report overall comprehension of the story
- Transfer Task
- PLS-5 UK

To test the effects of the intervention in order to answer the research questions of the study, two contrasts were created: (a) the two conditions with print book format versus the two conditions with e-book format (comparison of book mediums) (RQ1); and (b) the two conditions with performance teaching style versus the two conditions with interactive teaching style (comparison of teaching styles) (RQ2). Therefore, this entails a 2 (teaching styles) x 2 (book mediums) x 2 (pre, post) between-within subjects factorial design.

The majority of the data was analysed with analysis of covariance (ANCOVA) for the dependent variables' receptive vocabulary, expressive vocabulary and the standardised language test PLS-5 UK.

Here's a breakdown of the components:

- 1. Independent Variables (IVs):
 - Book Medium: This has two levels (print or e-book).
- Teaching Style: This also has two levels (interactive style or performance style).

- 2. Dependent Measures (DVs):
- Receptive Vocabulary
- Expressive Vocabulary
- 3. Covariate:

• Preexisting group differences: This is a variable that you want to control for, likely because there were differences between groups before the experiment or treatment occurred. Controlling for this variable helps to isolate the effects of the independent variables on the dependent measures.

ANCOVA is particularly useful when there are preexisting differences between groups that might affect the dependent variable. By using the preexisting group differences as a covariate, ANCOVA allows you to statistically remove the effects of these differences from the dependent variable, thus providing a clearer understanding of the effects of the independent variables.

In order to answer the first research question: Which type of book medium (i.e., e-book versus print book) produces better results in the development of vocabulary and story comprehension for children learning English as an additional language (EAL)?, the e-book and print book conditions were compared for each dependent variable. The data was analysed with ANCOVA for the dependent variables receptive, expressive vocabulary and the standardised language test PLS-5 UK while controlling for the effects of a covariate, which is likely the preexisting group differences, and a one-way ANOVA for story comprehension as results were obtained at pre-test only.

Similarly, to answer the second research question: *Which teaching style (i.e., interactive versus performance) better facilitates young children's (learning English as an additional language) vocabulary and story comprehension development,* the interactive and performance conditions were compared for each dependent variable. The data was analysed with one-way ANOVA for story comprehension, and ANCOVA was used for the dependent variables receptive, expressive vocabulary and the standardised language test PLS-5 UK while controlling for the effects of a covariate, which is likely the preexisting group differences.

The results section begins with a check for baseline differences. The first research question

is then analysed, beginning with the receptive and expressive vocabulary results, story comprehension, transfer task, and concluding with the standardised test results. Then the second research question is analysed.

Results

Research Question 1 - Comparison of Book Mediums: E-book versus Print Format

Receptive and Expressive Vocabulary - E-book versus Print Format.

In this ANCOVA analysis (Table L1), the impact of the e-book condition compared to the control condition, which is the print book, on receptive vocabulary as the DV was examined. The mean score for the e-book condition was 0.60 (SD=0.19), with an adjusted mean of 0.64. In contrast, the mean score for the control condition was 0.70 (SD=0.15), with an adjusted mean of 0.66. The ANCOVA results indicated a non-significant main effect of the condition on receptive vocabulary, as evidenced by a nonsignificant F-value of 0.743 (F(1, 57) = 0.743, p=0.39, $\eta_p^2=0.01$). Additionally, the partial eta squared value of 0.01 suggests that only 1% of the variance in receptive vocabulary scores can be attributed to the condition variable. Moreover, Cohen's d for the comparison between the e-book condition and the control condition. Despite the lack of statistical significance, these findings provide valuable insights into the comparative performance between the e-book and print book, indicating a minimal difference in receptive vocabulary outcomes between the two conditions.

Receptive vocabulary 95% CI F η_p^2 Adjusted **Pre-test Post-test** р Condition Ν Mean Mean Mean (SD)(SD)E-book 30 0.47 0.60 0.64 [0.60, 0.67]0.743 0.392 0.013 (0.19)(0.19)**Print book** 30 [0.62, 0.69]0.56 0.70 0.66 (control) (0.15)(0.15)

Table L1 Analysis of Covariance (ANCOVA) Results for Receptive Vocabulary Based onBook Medium

In the following ANCOVA analysis (Table L2 below), the primary objective was to investigate the influence of the e-book condition compared to the control condition on expressive vocabulary, serving as the DV. The mean score for the e-book condition was 0.83 (SD=0.66), accompanied by an adjusted mean of 0.95 and a corresponding 95% confidence interval ranging from 0.86 to 1.04. Conversely, the mean score for the control condition was 0.96 (SD=0.72), with an adjusted mean of 0.84. The ANCOVA yielded a non-significant result with F(1, 57) = 3.123, p=0.08, $\eta_p^2 = 0.05$, suggesting no statistically significant difference between the e-book and control conditions concerning expressive vocabulary outcomes. Nonetheless, the effect size, as indicated by the partial eta squared value of 0.05, demonstrates a modest proportion of the variance in expressive vocabulary scores attributable to the condition variable. Cohen's d for the comparison between the ebook condition and the control condition is approximately -0.19. This indicates a small effect size, with the negative sign indicating that the e-book condition has a slightly lower mean than the control condition. Despite the lack of statistical significance, these findings contribute to the understanding of the nuanced effects of e-book versus traditional instructional methods on expressive vocabulary acquisition.

| Expressive | vocal | bulary | | | | | | |
|-------------------------|-------|--------------------------|---------------------------|------------------|--------------|-------|-------|------------|
| Condition | N | Pre-test Mean (SD) | Post-test Mean (SD) | Adjusted Mean | 95% CI | F | р | η_p^2 |
| E-book | 30 | 0.61 (0.58) | 0.83 (0.66) | 0.95 | [0.86, 1.04] | 3.123 | 0.083 | 0.052 |
| Print book (control) | 30 | 0.83 (0.64) | 0.96 (0.72) | 0.84 | [0.75, 0.93] | - | | |

Table L2 Analysis of Covariance (ANCOVA) Results for Expressive Vocabulary Based

 on Book Medium

Implicit and Explicit Story Comprehension Results - E-book versus Print Format. (The implicit and explicit story comprehension results are the same as the results presented in the main text in Chapter 5. I have duplicated these results here to provide the reader with a comprehensive view of the results).

One-way ANOVA was performed to analyse the effect of book medium – e-book versus print book – on implicit, explicit and overall story comprehension of the story at posttest 1. These results were obtained after the intervention and three one-way ANOVAs were

conducted in order to obtain the results presented in Table L3. The results revealed that for the *Implicit* story comprehension test there was no statistically significant main effect for medium (e-book versus print book conditions, F(1, 56)=3.22, p=0.07, $\eta_p^2=0.05$. The effect size for e-books versus print books for the Implicit story comprehension test was d=-0.48 favouring the print book condition.

The results analysed for *Explicit* story comprehension and both tests combined revealed that there was a statistically significant effect (p<0.05) with the print book condition outperforming the e-book condition on both measures (explicit story comprehension: F(1, 56)=4.36, p=0.04, $\eta_p^2=0.07$; Both tests combined: F(1, 56)=4.32, p=0.04, $\eta_p^2=0.07$). The effect size for e-books versus print books for the Explicit story comprehension test was d=-0.56 favouring the print book condition.

Table L3 One-way ANOVA Results for Implicit and Explicit Story Comprehension (posttest 1)

| | | E-boo | k conditio | n | | Print bo | ok condi | tion | | | | |
|---------------------------------|----|----------------|---------------|-----------------|----|----------------|---------------|-----------------|------|------|------------|-----------|
| Dependent Variables | Ν | Mean (SD) | Std. Error | 95% CI | N | Mean (SD) | Std. Error | 95% CI | F | р | η_p^2 | Cohen's d |
| Implicit Story Comprehension | 30 | 0.85 (0.44) | 0.08 | [0.69, 1.01] | 28 | 1.06 (0.43) | 0.08 | [0.89, 1.22] | 3.22 | 0.07 | 0.05 | -0.48 |
| Explicit Story Comprehension | 30 | 0.78 (0.41) | 0.07 | [0.63, 0.92] | 28 | 1.00 (0.37) | 0.07 | [0.85, 1.15] | 4.36 | 0.04 | 0.07 | -0.56 |
| Both Tests combined | 30 | 0.81 (0.40) | 0.07 | [0.67, 0.96] | 28 | 1.03 (0.37) | 0.07 | [0.88, 1.17] | 4.32 | 0.04 | 0.07 | -0.57 |

PLS-5 results - E-book versus Print Format.

In this ANCOVA analysis (Table L4), the investigation centered on discerning the differential impact of the e-book condition relative to the control condition on performance as measured by the PLS-5 standardised test, serving as the DV. The mean score for the e-book condition was 67.84 (SD=13.72), accompanied by an adjusted mean of 66.56 and a corresponding 95% confidence interval ranging from 64.11 to 69.01. In contrast, the mean score for the control condition was 64.71 (SD=13.58), with an adjusted mean of 66.03. The ANCOVA revealed a non-significant result with F(1, 46) = 0.092, p=0.763, $\eta_p^2=0.00$), indicating the absence of a statistically significant difference between the e-book and control conditions concerning performance on the PLS-5 standardised test. Moreover, the partial eta squared value of 0.00 suggests that negligible variance in PLS-5 scores can be attributed to the condition variable. Cohen's d for the comparison between the e-book

condition and the control condition remains approximately 0.23, indicating a small effect size.

| PLS-5 standardised test | | | | | | | | | | | | | |
|-------------------------|----|--------------------------|---------------------------|------------------|----------------|-------|-------|------------|--|--|--|--|--|
| Condition | N | Pre-test Mean (SD) | Post-test Mean (SD) | Adjusted Mean | 95% CI | F | р | η_p^2 | | | | | |
| E-book | 25 | 68.20 (14.19) | 67.84 (13.72) | 66.56 | [64.11, 69.01] | 0.092 | 0.763 | 0.002 | | | | | |
| Print book (control) | 24 | 65.29 (13.10) | 64.71 (13.58) | 66.03 | [63.53, 68.53] | | | | | | | | |

 Table L4 Analysis of Covariance (ANCOVA) Results of PLS-5 Based on Book Medium

Research Question 2 - Interactive versus Performance Teaching Style

Receptive and Expressive Vocabulary - Interactive versus Performance Teaching Style

In this ANCOVA analysis (Table L5), the focus was on comparing the effects of interactive teaching style condition versus performance teaching style condition on receptive vocabulary, serving as the DV. The mean score for the interactive teaching style condition was 0.70 (SD=0.16), with an adjusted mean of 0.65 and a standard error of 0.01, accompanied by a 95% confidence interval ranging from 0.62 to 0.68. Conversely, the mean score for the performance teaching style condition was 0.59 (SD=0.18), with an adjusted mean of 0.64 and a standard error of 0.01, with a 95% confidence interval ranging from 0.61 to 0.68. The ANCOVA analysis resulted in a non-significant F-value of 0.071 $(F(1, 57) = 0.071, p=0.79, \eta_p^2=0.00)$, suggesting no statistically significant difference between the interactive teaching style and performance teaching style conditions concerning receptive vocabulary outcomes. Furthermore, the partial eta squared value of 0.001 indicates a minimal proportion of the variance in receptive vocabulary scores attributable to the teaching style condition variable. Cohen's d for the comparison between the interactive teaching style condition and the performance teaching style condition is 0.64. This indicates a moderate effect size, suggesting a meaningful difference between the two teaching styles in terms of their impact on receptive vocabulary.

| Receptive voca | bulary | 7 | | | | | | |
|-------------------|--------|--------------------------|---------------------------|------------------|--------------|-------|-------|------------|
| Condition | N | Pre-test Mean (SD) | Post-test Mean (SD) | Adjusted Mean | 95% CI | F | р | η_p^2 |
| Interactive style | 30 | 0.57 (0.16) | 0.70 (0.16) | 0.65 | [0.62, 0.68] | 0.071 | 0.790 | 0.001 |
| Performance style | 30 | 0.46 (0.17) | 0.59 (0.18) | 0.64 | [0.61, 0.68] | | | |

Table L5 Analysis of Covariance (ANCOVA) Results for Receptive Vocabulary Based on

 Teaching Style

In the following ANCOVA analysis (Table L6), the primary objective was to compare the effects of interactive teaching style condition against performance teaching style condition on expressive vocabulary, serving as the dependent variable. The mean score for the interactive teaching style condition was 1.05 (SD=0.73), with an adjusted mean of 0.96 and a standard error of 0.04, accompanied by a 95% confidence interval ranging from 0.87 to 1.04. Conversely, the mean score for the performance teaching style condition was 0.74 (SD=0.61), with an adjusted mean of 0.83, with a 95% confidence interval ranging from 0.74 to 0.91. The ANCOVA analysis yielded a statistically significant result with F(1, 57)= 4.510, p=0.03, η_p^2 =0.07, indicating a significant difference between the two teaching style conditions concerning expressive vocabulary outcomes. Furthermore, the partial eta squared value of 0.073 suggests that approximately 7.3% of the variance in expressive vocabulary scores can be attributed to the teaching style condition variable. Cohen's d for the comparison between the interactive teaching style condition and the performance teaching style condition is approximately 0.46. This indicates a moderate effect size, suggesting a meaningful difference between the two teaching styles in terms of their impact on expressive vocabulary. These findings underscore the importance of considering teaching style variations in educational contexts and provide valuable insights into their differential impacts on expressive vocabulary acquisition.

| Expressive voc | abular | у | | | | | | |
|----------------------|--------|--------------------------|---------------------------|------------------|--------------|-------|-------|------------|
| Condition | Ν | Pre-test Mean (SD) | Post-test Mean (SD) | Adjusted Mean | 95% CI | F | р | η_p^2 |
| Interactive style | 30 | 0.81 (0.65) | 1.05 (0.73) | 0.96 | [0.87, 1.04] | 4.510 | 0.038 | 0.073 |
| Performance style | 30 | 0.63 (0.57) | 0.74 (0.61) | 0.83 | [0.74, 0.91] | | | |

Table L6 Analysis of Covariance (ANCOVA) Results for Expressive Vocabulary Based

 on Teaching Style

Implicit, Explicit and Overall Story Comprehension Results - Interactive versus Performance Teaching Style

One-way ANOVA was performed to compare the effect of teaching style on story comprehension separately for the three story comprehension test scores (implicit, explicit, and combined, see Table L7) at post-test 1. The *Implicit* story comprehension analysis revealed that there was a statistically significant effect of teaching style on Implicit story comprehension (F(1, 56)=4.47, p=0.03, $\eta_p^2=0.07$) favouring the Interactive style. There was no statistical significant difference between the Interactive and Performance teaching styles for *Explicit* comprehension (F(1, 56)=2.76, p=0.10, $\eta_p^2=0.04$). When these two test results were combined and analysed in order to evaluate an overall score for comprehension the results were statistically significant (F(1, 56)=4.15, p=0.04, $\eta_p^2=0.06$). Overall, the three study instruments analysed for story comprehension demonstrated that Interactive scaffolding delivered superior outcomes compared to Performance scaffolding.

Table L7 One-way ANOVA Results of Implicit, Explicit and Overall Story ComprehensionResults Based on Teaching Style (post-test 1)

| | Iı | nteractive | e Teaching | g Style | Pe | rforman | ce Teachiı | ng Style | | | | |
|---------------------------------|----|----------------|---------------|-----------------|----|----------------|---------------|-----------------|------|------|------------|-----------|
| Dependent Variables | N | Mean (SD) | Std. Error | 95% CI | N | Mean (SD) | Std. Error | 95% CI | F | р | η_p^2 | Cohen's d |
| Implicit Story Comprehension | 28 | 1.07 (0.46) | 0.08 | [0.91, 1.24] | 30 | 0.83 (0.41) | 0.08 | [0.67, 0.99] | 4.47 | 0.03 | 0.07 | 0.55 |
| Explicit Story Comprehension | 28 | 0.98 (0.39) | 0.07 | [0.82, 1.13] | 30 | 0.80 (0.41) | 0.07 | [0.65, 0.95] | 2.76 | 0.10 | 0.04 | 0.44 |
| Both Tests combined | 28 | 1.02 (0.40) | 0.07 | [0.88, 1.17] | 30 | 0.81 (0.37) | 0.07 | [0.67, 0.96] | 4.15 | 0.04 | 0.06 | 0.54 |

Using Cohen's d, the effect size calculation for teaching styles for the Implicit story comprehension test revealed a medium effect size of d=0.55 and for Explicit story comprehension test revealed a moderate effect size of d=0.44, favouring the Interactive teaching style.

PLS-5 Results - Interactive versus Performance Teaching Style.

In this ANCOVA analysis (Table L8), the aim was to examine the differential effects of interactive teaching style condition compared to performance teaching style condition on performance as measured by the PLS-5 standardised test, serving as the dependent variable. The mean score for the interactive teaching style condition was 66.96 (SD=14.04), with an adjusted mean of 67.34 and a standard error of 1.16, accompanied by a 95% confidence interval ranging from 64.99 to 69.70. Conversely, the mean score for the performance teaching style control condition was 65.57 (SD=13.31), with an adjusted mean of 65.12 and a standard error of 1.24, with a 95% confidence interval ranging from 62.62 to 67.62. The ANCOVA analysis yielded a non-significant F-value of 1.692 (p =0.200), suggesting no statistically significant difference between the interactive teaching style and performance teaching style conditions regarding performance on the PLS-5 standardised test. Furthermore, the partial eta squared value of 0.035 indicates that approximately 3.5% of the variance in PLS-5 scores can be attributed to the teaching style condition variable. Cohen's d for the comparison between the interactive teaching style condition and the performance teaching style condition is approximately 0.10. This indicates a very small effect size, suggesting a minimal difference between the two teaching styles in terms of their impact on performance on the PLS-5 standardised test.

| PLS-5 standar | dised t | est | | | | | |
|----------------------|---------|------------------|------------------|----------------|-------|-------|------------|
| Condition | Ν | Mean (SD) | Adjusted Mean | 95% CI | F | р | η_p^2 |
| Interactive style | 26 | 66.96 (14.04) | 67.34 | [64.99, 69.70] | | | |
| Performance style | 23 | 65.57 (13.31) | 65.12 | [62.62, 67.62] | 1.692 | 0.200 | 0.035 |

Table L8 Results of Analysis of Covariance (ANCOVA) for the Standardised Test PLS-5in the post-test

Appendix M: Exploratory Analysis

Receptive and Expressive Vocabulary - E-book versus Print Format.

One-way ANOVA was performed for the receptive and expressive vocabulary test. More specifically, the data were analysed by running six one-way ANOVAs, one for each time point (pre, post1, post2) and one for each dependent variable (expressive vocabulary, receptive vocabulary). Table M1 following presents descriptive statistics of receptive and expressive vocabulary scores in the pretest, posttests, and second post-tests. The analysis revealed that there was a statistically significant main effect of medium (e-book versus print book) for the pre receptive vocabulary test ($F(1, 58) = 3.92, p = 0.05, \eta_p^2 = 0.06$) and the post receptive vocabulary test ($F(1, 58) = 4.69, p = 0.03, \eta_p^2 = 0.05$, favouring the print books for post-receptive vocabulary scores was *d*=-0.58, favouring the print book condition. For expressive vocabulary the results were non-significant with a *p*>0.05 for both pre and posttests. The effect size for e-books versus print books for the pre and posttests. The effect size for e-books versus print books for the pre and posttests. The effect size for e-books versus print books for the pre and posttests. The effect size for e-books versus print books for the pre and posttests. The effect size for e-books versus print books for the pre and posttests. The effect size for e-books versus print books for the pre and posttests. The effect size for e-books versus print books for the pre and posttests. The effect size for e-books versus print books for the post-expressive vocabulary test was *d*=-0.18, favouring the print book condition. Results from the second post-test revealed that there was no difference, neither for receptive nor expressive vocabulary at post-test 2, one month following the intervention.

| | | | E-boo | ok conditio | on | | Print B | Print Book condition | | | | | |
|-------------------------------|-------|--------------|----------------|-------------|--------------|--------------|----------------|----------------------|--------------|-------|------------|-----------|-------|
| Dependent Test N Variables | N | Mean (SD) | Std. Error | 95% CI | N | Mean (SD) | Std. Error | 95% CI | F | р | η_p^2 | Cohen's d | |
| Receptive Vocabulary | Pre | 30 | 0.47 (0.19) | 0.03 | [0.41, 0.53] | 30 | 0.56 (0.15) | 0.03 | [0.50, 0.62] | 3.92 | 0.05 | 0.06 | -0.52 |
| | Post1 | 30 | 0.60 (0.19) | 0.03 | [0.53, 0.66] | 30 | 0.70 (0.15) | 0.03 | [0.63, 0.76] | 4.69 | 0.03 | 0.07 | -0.58 |
| | Post2 | 27 | 0.59 (0.19) | 0.03 | [0.52, 0.65] | 24 | 0.65 (0.13) | 0.03 | [0.58, 0.72] | 1.929 | 0.17 | 0.03 | -0.36 |
| Expressive Vocabulary | Pre | 30 | 0.61 (0.58) | 0.11 | [0.39, 0.84] | 30 | 0.83 (0.64) | 0.11 | [0.61, 1.06] | 1.95 | 0.16 | 0.03 | -0.36 |
| | Post1 | 30 | 0.83 (0.66) | 0.12 | [0.58, 1.08] | 30 | 0.96 (0.72) | 0.12 | [0.70, 1.21] | 0.49 | 0.48 | 0.00 | -0.18 |
| | Post2 | 27 | 0.82 (0.66) | 0.14 | [0.53, 1.10] | 24 | 0.97 (0.79) | 0.14 | [0.67, 1.27] | 0.549 | 0.46 | 0.01 | -0.20 |

Table M1 One-way ANOVA Results for Receptive and Expressive Vocabulary Based on Book Medium (pre-test, post-test 1, post-test 2)

Receptive and Expressive Vocabulary - Interactive versus Performance Teaching Style. A series of one-way ANOVAs was performed for comparing the scores before and after the intervention in terms of receptive vocabulary and expressive vocabulary for the two teaching styles. More specifically, the data were analysed by running six one-way ANOVAs, one for each time point (pre, post1, post2) and one for each dependent variable (expressive vocabulary, receptive vocabulary). Table M2 presents descriptive statistics for the pre-test, post-test 1 and post-test 2 for receptive and expressive vocabulary as dependent variables and the teaching style, performance and interactive style, as independent variables. According to the results illustrated in Table 5.13, receptive vocabulary scores were statistically significant (p<.05) and there is a difference between teaching styles at pre and at post-test. More specifically, the analysis revealed that there was a statistically significant main effect of teaching style (interactive versus performance) for the pre receptive vocabulary test (F(1, 58) = 5.869, p = 0.01, $\eta_p^2=0.09$, d=0.66) and the post receptive vocabulary test (F(1, 58) = 5.042, p = 0.02, $\eta_p^2=0.08$, d=0.64). The effect size for teaching styles for the post-receptive vocabulary test revealed a large effect size of d= 0.64, favouring the interactive teaching style.

Following the first post-test, participants completed a second post-test after one month for the dependent variables receptive and expressive vocabulary. The second post-test revealed a statistically non-significant main effect for both dependent variables, suggesting that teaching style did not make a difference in terms of vocabulary skills.

| | Interactive teaching style condition | | | Performance teaching style condition | | | | | | | | | |
|--------------------------|--------------------------------------|----|----------------|--------------------------------------|--------------|----|----------------|---------------|--------------|-------|-------|------------|-----------|
| Dependent Variables | Test | Ν | Mean (SD) | Std. Error | 95% CI | Ν | Mean (SD) | Std. Error | 95% CI | F | р | η_p^2 | Cohen's d |
| Receptive Vocabulary | Pre | 30 | 0.57 (0.16) | 0.03 | [0.51, 0.63] | 30 | 0.46 (0.17) | 0.03 | [0.40, 0.52] | 5.869 | 0.019 | 0.09 | 0.66 |
| | Post1 | 30 | 0.70 (0.16) | 0.03 | [0.63, 0.76] | 30 | 0.59 (0.18) | 0.03 | [0.53, 0.66] | 5.042 | 0.029 | 0.08 | 0.64 |
| | Post2 | 26 | 0.64 (0.16) | 0.03 | [0.57, 0.71] | 25 | 0.59 (0.17) | 0.03 | [0.53, 0.66] | 0.921 | 0.342 | 0.01 | 0.30 |
| Expressive Vocabulary | Pre | 30 | 0.81 (0.65) | 0.11 | [0.58, 1.04] | 30 | 0.63 (0.57) | 0.11 | [0.41, 0.86] | 1.197 | 0.278 | 0.02 | 0.29 |
| | Post1 | 30 | 1.05 (0.73) | 0.12 | [0.80, 1.30] | 30 | 0.74 (0.61) | 0.12 | [0.49, 0.99] | 3.129 | 0.082 | 0.05 | 0.46 |
| | Post2 | 26 | 0.92 (0.78) | 0.14 | [0.64, 1.21] | 25 | 0.85 (0.67) | 0.14 | [0.55, 1.14] | 0.139 | 0.711 | 0.00 | 0.09 |

Table M2 One-way ANOVA Results for Receptive and Expressive Vocabulary Based on Teaching Style (pre-test, post-test 1, post-test 2)

Interaction Effect of Book Medium and Teaching Style

In this section, all possible combinations of the levels of the factors are investigated by analysing both factors (2 x 2) by time (pre, post) in order to evaluate main effects and interaction effects for each dependent variable. Therefore, in order to evaluate the main effects and the interaction effect of the two factors (book medium and teaching style) a series of two-way repeated measures ANOVAs were performed which included data from pre-test to post-test 1. It was important to evaluate the two variables (book medium and teaching style) with two time points (pre-test and post-test 1) first as most participants were present at both time points. Following this analysis, repeated measures ANOVA was performed that included the third time point, i.e., the second post-test which was administered to the entire sample resulting from factors such as pupils leaving the school or pandemic-related circumstances.

Receptive Vocabulary. A repeated measures ANOVA with two factors for receptive vocabulary (pre-test and post-test 1; 2x2x2) revealed non-significant effects for both teaching style effect and its interaction with book medium (Table M3).

Table M3 Repeated Measures ANOVA Results for Receptive Vocabulary of Book Mediumby Teaching Style (pre-test versus post-test 1)

| | | | | | Time | | | | | |
|---------------------------------------|-----------------------|----|----------------|---------------|--------------|--------|----------------|---------------|-----------------|--|
| | | | P | re | | Post 1 | | | | |
| Book Medium | Teaching Style | N | Mean (SD) | Std. Error | 95% CI | N | Mean (SD) | Std. Error | 95% CI | |
| E-book | Interactive style | 15 | 0.53 (0.18) | 0.04 | [0.44, 0.62] | 15 | 0.64 (0.19) | 0.04 | [0.55, 0.74] | |
| · | Performan ce style | 15 | 0.41 (0.19) | 0.04 | [0.32, 0.50] | 15 | 0.55 (0.20) | 0.05 | [0.44, 0.64] | |
| Print Book | Interactive style | 15 | 0.61 (0.15) | 0.04 | [0.52, 0.70] | 15 | 0.75 (0.11) | 0.05 | [0.64, 0.86] | |
| | Performan ce style | 15 | 0.51 (0.16) | 0.04 | [0.42, 0.60] | 15 | 0.64 (0.16) | 0.04 | [0.56, 0.76] | |
| ANOVA results | | | F | | р | | | η_p^2 | | |
| Book Medium Effect | | (| 0.173 | | 0.679 | | | 0.0 | 0 | |
| Teaching Style Effect | 0.051 | | | | 0.822 | | | 0.0 | 0 | |
| Book Medium x Teaching Style | | | 0.560 | | 0.458 | | | 0.0 | 1 | |

A two-way repeated measures ANOVA was then performed including the three time points, post-test, post-test 1 and post-test 2 (Table M4). Results revealed a p-value of 0.053 for book medium effect (F(1, 47)=3.940, p=0.053, $\eta_p^2=0.07$) indicating the results do not meet the threshold for statistical significance. While a p-value of 0.053 suggests that there is some evidence to suggest that the book medium may have an effect on children's receptive vocabulary scores favouring the print book, the results do not meet the traditionally accepted threshold for statistical significance. For teaching style effect and its interaction with book medium non-significant results were revealed.

| | Pre | | | Pos | st 1 | Post | 2 |
|----------------|-------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Feaching | Ν | Mean | SD | Mean | SD | Mean | SD |
| Style | | | | | | | |
| nteractive | 15 | 0.53 | 0.18 | 0.64 | 0.19 | 0.62 | 0.19 |
| style | | | | | | | |
| Performan | 12 | 0.41 | 0.19 | 0.54 | 0.20 | 0.55 | 0.19 |
| ce style | | | | | | | |
| nteractive | 11 | 0.60 | 0.14 | 0.75 | 0.11 | 0.67 | 0.13 |
| style | | | | | | | |
| Performan | 13 | 0.54 | 0.16 | 0.66 | 0.16 | 0.64 | 0.14 |
| ce style | | | | | | | |
| | | | F | | р | η_p^2 | |
| | | | | | | • | |
| l | | 3. | 940 | | 0.053 | 0.0 | 7 |
| | | | | | | | |
| Teaching Style | | 2. | 769 | | 0.103 | 0.05 | |
| | | | | | | | |
| Book Medium | | 0.163 | | | 0.688 | | 0 |
| x Teaching | | | | | | | |
| | | | | | | | |
| | Yeaching Style Interactive style erforman ce style erforman ce style | Yeaching Style N Interactive style 15 erforman 12 ce style 11 style 13 ce style 13 ce style 14 | Presente Presente Veaching N Mean Style 15 0.53 style 0.41 ce style erforman 12 outeractive 11 outeractive 11 outeractive 11 outeractive 11 outeractive 13 outeractive 13 outeractive 13 outeractive 13 outeractive 13 outeractive 13 | Free Veaching N Mean SD Style 15 0.53 0.18 style 15 0.41 0.19 ce style 0.60 0.14 style 11 0.60 0.14 style 13 0.54 0.16 ce style F 3.940 ce 2.769 0.163 | Fre Fre Fre Style N Mean SD Mean Style 15 0.53 0.18 0.64 style | Fre Post 1 Veaching N Mean SD Mean SD style 15 0.53 0.18 0.64 0.19 style 12 0.41 0.19 0.54 0.20 ce style 0.14 0.75 0.11 style 11 0.60 0.14 0.75 0.11 style 13 0.54 0.16 0.66 0.16 ce style F p 3.940 0.053 0.103 ce 2.769 0.103 0.688 | Free Fost 1 Fost 1 Fost 1 Veaching Style N Mean SD Mean SD Mean interactive 15 0.53 0.18 0.64 0.19 0.62 style |

Table M4 Repeated Measures ANOVA Results for Receptive Vocabulary of Book Mediumby Teaching Style (pre-test, post-test 1 and post-test 2)

Expressive Vocabulary. A repeated measures ANOVA with two factors for expressive vocabulary (pre-test and post-test 1; 2x2x2) revealed non-significant effects for book medium and its interaction with teaching style effect (Table M5). However, the main effect of teaching style revealed statistically significant results with F(1, 56) = 5.211, p = 0.02, $\eta_p^2 = 0.08$, which reveals that no matter what book medium is used for expressive vocabulary the teaching style matters (see Figures M1 and M2).

Table M5 Repeated Measures ANOVA Results for Expressive Vocabulary of BookMedium by Teaching Style (pre-test versus post-test 1)

| | | | | | Time | | | | |
|---------------------------------|-----------------------|-------|----------------|---------------|--------------|----|----------------|---------------|--------------|
| | | | Р | Post 1 | | | | | |
| Book Medium | Teaching Style | Ν | Mean (SD) | Std. Error | 95% CI | Ν | Mean (SD) | Std. Error | 95% CI |
| E-book | Interactive style | 15 | 0.58 (0.58) | 0.15 | [0.27, 0.90] | 15 | 0.83 (0.74) | 0.17 | [0.49, 1.18] |
| | Performan ce style | 15 | 0.64 (0.59) | 0.15 | [0.32, 0.95] | 15 | 0.83 (0.58) | 0.17 | [0.48, 1.17] |
| Print Book | Interactive style | 15 | 1.04 (0.66) | 0.15 | [0.72, 1.35] | 15 | 1.26 (0.68) | 0.17 | [0.92, 1.61] |
| | Performan ce style | 15 | 0.63 (0.57) | 0.15 | [0.32, 0.95] | 15 | 0.65 (0.63) | 0.17 | [0.30, 1.00] |
| ANOVA results | | | F | | р |) | | | η_p^2 |
| Book Medium effect | | 2.696 | | | 0.106 | | | | 0.04 |
| Teaching Style Effect | | 5.211 | | | 0.026 | | | | 0.08 |
| Book Medium x Teaching Style | | | 1.589 | | 0.2 | 13 | | | 0.02 |

Figure M1 Repeated Measures for Expressive Vocabulary of Book Medium by Teaching Style for the Performance Teaching Style (pre-test to post-test 1)







Following the analysis shown in Table M5, Table M6 presents a repeated measures ANOVA for expressive vocabulary which revealed that the interaction effect for the dependent variable expressive vocabulary, measured at three time-points, was not significant (F(1, 47) = 2.071, p = 0.15, $\eta_p^2 = 0.04$), as well as the effects of book medium and teaching style.

Table M6 Repeated Measures ANOVA Results for Expressive Vocabulary of BookMedium by Teaching Style (pre-test, post-test 1 and post-test 2)

| | | | | | Tiı | ne | | | |
|--------------------------------|-----------------------|-------|------|--------|-------|------|----------------|------|--|
| | | Pre | | re | Pos | st 1 | Post 2 | | |
| Book | Teaching | Ν | Mean | SD | Mean | SD | Mean | SD | |
| Medium | Style | | | | | | | | |
| E-book | Interactive style | 15 | 0.58 | 0.58 | 0.83 | 0.74 | 0.72 | 0.61 | |
| | Performan ce style | 12 | 0.71 | 0.63 | 0.87 | 0.65 | 0.94 | 0.71 | |
| Print Book | Interactive style | 11 | 0.98 | 0.72 | 1.21 | 0.75 | 1.21 | 0.92 | |
| | Performan ce style | 13 | 0.68 | 0.60 | 0.70 | 0.67 | 0.76 | 0.64 | |
| ANOVA results | | | 1 | ч Г | P | , | η_{I}^{2} | | |
| Book Me Effect | dium | 0.610 | | 510 | 0.439 | | 0.01 | | |
| Teaching Style 0.569 Effect | | 0.454 | | 0.01 | | | | | |
| Book Me Teaching | lium x Style | | 2.0 | 071 | 0.157 | | 0.0 | 4 | |

Story Comprehension (Implicit and Explicit). For implicit and explicit story comprehension results a series of two-way ANOVAs were performed at post-test 1. Specifically, two separate two-way ANOVAs were conducted to evaluate the interaction between the two variables (book medium and teaching style).

Implicit Story Comprehension. For implicit story comprehension (Table M7, Figure M3) the teaching style effect was statistically significant F(1, 54) = 5.223, p = 0.02, $\eta_p^2 = 0.08$, but not the book medium effect, F(1, 54) = 3.949, p = 0.052, $\eta_p^2 = 0.06$. The interaction effect of teaching style by book medium on implicit story comprehension was not significant, F(1, 54) = 1.450, p = 0.23, $\eta_p^2 = 0.02$.

| Rook | Toophing | N | Moon | Std Frror | 05% CI |
|---------------------|---------------------|----|--------|------------|--------------|
| Madia | f eaching St-rlo | 14 | (SD) | Stu. E1101 | 93 /0 CI |
| Mealum | Style | | (5D) | | |
| E-book | Interactive | 15 | 0.91 | 0.11 | [0.69, 1.13] |
| | style | | (0.47) | | |
| | Performance | 15 | 0.79 | 0.11 | [0.57, 1.01] |
| | style | | (0.42) | | |
| Print Book | Interactive | 13 | 1.26 | 0.18 | [1.03, 1.50] |
| | style | | (0.38) | | |
| | Performance | 15 | 0.87 | 0.11 | [0.65, 1.09] |
| | style | | (0.40) | | |
| ANOVA | | | F | р | η_p^2 |
| results | | | | | - F |
| Book Mediu | m | 3. | 949 | 0.052 | 0.06 |
| effect | | | | | |
| Teaching Sty | yle | 5. | 223 | 0.026 | 0.08 |
| Effect | | | | | |
| Book Mediu | m x | 1. | 450 | 0.234 | 0.02 |
| Teaching Sty | yle | | | | |

Table M7 Two-way ANOVA Results for Implicit Story Comprehension of Book Mediumby Teaching Style (post-test 1)

Figure M3 Implicit Story Comprehension of Book Medium by Teaching Style at Post-test 1



Teaching Style

Explicit Story Comprehension. The book medium effect for explicit story compression was statistically significant with a small effect size (F(1, 54) = 4.718, p = 0.03, $\eta_p^2 = 0.08$), which reveals that no matter what teaching style a participant is taught with the book medium makes the difference for explicit story comprehension. The interaction effect of book medium by teaching style on explicit story comprehension was not significant, F(1, 54) = 0.001, p = .976, $\eta_p^2 = 0.00$, which means that the effects of medium and teaching style on explicit story comprehension are not dependent on each other (Table M8, Figure M4).

| Book | Teaching | Ν | Mean | Std. Error | 95% CI |
|-------------------|-------------|----|---------------|------------|--------------|
| Medium | Style | | (SD) | | |
| E-book | Interactive | 15 | 0.87 | 0.10 | [0.67, 1.07] |
| | style | | (0.39) | | [0.07, 1.07] |
| | Performance | 15 | 0.69 | 0.10 | [0.48, 0.89] |
| | style | | (0.43) | | |
| Print Book | Interactive | 13 | 1.10 | 0.10 | [0.88, 1.32] |
| | style | | (0.36) | | |
| | Performance | 15 | 0.91 | 0.10 | [0.71, 1.11] |
| | style | | (0.37) | | |
| ANOVA | | | F | р | η_p^2 |
| results | | | | | - 1 |
| Book Mediu | m | 4. | 718 | 0.034 | 0.08 |
| effect | | | | | |
| Teaching Style | | 3. | 167 | 0.081 | 0.05 |
| Effect | | | | | |
| Book Mediu | m | 0. | 001 | 0.976 | 0.00 |
| x Teaching | | | | | |
| Style | | | | | |

Table M8 Two-way ANOVA Results for Explicit Story Comprehension of Book Medium byTeaching Style (post-test 1)



Figure M4 Explicit Story Comprehension of Book Medium by Teaching Style at Post-test 1

Standardised Test PLS-5 UK. The interaction effects of book medium by teaching style from the repeated measures ANOVA were not significant, F(1, 45) = 0.040, p = .842, $\eta_p^2 = 0.00$ (Table M9), suggesting that there is no standardised test effect.

Table M9 Repeated Measures ANOVA Results for Standardised Test PLS-5 UK of Book Mediumby Teaching Style (pre-test versus post-test 1)

| | | | | | Time | |
|---------------------------------|-----------------------|----|------------------|------------|------------------|------------|
| | Teaching | | | Pre | Po | st 1 |
| Book Medium | Teaching Style | Ν | Mean (SD) | Std. Error | Mean (SD) | Std. Error |
| E-book | Interactive style | 14 | 68.21 (14.20) | 3.75 | 68.71 (14.20) | 3.71 |
| | Performan ce style | 11 | 68.18 (14.88) | 3.75 | 66.73 (13.69) | 4.19 |
| Print Book | Interactive style | 12 | 64.17 (13.25) | 3.75 | 64.92 (14.18) | 4.01 |
| | Performan ce style | 12 | 66.42 (13.44) | 3.75 | 64.50 (13.46) | 4.01 |
| ANOVA re | esults | | F | | р | η_p^2 |
| Book Med Effect | ium | | 0.004 | | 0.953 | 0.00 |
| Teaching S Effect | Style | | 1.699 | | 0.199 | 0.03 |
| Book Medium x Teaching Style | | | 0.040 | | 0.842 | 0.00 |

Appendix N: Research Tools Selection Process

To choose the appropriate research instruments for the experimental investigation, it was crucial to review similar studies and examine the tools they used. This resulted in the development of Table N1, which presents the study tools utilised in the studies evaluated in the meta-analysis, as well as studies involving children learning EAL with the support of an e-book.

Vocabulary Research Tools

The vocabulary measures were created by building upon other studies that investigated the acquisition of expressive and receptive vocabulary in young children (e.g., Homer et al., 2014; Kelley & Kinney, 2017; Korat et al., 2014a; Takacs & Bus, 2016; Yang et al., 2022). Target words were selected from each story. All of the target words were categorised as Tier 2 words, which are regarded to be high utility and high frequency vocabulary words that can be beneficial for children to learn in order to improve their vocabulary and comprehension skills as suggested by Beck et al. (2013) and Beck et al. (2007). The process of selecting words involved carefully choosing wide-ranging and practical vocabulary (i.e., useful for story comprehension), seeking terms that conveyed ideas familiar to preschool children while enabling more precise communication (e.g., using "between" instead of "middle"). The focus was on words that children with limited vocabulary were unlikely to be familiar with but were probably familiar with among their peers who had larger vocabularies or older children. The criteria for selecting words were operationalized based on the following factors as in Spencer et al. (2012): (a) the word should not be uncommon in the vocabularies of pre-K children with limited knowledge; (b) it should appear relatively frequently in the discourse of adult speakers; (c) it should have a straightforward definition suitable for children (e.g., the definition of grin was to *smile*); and (d) it should be contextually relevant to the storybook.

Receptive Vocabulary

The receptive vocabulary test demonstrated resemblance to measures employed in prior studies (see Table 1; Kelley & Kinney, 2017; Korat et al., 2014a; O'Toole, 2015; O'Toole & Kannass, 2018; Sari et al., 2019; Zhou & Yadav, 2017; Silverman, 2007; Sun et al., 2019; Takacs & Bus, 2016; Yang et al., 2022). The receptive vocabulary test was created

in the same format as the standardised test Peabody Picture Vocabulary Test (PPVT-test; Dunn & Dunn, 1981). The children were asked to choose the illustration that best represents the word's meaning out of a set of four illustrations. The visual representation of the word did not precisely replicate the depiction in the story. The selected images consisted of recognisable things (e.g., pencil, chair) that were incorporated to sustain children's engagement (refer to Sénéchal et al., 1995).

Expressive Vocabulary

Previous studies investigating the effectiveness of e-books for improving expressive vocabulary skills in both typically developing children and children learning EAL assessed children's expressive vocabulary by asking children to provide a definition for words included in the story (Homer et al., 2014; Kelley & Kinney, 2017; Korat et al., 2014a; Silverman, 2007; Yang et al., 2022). The children were instructed to provide explanations for the meanings of the target words. For example, the child was asked: "What is the meaning of *suddenly*?".

Story Comprehension Research Tools

According to Florit et al. (2011), story comprehension is a complex process that requires the processing of explicit (i.e., information presented in the text) and implicit information (i.e., information inferable from the text / illustrations or from previous knowledge). Paris and Paris (2003) devised a narrative comprehension test specifically designed to evaluate the explicit and implicit story comprehension skills of young children. Explicit story comprehension (ESC) involves the recognition and understanding of the many components of a story, such as characters, setting, initiating event, issue, and solution. Implicit story comprehension (ISC) encompasses the ability to deduce the emotions of the story's main characters, establish cause-and-effect relationships, make predictions, and identify the underlying theme. ISC necessitates a more profound ability to understand and interpret stories. The objective of this study was to assess the effectiveness of e-books in enhancing young children's story comprehension skills, including both implicit and explicit aspects. Twelve prompted open-ended comprehension questions (Paris & Paris, 2003) were employed to assess children's story comprehension. Seven questions were related to explicit comprehension (characters, setting, initiating event, problem, and outcome or solution) and five to implicit comprehension (feelings, causal inference,

dialogue, prediction, and theme). Open-ended *Wh*-questions were created to measure children's story comprehension, a style of measurement consistent with prior work in e-book interventions and young children (Altun, 2021; Critelli, 2011; Karemaker et al., 2017; Kozminsky & Asher-Sadon, 2013; O'Toole, 2015; O'Toole & Kannass, 2018; Reich et al., 2019; Richter & Courage, 2017; Zhou & Yadav, 2017). In addition, children (during the implicit test) were allowed to refer to the story illustrations while answering the questions, which helped to reduce the impact of working memory on children's story comprehension performance (Cain et al., 2004). During the explicit test, children were asked to freely recall the answers and if they were not able to answer the open-ended question, I offered three multiple-choice options (e.g., Reich et al., 2019).

Table N1 Research Tools Used in Studies Related to E-books and Young Children

| | | Research Tools |
|---------------------------|-------------------------|----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Study and Year | Outcome measures | Details |
| Altun (2021) | Story Comprehension | The children were asked to respond to five questions to ascertain their story comprehension levels. The questions involved story |
| | (1 measure) | elements such as characters, settings, plots, problems, and outcomes. |
| | Vocabulary (Receptive) | The Turkish Receptive Language Test was used to assess the participating children's vocabulary levels of Turkish children aged 2- |
| | (1 measure) | 12 years old. |
| Broemmel et al. | Story Comprehension | The participant was given the book and invited to look at the pictures and tell the story. |
| (2015) | (1 measure) | |
| Critelli | Story Comprehension | A 10-item comprehension test which consisted of various questions regarding the story. |
| (2011) | (1 measure) | |
| Eng et al. (2020) | Story Recall | Children were asked questions that probed their memory for details about the story that fit the narrative reconstruction |
| | (1 measure) | criteria. |
| Homer et al. | Vocabulary (Expressive) | Children were asked to explain the meaning of a set of words. For example, experimenters asked, "Can you tell me what the word |
| (2014) | (1 measure) | 'secret' means'? What does it mean when something is secret?". |
| Ihmeideh (2014) | Vocabulary (Expressive) | Children were shown ten pictures of objects and were asked to say the word of the object. Afterward, children were asked a |
| | (1 measure) | follow-up question related to the picture shown. The raters corrected children's responses based on one of two domains: |
| | V | Knows the meaning of the word or does not know. |
| Karemaker et al. (2017) | (1 massure) | Distignary format |
| (2017) | | Story comprehension task that included 6 open ended questions about the story quests (1 inforential question 1 complex phrase |
| | (2 measures) | question and 4 factual questions) |
| | (2 measures) | Recall measure to assess children's recall of the e-book story. Children had to recall the story without the aid of the book |
| | | |
| Kelley and | Vocabulary (Expressive) | On the definitional measure, participants were asked to respond to open-ended definitional questions. |
| Kinney (2017) | (2 measures) | Participants were asked to respond to four ves-no questions. |
| | Vocabulary (Receptive) | On the recentive vocabulary measure participants were asked to select from a plate of four pictures a picture that illustrates the target |
| | (1 measure) | word. Pictures were high-quality photographs chosen to clearly represent the target word. |
| | Story comprehension | Story comprehension was measured using a story retell task. |
| | (1 measure) | |
| Korat, Levin, | Vocabulary (Expressive) | The children were presented orally with the 14 target words, one at a time, and were asked to explain their meaning. |
| Atishkin, | (1 measure) | |
| Turgeman | V | $\mathbf{D}_{\text{rescaled}} = \mathbf{D}_{\text{rescaled}} + \mathbf{D}_{\text$ |
| (2014a) | (1 masure) | Researchers created a receptive word meaning test in the same format as the PPV 1-test (Dunn & Dunn, 1981). The children were asked to choose the illustration that hest represents the word's meaning out of a set of four illustrations. |
| | (1 measure) | asked to choose the musication that dest represents the word's meaning out of a set of four musications. |

(continued)
Table N1 Research Tools Used in Studies Related to E-books and Young Children

| | | Research Tools |
|----------------------------------------|----------------------------------------|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| Study and Year | Outcome measures | Details |
| Kozminsky and Asher-Sadon (2013) | Story Comprehension (1 measure) | Four questions based on a picture from the story and two moral inference questions about the story. |
| | Vocabulary (1 measure) | Examined the child's understanding of difficult words in the story. |
| Lee (2020) | Vocabulary (Expressive) (1 measure) | In a multiple-choice vocabulary pretest, students were asked to select the meaning of 14 target words out of four choices [e.g., abandon: (a) to show off; (b) to give up; (c) to jump; and, (d) I don't know]. |
| Neuman et al. (2017) | Vocabulary (Expressive) (1 measure) | Five words from each story were chosen as target words. Children were first given a word, such as "gigantic" then given a sentence with the word in context. Children were asked to identify the meaning of the word. |
| | Story Comprehension (2 measures) | The child was asked to retell the story. A point was given for each story element described (e.g., setting (1), characters (1), events (3), plot or theme (1), resolution (1)). |
| | | Researchers created five pictures with event scenes from each story. Cards were mixed, and children were asked to sequence the story. |
| O'Toole (2015) | Vocabulary (Receptive) (1 measure) | The test consisted of a series of plates. Each plate contained four pictures of objects or actions (one per quadrant). Fourteen plates were constructed. Four of the 14 plates contained familiar objects (e.g., a pencil, a spoon, a teddy bear, and a cup). |
| | Story Comprehension (1 measure) | Seven Wh-questions were created in order to measure children's story comprehension. |
| O'Toole and Kannass (2018) | Vocabulary (Receptive) (1 measure) | The word learning test consisted of 14 plates, and each plate contained four pictures of objects or actions (one per quadrant). |
| | Story Comprehension (1 measure) | Story comprehension questions. Open-ended "wh-" questions were created to measure children's story comprehension. |
| Pearman (2008) | Story Comprehension (1 measure) | Upon completion of the reading, students performed an oral retelling following the cues of "Tell me about the story" or "Can you tell me the story that you just read?" |
| Reich et al. (2019) | Story Comprehension (3 measures) | The child was asked 14 questions about the book that were drafted specifically for this study. The first was a global recall question, "What was the book about?" followed by specific questions about the characters. For eight of the questions, children were asked to freely recall the answers. If they could not recall the answers correctly, the researcher read a list of multiple-choice options to select from. Children were asked to freely recall the answer and if they were not able to answer the open-ended question, we offered three multiple-choice options. |
| | | Sequencing of story events: children placed four images from the story in the order that they occurred. |
| | | Story events and characters. There were eight questions on characters or events in the story. Questions were presented verbally or with a picture. |
| | Vocabulary (Expressive) (1 measure) | Three questions about words presented in the book. Two asked meaning (e.g., <i>He was a veterinarian. But what is a veterinarian?</i>) with recognition and free recall and one asked for visual recognition of a word that is large and phonetically spelled in the book. |
| Richter and Courage (2017) | Story Comprehension (1 measure) | Children were assessed with nine questions that were asked after each reading. |

(continued)

Table N1 Research Tools Used in Studies Related to E-books and Young Children

| | | Research Tools |
|--------------------|-------------------------|------------------------------------------------------------------------------------------------------------------------------------------|
| Study and Year | Outcome measures | Details |
| Rvachew et al. | Story Comprehension | Story retell was elicited by showing the child the cover page of the book along with the following standard script. |
| (2017) | (1 measure) | |
| Sapsaglam et al. | Story Comprehension | The researcher asked children five questions regarding the story. |
| (2020) | (1 measure) | |
| Sari et al. (2019) | Story Comprehension | Researchers used a retelling procedure in which we prompted children to explain how the characters respond to various events. The |
| | (1 measure) | questions targeted the problem, the initiating event, the reaction, the solution, and the emotions of characters elicited by the problem |
| | | or solution. |
| | Vocabulary (Receptive) | Children completed the receptive test by selecting one of four pictures that corresponded to the target word spoken out loud by the |
| | (1 measure) | experimenter (similar to the procedure of the PPVT). |
| | Vocabulary (Expressive) | Expressive knowledge of the target words was tested with a sentence completion task in which children were asked to complete a |
| | (1 measure) | sentence presented with a picture from the story. |
| Smeets and Bus | Vocabulary (Receptive) | The Peabody Picture Vocabulary Test is a standardized test that was used to assess children's level of general receptive vocabulary. |
| (2015) | (1 measure) | Children were asked to select one of four pictures matching a word named by the examiner. |
| | Vocabulary (Expressive) | To assess children's level of general expressive vocabulary, the experimenter showed pictures while reading aloud incomplete |
| | (1 measure) | sentences. Children were asked to complete sentences such as "A bird flies in the ? [air]". |
| | Story Comprehension | Children were asked to retell one of the four stories they had heard during the intervention. |
| | (1 measure) | |
| Takacs and Bus | Vocabulary (Receptive) | This test was a multiple-choice test where children had to choose the corresponding picture from 4 options. |
| (2016) | (1 measure) | |
| | Story comprehension | Resembling the common activity of independently "reading" a familiar storybook, we asked children to retell the three stories using |
| | (1 measure) | the static illustrations of the stories. |
| Zhou and Yadav | Vocabulary (Receptive) | The target vocabulary test required children to select among four pictures the one picture that best represented the given target word. |
| (2017) | (1 measure) | |
| | Story Comprehension | The story comprehension test contained six open-ended questions. |
| | (1 measure) | |
| Zipke (2017) | Story Comprehension | Participants were asked to retell the story to the experimenter. |
| | (1 measure) | |
| | Vocabulary | Participants were asked to choose a salient word from an array of comparable words to see if they recognized the target word. |
| | (1 measure) | |

(continued)

Table N1 Research Tools Used in Studies Related to E-books and Young Children

| | | Research Tools |
|--------------------|----------------------------|----------------------------------------------------------------------------------------------------------------------------------------|
| Study and Year | Outcome measures | Details |
| Studies with child | ren learning EAL & e-books | |
| Silverman (2007) | Vocabulary (Receptive) | Students were presented with four pictures and asked to "Point to the picture that shows that someone is afraid". |
| | (1 measure) | |
| | Vocabulary (Expressive) | Children were asked without picture prompts, "What is?" or "What doesmean?". |
| | (1 measure) | |
| Silverman and | Vocabulary | For each target word, children were asked four yes-or-no questions. |
| Hines (2009) | (1 measure) | |
| Sun et al. (2019) | Vocabulary (Expressive) | The productive vocabulary test measured whether children could orally produce the target words in the context of the storybooks. |
| | (1 measure) | |
| | Vocabulary (Receptive) | The multiple-choice format of the test was similar to PPVT, were children were asked to choose the corresponding picture from four |
| | (1 measure) | options. |
| | Story Comprehension | Children were asked to retell the three stories using the static illustrations of the stories. |
| | (1 measure) | |
| Verhallen and | Vocabulary (Receptive) | Among three distracters, children chose the illustration that represented the target item. |
| Bus (2010) | (1 measure) | |
| | Vocabulary (Expressive) | Children filled in the last word of a stimulus sentence that the experimenter orally presented while the computer screen showed a |
| | (1 measure) | matching picture from the story. |
| Verhallen et al. | Story Comprehension | The researchers asked form children to retell the story with the help of story pictures. |
| (2006) | (1 measure) | |
| Yang et al. | Vocabulary (Receptive) | Children heard one English word or phrase from the story and selected which four pictures displayed on the screen best illustrate that |
| (2022) | (1 measure) | term. |
| | Vocabulary (Expressive) | Children were shown four big pictures consisting of 25 story-related items and asked children had to name it in English. |
| | (1 measure) | |
| | Story Comprehension | The comprehension test consisted of seven sets of questions, with closed-ended question and open-ended questions. |
| | (1 measure) | |